

Accessible information: collaborating with people with aphasia to develop an evidence-based template for health information.

Caroline Haw

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy

> The University of Sheffield Faculty of Medicine, Dentistry and Health Department of Human Communication Sciences

> > February 2017

Acknowledgements

Many people, from all spheres of my life, have supported me during the writing of this thesis, for which I would like to offer my heartfelt thanks.

I would not have started, persevered with, or completed this thesis without the inspiration, patience, and guidance of my mentor, supervisor and friend, Ruth Herbert. Thank you Ruth for the opportunity, and for your kindness, encouragement, tireless reading and commenting on drafts, and for teaching me so much about research.

My thanks are also due to my second supervisors, Shelagh Brumfitt, Catherine Tattersall, and Emma Gregory, each of whom has supported me through the PhD process in different and complementary ways. Thanks also to Bev Bennett, who commented on the qualitative aspects of my writing.

This whole endeavour would not have been possible without the wholehearted loving support and confidence of my family, who have sustained me and provided much-needed perspective, through the highs and the lows, from the beginning to the end. To all of you, thank you. To my husband Pat, for all your backstage support, incidental music, and your supreme knowledge of commas. To my daughters, Olivia and Rachel, for your "you can do it Mum!" and for being proud of me, as I am proud of you both. To all my parents, for their love and loyal belief in me, and particularly my Mum, Amanda Parkyn, who has always inspired me to reach just beyond my grasp.

I'm indebted to many friends, near and far, the ones I see often, and the ones I see rarely. Special thanks to my dear friends Janet Walmsley and Margaret Boddy for urging me on from the outset.

I am grateful to many colleagues, current and past, for their steadfast encouragement. I owe a particular debt of gratitude to my team mates Jo Burke, Emma Gibbs, Jenni Crisp, and Audrey Delaney who have been unswervingly generous in their support throughout.

The project described in this thesis was a creative collaboration. Thank you to everyone involved, Maggie Warburton and Aileen Donald at the Stroke Association, who provided generous funding for the research, my co-worker Cath Brown, Jon Dale, graphic designer, and the many colleagues who have been generous with their knowledge and time. Most of all, thank you to the people with aphasia who participated in the focus groups and who inspired me to complete the thesis.

I am fortunate to have had the support of Sheffield Teaching Hospitals NHS Foundation Trust in pursuing my research studies. Thank you to my managers Lynn Johnson who took the risk, and to Sue Pownall for supporting me to see it through to completion.

Sincere thanks for IT support in the department of Human Communication Sciences, especially the ever-helpful Jonathan Holroyd, who has rescued me from many word processing scrapes, and Katharina Albrecht, for kindly sharing your knowledge of formatting.

This project stems from a career working with people with aphasia. I hope this research does ultimately make a difference to the experience of people with aphasia and their families, for the many people I have met over the course of my career have provided the inspiration.

Funding statement

This research was undertaken with the support of grants from the Stroke Association (TSA FF 2010/04 and TSA FF 2011/01), and from South Yorkshire Collaboration for Leadership in Applied Health Research and Care (SY CLAHRC).

The materials developed in this project are the copyright of The Stroke Association and are reprinted here with the kind permission of The Stroke Association.

Potential conflict of interest. R. Herbert, C. Haw and E. Gregory co-authored Accessible Information Guidelines (2012) (<u>http://www.stroke.org.uk/professionals/accessible-</u> <u>information-guidelines</u>), which are based partially on the work reported here.

Disclosure statement

The author is in receipt of no financial gain from the publication of this work, nor from any products emanating from this research.

Abstract

People with aphasia require information about their health. Guidelines highlight the importance of providing information to this population in an accessible format to support their health literacy. Studies suggest, however, that people continue to feel under-informed after stroke, a particularly distressing predicament in the presence of language impairment.

This study aimed to develop a template for accessible information for people with aphasia. In Phase 1, literature regarding modified formatting for people with aphasia, facilitating language processing in aphasia, and information topics needed after stroke was reviewed, and principles for design extracted. These principles informed the prototype template, developed in collaboration with graphic designers.

In Phase 2, fourteen people with a range of aphasia types and severity collaborated in an iterative design process. Participants attended two facilitated focus groups, using Talking Mats[®] to give their views on the template, and on specific issues relating to modified formatting. Discussions were recorded, transcribed, and analysed using Framework analysis (Richie & Spencer, 1994). After each focus group, modifications to the template were made following feedback.

The collaborative, iterative design process resulted in the development of a template for conveying health information for people with aphasia. Analysis of the discussion data generated four themes: 'Visual access', 'The look of it', and 'Understanding', reflected the issues surrounding information access from the perspective of people with aphasia. The overarching theme, 'Everyone's different', captured the varied experience of aphasia for individuals.

The findings informed the final version of the template, including relevant topics, language structures to maximise comprehension of text, optimum number of concepts per page, facilitative features of typography, use of images, and layout of information.

In Phase 3, the template principles were translated into accessible information guidelines, which were user-tested, modified, and subsequently published by The Stroke Association.

Designing for people with aphasia requires attention to the visual, cognitive, conceptual, and language processing demands, and consideration of the experience of aphasia. Further implications for practice and research are outlined.

Contents

Acknowledge	nents	i
Abstract		iii
List of tables.		xiv
Table of figure	2S	xiv
Chapter 1 Ir	troduction and literature review	1
1.1 Intro	oduction to stroke and aphasia	1
1.2 Hea	Ith literacy	2
1.2.1	Systems to support accessibility of health information	3
1.3 Info	rmation after stroke	5
1.3.1	Topics and timing of information after stroke	5
1.3.1.1	Information topics	5
1.3.1.2	Timing of information	6
1.3.2	Information formats after stroke	6
1.3.3	Summary	7
1.4 Aph	asia and information after stroke	8
1.4.1	Aphasia	8
1.4.1.1	Comprehension of language	8
1.4.	1.1.1 Auditory comprehension	8
1.4.	1.1.2 Acquired dyslexia in aphasia	9
1.4.1.2	Language output in aphasia	10
1.4.	1.2.1 Spoken language output	10
1.4.	1.2.2 Written language output	11
1.4.1.3	Other cognitive deficits and aphasia	11
1.4.2	The challenges for people with aphasia and health information	12
1.4.3	Historical background to research into information access for people	with
aphasia		13
1.4.4	The current status of accessible health information for people with a	•
1 4 5	Autoria and baatth literaal	
1.4.5	Aphasia and health literacy	
1.4.6	Iterative design involving people with aphasia	
	rview of the study Study aims	
1.5.1 1.5.2	Study design	
1.5.2	, .	
1.3.2.1	THOSE I MARKEN AND A	

1.5.2.2	Phase 2	20
1.5.2.3	Phase 3	20
1.6 Summ	ary	22
•	se 1: Extraction of principles for accessible formatting from the existing	23
2.1 Introd	uction	23
2.2 Reviev	v of text formatting modifications	23
2.2.1 A	ims of literature review into text formatting modifications	23
2.2.2 Li	terature search methods	24
2.2.2.1	Databases and search terms	24
2.2.2.2	Study inclusion and exclusion criteria	24
2.2.2.3	Results of literature search and selection process	24
2.2.3 R	eviews of selected studies	25
2.2.3.1	Rose, Worrall, & McKenna (2003)	28
2.2.3.2	Ghidella, Murray, Smart, McKenna, & Worrall (2005)	29
2.2.3.3	Brennan, Worrall, & McKenna (2005)	30
2.2.3.4	Dietz, Hux, McKelvey, Beukelman, & Weissling (2009)	31
2.2.3.5	McKelvey, Hux, Dietz, & Beukelman (2010)	32
2.2.4 S	ummary of findings	32
2.2.4.1	Language	32
2.2.4.2	Typography	32
2.2.4.3	Images	33
2.2.4.4	Layout	33
2.2.4.5	Combined formatting features	33
2.2.4.6	Appraisal of the selected studies	33
2.2.5 Ir	nplications for the development of novel materials	35
2.2.5.1	Language	35
2.2.5.2	Typography	36
2.2.5.3	Images	36
2.2.5.4	Layout	37
2.2.5.5	Combined formatting features	37
2.2.6 S	ummary	38
2.3 Review	v of factors affecting language processing in aphasia	39
2.3.1 A	im of literature review into language processing in aphasia	39
2.3.1.1	Methods	39

2.3.2	Findings 1: Lexical factors affecting processing in aphasia	40
2.3.2.1	Psycholinguistic properties of single words	40
2.3.2	1.1 Imageability	40
2.3.2	.1.2 Concreteness	40
2.3.2	1.3 Frequency and familiarity	41
2.3.2	1.4 Word length	41
2.3.2	1.5 Individual variation in aphasia	41
2.3.2	1.6 Effects of grammatical class on processing in aphasia	42
2.3.3	Findings 2: Sentence processing in aphasia	42
2.3.4	Findings 3: Priming	43
2.3.5	Implications for the development of novel materials	44
2.3.5.1	Lexical terms	44
2.3.5.2	Sentence forms	45
2.3.5.3	Priming	45
2.3.6	Summary	45
2.4 Revie	ew of health information topics pertinent to people with aphasia	46
	Aims of literature review into health information topics for people with	47
•	Literature search methods	
2.4.2		
	Findings of the literature search	
2.4.3		
2.4.3.2		
2.4.3.2	Topics identified in the studies	
2.4.3.4	Topics selected for the template	
	Formatting features requiring further evidence	
	tiples for the design of Template 1	
	mary of chapter	
	ase 1: Development of the template	
	blishment of Phase 1 collaborative design team	
	Identification of graphic designers	
	Discussion of principles with the graphic designers	
	eration of text for the template	
	Topics	
	Text content	

3.2.2.1	Lexical terms	. 58
3.2.2.2	Sentence forms	. 58
3.2.2.3	Use of priming	. 59
3.3 Agre	eement with graphic designers on Template 1 design	. 59
3.3.1	Overall layout	. 59
3.3.2	Typography	. 60
3.3.3	Images	. 60
3.3.4	Number of images per concept	. 60
3.3.5	Language content	. 60
3.4 Sum	mary of design principles for the prototype template	. 61
3.5 Prov	vision of content to the graphic design team and Template 1 delivery	. 63
	elopment of materials for exploration of factors identified in the literature ing clear evidence.	
3.6.1	Further evidence for image style	. 63
3.6.2	Further evidence for number of images per concept	. 64
3.6.3	Further evidence for font sizes, font styles, and emphasis of key words	. 64
3.7 Sum	mary of template development	. 64
Chapter 4 P	hase 2: Methodological issues	66
4.1 Invo	lvement of people with aphasia in the development of information	. 66
4.2 Focu	us groups	. 67
4.2.1	Introduction to focus groups as a research method	. 67
4.2.2	Focus groups and people with aphasia	. 68
4.2.3	Consideration of alternative data collection methods	. 70
4.2.4	Summary	. 71
4.3 Met	hods to facilitate the participation of people with aphasia in focus groups .	. 71
4.3.1	Supported communication techniques	. 72
4.3.2	Talking Mats [®]	
4.3.3 in focus ք	Identification of principles for facilitating people with aphasia to participa groups	
4.3.3.1	. Feeling comfortable in the group	. 76
4.3.3.2	Supporting comprehension	. 76
4.3.3.3	Supporting expression of individual views	. 77
4.4 Disc	ussion of suitable methods for the analysis of focus group data	. 79
4.4.1	Background literature and discussion of data analysis methods	. 79
4.4.1.1	Analysis of discussion data	. 79
4.4.1.2	Analysis of the Talking Mats [®] scale	. 81

4.5	Sum	imary of chapter	82
4.5.	.1	Aims of Phase 2	82
Chapter	5 P	hase 2: Methods	84
5.1	Part	icipant details	84
5.1.	.1	Sampling	84
5.1.	.2	Inclusion and exclusion criteria	84
5.1.	.3	Recruitment	85
5.1.	.4	Ethical approval for the study	86
5.1.	.5	Constitution of the focus groups	86
5.1.	.6	Informed consent to participate	86
5.1.	.7	Aphasia profiling of participants	87
5.1.	.8	Overview of the participants	87
5.2	Stuc	dy design	89
5.3	Mat	erials and resources	. 89
5.3.	.1	Communication Support for participants with aphasia	. 89
5	5.3.1.1	Focus group leadership	. 90
5	5.3.1.2	2 Communication Facilitators	. 90
5	5.3.1.3	Communication resources	91
5.3.	.2	Talking Mats [®] materials	91
5	5.3.2.1	The Talking Mat [®] and visual rating scales	91
5	5.3.2.2	Talking Mats [®] practice items	91
5.3.	.3	Template 1 for focus group meeting 1	92
5	5.3.3.1	Variations in Template 1 presented to participants	92
5	5.3.3.2	Materials for exploration of text preferences	92
5.3.	.4	Template 2 for the second focus group meetings	93
5.4	Proc	cedure	93
5.4.	.1	Preparation for the focus groups	93
5.4.	.2	Communication support within the focus groups	94
5.4.	.3	Focus group meetings	95
5	5.4.3.1	Introductions and settling in	95
5	5.4.3.2	Introduction to the Talking Mats [®] framework	95
5	5.4.3.3	Presentation of samples of Template 1 for discussion	95
5	5.4.3.4	Presentation of materials comparing font size, style and emphasis	97
5.4.	.4	Compilation of feedback to the graphic designer 1 (first iteration)	97
5.4.	.5	Second focus group meetings	97

5.4.5.1	Preparation for the second focus group meetings	97
5.4.5.2	Identification of further aims for the second focus group meet	ings 98
5.4.5.3	Materials for the investigation of amount of information and e	•
font pret	ferences	
5.4.5.4	Presentation of samples of Template 2 for discussion	
5.4.5.4	4.1 Review of Template 2 presented as a set of cards	99
5.4.5.4	4.2 Amount of information and emphasis in font	100
5.4.6 C	ompilation of feedback to the graphic designer (second iteratio	n) 100
5.4.7 P	roduction of Template 3	100
5.5 Data r	ecording, preparation and analysis	100
5.5.1 C	Overview	100
5.5.2 A	udio recordings of discussions	101
5.5.2.1	Audio-recorded material	101
5.5.2.2	Transcription of audio recordings	102
5.5.2.3	Data preparation	103
5.5.2.4	Data analysis	103
5.5.2.4	4.1 Familiarisation	103
5.5.2.4	4.2 Identifying a thematic framework	103
5.5.2.4	4.3 Indexing	
5.5.	2.4.3.1 Methods for handling collaborative discussion data	104
5.5.2.4	4.4 Charting	104
5.5.	2.4.4.1 Methods for handling collaborative discussion data in	the analytic
pro	cess	
5.5.2.4	4.5 Mapping and interpretation	105
5.5.2.5	Rigour in the data analysis	105
5.5.3 T	alking Mats® data	106
5.5.3.1	Photographic material generated in the focus groups	106
5.5.3.2	Data preparation	107
5.5.3.3	Data analysis	107
5.5.3.4	Rigour	107
5.6 Chapt	er summary	107
•	, ticipants	
•	uction	
6.2 Langu	age and communication profiles of participants	108
6.2.1 G	iroup A participants	110

6.2.2	Group B participants	111
6.2.3	Group C participants	113
6.3 Sumi	mary	115
Chapter 7 Th	ematic framework	116
7.1 Then	natic framework	116
7.1.1	Development of the framework 1 - Data management	116
7.1.1.1	Identifying a thematic framework	116
7.1.1.2	Indexing	117
7.1.1.3	Charting	119
7.1.2	Development of the framework 2 – Descriptive account	119
7.2 Then	ne 1 - Visual access	121
7.2.1	Sub-theme 1 Seeing clearly	121
7.2.1.1	Size	121
7.2.1.2	Clarity	123
7.2.1.3	Emphasis	124
7.2.2	Sub-theme 2 Amount of information	125
7.2.2.1	Single concepts	125
7.2.2.2	Linked concepts	127
7.2.3	Presentation of Talking Mats [®] data	128
7.2.3.1	Size	128
7.2.3.2	Font style	129
7.2.3.3	Emphasis	129
7.2.3.4	Amount	130
7.2.4	Summary of Theme 1	
7.3 Then	ne 2 – The look of it	
7.3.1	Sub-theme 1 Acceptable	132
7.3.1.1	Acceptable to adults with aphasia	132
7.3.1.2	Acceptable to individuals	133
7.3.2	Sub-theme 2 Design	135
7.3.2.1	Image style	135
7.3.2.2	Colour and font	136
7.3.2.3	Consistent design	137
7.3.3	Presentation of Talking Mats [®] data	138
7.3.3.1	Image style	138
7.3.4	Summary of Theme 2	

7.4 The	me 3 – Understanding the information	140
7.4.1	Sub-theme 1 - The words	140
7.4.1.1	Aphasia and reading	140
7.4.1.2	2 Understanding the words	141
7.4.2	Sub-theme 2 The images	143
7.4.2.1	Impact of image style	143
7.4.2.2	2 Interpreting image meaning	147
7.4.3	Sub-theme 3 Words and images together	150
7.4.3.1	Words and images work together	150
7.4.3.2	2 Number of images per concept	152
7.4.4	Presentation of Talking Mats [®] data	154
7.4.4.1	Number of images per concept	154
7.4.5	Summary of Theme 3	154
7.5 The	me 4 - Everyone's different	155
7.5.1	Sub-theme 1 Experience of stroke and aphasia	156
7.5.1.1	L Knowledge of aphasia	156
7.5.1.2	2 Aphasia is individual	158
7.5.2	Sub-theme 2 Individual responses	159
7.5.2.1	I Identifying with the information	159
7.5.2.2	2 Emotional reactions to the materials	160
7.5.3	Sub-theme 3 Individual visual style	161
7.5.3.1	L Looking at information	162
7.5.3.2	2 Making sense with support	164
7.5.4	Summary of Theme 4	165
7.6 Cha	pter summary	165
Chapter 8 O	outcomes of Phase 2 and design of Template 3	166
8.1 Res	ults of focus groups: feedback to graphic designer	166
8.1.1	First iteration	166
8.1.2	Second iteration	167
	final template	
	pter summary	
	hase three: Development of accessible information guidelines	
	oduction	
	elopment of the draft guideline	
9.2.1	Aims	

9.2.	2	Methods for translating the template criteria into guidelines	. 172
9	.2.2.1	Development of the five step approach	. 172
9.2.	3	Production of the draft guidelines	. 176
9.3	User	-testing of the draft guidelines	. 177
9.3.	1	Introduction	. 177
9.3.	2	Methods	. 177
9	.3.2.1	Participants	. 177
9	.3.2.2	Recruitment	. 177
9	.3.2.3	Ethics and Consent	. 177
9	.3.2.4	Study design	. 178
9	.3.2.5	Materials	. 178
9	.3.2.6	Procedure	. 179
9	.3.2.7	Data management and analysis	. 179
9.3.	3	Results	. 179
9	.3.3.1	Thematic Framework	. 179
T	heme	1 - Need for guidelines	. 180
T	heme	2 - Content	. 182
Т	heme	3 – Accessibility	. 183
T	heme	4 - Usability	. 184
S	umma	ıry	. 186
9	.3.3.2	Results of the qualitative analysis and modifications to the guidelines.	. 187
9.4	Inter	pretation of main findings	. 190
9.5	Chap	ter summary	. 192
Chapter	10	Discussion of findings	193
10.1	Aims	of the study	. 193
10.2	Sum	mary of main findings	. 194
10.3	Desi	gning health information for people with aphasia	. 195
10.3	3.1	Topics	. 196
10.3	3.2	Language	. 197
10.3	3.3	Amount of information per page	. 199
10.3	3.4	Typography	. 201
10.3	3.5	Images	. 203
10.3	3.6	Layout	. 206
10.3	3.7	Acceptability	. 208
10.3	3.8	Summary	. 211

10.4	Indiv	vidual differences	211
10.4	4.1	Individuals with aphasia	212
10.4	4.2	Individual approaches to information content	213
10.4	4.3	Support with accessing information content	214
10.5		hods for involving people with aphasia in developing accessible information in the second second second second	
10.	5.1	Participants	215
10.	5.2	Materials	216
10.	5.3	Focus group methods	217
10.	5.4	Data analysis methods	219
10.6	Qua	lity, rigour and transferability of findings	220
10.0	5.1	Validation of findings	221
10.0	5.2	Transferability of findings	221
10.0	5.3	Reflection on strengths and limitations	222
10.7	Imp	lications for practice and future research	224
10.3	7.1	Implications for practice	225
10.7	7.2	Implications for further research	226
Reference	ces		228
Appendi	ces		248

List of tables

Table 2.1 Studies included in the literature review of modified formatting	. 26
Table 2.2 Summary of findings of facilitative formatting features identified in literature	
review	. 38
Table 2.3 Summary of facilitative features of language derived from language processing	
literature	
Table 2.4 Summary of studies included in the literature overview of topics relevant to	
people with aphasia	. 49
Table 2.5 Information topics required after stroke, classified by ICF (2001) domains	. 52
Table 2.6 Studies contributing to the evidence for topics in each of the four ICF (2001)	
domains	. 52
Table 2.7 Principles derived from the literature for the design of Template 1	. 54
Table 3.3.1 Design principles for Template 1	. 62
Table 4.1 Principles for facilitating participation of people in focus groups	
Table 5.1 Overview of the participants	
Table 5.2 Phase 2 Iterative design process	
Table 5.3 Amount of audio data from each group at each meeting in hours, minutes, and	
seconds	
Table 5.4 Amount of audio data from each topic in each group, in minutes and seconds.	102
Table 5.5 Number of Talking Mats [®] data from each group at each meeting	106
Table 6.1 Language assessment data for all participants	109
Table 7.7.1 Initial thematic framework	117
Table 7.7.2 Initial thematic framework for indexing the data	118
Table 7.7.3 Thematic Framework of the qualitative data	120
Table 7.7.4 Frequency of font size ratings	129
Table 7.7.5 Frequency of font style ratings	129
Table 7.7.6 Frequency of emphasis style ratings 1	130
Table 7.7.7 Frequency of emphasis style ratings 2	130
Table 7.7.8 Frequency of number of concepts per page ratings.	131
Table 7.7.9 Frequency of image style ratings	139
Table 7.7.10 Frequency of ratings for single or multiple images per concept	154
Table.8.1 Final checklist of template design criteria	169
Table 9.1Template criteria to guideline – sentences	173
Table 9.2 Template criteria to guideline – vocabulary	174
Table 9.3 Template criteria to guideline - layout, typography, images	175
Table 9.4 Template criteria to guideline - making a set	
Table 9.5 Thematic framework of the survey data	180
Table 9.6 Changes to the Guidelines document following user-testing	189

Table of figures

Figure 1.1 Summary of Phase 1 and 2 study design	. 21
Figure 3.1 Example showing application of the agreed design principles into Template 1 .	. 61

Chapter 1 Introduction and literature review

The availability of accessible information for people with aphasia is an essential component of healthcare provision after a stroke (Intercollegiate Stroke Working Party, 2016). This chapter introduces background and contemporary literature and developments pertinent to the field of accessible information for people with aphasia, with the aim of setting the context for the thesis.

1.1 Introduction to stroke and aphasia

Stroke is a complex medical condition with multiple psychological, cognitive, language, and social consequences. The total incidence of stroke in the UK is over 100,000 new cases per year (Stroke Association, 2017). In the general population, the prevalence of stroke is reported as 1.5 - 1.75% (Mant, Wade, & Winner, 2004), translated by the Stroke Association as representing 1.2 million people who have survived a stroke. Of these people, approximately 50% live with ongoing physical, cognitive, and linguistic disabilities, and approximately 33% experience post-stroke depression (Stroke Association, 2017).

Aphasia is an acquired communication disorder caused by focal damage to the languagedominant hemisphere of the brain, affecting approximately one third of people following a stroke. Aphasia occurs in 52% of all stroke cases at the acute stage, 29% at three months, and persists beyond six months after stroke in 15% of cases (Mant et al., 2004). Currently in the UK, approximately 350,000 people are estimated to have aphasia resulting from stroke (Stroke Association, 2017).

Aphasia can affect all modalities of language function, including auditory comprehension, spoken language output, reading, writing, and gesture, with implications for people's ability to communicate functionally, and to participate in their family and social environments. Despite the use of a single diagnostic term, aphasia varies widely, reflecting overall stroke and aphasia severity, individual patterns of language impairment, and related stroke consequences such as cognitive impairment.

The impact of aphasia is extensive and enduring, both for those directly affected, and for their families and friends (Avent et al., 2005). It is associated with reduced independence (Cruice, Worrall, & Hickson, 2011), and, overall, impacts more severely on people's quality of life than stroke without aphasia (Hilari, 2011). People with aphasia are more likely to be depressed than their non-aphasic peers (Hilari, Needle, & Harrison, 2012), and have higher levels of stress

(Stroke Association, 2013). Aphasia can also lead to a reduction in quantity and quality of social activity, including loss of friendships (Northcott & Hilari, 2011), and this is worse for those with severe aphasia (Parr, 2007).

Recent research has endeavoured to show positive approaches taken by individuals with aphasia, revealing acceptance and resilience to the challenges of living with aphasia (e.g. Bright, Kayes, McCann, & McPherson, 2013; Brown, Worrall, Davidson, & Howe, 2010; Cruice, Hill, Worrall, & Hickson, 2010; Worrall et al., 2010). The reciprocal benefits to both individuals and services have been demonstrated by studies reporting a partnership approach with people with aphasia, in which the expertise of those living with the condition to inform and educate is valued (e.g. Swart & Horton, 2015).

1.2 Health literacy

Health literacy refers to the understanding a person has of their health, through a process of obtaining, understanding, and using health information (Kickbusch, Pelikan, Apfel, & Tsouros, 2013). It is based on a body of evidence that has demonstrated that people who are informed are better able to make wise decisions about their health, and to develop resilience in the face of ill-health or disability (Nutbeam, 2000).

Access to health information has been demonstrated to benefit patients' ability to understand their condition (Coulter & Ellins, 2006; McPherson, Higginson, & Hearn, 2001) and their ability to interact with staff who provide health services (Coulter, Entwistle, & Gilbert, 1999). Betterinformed patients have been found to follow treatment plans more assiduously (Mansoor & Dowse, 2003), to be more productively involved in decisions about their care (Stacey, Légaré, Col, et al., 2014), and to be more autonomous and able to engage in self-management (Murray, Burns, See, Lai, & Nazareth, 2005). Reduced anxiety (Humphris, Ireland, & Field, 2001) and increased satisfaction with services (Tomkins, Siyambalapitiya, & Worrall, 2013) are also associated with access to health information. Involving service users in developing health information relevant to their condition has resulted in better uptake, with information more appropriately targeted in terms of both content and format (Bunge, Muehlhauser, & Steckelberg, 2010).

Nutbeam (2000) identified three levels of health literacy, representing the skills needed to engage with health information and services, a model which has been adopted in the UK by the Royal College of General Practitioners (Rowlands, Protheroe, Price, Gann, & Rafi, 2014). The first level, functional health literacy, refers to understanding basic health information; the second, interactive health literacy, entails being able to discuss the information in order to ask questions and make decisions. The third level, critical health literacy, involves selfmanagement, taking control of one's own health. Inherent in this definition is a reciprocal responsibility between individuals and service providers: individuals to engage with their health, with information and services, and service providers to make information and services accessible to all service users, for example through the provision of accessible information (Paasche-Orlow, McCaffery, & Wolf, 2009).

Access to health information to support all levels of health literacy is vital, since research identifies that poorer health outcomes are associated with low health literacy (Berkman, Sheridan, Donahue, Halpern, & Crotty, 2011; Magasi, Durkin, Wolf, & Deutsch, 2009; Serper et al., 2014; Wolf, Gazmararian, & Baker, 2005). Interventions to improve health literacy are described at the level of countries, communities and individuals, addressing the needs of both whole populations and groups, and individuals within their specific contexts. However, as Pleasant, McKinney, and Rikard (2011) note, measuring health literacy poses significant challenges, since it is a broad concept, which may vary within and between individuals, dependent on context. They propose that measurement should include objective assessment of the different aspects of accessing, understanding, appraising, and using health information, as well as literacy and numeracy.

The growing evidence for the impact of health literacy on healthcare outcomes is represented by a significant body of literature and guidance for healthcare staff regarding effective communication with patients (e.g. Ley, 1988; Street, 2009) and the production of written information for patients (e.g. Duman, 2003; Hoffmann & Worrall, 2004). Research has addressed the information formatting needs of specific groups, for example people with low literacy (e.g. Doak, Doak, & Root, 1996), adults with learning disabilities (e.g. Rodgers & Namaganda, 2005), and people with aphasia (e.g. Rose, Worrall, & McKenna, 2003).

Coulter, Entwistle, and Gilbert (1998) identified the need to include people affected by a health condition in developing information, and this principle is carried forward in guidance aimed at health care professionals producing information for their patient groups (e.g. Duman, 2003; Department of Health, 2003). Within these guidance documents, readers are advised to consider the needs of those with intellectual or sensory disabilities, such as people with a learning disability, hearing impairment, or low vision, but the specific disabling effects of acquired language disorders are not represented.

1.2.1 Systems to support accessibility of health information

Health literacy is a moral imperative, underpinned by concepts of human rights, accessibility, inclusion, empowerment, and equality of opportunity. Such principles are embedded in the

aims of the United Nations, which provides guidance to member states on their legal obligations in relation to the rights of citizens, including those with disabilities (Convention on the Rights of Persons with Disabilities, 2008). The World Health Organisation (2011) identifies the right to health as a basic human right, with accessibility, including information accessibility, as one of its core principles.

As a human right and as a prerequisite to health literacy, accessibility of health information forms part of public health strategy and is included within wider policies in the USA and the UK. In the United States, the U.S. National Action Plan to Improve Health Literacy (2010) includes the target of providing health information which can be understood by all people.

In the United Kingdom, several legal and policy systems are now in place to support individuals, regardless of disability, to take an active role in their own health. Firstly, the Equality Act (2010) provides a legal framework to protect the rights of people with so-called "protected characteristics" (p.4) including those with communication disabilities. Secondly, the UK government's strategic white paper "Equity and excellence: Liberating the NHS" (DOH, 2010) articulates an aim for all patients to have access to information and be involved in all decisions about their care in order to take responsibility for their health. Finally, the recently launched NHS England Accessible Information Standard (2015) supports this aim, requiring health and social care organisations to provide communication support and appropriately accessible information to any adult with a communication disability. Compliance with this standard requires services systematically to identify, record, highlight, and share the communication needs of their individual service users. The standard primarily aims to support individuals with long-term communication disability arising from a sensory loss, such as hearing or visual impairment, or a learning disability, but those with or other, less prevalent impairments, such as aphasia, are included. The scope of the standard does not extend to people with low literacy, dyslexia or English as a Second Language. Nor does it encompass support for decision-making, which falls within the scope of the Mental Capacity Act (Office of Public Sector Information, 2005), or aspects of the physical environment, such as signage in public buildings.

As Bunning and Horton (2007) propose, there are commonalities in the experience of having a communication disability, which this standard now recognises. However, to address specific needs appropriately, modifications to information must be suitable for individuals, with specific alterations to communication and information tailored to the needs of each group. Currently the standard mandates the provision of accessible information, but no evidence-based guidance for the different groups within its scope is provided. As noted by Rafi, Sullivan, and Mathers (2016), specific resources to support the diverse needs of different groups of

people affected by a wide range of communication impairments, including, for example, those with learning disabilities, sensory impairments, dementia, or aphasia, will need to be made available to organisations providing health and social care to achieve the requirements of the Accessible Information Standard.

1.3 Information after stroke

In this section, the information needs of people after a stroke will be outlined, in terms of information required, optimal timing of information provision, and the range of formats used. The research literature on this topic relates to the population of people after a stroke in broad terms without specifying the nature of their impairments. For example, some studies have included people with aphasia, but their specific contributions are not separated out in the reporting of findings.

1.3.1 Topics and timing of information after stroke

In the United Kingdom, the period following a stroke may involve emergency admission to hospital for diagnosis and hyper-acute medical care, ongoing acute medical treatment, and early hospital rehabilitation. Hospital care is usually followed by transfer home, with community rehabilitation services (Intercollegiate Stroke Working Party, 2012). Progression through these different phases of care in hospital and at home will vary, determined by individual needs and local service provision. The disabling consequences of stroke are life-long in many cases, persisting well beyond the time allocated by care providers. Throughout the aftermath of a stroke, people need information. The importance of information is recognised in UK national Stroke guidelines, which specify that all patients and their families should receive information tailored to their individual needs and formatted in a way they can understand at each stage of the stroke care pathway (Intercollegiate Stroke Working Party, 2012:62; National Institute for Health and Care Excellence (NICE), 2013:26).

1.3.1.1 Information topics

Specific information needs following a stroke are wide-ranging, since the consequences of stroke can be both varied and long-lasting. A systematic review of the education needs reported by patients and carers after stroke by Hafsteinsdottir, Vergunst, Lindeman, & Schuurmans (2011) reflects the findings of 21 selected studies, identifying a range of topics. These include medical concerns such as the cause of stroke, its diagnosis, prognosis, and treatment, and how to prevent further strokes. Information is also needed to address practical concerns, such as dealing with medication and supporting rehabilitation. The need to be informed about what to expect and how to help with psychological and emotional issues, such

as fatigue and depression, and cognitive issues such as memory and concentration, are identified. Finally, information is required relating to finance and benefits, access to services and social activities, driving after stroke, and return to work.

1.3.1.2 Timing of information

People report a need for different types of information at different stages after a stroke, and this has been found to be of most value when it is individualised, personally relevant information, rather than generic (Wiles, Pain, Buckland, & McLellan, 1998). Linked with this, the timing of such information has been found to be an important factor, with people reporting that they want to receive information at a time when they are able to process it (Eames, Hoffmann, Worrall, & Read, 2010). The need for information to be repeated is also highlighted in the literature, to provide multiple opportunities for people to apply the information to their current circumstances (Danzl et al., 2016).

Whilst the literature reports preferences for information to be provided responsively and individually, some studies also report participants expressing dissatisfaction with services where information was not provided systematically (e.g. Danzl et al., 2016; Eames et al., 2010). As Eames et al. observe, this reflects a tension in the expectations of patients and family members, and provides a challenge for healthcare providers seeking to answer the questions that patients and families ask, as well as educating them in the areas where they do not yet have awareness.

1.3.2 Information formats after stroke

In clinical practice in the developed world, information after stroke is provided by healthcare services in a range of different ways, and the literature in this area reflects its multi-faceted nature. The formats described in the literature include those which are provided as an intervention, such as written information packages (e.g. Mant, Carter, Wade, & Winner, 1998) or education sessions covering predetermined topics (e.g. Forster et al., 2012b). The format of such provision may include, for example, stroke information leaflets, posters, or web-based information, and the use of individual and group teaching methods.

Verbal provision of information also arises within the context of service provision, during, for example, consultations, daily care, or rehabilitation sessions. Such interactive information provision may be conceptualised as part of communication, and is not generally reflected in the information as intervention studies, but is captured by studies recording the experiences of people after stroke. For example, as Eames et al. (2010) found, the reported experience of patients and family members reflects the whole, rather than separate elements of their care.

And as Anderson and Marlett (2004) argue, the skilful use of communication during stroke rehabilitation can serve to inform patients and family members in a positive way.

Effectiveness of information provision depends on individual preferences and needs. Forster, Smith, and Young (2012a) found that an active style, in which people could check their understanding by asking questions, was more effective when aiming to increase knowledge about stroke. This echoes the findings of preference surveys (e.g. Eames, Hoffmann, Worrall, & Read, 2011; Wachters-Kaufmann, Schuling, The, & Meyboom-de Jong, 2005) which showed that people valued time and attention to their personal concerns.

To accommodate individual preferences, research has explored the use of information packages which provide a bespoke combination of different formats. For example, Eames, Hoffmann, Worrall, Read, & Wong (2013) found increased satisfaction with information provision and greater ability to seek information independently among people who received a tailored package of computer-generated written information and the opportunity for verbal follow-up.

Across the literature, people express a preference for a combination of verbal information, backed up with written resources, which they may keep for reference (Eames et al., 2011). Written information is also valued as a support for talking things through during face-to-face consultations (Wachters-Kaufmann et al., 2005). This written information is essential, since as participants in the study by Eames et al. (2011) reported, memory for verbal information alone is vulnerable: "[They] may have told me but [the information has] gone in and out...didn't register." (p.75).

Despite the reported need to provide written information, however, the literature suggests that there may be a tendency for healthcare providers to give only verbal information (Eames, Hoffmann, McKenna, & Worrall, 2008; Hoffmann, McKenna, Herd, & Wearing, 2007). This may relate to a lack of suitable materials for healthcare staff to provide. In addition, patients and families may discard written materials that they find unsuitable for them.

1.3.3 Summary

Information provision after stroke has been the focus of a growing volume of research, providing valuable evidence regarding the topics, timing, and formats of information needed by patients and their families. The literature tells, however, of longstanding concerns expressed by both healthcare providers and stroke survivors and their families, that information provision remains inadequate (Hanger, Walker, Paterson, McBride, & Sainsbury, 1998; Hanger & Wilkinson, 2001; Rodgers, Bond, & Curless, 2001). The Stroke Association (McKevitt et al., 2011) report that 54% of people were dissatisfied with the amount of information they received after their stroke. The needs of people post-stroke with retained language and communication skills are therefore not being fully met. People with aphasia after stroke are likely to be further disadvantaged with regard to information provision, and indeed Brady, Fredrick, and Williams (2013) report that few studies have attended to the specific needs of those with language impairment after stroke.

1.4 Aphasia and information after stroke

People with aphasia after a stroke have the same need for information as those with stroke without aphasia, and in addition need information about aphasia. Impairments to language processing, including reading, present significant barriers to access to information for those with aphasia, their families, and healthcare providers. In this section, language processing in aphasia will first be described, then related to the issue of health information, including health literacy.

1.4.1 Aphasia

Aphasia is an acquired disorder of language processing which may affect speaking, understanding, reading, and writing. It affects individuals differently in terms of language modalities impaired and functional communication ability, and it varies in severity. Improvements can occur over time, but it is viewed as a chronic condition, with enduring effects on language and communication. The characteristics of aphasia are manifold, and a detailed description is beyond the scope of this thesis. The following section is provided to orientate the reader to the range of difficulties faced by people with aphasia when accessing spoken or written information.

1.4.1.1 Comprehension of language

Aphasia may affect comprehension of spoken and written language, and may vary between individuals depending on the type of stroke and extent of the lesion (Damasio, 1981).

1.4.1.1.1 Auditory comprehension

Auditory comprehension deficits are common in aphasia and these are manifested in terms of lexical comprehension, semantic impairments, and sentence processing. Syndromes such as Wernicke's aphasia are particularly associated with impairments in auditory comprehension, in which individuals may have difficulty processing the sounds in words or accessing the word meaning. In healthcare, people with aphasia may have difficulty understanding information due to lexical, semantic, and sentence comprehension impairments. For example, the vocabulary used to explain the details of stroke and its treatment may be particularly challenging in the presence of aphasia, since medical terms are frequently long and unfamiliar. Technical medical terms, describing specialist procedures or medication, for example, may be difficult to comprehend. Medical vocabulary may also be new to someone who has no previous experience of stroke; for example, terms such as 'aphasia' or 'dysphagia' may have no meaning for them. Terms may be used to convey uncertain or previously unknown concepts, such as possible recovery, future care needs, or services. Terms such as 'rehabilitation' may have different semantic associations, presenting a difficulty in comprehending what such a term might mean in the person's current situation. Within stroke care, metaphorical language is used, for example 'stroke journey', 'stroke pathway', 'goals', presenting possible problems for people with aphasia, who may interpret these terms literally (lanni, Cardillo, McQuire, & Chatterjee, 2014).

Impairments to sentence comprehension can result in difficulties understanding health information, since healthcare staff may explain complex concepts using correspondingly complex language structures. Overall, sentence complexity is a predictor of difficulty, with more non-canonical and complex structures being harder to understand (Caplan, Waters, DeDe, Michaud, & Reddy, 2007). Trying to comprehend information presented verbally may be challenging for someone with aphasia, particularly if no accommodation is made for their syntactic impairment.

1.4.1.1.2 Acquired dyslexia in aphasia

Of particular importance to people's access to information in healthcare settings is the fact that most people with aphasia present with acquired deficit in reading. According to theories of reading comprehension (e.g. Marshall & Newcombe, 1973) there are several stages in access to written word meaning, with damage to any stage resulting in reading comprehension difficulties.

Sentence reading requires an ability to understand the individual words, as outlined above. In addition, use of syntactic rules and the assignation of semantic roles to the different parts of the sentence is needed to derive the propositional meaning (Ellis, 1984, reprinted 1989), with challenges mirroring those of spoken sentence comprehension. Text reading has recently attracted more interest within aphasiology, found to require additional cognitive skills such as working memory and making inferences (e.g. Chesneau & Ska, 2015; Meteyard, Bruce, Edmundson, & Oakhill, 2015).

Impairments in visual orthographic analysis result in peripheral dyslexias. For example, neglect dyslexia and attentional dyslexia result in errors due to visual processing of letters at the initial or final position, or due to intrusion of letters from neighbouring words. Letter-by-letter reading involves sequential single letter reading, without these being integrated and recognised as a whole word unit. In visual dyslexia, words are mistaken for visually similar word forms. Peripheral dyslexias may or may not co-occur with aphasia; in contrast, central dyslexias which arise from lexical and semantic levels of processing are more usually associated with aphasia (Riley & Kendall, 2013).

Written information materials may present difficulties for reading at any of the levels outlined above. As with spoken information, the vocabulary choices may be challenging for people with aphasia if medical terminology is used without explanation. Similarly, use of syntactically complex forms, such as passive forms or long sentences, may be difficult to read. Challenges to memory and visual processing may be presented if materials are lengthy or poorly reproduced. Since written language is used in a variety of information media, difficulties with reading can affect people's ability to understand such formats as patient information leaflets, public posters and displays, and web-based information.

1.4.1.2 Language output in aphasia

1.4.1.2.1 Spoken language output

Word finding difficulties are one of the most prevalent features of aphasia (Hickin, Best, Herbert, Howard, & Osborne, 2002) and range in severity and type from non-retrieval to occasional delays to overt errors. Such errors may be related to the target in terms of form (e.g. word length), sound, or meaning.

Sentence production requires the conceptualisation of an event, the creation of an argument structure, and the selection of the correct syntactic frame (e.g. Garrett, 1980). Individuals with aphasia may have specific difficulties generating syntactic structures, or mapping the argument structure to the sentence form (Marshall, 2013), resulting in reduced fluency and a reduced range of structures available to them.

In the context of information exchanges, individuals may therefore have difficulty retrieving the vocabulary and sentence structures necessary to identify their concerns or to ask specific questions about their health. Communication partners may also fail to interpret the intended meaning of the speaker with aphasia, due to errors in production by the person with aphasia.

1.4.1.2.2 Written language output

In addition to impaired motor skills resulting from stroke, difficulties with writing may mirror impairments in spoken language output. At the single word level, difficulties arise with word retrieval and spelling; beyond single words, people might experience problems with syntax and text level writing. Such difficulties affect people's ability to operate independently in the realm of written language, for example writing letters, using email, texting, or completing forms.

1.4.1.3 Other cognitive deficits and aphasia

Cognitive changes after stroke are widely reported, with impairments to individual or multiple cognitive domains including attention, memory, visuospatial processing, language, and executive function (Helm-Estabrooks, 2002), any or all of which may affect individual's ability to interact with information. Cognitive impairment after stroke is not limited to the acute stages, suggesting that ongoing specific difficulties may combine with language impairments to hinder access to information (Cumming, Marshall, & Lazar, 2013). As stated above, visual processing deficits, including hemianopia and peripheral dyslexias, can affect ability to process written information formats.

Traditional accounts of aphasia have emphasised the linguistic nature of the condition (e.g. Edelman & Greenwood, 1992). Recent investigations, however, indicate that aphasia can cooccur with impairments to other cognitive functions such as attention (Heuer & Hallowell, 2015; Murray, 2012), executive function (Frankel, Penn, & Ormond-Brown, 2007), and shortterm memory (Kalbe, Reinhold, Brand, Markowitsch, & Kessler, 2005). The presence of such cognitive deficits, combined with the purely linguistic impairments evidenced in acquired dyslexia, can be seen to present additional challenges to accessing written information formats in healthcare.

Reading is a cognitively challenging activity, requiring a combination of visual skills and selective and sustained attention, to focus on the text; working and short-term memory, to link sections of text and draw inferences; and self-monitoring, to ensure continued focus on the task (Rayner & Pollatsek, 1989). Memory of the text, both in the short and long term, is needed in order to retain information content. According to Alvarez and Emory (2006), these basic cognitive skills underpin higher order cognitive processes involved with planning and problem solving, encompassed by the term executive function, skills which are also used when approaching a task such as reading in a strategic manner. Strategic approaches to reading employed by people with aphasia have been found to influence reading comprehension and enjoyment (Lynch, Damico, Abendroth, & Nelson, 2013). For example, self-pacing (Caplan et

al., 2007) results in better comprehension than fast reading, but depends on an individual having adequate executive function control to use the strategy.

Given the above potential language and other cognitive processing difficulties which may present in aphasia, it is unsurprising that people with aphasia have difficulties accessing written and spoken information. The specific difficulties faced by people with aphasia will be further considered below.

1.4.2 The challenges for people with aphasia and health information

Information is provided in different formats in stroke care, as discussed in section 1.3.2 above, but all depend on spoken and written language, and hence are inherently demanding for people with aphasia. These are lasting challenges in many cases, as aphasia is often a lifelong condition.

An early challenge for understanding information relates to the sudden onset of stroke, after which people may find themselves unable to understand others, speak, read, or write, a situation which people have reported as finding frightening (Parr, Byng, Gilpin, & Ireland, 1997). In such stressful and unwanted situations, unfamiliar and conceptually difficult information can be particularly challenging, and studies have reported people feeling too overwhelmed to be able to absorb new information (Eames et al., 2010).

Public awareness of aphasia is low (McCann, Tunnicliffe, & Anderson, 2013; Simmons-Mackie, Code, Armstrong, Stiegler, & Elman, 2002), therefore developing an understanding of aphasia in itself presents a challenge to new patients and families, who may well have had no previous knowledge of the condition.

The skills of healthcare staff play an important role in supporting people with aphasia to access information during such interactions. National guidelines (Intercollegiate Stroke Working Party, 2012) indicate the need for all staff to be aware of aphasia, and trained in the practical strategies to support interactions. Such training has been demonstrated to have a positive influence, for example Carling-Rowland, Black, McDonald, & Kagan (2014) report the increased reliability of social workers when assessing the mental capacity of people with aphasia after practical training.

Guidelines produced for the general population are not always appropriate for the needs of people with aphasia. For example, Duman (2003) proposes the use of question forms, to encourage engagement with the materials. This technique may, however, be a barrier to understanding for people with aphasia, given that non-canonical forms are known to be more difficult to comprehend. In sum, information may be inaccessible due to a combination of the constellation of impairments associated with aphasia, the nature, formats, and timing of the information, and the environment, including the communication skills of those around the individual.

1.4.3 Historical background to research into information access for people with aphasia

The social model of disability understands disability as it is constructed within wider society (Oliver, 1996). Applied to aphasia, it removes the focus of the impairment from the individual, and relocates it within the interactions that the individual has in the society they live in, including with other people and in the wider environment (Byng & Duchan, 2005). With regard to access to health information, the skills and knowledge of healthcare providers, and the materials used to provide information, are critical factors in enabling people with aphasia to access information, and so they are the focus of the research in this field.

Developed principally in the USA, Canada, and the UK from the early 1990s, the social model of disability informed developmental work into communication environments and communication access. The literature surrounding accessible information for people with aphasia has its origins in this important body of work.

Within the area of everyday conversation, Kagan and Gailey (1993) introduced the concept of "communicative accessibility" (p.204). They developed the idea of training conversation partners to adopt specific communication strategies to assist people with aphasia to participate in conversational interactions. Kagan and Kimelman's (1995) investigation of aphasia and informed consent introduced the concept of communication access, posing the question "do we provide subjects with information in an accessible format?" (p.67). Through this work, they identified that written information could be modified to maximise the reading comprehension of participants with aphasia.

Further research into the lived experience of aphasia took place subsequently. In the UK, Parr et al. (1997) investigated the experience of aphasia from the perspective of those living with the condition through in-depth interviews with fifty people with chronic aphasia. Eight overarching themes identified through Framework analysis characterised the complexity of the individual and shared experience of aphasia. One of these themes, *"Everything seems a secret*': information and aphasia" (p.87) highlights the difficulties experienced by people with aphasia in accessing information after their stroke. In health literacy terms, the study found that both obtaining and understanding information were problematic for people with aphasia, demonstrating powerfully the need for healthcare providers to provide accessible information

to people with aphasia after stroke. These findings provided a categorically negative response to the question posed by Kagan and Kimelman two years previously.

As a result of Parr et al.'s (1997) findings, subsequent work involved people with aphasia in developing information resources. Parr, Pound, Byng, & Long (1999) involved people with aphasia in the development of an information resource about aphasia. This was a ground-breaking achievement due to both the involvement of people with aphasia in its production and the novel presentation, which included images as well as text, use of bold font, and colour-coding to delineate different topics. Rose et al. (2003) later described this formatting as "aphasia-friendly principles: (1) use of simple words and short sentences; (2) use of large and standard font; (3) use of white space; and (4) use of relevant pictures." (p.950).

Later work at the charity Connect, in the UK, continued to use these principles whilst developing a model for including people with aphasia in the operational management of the organisation (Parr, Pound, & Hewitt, 2006; Pound, Duchan, Penman, Hewitt, & Parr, 2007). Working in collaboration with people with aphasia, these authors also produced guidance for good practice in involving people with aphasia in research (e.g. Swinburn, Parr, & Pound, 2007) and training resources to educate organisations about aphasia (e.g. Parr, Wimborne, Hewitt, & Pound, 2008). 'Aphasia-friendly' principles are promoted in these resources as part of the overall aim of maximising communication access for people with aphasia. The principles underlying aphasia-accessible formatting are stated in very broad terms only, however, raising a risk of mis-interpretation by consumers in their use of the guidance to produce new information materials for people with aphasia.

The broadening scope of aphasia researchers and clinicians working within a social rather than a medical model of disability (e.g. Kagan & LeBlanc, 2002; Pound, Parr, Lindsay, & Woolf, 2000) is consistent with the World Health Organisation ICF (2001) model, which seeks to provide a means of describing the health of individuals within their personal context. Of particular importance in this discussion is the concept of the environment, which can be classed as either enabling or disabling, due to the presence of either facilitators (factors which improve functioning) or barriers (which limit functioning). According to the ICF (2001) classification, the environment includes any aspect of the physical environment, people (including their skills and attitudes), and the services (including systems and policies) which an individual may access.

Using this ICF (2001) model, Howe, Worrall, & Hickson (2004) identified the concept of an "aphasia-friendly environment" (p.1015), in which facilitators to participation are maximised and barriers minimised, and they reviewed research which addressed the different aspects of the environment listed by the ICF (2001). They found limited specific evidence about how to

modify environments for people with aphasia, calling for further research to both understand the issues from the perspective of those with aphasia, and to specify the barriers and facilitators across all areas, including the modification of written information. A later review using the same framework, with reference to the acute hospital environment, by O'Halloran, Hickson, & Worrall (2008), found evidence that the knowledge, skills and attitudes of healthcare staff significantly affected the experience of people with aphasia whilst in hospital, and this was generally reported from a negative perspective.

Within the field of written information, Worrall, Rose, Howe, McKenna, & Hickson (2007) discuss accessibility of written information, calling for further research into the ways of formatting written information for people with aphasia. Since the integration of the social model and the ICF into aphasia care, however, and the conceptualisation of aphasia-friendly environments and formatting, relatively little research has been conducted to examine the impact of this approach on people with aphasia.

A number of studies have investigated the impact of modifying written information for people with aphasia (Brennan, Worrall, & McKenna, 2005; Dietz, Hux, McKelvey, Beukelman, & Weissling, 2009; Ghidella, Murray, Smart, McKenna, & Worrall, 2005; McKelvey, Hux, Dietz, & Beukelman, 2010; Rose, Worrall, Hickson, & Hoffmann, 2011a, 2011b, 2012; Wilson & Read, 2016). Such research has taken a largely positivist approach, seeking to establish clear evidence for specific aspects of formatting which are either preferred by people with aphasia, or beneficial to comprehension. Only one study (Rose et al., 2003) has demonstrated that the use of combined modified formatting of written information can facilitate the comprehension of health information in people with aphasia.

The body of work surrounding communication access to written information by people with aphasia has led to an apparent perception amongst healthcare staff in the UK and elsewhere that effective and evidence-based principles have now been established. Guidelines have now been produced for making information accessible to people with aphasia (Herbert, Haw, Brown, Gregory, & Brumfitt, 2012), representing an increasing resource available to healthcare staff. These were produced in collaboration with people with aphasia, but their effectiveness has not been rigorously tested.

Materials are now being published, using methods of formatting described in the literature, which have been assumed effective in ensuring accessibility to people with aphasia (e.g. Carling-Rowland et al., 2014; Jayes & Palmer, 2014; Pearl, 2014). The term 'aphasia-friendly' is widely used in both research and clinical practice, suggesting that there is agreement about what this means and how it may be delivered. It is not the case, however, that effective methods for providing access to information for people with aphasia have been firmly established.

1.4.4 The current status of accessible health information for people with aphasia

Recent research into access to information suggests that people with aphasia want information (Rose, Worrall, Hickson, & Hoffmann, 2010; Worrall et al., 2010; Worrall et al., 2011) and report higher levels of satisfaction and support when they feel informed (Hilari & Northcott, 2006; Tomkins et al., 2013). A recent survey into how people with aphasia obtain information (Hinckley, Hasselkus, & Ganzfried, 2013) found, however, that people with aphasia continue to report that they do not receive the information they need after a stroke, suggesting that there has been little improvement since Parr et al.'s (1997) study.

The research suggests that people with aphasia experience greater difficulties obtaining and understanding information (Cottrell & Davies, 2004; Eames et al., 2010) and are less likely to receive written information than other stroke patients (Eames, McKenna, Worrall, & Read, 2003; Knight, Worrall, & Rose, 2006). When information is provided, it is frequently written in ways which are not accessible (Aleligay, Worrall, & Rose, 2008; Eames et al., 2003; Rose, Worrall, McKenna, Hickson, & Hoffmann, 2009). There is some evidence that people with aphasia are not treated as independent consumers of health care. For example, Knight et al. (2006) observed that in an acute stroke setting, people with aphasia were given information only when a family member was present.

The provision of information accessible to individuals is needed from the outset of their stroke and into the longer term (Parr et al., 1997). However, as Eames et al. (2008) found, less than half the 119 community stroke services they surveyed in Australia considered the specific information needs of people with aphasia, suggesting that there remains a continuing lack of awareness and skill in this area. There are also several reports of a lack of supported or accessible information to facilitate people living independently (Brown et al., 2010; Hilari & Northcott, 2006; Hinckley et al., 2013; Parr, 2007). This means that people are less likely to receive information about services or activities, fuelling a cycle of social isolation and poor health-related quality of life outcomes, as described by Hilari et al. (2012).

Kagan and Kimelman (1995) first asked: "do we provide subjects with information in an accessible format?" (p.67). Parr et al.'s (1997) account suggested that the answer was no. Following their findings, significant developments in the concept of communication access were made, and from this an accumulation of work has led to an acceptance of the concept of 'aphasia-friendly' information. However, this work has not yet had a significant impact on enabling people with aphasia to feel informed.

1.4.5 Aphasia and health literacy

Aphasia can be viewed as a barrier to health literacy, since it has a direct and negative effect on obtaining, understanding, and using health information. UK Stroke guidelines and wider policies include consideration of those affected by aphasia, specifying that information should be provided in accessible formats, attending to individual communication needs (NHS England, 2015; Intercollegiate Stroke Working Party, 2012; NICE, 2013). Guidance has now been provided (e.g. Herbert et al., 2012), but the reported experiences of people with aphasia do not yet suggest there has been an impact on the health information they receive. Two factors may account for this. Firstly, research into the social consequences of aphasia reports the low public awareness of aphasia (McCann et al., 2013; Simmons-Mackie et al., 2002), with a consequence that there is a lack of awareness of the needs of those with aphasia. As Parr (2007) noted, the Disability Discrimination Act (1995) made provision for people with a communication disability, but lack of awareness of the condition remained a barrier to its proper implementation. Secondly, translating guidance into good practice is a complex process, requiring both time and resources to ensure successful implementation (e.g. Graham et al., 2006). It is likely that making published guidelines (e.g. Herbert et al., 2012) available in isolation not sufficient, and that service providers require training to increase both their awareness of aphasia, and their knowledge of accessible formatting for this population.

1.4.6 Iterative design involving people with aphasia

Supporting the health literacy of people with aphasia is complex. In order to address the concerns outlined, further details of both accessible formatting modifications and the responses of people with aphasia to modified materials is needed. To achieve this aim, one possible option is to involve people with aphasia as collaborators in the design of materials in which their views determine the resulting design. Such an approach is consistent with the concept of "inclusive design" (Cruice, 2007:6), whereby people affected by a condition contribute to the design of products or environments which they are likely to use.

Previous work in the area of accessible information has involved people with aphasia either as active collaborators in an organisational context in developing new information resources (e.g. Parr et al., 1999, 2004; Cottrell & Davies, 2006), or as research participants responding either to controlled stimuli (e.g. Brennan et al., 2005), to surveys (e.g. Ghidella et al., 2005), or to interviews (e.g. Rose et al., 2011a). None of these approaches has resulted in the dissemination of clear findings which can be used when producing new accessible information.

An iterative design process provides clarity in the steps taken towards the final design output. As described by Sears and Lund (1997), this process follows several stages, depending on the complexity of the target design. In the first stages, a prototype design is developed, based on the needs of, and tested with, the target users. In later stages, the design is refined and retested in a cyclical process.

In recent years, several studies have reported the process of co-design and participatory design with people with aphasia, towards a variety of aims. Such studies include the development of novel therapy tools (Galliers et al., 2012; Wilson et al., 2015), modified social networking and email platforms (Al Mahmud & Martens, 2013; Miller, Buhr, Johnson, & Hoepner, 2013), and devices to support communication (Al Mahmud, Limpens, & Martens, 2013; Moffatt, McGrenere, Purves, & Klawe, 2004).

Whilst these studies vary in their aims, they all seek to create new designs through genuine collaboration between the "end-users" of the product (Wilson et al., 2015:21), design professionals, and clinical researchers. The sharing of expertise between collaborators is viewed as central to this process. The meaningful involvement of people with aphasia in such a process necessitates consideration of how to engage people so that they can understand the aims and objectives and contribute their opinions.

1.5 Overview of the study

This thesis reports the methods and findings of a project which took place between October 2010 and July 2011.

1.5.1 Study aims

The overarching aim of this study was to design a means of formatting health and related information within a template which is maximally accessible to people with aphasia.

Further aims identified to achieve this included:

- Identification of the knowledge base in three critical domains to inform the initial design of a prototype template for presenting information to people with aphasia. These three domains are:
 - a. Evidence regarding the impact of formatting modifications on people with aphasia.
 - b. Evidence concerning the lexical and syntactic means of facilitating comprehension of the written word for people with aphasia.

- c. Evidence surrounding the priorities for health information topics required by people with aphasia.
- Collaborating with people with aphasia and graphic designers in an iterative design process, seeking the views of people with aphasia to develop the prototype template and attempting to understand the issues of accessible formatting from the perspective of people with aphasia.
- Developing practical guidelines for use by health professionals, researchers, family members or others working with people with aphasia producing new information materials.

1.5.2 Study design

The study was conducted in three phases.

Phase 1 included scrutiny of the literature and subsequent development of the prototype template in collaboration with graphic designers and healthcare professionals, and is described in Chapters 2 and 3.

Phase 2 involved a collaboration with people with aphasia and the graphic designers in an iterative design process, culminating in the final version of the template, and is described in Chapters 4 - 8.

Phase 3 involved the translation of the findings of Phase 2 into practical guidelines, usertesting and further development of the guidelines. This phase, which resulted in the publication of accessible information guidelines, is outlined in Chapter 9.

1.5.2.1 Phase 1

Phase 1 involved two stages. Stage 1 involved scrutiny of the existing evidence in the research literature in order to extract evidence based principles from three bodies of evidence to inform the development of the prototype template. The three areas investigated were:

- Existing published evidence concerning the effectiveness of formatting modifications used to facilitate the access of people with aphasia to written health and related materials. This includes what has been investigated in terms of both the impact on people with aphasia's comprehension and the preferences of people with aphasia for written information and modified formatting.
- Evidence from research into language processing in aphasia regarding the lexical and syntactic parameters that enable people with aphasia to access the meaning of written language.

3. Evidence about the health information topics required by people with aphasia.

In Stage 2, the principles extracted from the three reviews were carried forward into the design of the prototype Template 1, devised in collaboration with a graphic design team, and with advice from relevant healthcare professionals.

1.5.2.2 Phase 2

Phase 2 involved a collaboration with people with aphasia and the graphic designers in an iterative design process. A focus group design was selected, bringing people with aphasia together to review the template and to gather their views on accessible formatting. For this collaboration, suitable methods for maximising the involvement of all participants in the design process were developed through consideration of previous literature described in Chapter 4.

The same people participated consistently throughout the iterative design process, each participant attending two focus group meetings. During each focus group meeting, qualitative and rating scale data were collected.

In the first focus group meeting, people with aphasia provided structured feedback on Template 1. Following this meeting, feedback provided to the graphic design team informed the development of the second version of the prototype, Template 2. In the second focus group meeting, people with aphasia provided feedback on Template 2, after which feedback to the graphic design team informed the third and final template, Template 3.Figure 2.1 below summarises the study design.

1.5.2.3 Phase 3

In Phase 3, the design team collaborated to translate the principles embodied in the final Template into practical guidelines for use by anyone making accessible information for people with aphasia. The draft guidelines were then user-tested by a mixed group of participants, who gave their views on the draft guideline via an online survey. The data collected from the survey was used to inform modifications to the guidelines, which were subsequently submitted for publication.

Figure 1.1 Summary of Phase 1 and 2 study design



Phase 3 Development, user-testing, and completion of accessible information guidelines, submitted for publication.
1.6 Summary

Current literature highlights the need for information for people after a stroke, showing that access to appropriate and timely health information has a positive impact on their recovery, their engagement and motivation with rehabilitation, and ultimately on their psychological adjustment to stroke and wellbeing (Forster et al., 2012a). However, it is acknowledged that patients and their families remain dissatisfied with the amount of information they receive (e.g. McKevitt et al., 2010). This is a recurring theme in the literature, and strongly stated for people with aphasia, who reportedly do not feel adequately informed about their health (Parr et al., 1997, Tomkins et al., 2013).

Research has been conducted in healthcare settings to address patient information needs, resulting in guidance for professionals producing information (e.g. Duman, 2003). There are two strands to this guidance. Firstly, advice regarding design and production is provided, and secondly, it is recommended that people affected by a health condition should be involved in producing information. Within this literature, however, the challenges of designing for or collaborating with people with aphasia are not addressed.

The need to adapt information for people with aphasia falls within the scope of the Equality Act (2010) and the Accessible Information Standard (2015), but a lack of evidence-based guidelines means that implementing such legislation is left to the interpretation of local services.

Further literature addresses the information needs of patients and their families after a stroke (e.g. Forster et al., 2012a; Hafsteinsdottir et al., 2011; Hanger et al., 1998), and this has contributed to national guidance (e.g. NICE, 2013). Within this literature, it is acknowledged that people with aphasia require adapted information. There is, however, an implicit assumption about the ability of speech and language therapists to provide appropriately modified information, with no clear evidence-based guidance for how to produce accessible information for people with aphasia. Whilst there is a body of research into accessible information (e.g. Rose et al., 2003, Brennan et al., 2005), different methodologies and research aims mean that findings are not always consistent, and cannot be easily translated. Despite this, the concept of 'aphasia-friendly' information is currently widely used, and new materials are being published which follow the broad principles modelled by Parr et al. (1999) and further specified by Rose et al. (2003). There remains, however, limited evidence into the impact of these formatting modifications on the experience of information or comprehension by people with aphasia.

22

Chapter 2 Phase 1: Extraction of principles for accessible formatting from the existing research

The research outlined in Chapter 1 demonstrates the need people with aphasia have for accessible health and related information, but highlights an ongoing dissatisfaction with current provision of information materials. This study therefore aimed to work with people with aphasia to develop more accessible formats within a novel template. Three strands of knowledge were considered necessary to the task of generating such a template. The purpose of this chapter is to review the existing literature in each of the three areas identified in Section 1.5: the effectiveness of formatting modifications, factors to facilitate written language processing, and health information topics needed by people with aphasia.

2.1 Introduction

The aim of these reviews is to identify principles of best practice for designing new health information for people with aphasia, including accessible formatting and language, and relevant topics. These three reviews form the first part of Phase 1 of this investigation, the outcome of which is destined to inform the design and formatting of a prototype template for health information, to trial with participants with aphasia in Phase 2 of this study. These initial literature reviews were conducted between October 2010 and January 2011.

2.2 Review of text formatting modifications

The first literature review is concerned with the research evidence for modifications to features of formatting which have been explored with a view to increasing the accessibility of information materials to people with aphasia.

2.2.1 Aims of literature review into text formatting modifications

This review has two related aims. The first is to identify the existing research literature on the impact of formatting modifications to written health materials on people with aphasia. The second aim is to establish the current evidence base regarding the effectiveness of specific modifications on access to information for people with aphasia, in order to extract principles to be applied in the development of the prototype template.

2.2.2 Literature search methods

2.2.2.1 Databases and search terms

Two electronic databases, Web of Science and Scopus, were searched, using a range of different search terms, reflecting the requirements of the review.

The following search terms, visible in any or all of title, key words, abstract: aphasia OR dysphasia AND accessib* OR read* OR prefer* AND information OR print* OR writ* OR internet OR web OR comput*.

A second search combining the term 'aphasia-friendly' with aphasia OR dysphasia was conducted to ensure coverage.

2.2.2.2 Study inclusion and exclusion criteria

To be included in this review studies had to meet the following criteria. Studies had to report novel empirical data gathered from participants who were adults with acquired aphasia, with no other neurological conditions. If studies included other sub-groups such as people with dementia or healthcare professionals, the data from the stroke aphasia sub-group had to be clearly differentiated and only those data were considered for the review. The studies had to examine participants' comprehension of or their views of functionally relevant, written information materials, either printed or computer-mediated, which had been modified to include clearly described formatting changes hypothesised to facilitate access for people with aphasia. Studies had to report clear outcomes of providing such information to people with aphasia. These included the impact on participants' comprehension of the formatting modifications, and/or participants' reported preferences for specific formatting features. Studies reporting only the views of proxies such as family members, or only the views of healthcare professionals, were excluded. There was no restriction on study design, publication date or geographical location, but publications not in English were excluded. Review articles were retained for citation checks but were not included.

2.2.2.3 Results of literature search and selection process

The initial search yielded 10,406 and 5,731,009 results on Scopus and Web of Science respectively. The search was therefore refined using a sequence of the terms aphasia AND accessib*, and 'aphasia-friendly'. This resulted in the retrieval of 110 articles.

Articles were selected for review based on an initial reading of the title, the abstract, and where necessary, the full article.

Articles were discarded if they were duplicates or did not meet the inclusion and exclusion criteria outlined above. For example, articles were discarded which did not report materials specifically designed to explore accessibility or readability, failed to provide novel data, or which reported outcomes which were not related to comprehension, recall or preference of modified materials. Several studies were retrieved which reported the use of aphasia-friendly principles, but did not specifically examine the effect of the modifications on the participants (e.g Egan, Worrall, & Oxenham, 2004; Kerr, Hilari, & Litosseliti, 2010) and therefore were not included. Two review articles were also found (Howe et al., 2004 and Worrall et al., 2007), reflecting the studies previously undertaken, but were excluded since they did not provide new data.

The literature search resulted in the retrieval of three studies which provided new empirical or qualitative data on either comprehension or acceptability of modified information. These were: Rose et al. (2003); Ghidella et al. (2005); Brennan et al. (2005). A further two articles were retrieved following scrutiny of the citations of the selected articles: Dietz et al. (2009) and McKelvey et al. (2010).

2.2.3 Reviews of selected studies

Using the criteria identified above, five studies were selected for this review. See Table 2.1 below for a summary of the studies.

	Authors	Number of participants	Participant characteristics	Study design	Procedures	Findings
1	Rose et al. (2003)	12	Mild - moderate aphasia, 1;10 - 9;5 years post onset.	Comprehension/knowledge testing of two matched groups. Health information.	Participants answered knowledge questions after reading 'aphasia- friendly' and unmodified health leaflets.	AF format resulted in 11.2% increase in comprehension over control format.
				Preference ratings.	Rated different leaflets.	Over 50% preferred control format.
2	Ghidella et al. (2005)	6 PWA 18 SLT	Mild - moderate aphasia, 1 - 12 years post onset.	Survey. Quantitative rankings of websites.	Participants rated five aphasia websites for quality and perceived accessibility.	PWA ratings of accessibility agreed with those of researchers, but their overall preferences did not completely reflect this.
3	Brennan et al. (2005)	9	Mild - moderate aphasia, 3 - 11 years post onset.	Comprehension testing of individual participants. Non-health topics.	Cloze task in six different conditions and across reading grades 5, 6 and 7: Control (no modifications); simple words and short sentences; large print; increased white space; with pictures; combined AF features.	At Grade 6, significant effect of all separate (except pictures) and combined modifications.

Table 2.1 Studies included in the literature review of modified formatting

	Authors	Number of participants	Participant characteristics	Study design	Procedures	Findings
4	Dietz et al. (2009)	7	Moderate - severe Broca's aphasia, 2 - 6 years post onset.	Comprehension testing of individual participants. Non-health topics.	Comparison of high context versus low context photographs versus control (no picture).	Significant increase in comprehension with high context over low context photographs. No difference between high context photographs and control. Faster response times in control.
				Preference ratings.	Participant ratings of picture type.	Preference for high context over low context photographs.
5	McKelvey et al. (2010)	8	Severe aphasia, 0;4 - 19;6 years post onset.	Comprehension testing of individual participants. Non-health topics.	Spoken and written single word to picture matching task: personally relevant contextualised vs non- personally relevant contextualised photographs vs non- personally relevant iconic images.	Significant increase in comprehension with personally relevant contextualised photographs.
				Preference ratings.	Participant ratings of picture type.	Preference for personally-relevant contextualised photographs.

AF = Aphasia Friendly; SLT = Speech and language therapists

2.2.3.1 Rose, Worrall, & McKenna (2003)

Rose et al. (2003) were the first research team to define and test the impact of modified health information on people with aphasia. Building on previous work by Parr et al. (1999) and others, they defined "aphasia-friendly principles" as "(1) use of simple words and short sentences; (2) use of large and standard font; (3) use of white space; and (4) use of relevant pictures." (p.950). The study aimed to examine both the impact of these combined factors on participants' comprehension of health information materials, and participants' preferences for modified or original formats.

Twelve participants with mild to moderate chronic aphasia were asked to read health information on four different topics. These were presented in original and modified formats, the latter produced following the above-defined principles. To determine gain in knowledge, which would imply comprehension of the materials, the authors developed twelve reversed pairs of yes/no questions about the health topics covered; these were posed one month before the test session and immediately after reading all the information. Participants were also asked to rate whether they were highly certain (3), mildly certain (2), or uncertain (1) of their responses. After looking at all the leaflets, participants were asked to state their preferred type of leaflet for each of the topics.

Group data showed that participants' knowledge increased significantly after reading any type of information. However, this was significantly greater after reading the adapted materials, with participants expressing more confidence in their responses. Whilst this is an encouraging finding, it may be that caution is required. Whether participants did indeed understand more is questionable, given the use of a yes/no question format, with both spoken and key written word input. Arguably, features of the aphasia-friendly materials, such as large font and short sentences, might facilitate the participants to utilise visual recognition of key words to select correct responses, rather than deeper processing of the information. And whilst the authors reasoned that providing information on topics not directly relevant to the participants would reduce interference of prior knowledge, it is possible that providing novel, possibly emotionally distressing information might in itself heighten attention, reinforcing recognition memory.

Despite the positive impact on comprehension of the modified materials, over half of the participants preferred the original leaflets. A quarter of participants varied in their preferences between the modified and original materials, but it is not reported whether there was any pattern for preferences between the different topics; that is, whether people with aphasia preferred modified information on the topics of personal relevance, compared with other

health topics. As the authors note, this finding reveals a level of complexity in this area requiring further qualitative research to understand in greater detail the perspectives of people with aphasia.

2.2.3.2 Ghidella, Murray, Smart, McKenna, & Worrall (2005)

Websites about aphasia designed for people with aphasia were examined in this study for their overall quality, readability, and accessibility. Six people with mild to moderate chronic aphasia and eighteen speech and language therapists (SLTs) participated.

Accessibility and quality of the websites were determined by the authors based on previous research evidence, for example Rose et al. (2003). Readability scores were calculated by the authors using a 500-word sample from each website.

Participants with aphasia were asked to rate the accessibility of the websites via an eight question survey addressing broad issues of content, language, design, navigation and overall usefulness. Having completed the surveys for all five websites, participants were then asked to place them in order of overall preference. The SLTs completed a "similar" (p.1139) accessibility survey and answered questions about quality of the websites.

Contrary to their hypotheses, the authors found that websites designed for users with aphasia were not necessarily accessible or of high quality, based on their evaluation. Aphasia Help, designed with input from people with aphasia, scored highest for both quality and accessibility. All other websites varied between their quality and accessibility scores. For example, Speakability scored highly for accessibility, but low for quality. SLTs' ratings of quality and accessibility were similar to those of the researchers.

Ratings of accessibility between the researchers, the SLTs and people with aphasia showed some agreement about the most and least accessible websites, but the people with aphasia expressed some different opinions. For example, Aphasia Help, rated highest for accessibility by researchers and SLTs, was rated third most accessible by people with aphasia. The Aphasia Center of California, rated second most accessible by people with aphasia was, in contrast, rated poorly for accessibility by SLTs and researchers.

There were also slight differences between the people with aphasia's accessibility ratings and their ranked preferences, similar to the findings of Rose et al. (2003). Issues of tone or childishness are considered by the authors to account for this, although this is speculative, since, as they acknowledge, the fixed response format did not allow participants to elaborate on their responses. It is noteworthy that accessibility, the factor considered most important by researchers, may not be of primary interest to users when rating websites. The authors note that most of the participants were daily users of the internet, so it may be that the participants' familiarity with internet sites for general use influenced their preferences.

2.2.3.3 Brennan, Worrall, & McKenna (2005)

The positive findings of Rose et al. (2003) prompted Brennan et al. (2005) to test the impact of the same four aphasia-friendly principles identified in the preceding study on the comprehension of people with aphasia. They hypothesised that people with aphasia would understand more content where any or all of the aphasia-friendly principles were used. Nine people with chronic mild to moderate aphasia from a university aphasia group took part in the study.

The authors devised materials based on a published reading assessment tool (Thurstone, 1990). Subjects were required to read a paragraph and complete the final sentence with a multiple choice word or phrase (cloze task), where correctly completing the sentence indicates comprehension of the preceding paragraph. These materials were divided into sets of stimuli at reading grades 5, 6, and 7, as defined by the published assessment tool. The authors then formed each of these into six further subsets, modifying the original materials into the following conditions: control, simple language and vocabulary, large font size, increased white space, addition of pictures, and all four principles combined. A total of 90 paragraphs were presented to participants over a two-session testing period, starting with the Grade 5 set. The application of a discontinue rule meant that if participants failed either all five Grade 5 paragraphs, or four consecutive paragraphs in any of the subsets of Grades 6 or 7, no further paragraphs in that subset were shown.

Results at Grade 5 were discounted due to a ceiling effect found for these materials, which, the authors propose, left little room for any effect of the modifications. Grade 7 prompted the discontinue rule for several participants, meaning that the data is skewed towards the more able readers rather than the whole sample. Reporting of the results is focused on the Grade 6 group findings. At Grade 6, the group data showed a significant positive impact on comprehension of all modifications except the addition of images. The four principles applied together, however, did not produce any significantly greater benefit to comprehension than the separately applied principles of simplified language and large font, which had the greatest impact.

Contrary to their hypothesis, the authors found that adding images to unmodified text did not aid comprehension. Participants were observed to persist with looking at the materials with added pictures, leading the authors to question whether there may be cognitive (distracting), engagement (irritating), stylistic, or design (pictures not attached to cloze answers) issues affecting their utility. They do not question why images appeared to have no negative impact in the aphasia-friendly condition, in which they would have been equally distracting, irritating, of the wrong style, or badly placed. An alternative explanation could be that the addition of images to unmodified text places a processing load on participants as they attempt to integrate linguistic and visual stimuli at different levels of complexity. This increased processing could also account for the longer looking time observed.

2.2.3.4 Dietz, Hux, McKelvey, Beukelman, & Weissling (2009)

In the previous studies, the images presented in the information materials were reported as a single entity, without consideration of image style as a sub-factor of accessibility. Dietz and colleagues argued that the line drawings used in the study by Brennan et al. (2005) could have presented a processing challenge for people with aphasia, thus accounting for the finding that pictures did not support comprehension of unmodified text.

Seven people with moderate to severe chronic aphasia participated in the study. The authors devised materials to reflect the study aims of comparing comprehension, response times and preferences of three conditions: high context photographs, low context photographs, and a "no context" (p.1057) text only control.

To develop the materials, the researchers first selected photographs, then composed reading passages to accompany these for testing in the three conditions. The reading passages were all controlled for linguistic variables and ease of reading, none exceeding Flesch-Kincaid Grade level 2.5. Participants were shown each reading passage in turn, along with either two high context photographs, two low context photographs, or no photographs. Following each passage, participants were asked nine fixed-response questions, each of which was recorded for accuracy and response time. Preferences for the materials were gathered via a 1-5 Likert scale, recording the degree to which the participants considered that the photographs either were or would have been helpful for understanding the passages.

This study found that the high context photographs conferred a significant benefit to comprehension over the low context photographs and a non-significant benefit over the control condition. Response times were longest in the condition associated with greatest comprehension, and shortest in the control condition. Participants reported a preference for the high over the low context photographs, and reported they would have preferred to see photographs where none were shown.

2.2.3.5 McKelvey, Hux, Dietz, & Beukelman (2010)

In a further exploration of image style, McKelvey et al. set out to compare responses to personally relevant, non-personally relevant and pictographic symbols. The aim of the study was to identify picture materials for alternative and augmentative communication (AAC). The potential transferability of findings to health information is perhaps questionable; however, it is of interest to this review as it deals with picture processing and the preferences of people with aphasia. This study examined accuracy of responses to photographs in a single word to picture matching task. The authors hypothesised that participants would both prefer and be more accurate in matching single words to "personally relevant, contextualised photographs over non-personally relevant, contextualised photographs and non-contextualised, iconic images." (p.24). Participants' preferences for the different image styles were also explored.

Participants were eight adult AAC users with severe chronic aphasia. In keeping with the AAC context, materials were created by the researchers and individualised to each participant. Each participant was shown sixteen individual concepts in three conditions: personally-relevant, contextualised photographs; non-personally relevant, contextualised photographs; and non-contextualised symbols. Participants were asked to select the best picture for the concept. Comprehension of the concept word was maximised via use of both spoken and written input.

As a group, response accuracy was highest with personally relevant items, and lowest with symbols, although the variation in responses was wide. In addition, participants expressed a strong preference for personally relevant, contextualised photographs, particularly for nouns and proper nouns.

2.2.4 Summary of findings

The current evidence includes preference and comprehension data, and congruence between these varies.

2.2.4.1 Language

A positive impact of simple vocabulary and syntax on comprehension was found by Brennan et al. (2005), but methodological difficulties mean that the findings are difficult to interpret. Overall, it was found that the language component of information materials was underspecified in relation to the complexity of aphasic language processing impairments.

2.2.4.2 *Typography*

The use of larger font size is supported by studies exploring comprehension of people with aphasia, both as an isolated formatting feature (Brennan et al., 2005), and as part of combined

aphasia-friendly formatting (Rose et al., 2003). No studies have examined different methods for emphasising key words in text.

2.2.4.3 Images

Images have been demonstrated to aid comprehension only in the context of other combined features including reduced complexity of vocabulary and syntax (Brennan et al., 2005; Dietz et al., 2009). Mixed findings concerning study participants' preferences for use of images has been reported, with variation between studies addressing images specifically (Dietz et al., 2009; McKelvey et al., 2010) and those addressing combined formatting features (Rose et al., 2003; Ghidella et al., 2005).

2.2.4.4 Layout

No studies have empirically explored variations in the layout of text and images, and no clear evidence supports the provision of single or multiple images with text.

2.2.4.5 Combined formatting features

Only one study, Rose et al. (2003), explored the impact on comprehension in relation to health information, reporting that the use of aphasia-friendly formatting increased knowledge by 11.2% over traditional leaflets. The studies reviewed varied in focus, addressing health related information (Rose et al., 2003; Ghidella et al., 2005), general interest topics (Brennan et al., 2005; Dietz et al., 2009), or personally relevant information (McKelvey et al., 2010). The question of whether health information should be differently formatted from other reading materials was not addressed in these studies. As discussed, the inherent difficulty of health concepts may be an additional consideration compared to other types of information. Although researchers noted that modified formatting resulted in longer documents, no studies addressed the formatting of a set of related information concepts.

The findings of preference surveys and rankings suggest that although some people with aphasia may acknowledge the utility of modified formatting to facilitate their comprehension of written material, many do not wish to receive adapted written information materials. None of the studies reviewed incorporated a qualitative element to explore the issue of accessibility and acceptability of materials in greater depth.

2.2.4.6 Appraisal of the selected studies

The five studies reviewed here represent the published literature into the specifics of formatting information accessibly for people with aphasia. The small number of studies retrieved demonstrates that, although accepted within clinical practice, the research evidence base is not extensive.

The studies have varying aims and methods, but are all exploratory in nature. A desire to achieve consensus with respect to formatting features is visible, despite displaying individual differences amongst participants. Studies are therefore also linked by a will to understand the preferences of readers, showing sensitivity to their changed abilities since onset of aphasia. All studies adopted quantitative methods to explore pre-determined questions, testing comprehension or seeking preferences.

All the studies have small sample sizes (n = 6 - 12), with only forty-two people with aphasia represented in the studies in total. A wide range of ages are represented (28 - 86 years), but participants' other characteristics are more limited. Fewer women (17) than men (25) were involved, and average educational level is high (13 years). Although it is recognised that people have different information needs at different stages after stroke, these studies do not capture this aspect, with the time since onset being over three years. The range of aphasia severity represents largely those with mild-moderate impairments. Only McKelvey et al. (2010) sought those with severe expressive aphasia, since the study focussed on AAC, which is likely to be more used by those with severe aphasia.

Most studies used the Aphasia Quotient (AQ) to provide profiles of participants' aphasia. In the context of addressing participants' responses to written materials however, this may not be the most useful measure, since it is calculated from scores of the oral language subtests of the Western Aphasia Battery (WAB) (Kertesz, 2007). Similarly, the Rapid Estimate of Adult Literacy in Medicine (REALM) (Murphy et al., 1993) used by Rose et al. (2003) is of debatable value with a sample with aphasia, since it produces a reading grade based on a person's ability to read aloud medical terms. It is standardised on a non-aphasic population, and since any errors in output are scored as incorrect, reading grades may appear lower for people with aphasia than actual reading ability. Dietz et al. (2009) used supplementary reading assessments such as the Reading Comprehension Battery for Aphasia (RCBA) (LaPointe & Horner, 1998). Where participants are described only in averages, these are, however, of limited value, particularly since little reference is then made to any correlation between reading ability and responses. As Dietz et al. propose, wider profiling of participants with increased detail on cognitive functioning would allow greater opportunity to link findings with individual formatting needs.

Across all studies, a variety of materials were used, including relevant and non-relevant health leaflets, topics of general interest, and personally relevant information. The limited description of materials in several of the studies means that credibility and replicability are limited. All researchers developed new measures to test comprehension, comparing variables of interest to their research aims. Such a range of individual approaches raises challenges when comparing findings across the studies. The use of surveys and rankings has produced quantifiable data. This may, however, have resulted in a narrow reporting of the views of people with aphasia, since such methods do not allow participants to elaborate on their responses (e.g. Ghidella et al., 2005).

Finally, none of the studies reported providing feedback to participants on how their input may have influenced the design of information materials.

This is a multifaceted area of research, and the variety of methods attempted bear witness to this. In the following section, the implications for the development of a new template for health information for people with aphasia will be considered.

2.2.5 Implications for the development of novel materials

The literature review undertaken aimed to identify principles to be used to create a new, evidence-based template for presenting information in the most accessible format for people with aphasia.

There is some evidence for the impact of modified formatting on comprehension of materials and some evidence that people with aphasia perceive modified formatting as both beneficial to their comprehension and an acceptable format for the provision of written information. The range of study designs, research aims, and methods used, however, limit the strength of the evidence for both the acceptability of modified formatting and some specific aspects of the modifications to information. In this section, the implications of the literature review findings to devise the prototype template will be discussed in relation to five key areas: language, typography, images, layout, and combined features.

2.2.5.1 Language

Despite the large amount of research into language processing in aphasia, the language modifications within the literature on accessible information are described in broad terms only. For example, Rose et al. (2003) specify "simple words and short sentences" (p.950). Whilst both Brennan et al. (2005) and Ghidella et al. (2005) identify background information about aphasia, neither provides any detail on the processes involved in the linguistic modifications to materials.

The Flesch-Kincaid Reading Grade level (Flesch, 1948) was used as a measure of text reading difficulty in all the quantitative studies, with the exception of McKelvey et al.'s (2010) single word study. This formula is based on the length of words and sentences used in the text and provides a guide to the North American school grade level which would be required to read text. Whilst this is an established tool, it accounts for quantitative features of sentence length

and word letter length only. Psycholinguistic variables such as word frequency, word imageability, or syntax are not accounted for in this measure. Since aphasic language impairment is subject to other factors than non-impaired language, such a measure cannot address the reading requirements of people with aphasia.

Only one study (Brennan et al., 2005) isolated simple language and vocabulary as a condition in their investigation, but demonstrated its positive impact on comprehension at reading grade 6 only.

2.2.5.2 Typography

Explorations of typographical features suggest there is evidence for the use of larger font sizes. Brennan et al. (2005) found that a large font size resulted in greater comprehension of written material presented at reading grade 6.

There is, however, no evidence about the impact on comprehension of either different font styles, or of different methods of emphasis in font, and further investigation is warranted into how these features can be used in accessible design.

2.2.5.3 *Images*

Research into the use of images has dominated this field of study, but widely varying study aims and methods and findings make it difficult to synthesise the evidence.

Rose et al. (2003) found varied responses to the use of images in modified health information materials, with some participants reportedly perceiving images as helpful and others finding them unacceptable.

Overall, the effectiveness of images in facilitating comprehension remains inconclusive. Images do not support comprehension when added to unmodified text (Brennan et al., 2005), but do appear to add value where the text and images were composed together (Dietz et al., 2009). For people with more severe language impairment, a greater impact on comprehension is shown when personally relevant, high context photographs are used (Dietz et al., 2009; McKelvey et al., 2010).

Overall, the evidence to date points to a need for further clarification of the use of images. Exploration of responses to images in general and to different image styles is needed to determine their acceptability and perceived usefulness for people with aphasia when accessing written information materials.

2.2.5.4 Layout

The literature review yielded limited evidence about specific layout of information. This may in part be due to the different terms used to describe this aspect of formatting. For example Rose et al. (2003) discuss the impact of modified formatting on document length, but others, for example Brennan et al. (2005), do not refer to this, focusing instead on individual features such as line spacing, rather than the layout. The question of amount of information per page was not addressed specifically by any studies.

The use of increased white space has been shown to be effective in supporting comprehension at Reading Grade 6 (Brennan et al., 2005), but no other evidence about the impact on comprehension of layout was retrieved from the literature. The findings of studies focusing on images signal the need to consider more carefully how images are presented with text in the context of the overall layout.

Specific features of layout remain underspecified for the purposes of producing a new template. Further exploration of people with aphasia's response to the layout of modified information is needed, particularly concerning the number of images in relation to the text.

2.2.5.5 Combined formatting features

The reported preferences of people with aphasia suggest an ambivalence towards the use of combined formatting features, with a tension reported in the literature between the perceptions of people with aphasia in relation to accessibility versus acceptability of combined features of modified information.

A positive impact on comprehension of these combined features was demonstrated initially in relation to health information (Rose et al., 2003), but no subsequent studies have provided categorical support for combining larger fonts, increased white space, and images accompanying text. For example, Brennan et al. (2005) found that the combined features did not produce any significantly greater benefit to comprehension than providing simplified text or large font sizes. This lack of support stems largely from the study aims, which have focussed attention on individual formatting features, rather than exploring the impact of combined features benefit of a considered pairing of text and images (Brennan et al., 2005; Dietz et al., 2009) where images are linked with simplified text. Table 2.2 below provides a summary of the findings of the literature review regarding facilitative formatting features.

 $Table \ {\it 2.2 Summary of findings of facilitative formatting features identified in literature review}$

Language	Beneficial impact of simple words and short sentences at RG6* (Brennan et al., 2005).
Typography	
Font size	Positive impact of 24pt font compared with unspecified control (Brennan et al., 2005).
Font style	No studies of font style and comprehension.
Emphasis	No studies of emphasis and comprehension.
Images	Line drawings beneficial as part of combined AF format at RG6 only (Brennan et al., 2005).
	Significant impact of high context, personally relevant photographs over low context photographs at max RG2.5, but non-significantly better than control (Dietz et al., 2009).
	Significant impact of personally relevant contextualised photographs over non-personally relevant photographs or iconic images in accuracy of single word to picture matching in people with severe aphasia (McKelvey et al., 2010).
	Preference expressed for use of contextualised photographs (Dietz et al., 2009) and of personally relevant, contextualised photographs (McKelvey et al., 2010).
Layout	Increased line spacing in text improved comprehension at RG6 (Brennan et al., 2005).
Combined factors	Combined AF features result in 11.2% increase in comprehension of health information (Rose et al., 2003). However, Rose et al. (2003) found that over 50% preferred unmodified format and Ghidella et al. (2005) report that participants' perceptions of accessibility of websites differed from their preferences.

*RG = Reading Grade

2.2.6 Summary

This review found a small evidence base. There is very limited evidence regarding effects on comprehension, with only one study showing a clear advantage for health information materials presented in a modified format (Rose et al., 2003). Reported preferences for

specially formatted materials remain varied, with differences reported between participants' perceptions of accessibility and their preferred written and web-based formats.

Principles for some aspects of typography, use of images, layout, and combined features have been identified, and there is evidence to support the provision of short sentences. As highlighted, however, the studies reviewed did not provide evidence concerning linguistic factors known to affect language comprehension in aphasia. In order to derive principles of language use for the proposed template, therefore, the following review will address the knowledge base regarding language processing in aphasia.

2.3 Review of factors affecting language processing in aphasia

The second literature review addressed research literature on factors affecting language processing in aphasia. Given the large scale of this literature, and the confines of the thesis, the review will focus on established knowledge, aiming to identify factors which facilitate lexical and syntactic aspects of language processing, and which can be translated into principles for the template.

Two previous reviews of facilitating reading in aphasia were retrieved, which aimed to identify language processing characteristics of aphasia, and to apply these to the provision of simplified text (Singh, Gedeon, & Rho, 1998; Devlin & Tait, 1998). Whilst these reviews provide useful syntheses of language in aphasia and its possible impact on the reading of web-based text (Singh et al., 1998) and newspapers (Devlin &Tait, 1998), these do not appear to have been translated into tangible materials for people with aphasia. Further investigation of factors to facilitate language processes is therefore required, which can be applied to written health information for people with aphasia.

2.3.1 Aim of literature review into language processing in aphasia

The aim of this review is to identify relevant literature, which can be used to specify lexical and syntactic aspects of written language which are known to facilitate the language processing of people with aphasia.

2.3.1.1 Methods

Due to the volume of possible research literature, this review does not attempt to be comprehensive, but aims rather to provide an overview based on key texts and seminal research findings. Relevant literature was retrieved based on clinical expertise, with a view to possible applications of the knowledge base to the development of text for the proposed template. Clinical expertise is acknowledged to form one facet of the evidence base for an area, such as aphasia, and is considered when formulating clinical guidelines (e.g. Simmons-Mackie et al., 2017).

Three areas of the evidence base for language processing were considered important to this review: factors affecting single word processing, sentence processing factors, and evidence into the utility of priming to facilitate processing in people with aphasia.

2.3.2 Findings 1: Lexical factors affecting processing in aphasia

At the single word level of language processing, two factors were considered highly significant in terms of identifying facilitators to processing: psycholinguistic variables and grammatical class.

2.3.2.1 Psycholinguistic properties of single words

According to cognitive neuropsychological models of language processing (e.g. Ellis & Young, 1988), factors known to affect lexical processing relate to the different levels of processing, namely semantic and phonological processing. The psycholinguistic properties of lexical items have been found to affect processing speed and accuracy of processing in people with aphasia. Lexical processing pertaining to the semantic system is affected by the imageability or concreteness of lexical items, whereas phonological processing is affected by features such as frequency, familiarity, and word length (e.g. Cole-Virtue & Nickels, 2004; Coltheart, 1981; Nickels & Howard, 1995, Kay et al., 1997).

2.3.2.1.1 Imageability

Imageability effects are linked with semantic processing, with higher imageability words more likely to be understood than lower imageability items (Coltheart, 1981; Franklin, 1989). In reading, imageability is associated with access to the semantic system via orthography, with higher imageability items more easily processed (Ellis & Young, 1988).

2.3.2.1.2 Concreteness

There is a link between imageability and the degree to which words are considered concrete or abstract, since concrete terms are inherently more easily imageable than abstract terms (Bird, Howard, & Franklin, 2003). Franklin, Howard, & Patterson (1994) report a significant deterioration in comprehension of abstract terms amongst some people with aphasia. Metaphors and idioms use words in their non-literal meanings to convey meanings which are established through shared understanding within a population. In healthcare, metaphors are considered a useful means of conveying concepts such as a patient's 'stroke journey', however, research demonstrates that such non-literal use of lexical terms is more difficult to process in aphasia (Papagno & Caporali, 2007).

2.3.2.1.3 Frequency and familiarity

The effects on comprehension of word frequency are well-documented in aphasia, with more frequently accessed terms found to be understood more rapidly and accurately than their less frequently used counterparts (Coltheart, 1981; Laiacona, Luzzatti, Zonca, Guarnaschelli, & Capitani, 2001).

In reading, word frequency is associated with access to the orthographic input lexicon, with higher frequency items more likely to be recognised as real words (Ellis & Young, 1988).

There is a known link between the concepts of personal word frequency and familiarity (e.g. Gernsbacher, 1984), since words which are more familiar are likely to be more frequently accessed. In the realm of healthcare information, this aspect of frequency is important, since terms which are highly familiar and frequently accessed by healthcare professionals may be either unfamiliar or novel to patients. This links with lexicality, the variable determining whether a term is a real word or a non-word. In healthcare, people may be encountering medical terminology for the first time. For example, the term 'Warfarin' will be frequently accessed by a pharmacist, but may be a new term for a person experiencing stroke care for the first time. For a person with acquired dyslexia, this term may have no representation within their orthographic input lexicon, rendering it incomprehensible to them through the expected reading channels (Ellis 1984).

2.3.2.1.4 Word length

Increasing word length in nouns is associated with greater processing difficulty, with shorter words typically processed with greater ease (Nickels & Howard, 2004).

In reading, effects of word length are associated with a deficit at the level of visual orthographic analysis, which may in turn impact on a person's ability to perform any reading task (Marshall & Newcombe, 1973). Providing short words in text can therefore be seen to maximise a person's access to the meaning of the text.

2.3.2.1.5 Individual variation in aphasia

The outline above reports findings from the literature which are applicable to the broad population with aphasia, but it is important to note that individual differences in aphasia presentation are widely reported, due to factors including the overall severity of stroke and aphasia, and the range and specific patterns of impairments (Franklin, 1989; Lesser & Milroy, 1993;Nickels & Howard, 1995). In relation to processing the written word, variability occurs due to deficits at different levels of processing in terms of visual, lexical and semantic processing (Coltheart, Rastle, Perry, Langdon, & Ziegler, 2001). Cases of individuals with aphasia presenting with reverse effects of the above psycholinguistic variables have been reported by researchers, for example, for frequency (Marshall, Pring, Chiat, & Robson, 2001) and word length (Best, 1995).

2.3.2.1.6 Effects of grammatical class on processing in aphasia

Evidence from the aphasia literature suggests that content words such as nouns and verbs are more easily processed than grammatical function words, such as pronouns (Berndt & Caramazza, 1981). Difficulties processing nouns and verbs can arise due to the lexical factors discussed above, but difficulties with verbs may be more pronounced, since the exact meaning of a verb is dependent on the syntactic and semantic context in which it is used, an issue discussed further below.

Difficulties processing function words arise for different reasons, depending on aphasia type. For people with Broca's aphasia, grammatical function words are considered to be more difficult to process because they are governed by syntactic rules (Berndt & Caramazza, 1981). For people with semantic impairments, comprehension is compromised by the low imageability of function words (Bird, Franklin, & Howard, 2002).

Proforms can present specific difficulties in processing since their comprehension is dependent on identification of the previous referent, a different lexical item. The meaning of the proform must therefore be inferred by the reader (Raynor & Pollatsek, 1989). Proforms such as pronouns are inherently of low imageability, since their meaning relates to a different lexical term, which must be tracked in the sentence.

2.3.3 Findings 2: Sentence processing in aphasia

Research into sentence comprehension in aphasia has established that deficits in syntactic processing are more likely to disrupt comprehension of sentence forms which do not conform to highly frequently occurring canonical structures (Caramazza & Zurif, 1976). As discussed by Schwartz, Saffran and Marin (1980), this is viewed as a "... dissociation between lexical and syntactic aspects of comprehension..." (p.249). In such cases, people with agrammatism parse sentences using the expected, canonical structure, processing the individual lexical items in the order in which they appear. This can also result in difficulties in mapping the surface structure of the sentence to the meaning, particularly where the semantic roles of nouns are ambiguous, for example in passive sentences (Caramazza & Zurif, 1976).

Resulting difficulties arise for people with aphasia when attempting to process reversible sentences, in which the subject and object of the sentence can appear in opposing positions in a sentence, but have semantically different roles. Sentence structures which incur a change in

canonical word order, such as question forms and passive constructions, have also been found to present processing challenges (Caplan et al., 2007). Such difficulties are pertinent to accessing health information, since formatting guidelines promote the use of a question and answer format, whereby information topics are introduced via the interrogative form (e.g. Duman, 2003).

Syntactically complex sentences involving more than one clause are more difficult to process with aphasia, since they place demands on both syntactic processing and working memory. Longer sentences are not necessarily more challenging in aphasia, since it is the syntactic complexity, rather than the sentence length per se, which has the greatest impact on processing (Marshall, 1995; Marshall, 2002). However, extra sentence length adds to the overall cognitive load experienced by a reader with aphasia, who may have additional impairments to non-linguistic cognitive domains, such as attention and working memory (Helm-Estabrooks, 2002).

2.3.4 Findings 3: Priming

A body of literature regarding the effects of structural and semantic priming on language processing supports the idea that targeted repetition of input may be utilised to enhance the reduced capacity for language processing found in aphasia (e.g. Hartsuiker & Kolk, 1998; Pickering & Ferreira, 2008).

Structural priming refers to the facilitation of processing of specific sentence structures following exposure to that same or similar sentence structure (Pickering & Ferreira, 2008). Lexical boost is associated with structural priming, but refers specifically to the additional priming that occurs due to the repetition of content words within the sentence frame. This has been demonstrated strongly in relation to verbs but applies also to nouns (Pickering & Branigan, 1998).

Repetition priming refers to the same effect, but in this case the prime is the target word itself. This is based on the finding that a stimulus is easier to process if it has recently been experienced (Cave, 1997).

These priming techniques are distinct from semantic priming, which refers to the greater ease in processing a lexical term when this has been preceded by a semantically related term, performed under experimental conditions (McNamara, 2005).

Studies of priming with people with aphasia report that subjects were found to have increased access to syntactic forms which they were exposed to repeatedly. Hartsuiker and Kolk (1998) propose that the use of repeated syntactic structures and lexical items has a facilitative effect

by reducing the processing effort needed by an individual when processing language. Saffran and Martin (1997) found that participants with aphasia were more able to produce passive structures after being exposed to such structures repeatedly. Hartsuiker and Kolk (1998) found, in addition, that priming exerted a stronger effect on study participants with aphasia than on participants without aphasia. In both studies, priming had some lasting effects, leading the authors to postulate that priming facilitates access to language structures, which had previously been restricted by the presence of aphasia.

Studies into facilitating word finding in aphasia have shown the value of repetition priming as a means of improving study participants' ability to retrieve stimulus items (Best, Herbert, Hickin, Osborne, & Howard, 2002; Hickin et al., 2002; Nickels, 2002).

Research into priming with people with aphasia has focused on the impact of structural priming on spoken word production. It is, however, possible to argue that facilitating production affects comprehension, due to increased language processing activity, and, as proposed by Pickering and Ferreira (2008:453), "Priming occurs between comprehension and production, and so it is reasonable to hypothesize that comprehension and production involve the same representations."

In the following section, the principles taken forward from this knowledge for application to the development of the template will be outlined.

2.3.5 Implications for the development of novel materials

It was argued in Section 2.2 that within the literature relating to the formatting of information for people with aphasia, the optimal means of modifying written language was underspecified. The preceding overview of knowledge regarding language processing in aphasia provides evidence that aspects of lexical-semantic and syntactic processing may be used to specify the written language input on the Template developed in this study.

In this section, the principles which determine the choice of lexical terms and sentence forms, and the use of structural and semantic priming to maximise reading comprehension, will be described.

2.3.5.1 Lexical terms

The choice of lexical terms in the template will be governed by the evidence from the literature which shows that lexical processing is in general easier in aphasia when words are of high imageability, concrete rather than abstract, high frequency and familiarity, and short in length. Factors known to aid processing include the selection of content, rather than function words. Avoiding idioms and metaphors is also known to aid processing. The specific

processing difficulties associated with pronouns will be avoided by repetition of nouns. It must be acknowledged that in the context of providing accurate health information, novel or unfamiliar terms which are highly pertinent to the condition are required. For example, to explain adequately the presence of aphasia, the term 'aphasia' must be used, but this requires explanation.

2.3.5.2 Sentence forms

Evidence from syntactic processing in aphasia supports the provision of single clause, canonical sentence forms. This principle implies that passive, interrogative, or other moved structures will not be included. Other complex structures, such as embedded clauses, will be avoided. Sentences will be short, thereby minimising the impact of working memory deficits on text reading.

2.3.5.3 *Priming*

Evidence regarding the beneficial impact of structural and lexical priming can be translated into principles for devising maximally accessible written text. With regard to structural priming, the use of repeated, canonical structures is adopted to minimise the processing load and maximise syntactic comprehension. The concept of lexical boost is adopted, with content words (nouns and verbs) repeated across sentences. An advantage of repeating content words within sentences is that it means that proforms are not required.

With regard to repetition priming, the use of the same content words repeatedly in the template materials maximises opportunities for readers to process their meaning. For example, repetition of the same lexical term in a heading and within the subsequent sentences.

The example below shows the use of the repeated canonical subject – verb – object sentence structure, with repetition of the term 'brain' through the heading and the sentences.

Heading	The brain
Sentence 1	Your brain controls your body.
Sentence 2	Your brain controls your speech

Finally, semantically related words will be included in the materials in order to maximise any semantic priming.

2.3.6 Summary

The overview of the language processing literature in aphasia has demonstrated its application to the development of maximally accessible text for people with aphasia within the proposed template. This is summarised in Table 2.3 below.

	Use	Avoid	
Lexical terms	Content words (Berndt &	Proforms (Berndt &	
	Caramazza, 1981)	Caramazza, 1981; Bird,	
		Franklin, & Howard, 2002)	
	High imageability, concrete, high	Abstract terms (Franklin et al.,	
	frequency, familiar, short words. (e.g. Coltheart, 1981; Nickels &	1994)	
	Howard, 2004)	Metaphors and idioms	
		(Papagno & Caporali, 2007)	
Sentence forms	Single clause, frequently-	Non-canonical structures,	
	occurring canonical structures	such as passive or	
	Short sentences (Caramazza &	interrogative forms (Caplan et	
	Zurif, 1976)	al., 2007)	
		Embedded clauses (Marshall,	
		2002)	
Structural priming	Repeated canonical forms (Saffran	& Martin, 1997; Hartsuiker &	
	Kolk, 1998)		
Lexical boost	Repeat content words, specifically verbs and nouns (Pickering &		
	Ferreira, 2008)		
Repetition priming Repeat terms throughout template		2	
Semantic priming	Present information within semantic categories (McNamara, 2005)		
Jemantic prinning			

Table 2.3 Summary of facilitative features of language derived from language processing literature

Note that in the context of health information, certain terms which do not conform to the ideal in terms of their psycholinguistic properties are unavoidable, in order to convey accurate information, as discussed in Section 2.3.5.1.

Principles for the formatting and language components of the template have been established through the scrutiny of relevant literature. The final strand of knowledge required to inform the template development concerns the information content.

2.4 Review of health information topics pertinent to people with aphasia

In order to develop the proposed template, details of the information content of relevance to the intended audience were required. As noted in Section 2.2, previous studies in this area have incorporated a variety of relevant and non-relevant health information topics, general

interest, or personally relevant information. As Rose et al. (2011a) observed, participants with aphasia displayed greater engagement with information materials which were of direct, personal interest.

2.4.1 Aims of literature review into health information topics for people with aphasia

This review aimed to examine the literature to identify health information required by people with aphasia arising from stroke for the proposed template.

2.4.2 Literature search methods

Two electronic databases, Web of Science and Scopus, were searched, using the following two search terms, visible in any or all of title, key words, abstract: stroke AND aphasia AND information, and "information OR questions OR topics". Due to the large number of articles generated (over 17,000), the search was refined to include the following search terms: "stroke education" OR "stroke information" AND aphasia.

Articles were selected based on an initial reading of the title, and where necessary, the abstract. In addition to a library search, articles cited by others were examined for potential relevance. This review was conducted between December, 2010 and January, 2011.

2.4.2.1 Criteria for inclusion and exclusion of studies.

To be included in this review studies had to meet the following criteria. Studies had to report novel empirical data gathered from participants who were adults who had sustained a stroke and who presented with acquired aphasia, with no other neurological conditions.

The studies had to examine participants' views on which topics of information they thought should be provided to people who had sustained a stroke.

Studies had to report clear outcomes including details of the topics and sub-topics participants requested information about. Studies which referred to patient education and information, but which did not specifically seek the outcomes described above were excluded. Examples of studies which were excluded from this review included those which examined the process of information provision in healthcare settings, those which reported participants' responses to their own experiences of receiving information, and studies investigating the impact of information on patient and carer outcomes.

Due to the paucity of studies retrieved regarding the specific information topics required by people with aphasia, studies which sought the views of the stroke population as a whole were included.

There was no restriction on study design, publication date or geographical location, but publications not in English were excluded.

2.4.3 Findings of the literature search

2.4.3.1 Studies identified

Nine studies were identified through the search strategy, the main characteristics of which are shown in Table 2.4 below.

Authors	Participants	Data collection methods
Hanger & Mulley (1993)	Any person making information enquiry. 1,397 people asked 1,908 questions.	18-week record of all enquiries made to 8 Stroke Association UK advice centres. Quantitative, descriptive results reported.
van Veenendaal et al. (1996)	n = 35 stroke patients, 39 family members and 43 health professionals. Mean 18 months post onset.	Questionnaires sent out to stroke survivors, family members and health professionals. Received and desired information topics. Quantitative, descriptive analysis.
Parr et al. (1997)	n = 50 people with aphasia, 1/3 with severe aphasia, all between 5 – 18 years post onset.	Individual interviews, Framework analysis of interview data. Information formed one major theme.
Hanger et al. (1998)	n = any stroke patient: 60 at 2 weeks, 111 at 6 months, 72 at 2 years	Prospective recruitment of all patients attending one hospital in four-month period. Individual interviews at 2 weeks, 6 months, and 2 years post stroke.
Wiles et al. (1998)	n = 31 stroke patients and their carers.	Interviews with patients, carers or jointly at three stages: in hospital; up to 1 month after going home; 2-12 months post-discharge.
Wachters- Kaufmann et al. (2005)	n = 33 patients, 27 carers. 15 months post onset.	Telephone survey as part of a larger study. Quantitative data, descriptive analysis.
Avent et al. (2005)	n = 16 family members of people with aphasia, mean 5.5 years post onset.	Focus groups. Asked to discuss their experiences at three stages after stroke: at stroke onset – during rehabilitation – in chronic phase. Framework analysis.
Hare et al. (2006)	n = 27, recruited via primary care, 9 months – 22 years post onset.	Focus groups, home visits to severely disabled participants. Participants were asked what they want from community services. Thematic analysis. Information was one of three main themes.
Kerr et al. (2010)	n = 12 people living with stroke, 5 with aphasia, 7 without aphasia, 6 – 15 months post onset.	Focus groups. Participants asked what stroke information they would like to see on a website, and to identify information needs at different stages after stroke. Content analysis.

Table 2.4 Summary of studies included in the literature overview of topics relevant to people with
aphasia

2.4.3.2 Methods used in the studies

Different methods were used to gather data, including questionnaires (van Veenendaal, Grinspun, & Adriaanse, 1996), face-to-face interviews (Hanger et al., 1998; Parr et al., 1997; Wachters-Kaufmann et al., 2005; Wiles et al., 1998), telephone interviews (Wachters-Kaufmann et al., 2005), and focus groups (Avent et al., 2005; Hare, Rogers, Lester, McManus, & Mant, 2006; Kerr et al., 2010). One study (Hanger & Mulley, 1993) asked UK Stroke Association information centres to record all requests for information during a set period. Studies varied in whether they recruited only service users, family members, or a combination.

Participants were questioned about their information needs at different stages post stroke, depending on the study aims. Data collection time points included during acute hospital admission, on transfer home, and in the longer term post stroke. Overall, studies found that although the focus of information needs changed somewhat with passing time and changing circumstances, people continued to have information needs for many years after a stroke, and these included both unresolved questions and new information requirements.

2.4.3.3 Topics identified in the studies

The studies revealed both the wide range of information needs and a considerable overlap in findings between studies regarding the desired topics, suggesting some consistency for people experiencing stroke, with and without aphasia. In order to synthesise the findings, the types of information people need are presented here categorised according to the ICF (2001) domains of impairment, activity, participation, and well-being.

Regarding impairment information, all studies reported that people expressed a need for basic knowledge about their stroke, citing a lack of understanding about the physical causes and manifestations of a stroke. People wanted to know what to expect in terms of medical treatment, rehabilitation, and overall recovery, and their need to understand risk factors to prevent future strokes was reflected across several studies. People also wanted information about associated impairments specific to themselves, such as headaches, pain, abnormal sensation, and vision. Information was required repeatedly at different stages, reflecting that stroke can have lasting consequences, and that information needs change with individual patterns of recovery and circumstance. Avent et al. (2005) report the need for the term aphasia to be used to name the condition and for this to be shared to increase local and public awareness.

Information needs relating to activity tended to focus on rehabilitation, going home from hospital, and the longer term. People wanted information to help them to go home and be safe there, with advice about practical caring and coping, equipment, and services to support

this. They wanted to understand and engage with rehabilitation to help regain functions such as communication within families. A desire for information about issues such as memory, concentration, communication in aphasia, and sex suggests that people prefer to be prepared for potential difficulties, reinforcing the importance of information in helping people to be in control of their own lives.

In the longer term, information needed to facilitate social participation was wide-ranging, reflecting the breadth of daily life. People required information which would enable them to regain and then maintain their independence as members of society, including information about finances, and assistance with claiming benefits and entitlements, self-help, managing health, and resources to enable planning for the future. Information was needed to allow people to continue to live in their own homes, for example information about help with shopping, and adaptations to property. People wanted to know about services in the community, such as health and voluntary services which could support their independent living. In terms of participating in social activities, people needed information about getting out, including information about accessing public transport and returning to driving, including the Blue Badge scheme. They wanted information about social and support groups in the community, with people with aphasia particularly expressing a wish to meet others with the same condition. People wanted to pursue leisure activities and take holidays, and sought advice about how to achieve this. Information about return to work or future employment was desired by some people.

Within the domain of well-being, people wanted information about the psychological and emotional impact of stroke, including direct consequences such as emotionalism, and reactions such as depression, anxiety and panic. Support to understand issues such as frustration, fatigue, loss of initiative, and decreased confidence was sought, together with information about specific problems, such as personality changes. People wanted information about how to cope with stress after a stroke, and how to deal with family tensions in the light of changed circumstances. Overall, such information was sought in order to enable people to adjust and cope with the changes brought about by stroke. Throughout all stages, but most particularly in the longer term, people wanted to know how to access psychosocial support and counselling. Table 2.5 below summarises the information topics required, classified using the ICF (2001) domains.

ICF (2001) domain	Information topics required
Impairment	What is a stroke/TIA, causes of stroke
	How to prevent strokes
	Risk of recurrence
	Headaches, pain, abnormal sensation, vision
	Aphasia information, generic and specific to individuals
Activity	Memory and concentration
	Communication difficulties
	Sex
	Balance
	Going home from hospital
	Equipment
	Exercise
Participation	Driving, dealing with DVLA and Blue Badge criteria
	Finances, benefits and entitlements, and help with completing
	claim forms
	Employment
	Awareness raising for public
	Accessing local health, voluntary services
	Bus passes
	Shopping and help with shopping
Well-being	Depression, anxiety, panic
	Fatigue
	Personality changes
	Emotionalism
	Frustration, decreased confidence
	Family tension
	Coping with stress
	Psychosocial support and counselling
	Positive expectations and need to retain hope
	Long term emotional and psychological support

Table 2.5 Information topics required after stroke, classified by ICF (2001) domains

Table 2.6 below provides a summary of the research studies which contributed evidence on topics within each of the four domains of the ICF (2001).

ICF (2001) domains	Studies contributing to findings
Findings relating to impairment , activity,	Hanger & Mulley (1993)
participation, and wellbeing	Parr et al. (1997)
	Hanger et al. (1998)
	Wiles et al. (1998)
	Hare et al. (2005)
	Kerr et al. (2010)
Findings relating to impairment, activity, and	Wachters-Kaufman et al. (2005)
participation	
Findings relating to impairment , activity, and	van Veenendaal et al. (1996)
wellbeing	Avent et al. (2005)

Table 2.6 Studies contributing to the evidence for topics in each of the four ICF (2001) domains.

As shown, six of the nine studies contributed evidence to all four ICF domains, with the remaining three studies contributing to three of the four domains. As discussed above, the information topics required after stroke are both broad and detailed in scope, relating to both general and personally relevant issues.

2.4.3.4 Topics selected for the template

For the purpose of the current study, topics were selected from the above for their relevance to all potential participants, and their likelihood of engaging people in focus group discussions. Topics were selected which would be applicable to participants who were no longer in receipt of local stroke services, and who were already living successfully with aphasia in their local communities. Topics applicable to people still requiring services, or containing specific local information, were not considered appropriate for these materials.

Three broad topics were selected for the template. Basic information about what a stroke is and how it is caused in the body was the most frequently cited area where people expressed a need for information, yet continued to report a lack of understanding. The topic of aphasia was highly relevant, since it affected all participants. Finally, information about emotional consequences of stroke was frequently reported as a high information need in the longer term, a stage after stroke currently applicable to the study participants.

2.4.4 Formatting features requiring further evidence

The literature review into formatting modifications presented in Section 2.2 revealed areas for which the evidence base is not yet clearly established.

With regard to typography, research has shown the value of large fonts. However, evidence remains incomplete regarding font size in relation to page size, and no evidence supports the use of specific font styles, or specific styles of emphasis of key words in text. A lack of evidence about optimum image styles emerged from the research literature, indicating a need for further exploration of this area. The literature review highlighted that the specific layout of text and images within the design of information materials was not a focus for study. Clear evidence could therefore not be retrieved to specify the optimum number and positioning of images in relation to text. Finally, no evidence emerged to identify how to present a set of related information concepts.

2.5 Principles for the design of Template 1

Principles to be applied to the design of the prototype Template 1 were derived from the three literature reviews, as detailed above. These principles are presented in Table 2.7 below, and show the features of language, formatting and topics to be operationalised in the template.

Language	
Lexical terms	Use content words (e.g. Berndt & Caramazza, 1981). Use high frequency, high imageability short words (e.g. Coltheart, 1981; Nickels & Howard, 2004).
	Avoid abstract terms (Franklin et al., 2004). Avoid metaphors and idioms (Papagno & Caporali, 2007). Avoid proforms (e.g. Berndt & Caramazza, 1981).
Sentence forms	Use single clause, frequently-occurring canonical structures (Caramazza & Zurif, 1976). Avoid non-canonical structures, such as passive or interrogative forms (Caplan et al., 2007). Avoid embedded clauses (Marshall, 2002).
Priming	Use repeated canonical forms (Saffran & Martin, 1997; Hartsuiker & Kolk, 1998). Repeat content words, specifically verbs and nouns (Pickering & Ferreira, 2008). Use the same terms repeatedly. Present information within semantic categories (McNamara, 2005).
Readability	Reading Grade 5 or lower (Aleligay et al., 2008).
Typography	
Font size Font style Emphasis	Large and very large font sizes (Brennan et al., 2005). No evidence – needs further exploration. No evidence – needs further exploration.
Images	Contextualised photographs preferred (Dietz et al., 2009). Personally relevant, contextualised photographs preferred (McKelvey et al., 2010).
Layout	Increased spacing between lines of text (Brennan et al., 2005).
	Increased comprehension of health information via combined formatting features of large font sizes, use of images, increased white space and simple language (Rose et al., 2003). People with aphasia perceive combined formatting features as more accessible than unmodified formats (Ghidella et al., 2005).
Topics	Information about stroke and the causes of stroke (see Table 2.5). Aphasia (see Table 2.5). Emotional consequences of stroke (see Table 2.5).

2.6 Summary of chapter

This chapter has provided reviews of the literature across three separate areas of knowledge: accessible formatting for people with aphasia, the knowledge base concerning language processing in aphasia, and the information topics required by people with aphasia after stroke. These areas were selected for scrutiny, because each was considered to contribute to defining the principles for a new template to convey health information for people with aphasia.

Through this process of review, principles for formatting, use of language, and relevant topics have been identified for the proposed template, and areas identified as requiring further evidence highlighted.

The following chapter comprises the second part of the template development, in which the details of how these principles were carried forward into the template design will be outlined.

Chapter 3 Phase 1: Development of the template

This chapter describes how the strands of evidence identified in Chapter 2 were integrated into the prototype template (Template 1), in a collaborative design process with the graphic designers.

3.1 Establishment of Phase 1 collaborative design team

The design team in Phase 1 of the study included the author, who is an experienced speech and language therapist (SLT), a research SLT, a family member of a person with aphasia who is also an experienced teacher of English, and a team of professional graphic designers. Healthcare professionals, including a stroke physician, an occupational therapist, physiotherapist, stroke nurse, and clinical psychologist were involved on a consultative basis.

3.1.1 Identification of graphic designers

The author and research SLT selected the graphic designer by viewing the work of local designers via their websites, rating their work for clarity and simplicity of design, and gauging their interest through discussion of the project.

3.1.2 Discussion of principles with the graphic designers

A first meeting of the design team was arranged in order to discuss the design brief. The agenda included background discussion of aphasia and access to information, an introduction to the work required, and specific issues of designing information for people with aphasia. To facilitate this discussion, current, freely available information materials used in stroke care settings were collected. These included published materials (e.g. Parr et al, 1999, 2004; Cottrell & Davies, 2006); web-based information produced for people with aphasia (e.g. Aphasia Now www.aphasianow.org ; Aphasia Help www.aphasiahelp.org; Speakability www.speakability.org.uk ; Connect www.ukconnect.org, and general stroke information resources (e.g. Stroke Association leaflets).

Principles for the design of the prototype template were agreed at this meeting. These principles were based on the evidence extracted from the literature review, and are summarised in Table 2.7. The lead graphic designer provided professional opinions on best practice in graphic design.

It was agreed that the researchers would provide the key concepts for the template to the graphic designers as a series of short sentences, composed according to the principles

identified from the review of factors affecting language processing in aphasia, and categorised within topic areas identified in the literature review reported in Sections 2.3 and 2.4 (see Tables 2.3 and 2.5).

The graphic designers were asked to use their professional skills to create engaging, welldesigned materials, suitable for adults, taking into account the nature of language processing in aphasia and possible visual and cognitive difficulties arising from stroke (Helm-Estabrooks, 2002).

3.2 Generation of text for the template

Text for the template was devised by the researchers in line with the principles derived from the reviews of formatting modifications, factors affecting language processing in aphasia, and information topics required. Initial drafts of the text were refined with input from health professionals to ensure the accuracy of all concepts.

3.2.1 Topics

Topics selected for Template 1 included information about stroke and the causes of stroke, and information about what aphasia is. Further sentences were composed on the topics of emotional consequences of stroke.

Accuracy of information was checked using current, publicly available information resources (e.g. Stroke Association factsheets) and with healthcare professionals. The researchers then considered ways of conveying information to maximise comprehension of fundamental concepts. Through discussion, the researchers agreed to present background concepts first, with new concepts following on from this basic explanation. For example, the concept of stroke illness was explained by first outlining the role of the brain in relation to body functions.

Secondly, it was agreed to present information within three broad categories, representing background explanation, facts about the specific limitation to function (e.g. aphasia, emotions after stroke), and information to support readers, titled 'what helps'.

An example of each of these type of statements in the set of information concepts about aphasia included the following (see Appendix A for further examples):

background information:	language is talking and understanding
fact:	it is difficult to say words
what helps:	practising the words again and again helps
3.2.2 Text content

Principles derived from the literature reviews into modified formatting and language processing in aphasia were followed throughout the development of the text for the template. In this section, the process for selecting lexical terms, constructing sentences and the application of evidence regarding priming in aphasia is described.

3.2.2.1 Lexical terms

Selection of vocabulary items took account of the psycholinguistic variables of length, frequency and imageability. For each concept, a range of possible terms was identified, and these were each considered in terms of their imageability, concreteness, frequency, and likely familiarity. Abstract, metaphorical, or technical terms were avoided where possible. The British National Corpus (Leech, 2001) was consulted to ensure choice of high frequency items, for example 'makes' rather than 'produces', 'talking' rather than 'speech'. Ideally, the shortest terms were selected, although this was not always possible, due to the specific nature of the vocabulary involved. For example, the word 'haemorrhage' is specific to the explanation of one type of stroke, and cannot therefore be substituted, but needs to be explained.

Given that the possible choice of terms was limited by the topic areas, this was a pragmatic process, and the researchers discussed vocabulary choices to ensure that all variables had been considered, and that the choices were consistent with the context of the information topics.

Within each topic, subtopics were identified and a superordinate term selected for each subtopic. Repeated forms were used through the repetition of key content words in both the banner heading and the sentence (see example provided below).

3.2.2.2 Sentence forms

Sentences were constructed which consisted of single clause, canonical structures within the identified topics and subtopics. As with the lexical terms, the researchers discussed the construction of sentences to ensure that they both adhered to the principles identified in the literature and were pragmatically meaningful. Once the sentences had been agreed, the researcher used Flesch-Kincaid readability software to generate a reading grade for each set of sentences within a subtopic. These were then checked to ensure that all sentences used had a Reading Grade of five or lower, the level advised for text aimed at people with aphasia (Aleligay et al., 2008).

3.2.2.3 Use of priming

The principle of repeating key words and sentence structures was followed throughout the text development. The following two examples show how the principles of priming were used to convey key concepts about the brain and blood flow. Example 1 shows the use of structural and lexical priming for a subject – verb – object sentence, with repetition of the initial subject noun and the verb. In the second example, the first sentence introduces the concept; the initial noun is then repeated throughout the sequence, with structural repetition seen in the second and third sentences. In both examples, the heading identifies the superordinate term, which is repeated in the sentences.

	Example 1	Example 2
Heading	The brain	Your blood
Sentence 1	Your brain controls your body.	Arteries are blood vessels.
Sentence 2	Your brain controls your arms and legs.	Arteries carry blood.
Sentence 3	Your brain controls your speech.	Arteries carry blood to the brain.

Construction of the sentences maximised the chances of structural and lexical priming occurring. See Appendix A for further examples of information sentences generated for Template 1. Examples showing how the text was incorporated into the overall template design are provided below and in Appendices B and C.

3.3 Agreement with graphic designers on Template 1 design

Decisions taken by the design team were based on the principles generated from the literature review and in discussions regarding the optimum means of incorporating these principles into a physical template, which could be presented to people with aphasia in the focus groups.

3.3.1 Overall layout

In the light of the evidence concerning language processing in aphasia (see Section 2.3), the researchers identified the idea of presenting only one concept per page. As a result, the design team agreed to present each concept within a single white space, which was printed onto single cards measuring 170mm by 110mm.

The overall design of this white space included a coloured banner with heading text printed in bold black reversed into white and aligned to the left side. Below this banner, the information was conveyed in a single sentence, printed in black and aligned to the left side. Below the sentence, one, two, or three images were placed centrally. Evidence regarding line spacing was not required, due to the decision to present single sentences per page.

3.3.2 Typography

Large font sizes were selected to reflect research findings which suggest that people with aphasia comprehend more easily when presented with material printed in large or very large fonts (see Section 2.2.5). Thus, for the banner headline, font size 24pt was selected, and for the content sentence below this, 14pt was selected. The substantial difference in font sizes was selected to ensure that the difference between them could be easily perceived. Since no research evidence provided guidance on the use of serif or sans serif fonts, the professional opinion of the lead graphic designer's regarding current graphic design principles was sought. He advised the use of a sans serif font, and in addition he proposed the use of Vectora font style. Given that the literature review yielded no clear evidence for preference or effectiveness of emphasising key words, the design team agreed to present the Template 1 information without emphasis.

3.3.3 Images

Images were used throughout to illustrate the written information. Photographs and line drawings were both selected as suitable illustrative styles for the information content, since no consensus for the superiority of line drawings or photographs had emerged from the literature (see Section 2.2.5.3).

For people with severe aphasia, personally relevant photographs were found to be both easier to process and more highly rated (McKelvey et al., 2010), but the applicability of this finding to generic health information is limited, and for the purposes of the template design, this was not carried forward. It was agreed that the graphic designers would select images from a professional photographic library. Line drawings were then produced by the graphic designers from the photographs selected.

3.3.4 Number of images per concept

Evidence regarding number of images to present is unclear, but the mixed results of comprehension studies highlight the need for coherence between appropriately modified text and meaningful images (Brennan et al., 2005, Dietz et al., 2009).

3.3.5 Language content

As described above, the text to be included in the template design was devised by the research team, and each card included a single word or short phrase banner heading, to orientate readers to the topic. It was agreed that precise placement of the written sentence would be decided at point of design by the graphic designers, taking into consideration other aspects of the design, such as the placement of the images.

3.4 Summary of design principles for the prototype template

The final principles identifying the design criteria for Template 1 were agreed through the process described above and are summarised in Table 3.2 below. Figure 3.1 below shows an example of how the design principles described in this chapter were applied to produce a single card (shown actual size) conveying one information concept about the function of the brain. Further examples of the template are provided in Appendices B, C, and D.

Figure 3.1 Example showing application of the agreed design principles into Template 1



© The Stroke Association.

Language	Banner headings – single word or short phrase to orient to the topic covered by the sentence.			
Lexical terms	Use content words. Use high imageability, high frequency, short words. Use familiar words.			
	Avoid abstract terms. Avoid metaphors and idioms. Avoid proforms.			
Sentence forms	Use single clause, frequently-occurring canonical structures. Use short sentences. Avoid non-canonical structures, such as passive or interrogative forms. Avoid embedded clauses.			
Priming	Use repeated canonical forms. Repeat content words, specifically verbs and nouns. Use the same terms repeatedly. Present information within semantic categories.			
Readability	Flesch Kincaid readability software to be used to monitor reading grades, ensuring RG5 or lower throughout.			
Typography Font size	Banner heading - very large font (24). Information content sentence - large font (14).			
Font style Emphasis	Banner heading - Vectora Black. Information content sentence - Vectora Roman. Not used in Template 1.			
Images	Each sentence to be accompanied by one, two or three images to illustrate meaning. Photographs to be selected by graphic designer from professional photo libraries. Line drawings to be produced by designers from photographs.			
Layout	A white space measuring 170mm by 110mm to contain the content. White background to the overall space. Coloured banner with heading text printed in bold black reversed into white and aligned to the left side. Single sentence below banner to convey information content, printed in black and aligned to the left side. One, two, or three images placed centrally, below sentence. Borders to separate areas of text and related images. Consistent use of colour used in headings and images within specific topics.			
Topics	Basic information about stroke and the causes of stroke. Background information to aphasia. Emotional consequences of stroke.			

3.5 Provision of content to the graphic design team and Template 1 delivery

The researchers provided the content to the graphic designers in the form of a sequence of written sentences, categorised within the topic areas of 'What a stroke is' and 'What aphasia is'. The graphic designers were encouraged to seek clarification for any concepts which might be unclear or present design difficulties.

The graphic designers delivered a full set of Template 1 prototype materials in advance of the first focus group meeting.

3.6 Development of materials for exploration of factors identified in the literature review lacking clear evidence.

In Section 2.4.4, factors relevant to the design of Template 1, about which there was little clear agreement in the research literature, were highlighted. In this section, the methods and materials selected by the researcher for further exploration of these factors are outlined. The graphic designers were asked to provide variations in design to allow for further focus group exploration of two issues identified in the literature review as requiring further evidence: image style and number of images per sentence.

3.6.1 Further evidence for image style

To explore participants' responses to different image styles, the graphic designers were asked to provide two different versions of Template 1, one version including photographs, the other including line drawings.

Examples of Template 1 showing variations in image style are shown below. Images \mathbb{C} The Stroke Association.



Photographic style

Line drawing style



3.6.2 Further evidence for number of images per concept

To explore participants' responses to variations in the number of images per concept, the graphic designers were asked to provide two versions of Template 1. In the first version, the sentence was illustrated with a single image. In the second version, the sentence was illustrated by up to three images, to illustrate each key word in the sentence. To accommodate the separate images, the sentence was separated out, with space between each of the key concepts. Examples of Template 1 showing variations in number of images per concept are shown below. Images © The Stroke Association.



3.6.3 Further evidence for font sizes, font styles, and emphasis of key words

The literature review revealed a lack of clarity regarding optimum font sizes, font styles, and means of emphasising key words in text. The researcher therefore prepared materials to present to participants. These are described in Chapter 5, and examples are shown in Appendix E.

3.7 Summary of template development

In this chapter, the collaborative design of the Template 1 prototype has been described. This has demonstrated how three strands of knowledge, namely the findings of the literature review concerning the effectiveness of formatting modifications on people with aphasia's access to written information; the specific research evidence concerning factors which facilitate people with aphasia's processing of lexis and syntax; and the evidence from the literature review regarding the topics about which people with aphasia state they require information, were synthesised in the formulation of a new template for accessible information. The development of Template 1 materials to explore factors identified as needing further clarification from the research literature has also been described.

In the following chapter, consideration is given to suitable methods for collaborating with people with aphasia in a further, iterative design process.

Chapter 4 Phase 2: Methodological issues

The previous two chapters have detailed Phase 1 of the current study, including a review of the evidence and the process of designing an initial prototype Template 1 according to research evidence. Phase 2 of this study involves people with aphasia in a collaborative iterative design process to develop and refine this template. In this phase, people with aphasia will be involved in reviewing current best practice, and providing new insights into areas identified without robust research evidence.

As identified in Chapter 1, health literacy research identifies that patients should be involved in the development of new health information to ensure that it meets the needs of the target population in content, design, and accessibility (Bunge et al., 2010; Coulter et al., 1998).

The purpose of this chapter is to consider suitable methods for involving people with aphasia in the iterative design process. A brief review of previous methods used to involve people with aphasia in accessible information will be outlined. The rationale for the choice of data collection methods will then be described, including the techniques for facilitating the participation of people with aphasia. Lastly, issues relating to data analysis will be considered.

4.1 Involvement of people with aphasia in the development of information

The literature concerning the development of information aimed at people with aphasia has involved the target population in a range of ways. At the charity Connect in the UK, people with aphasia participated in focus groups to develop the Aphasia Handbook (Parr et al., 1999). Since the aim of the project was to develop information resources which would be accessible to people with aphasia, it is not reported as a research project in the usual sense. As a result, it is difficult to evaluate the involvement of people with aphasia in this work, since no published account is available.

As identified in Chapter 2, people with aphasia have participated in research studies addressing the impact of modified formatting on the comprehension of written information materials (e.g. Brennan et al., 2005; Dietz et al., 2009; Rose et al., 2003). Of these, only one study examined the impact of modified formatting on health information (Rose et al., 2003).

The opinions of people with aphasia have also been gathered in studies which aimed to understand their views of modified formatting (e.g. Ghidella et al., 2005; Rose et al., 2003) or their preferences with regard to images (Dietz et al., 2009; McKelvey et al., 2010). These studies have added to the evidence regarding accessible formatting, but none described a process of incorporating the findings in the design of new materials. The present study seeks to close the gap between research and design, by involving participants in a collaborative process. One means of achieving this is to employ methods developed in organisations such as Connect, and apply these in a research context, with the methods and analysis clearly described.

4.2 Focus groups

4.2.1 Introduction to focus groups as a research method

Focus groups are a form of group discussion, in which the views of people are expressed and collected within a natural human context, that is, a group. Morgan (1997) provides the following definition:

"...a research technique that collects data through group interaction on a topic determined by the researcher. In essence, it is the researcher's interest that provides the focus, whereas the data themselves come from the group interaction." (p.6).

As Stewart, Shamdasani, and Rook (2007) emphasise, the purpose of focus groups is to both canvass participants' opinions and seek more detailed reasons for these, though group discussion. In health care research, focus groups have been used extensively. As Kitzinger (2006) notes, such groups can be naturally supportive to participants who have health conditions in common, producing an environment in which people say more, and are often more critical than they would be in interviews. Through this process, focus groups generate data which are rich and varied (McLafferty, 2004).

As described by Merton, Fiske and Kendall (1990), effective focus groups should meet four criteria: firstly, they should address all the researcher's questions, but raise more issues; secondly, they should be focused on the issues; thirdly, they should involve all participants in a thorough discussion, addressing the issues in depth; finally, the researcher should be able to understand participants' responses through the sharing of individual opinions.

The challenge in conducting a group with people with communication disability is to ensure that the participants can follow the discussion and express their views. Finch and Lewis (2003) report that the use of techniques to enable participation is perceived as more acceptable in groups, in which looking at information together, or sharing experiences, is more natural and enjoyable, and less forced, than on one's own with a researcher. They highlight the need for "co-moderation" (p.196) when using such techniques; that is, allocating one leading moderator and other supporting facilitators. Such a structure allows the group moderator to retain their leadership of the group overall, and to preserve the flow of the group topic, while the facilitators provide support to the participants in doing the activities. Relevant techniques for participants with aphasia include the use of flipcharts to list issues, and providing visual materials.

4.2.2 Focus groups and people with aphasia

In this section, the methods used to involve people with aphasia in focus groups will be reviewed, to determine the feasibility of using focus groups methodology in an iterative design process with people with aphasia for the present study.

Few studies report strategies for including people with aphasia in focus groups. Four studies were reviewed for the methods used to support the specific language and communication needs of participants with aphasia (Dalemans, Wade, van den Heuvel, & de Witte, 2009; Garcia, Barrette, & Laroche, 2000; Kerr et al., 2010; Murphy, 2006). Of these, three studies conducted focus groups as a primary method for collecting qualitative data. Garcia et al. (2000) included people with aphasia, speech and language therapists, and local employers in focus groups to seek opinions about return to employment after stroke by people with aphasia. Murphy (2006) gathered the views of different stakeholders regarding communication between people with aphasia and General Practice staff. Kerr et al. (2010) involved people with and without aphasia in focus groups to decide on topics for stroke information for a local website.

Garcia et al. (2000) recruited fourteen people with mild or mild to moderate aphasia from an aphasia centre and via local speech and language therapists. It is unclear whether separate focus groups were conducted for the participants with aphasia. People with aphasia were accommodated in smaller groups of between two and six participants, in which the group facilitator is reported to provide communication support. In the case of the group of six people with aphasia, additional communication support was provided by a social worker. The authors state that "Participants were encouraged to use all methods of communication at their disposal" (p.275), but no detail is provided of the skills of the facilitators, or of the specific facilitation or communication techniques employed.

Murphy (2006) recruited two groups of people with aphasia from local support groups. No detail is provided on the numbers per group, or the participants, except that they are said to have three key word level comprehension. Talking Mats[®] were used to provide structure and symbols for participants to refer to. Although the exact method for this is not made clear, the reader can infer from the presentation of results that a single group Talking Mat[®] was used,

with all participants using the same mat and its accompanying symbols as a point of reference. In this study, the discussion data were then analysed, rather than the Talking Mat[®] itself, which functioned as a prop for group communication.

Kerr et al. (2010) report a focus group of five people with aphasia. Facilitation of the focus group in this study was undertaken by the researcher, with no other support staff reported. Kerr et al. refer to the use of pictures to illustrate key points in the group discussions between participants with and without aphasia, but no further detail is provided on how people with aphasia were supported to participate.

In these three studies, clinical interpersonal skills and techniques (e.g. encouraging all means of communication, provision of pictures and written words to support comprehension, use of Talking Mats[®]) were reported as means of supporting the participation of people with aphasia in focus groups addressing varied research aims. The reporting of such measures appears to assume that they increased the accessibility of the process to participants with aphasia, with research findings forming the focus of interest for the researchers. It can be argued, however, that a lack of clarity in reporting the means of ensuring access to the research for the participants is to the detriment of the rigour, and thus to the reliability and validity, of the findings.

One further study examined methods for involving people with aphasia in focus groups, within the context of an existing research project. Dalemans et al. (2009) used focus groups as a means of verifying results obtained from two other methods of data collection, namely structured diaries kept by participants with aphasia and family members, and semi-structured interviews addressing social participation. The participants in the focus group study are described as ten people with aphasia and nine family members, a subgroup of the larger study. The authors do not report the staffing involved in conducting a focus group of nineteen people with mixed communication abilities, implying that there was one interviewer for this large group. In keeping with the study aims, the authors identify four strategies used to ensure the participation of people with aphasia: firstly, a report from the previous stage of the research was sent to all participants in an 'aphasia-friendly' style, following principles identified by Brennan et al. (2005); secondly, PowerPoint was used to present these findings to the group; thirdly, during the subsequent discussion, pictures and written words from the presentation were again shown on the PowerPoint screen; finally, the group agreed ground rules regarding taking turns to talk, listening, and pace. These measures show an attempt to address the generic language processing needs of the participants, focusing particularly on supporting comprehension. Support to express views in the focus group discussion is, however, limited to the statement, "The interviewer tried to encourage each person with aphasia to express

69

themselves, by addressing them personally." (p.952). In a group of nineteen people with one facilitator, it is difficult for the critical reader to imagine that this provided sufficient support to people with aphasia to communicate their views.

Murphy (2006) comments on the usefulness of Talking Mats[®] in helping participants to understand the questions asked and in promoting a group discussion. The study aimed to record the group response, with the purpose of achieving consensus, therefore no commentary on the quality of participants' individual responses is provided. The use of a single Talking Mat[®] provides structure to the discussion, and, as with the use of pictures and written words made by Kerr et al. (2010) and Dalemans et al. (2009), can be seen to provide generic language processing support to the whole group. The communication needs of individuals within the group, however, are not attended to. It is possible that in these three studies (Dalemans et al., 2009; Kerr et al., 2010; Murphy, 2006) the researchers have overestimated the power of pictures to aid comprehension, and underestimated the need for communication support to individuals in a group. In Garcia et al.'s (2000) study, individual communication strengths were emphasised by facilitators, and valuable data was generated, however, the lack of detail in the reporting compromises the transferability of the methods to other studies. Dalemans et al. report on the success of their measures to facilitate participation. Whilst their researcher-centred report is positive, a more convincing claim would have been made had the experience been reported from the viewpoint of the participants.

4.2.3 Consideration of alternative data collection methods

Other possible methods for data collection are interviews and observation, and each will be briefly considered here. Structured interviews provide an opportunity to gather detailed opinions from individuals. In the aphasia literature, some researchers have suggested that such interviews are more suitable as a data collection method for participants with aphasia (e.g. Howe et al., 2004). In an individual interview, Morgan (1997) suggests that sharing the burden for developing the topic with one interviewee encourages participants to say more, generating more data for the researcher. People with aphasia may, however, find such a situation uncomfortable, placing undue pressure on them to speak. Since people with aphasia report that increased pressure to speak reduces their ability to speak (Edelman & Greenwood, 1992), there may in fact be a negative impact on the data collection. As Finch & Lewis (2003) state, "Social context creates spontaneity" (p.171) in which a greater opportunity for the exchange of ideas is fostered within a focus group. They provide an example of a study on attitudes towards disability amongst members of the public. A "layered" (p.186) discussion through the skilled facilitation of the group emerged, based on the skill of the moderator in picking up

themes within the group as it unfolded, and offering these for further discussion. They argue that these rich data would not have emerged in the course of individual interviews.

A further consideration concerning the choice of methods relates to the aims and nature of the research. Northcott & Hilari (2011) conducted individual interviews with people with aphasia to explore issues around loss of friends after a stroke, a personal and sensitive issue which people might be reluctant to discuss in a group. In contrast, the focus group studies discussed in Section 4.2.2 gathered opinions, drawing on the insights and expertise of people with aphasia, without asking participants to share their personal experiences in any detail. Such an approach is relevant here, since the current study does not seek in-depth personal information or experiences, but opinions based on the experience of having aphasia.

An alternative method, participant observation, could be used to collect data on how people interact with information. However, since the aim is to seek opinions, natural observation of participants is less informative to the research aims than hearing what participants have to say on the topic. In addition, the planned iteration of feedback to the graphic designer for this project needs to be prompt and clear, which would not be possible if detailed observation over time were recorded (Morgan, 1997).

4.2.4 Summary

The previous discussion supports the feasibility of involving people with aphasia in focus groups, but further consideration is needed to identify methods for maximising the inclusion of all participants. Specific issues include the use of evidence-based techniques to support communication, the numbers of participants per focus group, and how best to staff a focus group, including the potential roles of a lead moderator and the availability of trained communication facilitators within the group.

4.3 Methods to facilitate the participation of people with aphasia in focus groups

The review of literature reporting strategies to involve people with aphasia in focus group research showed that, whilst researchers have made efforts to facilitate participation, a lack of clarity surrounded the methods used to achieve this. In some cases, there was an apparent mismatch between the methods reported and the outcomes achieved. For example, Dalemans et al. (2009) report as successful a focus group involving nineteen people with aphasia, with minimal language support.

For the purposes of collaborating with people with aphasia in an iterative design process, greater clarity regarding communication support for participants with aphasia is needed, both

in terms of individual participant facilitation and the use of visual materials. In this section, findings from the literature which inform the development of principles for facilitating the participation of people with aphasia in focus groups will be outlined.

4.3.1 Supported communication techniques

The application of the social model of disability to aphasia provides a useful context to this study, which seeks to collaborate with participants with aphasia as partners in an iterative design process rather than as recipients of a procedure or intervention (Kagan & LeBlanc, 2002). In order to build a successful partnership, the communication between participants and researchers needs to function as a two-way interaction, in which the communication challenge is shared. According to the social model of disability, the person with the impairment (aphasia) is disabled by a society, which lacks both the awareness and the communication skills to include people with aphasia (Pound et al., 2007).

In the domain of communication disability, "supported conversation for adults with aphasia" (Kagan, 1998: 817) refers to a set of techniques which can be used when interacting with people with aphasia. Such techniques are described as "communication ramps" (Kagan and Gailey, 1993:206), since they mirror the notion of a physical ramp for a person who may otherwise be excluded by a physical disability from, for example, a building. Supported conversation techniques were developed through observation and analysis of people attending an Aphasia Centre in Ontario, Canada. The approach is underpinned by the philosophy that aphasia masks the valid thoughts and ideas of people with the condition. The skilful use by conversation partners of techniques to support conversations with people with aphasia reveals the underlying competence of the person with aphasia, who is then able to express their opinions.

Supported Conversation, as described by Kagan and Gailey (1993), is chiefly concerned with what they term "maintaining the feel and flow of conversation" (p.214), achieved through the use of techniques to facilitate both understanding and expression. Conversation partners of people with aphasia are trained in a range of techniques including pragmatic, communication, language, and practical skills. Pragmatic skills include, for example, giving a clear indication of a new topic, attending to comprehension difficulties, and giving time to process. Communication skills incorporate the ability to use, for example, gesture and pantomime, communicative drawing, and using these simultaneously if necessary. Language skills include, for example using closed questions, re-capping or verifying, and suggesting possible word choices to the person with aphasia. Lastly, practical skills for example, finding physical resources such as images to use within conversations, are needed.

Most of the techniques can be used to facilitate either comprehension or expression. For example, gesture, drawings, or visual resources may be used by the communication partner to illustrate a concept, or by the person with aphasia to convey an idea. Conversation partners maximise comprehension by routinely using other modalities (e.g. writing single words, drawing, gesture) whilst talking in conversation, and seek ways to allow an individual to finish what they are trying to express. In Supported Conversation, communication partners are trained to take account of the skills that the person with aphasia brings to the conversation.

There is an obvious overlap between Supported Conversation and Total Communication (Lawson & Fawcus, 1999), an approach in which communication by any means is encouraged to maximise communicative effectiveness. As Lawson and Fawcus describe, Total Communication was developed in aphasia clinics to enable an individual to express their thoughts and ideas, based on the idea that all channels of communication are equally valid. In this approach, the emphasis lies with the person with aphasia developing their communication skills to compensate for their language impairment.

These approaches do not only serve to improve communication between people with aphasia and others. By participating in genuine social interaction, people can express themselves as a whole person. In research to explore a partnership approach to communication participation, Lyon et al. (1997) paired ten people with aphasia and their caregivers with volunteers to support their communication in community activities. In the first phase, people with aphasia and their communication partners established their relationship and developed communication strategies; in the second, they jointly planned and undertook communitybased activities. As well as finding the expected benefits to communication and psychological well-being, the authors also noted qualitative changes which emerged through analysis of diary entries kept by the participants. This led the authors to propose that when people with aphasia were able to participate in activities in a fully absorbed fashion, they "... appeared to temporarily suspend any awareness that there were communicatively disabled." (p.703). This finding was striking, and supports the idea that communication can still feel natural for people with aphasia if it is well facilitated.

The benefits of specific strategies to support communication of people with aphasia within a research context are further supported by the findings of a pilot study carried out by Luck and Rose (2007), who found that when participants were appropriately supported to communicate in individual interviews, they provided more data than when an open-ended interview questioning style was used. The four strategies used were identified from thematic analysis of interview data with four participants. Firstly, there was an overt recognition of the possible impact of aphasia at the outset of the interview; secondly, the interviewer checked with the

participant that they had understood correctly, termed "interviewer clarification" (p.214); thirdly, they supported participants with encouragement, time to think, and suggestions of possible words to use; finally, they used "interviewer-generated ideas" (p.214) as a way of giving the person with aphasia something to respond to, rather than participants having to generate their own ideas.

Supported Conversation techniques validate the expertise of a person with aphasia, suggesting that data collected using such techniques is likely to reflect participants' genuine, individual opinions. The approach also provides an evidence-based approach to facilitating individuals with aphasia to participate, adding rigour to the research study. Since Supported Conversation techniques have been developed in the context of individual interactions, it is important to consider the provision of individual communication facilitators within the group.

In the following section, the use of visual materials to support individuals to participate will be discussed, principally with reference to Talking Mats[®].

4.3.2 Talking Mats®

The concept of Talking Mats[®] was developed at the University of Stirling by speech and language therapy researchers, and was initially developed for people with communication disability due to cerebral palsy, and later for people with motor neurone disease (Murphy, 1998, 1999). It has since been used in clinical and research contexts with adults and children with a wide range of communication impairments, including people with aphasia (Gillespie, Murphy, & Place, 2010; Murphy, 2000), intellectual impairment (Murphy & Cameron, 2008), and dementia (Murphy, Gray, van Achterberg, Wyke, & Cox, 2010)

The Talking Mat[®] concept consists of a structured approach to communication, in which the placement of symbols or pictures on a mat allows people with communication disabilities to express their views. The three main components to the materials needed are a topic or question, represented by an appropriate image, a rating scale representing positive, neutral or negative responses, and a set of options, also represented by appropriate images. The options images are then placed on the scale by the person with the communication disability, to indicate their response to each option. For example, in a discussion about food preferences, the topic would be represented via a picture of a meal; the rating scale would reflect likes or dislikes, and the options would include images of a range of possible food choices for that meal. The person would then place the options cards on the mat according to their like or dislike for each one. A more complex discussion could involve a topic of life decisions after a stroke, with options reflecting where a person might choose to live, and what care they might require. Following the interaction, the healthcare professional takes a photographic record of

the completed mat. This record can then be used for further conversations, for sharing with others, or for recording for research purposes.

Talking Mats[®] have been used in individual interactions (e.g. Murphy et al., 2010) and in groups (Murphy, 2006) to achieve a variety of aims involving the expression of views and preferences. It is important to note that Talking Mats[®] are a tool for conversations which can be thoroughly pre-planned by the researcher or healthcare professional. They are designed to enhance communication of choices and opinions, but are not a tool for spontaneous conversation, since the framework relies on the use of prepared materials. The utility and effectiveness of Talking Mats[®] has been reported in the research literature.

Murphy (2000) trialled the use of Talking Mats[®] with twelve participants with receptive and expressive aphasia, ranging from mild to severe, using the framework to explore participants' opinions about their quality of life. She found that in participants with stable neurological symptoms, all were able to participate in the interactions and express their views, and concluded that this is a suitable approach for use with people with aphasia in clinical and research interactions.

A later study by Murphy, Tester, Hubbard, Downs, and MacDonald (2005), involved ten frail, older care home residents in Talking Mats[®] interviews about their current life. Similar to Murphy's earlier (2000) findings, they report that the participants enjoyed using this framework, and that it enhanced their ability to give their personal views. The authors comment that the use of a scale provided a means of asking questions in a neutral way, resulting in authentic responses.

Murphy and Cameron (2008) studied the interactions of forty-eight young adults with intellectual disabilities in separate individual interviews, communicating via their usual means versus Talking Mats[®]. The findings suggest that the use of Talking Mats[®] improved both the person's understanding of the issues discussed as well as the researcher's understanding of the person's views, and the apparent confidence, engagement, and satisfaction of the person with intellectual disability. They also found that the interaction lasted longer, with sustained attention and more topics discussed than in the usual communication condition.

Murphy et al. (2010) report a similar finding, in a study of conversations with people at different stages of dementia. Thirty-one people with dementia were recruited from local services and grouped according to carer description into early, moderate, and late stage dementia. Each person was interviewed by a researcher in three separate conditions at different times: Talking Mats[®], structured, and unstructured conversations on the topics of activities, people, environment, and self. They found that the use of Talking Mats[®] resulted in

a reduction in perseveration and more sustained attention to the topic than either structured or unstructured approaches, particularly in those individuals with moderate or late stage dementia.

In summary, Talking Mats[®] have been found to provide effective support to people with a range of communication disabilities for communicating opinions and choices within a structured context. The applicability to the current study suggests that Talking Mats[®] may provide both a means of supporting communication, and collecting individual data during the focus groups. As noted by Murphy et al. (2005), this approach should be used together with other methods for facilitating communication with people with specific communication disabilities. It is notable that in all the above studies, Talking Mats[®] have supported individual interactions, therefore the application of an individual Talking Mat[®] framework to a group context requires consideration.

4.3.3 Identification of principles for facilitating people with aphasia to participate in focus groups

The above discussion has demonstrated how people with communication disabilities, including aphasia, can participate in a range of activities, including informal conversations and more structured research, to give their opinions. In this section, the principles for the full participation of people with aphasia in focus groups will be identified, with reference to this literature.

4.3.3.1 *Feeling comfortable in the group*

In order for people with aphasia to share their opinions, it is important for them to feel comfortable and supported within an environment which recognises the presence of aphasia. Evidence from the literature supports that an overt recognition of aphasia within the group from the outset is important (Kagan & Gailey, 1993; Luck & Rose, 2007; Lyon et al., 1997).

The principles of Supported Conversation also include the assumption that a person with aphasia is a competent adult, with their own expertise on the subject of aphasia.

4.3.3.2 Supporting comprehension

In order to participate fully, people with aphasia need to understand the aims of the focus groups and the proposed overall plan for the meeting. In addition, they need to understand the purpose and procedure for each of the activities within the focus group to ensure that their responses are cogent. Finally, they need to be able to follow the ensuing discussion so that they can hear the views of other members of the group. Supported Conversation techniques encompass a range of verbal and non-verbal strategies. A key factor in the implementation of such techniques is that the communication partner should be able to deploy the various techniques flexibly, responding to the needs of the person with aphasia both proactively and reactively. That is, resources such as visual materials must be prepared ahead, and be available for use in the interaction as necessary in order to achieve the aim of "maintaining the feel and flow of conversation" (Kagan & Gailey, 1993:214).

Two studies reviewed above reported that visual materials were used to provide support to the participants' understanding of the objectives and procedures (Dalemans et al., 2009; Kerr et al., 2010). Dalemans et al. (2009) report the presentation of a PowerPoint summary of the issues to the whole group for discussion, but neither study specifies exactly how the visual resources were used in the course of the focus group discussions. The presence of only one facilitator in these studies may have reduced the potential for flexible deployment of the resources.

4.3.3.3 Supporting expression of individual views

The purpose of focus groups is to collect individual views, therefore establishing the means for people with aphasia to communicate their opinions is critical. People with aphasia need time to think and time to formulate their own opinions in response to the focus group topics. They may also need support to contribute to the discussion, possibly reviewing their opinions in the light of what others say.

Talking Mats[®] have been shown to be an effective tool in supporting people with communication disabilities, including aphasia, to express their views (Murphy, 2000; Murphy et al., 2005; Murphy, 2006; Murphy & Cameron, 2008; Murphy et al., 2010). The framework provides a means for people to consider the issues and options on an individual basis, interacting physically with the materials in a defined physical space.

In the case of the study by Murphy (2006), all participants in the group were facilitated to interact with a single Talking Mat[®], with the aim of gaining consensus. Where individual opinions are sought, the presence of the physical mat and its contents then provide a point of reference for further discussion. The Talking Mats[®] approach is consistent with the strategy proposed by Luck and Rose (2007) of providing materials for participant to respond to, rather than expecting participants to generate ideas. In this context, therefore, the provision of an individual Talking Mat[®] for each participant, and an individual facilitator to support the use of this, is indicated.

Evidence suggests that techniques to facilitate individual expression of views in one-to-one interaction are now well established (Kagan & Gailey, 1993; Lawson & Fawcus, 1999; Luck & Rose, 2007: Lyon et al, 1997). Although none of the reviewed studies, aside from Garcia et al. (2000), provided additional communication support, this principle is proposed by Finch and Lewis (2003), who advocate the allocation of a lead moderator and additional facilitators to support individuals in the group. A summary of the principles for facilitating communication, and the means to achieve these in the focus groups, is shown in Table 4.1 below.

Principle	Means	Evidence
All participants to feel comfortable and supported, empowered to express their views freely.	Acknowledgement of aphasia. Overt acknowledgement of competence and expertise as people living with aphasia. Communication support provided throughout group process.	Kagan & Gailey (1993), Lyon et al. (1997), Luck & Rose (2007).
All participants to understand the aims and overall plan of the groups.	Use of visual resources.	Dalemans et al. (2010).
All participants to understand the aims and procedures for each activity.	Use of Talking Mats® framework.	Murphy (2000), Murphy et al. (2005), Murphy (2006), Murphy et al. (2010).
	Use of Supported conversation techniques and Total Communication.	Kagan & Gailey (1993), Lawson & Fawcus (1999).
All participants to be able to express their individual opinion.	Use of Talking Mats [®] framework.	Murphy (2000), Murphy et al. (2005), Murphy (2006), Murphy et al. (2010).
- p	Use of Supported Conversation Techniques and Total Communication.	Kagan & Gailey (1993), Lawson & Fawcus (1999).
All participants to be able to respond in the discussion.	Additional facilitation provided for individuals.	Garcia et al.(2000), Finch & Lewis (2003).

Table 4.1 Principles for facilitating participation of people in focus groups

4.4 Discussion of suitable methods for the analysis of focus group data

The previous section identified methods to support the communication of participants in the iterative design process. This was done to ensure that the methods used are maximally productive in generating useful data for subsequent analysis, contributing to the overall trustworthiness of the findings.

In this section, the issues relating to the analysis of focus group data will be discussed, firstly with regard to the qualitative discussion data, and secondly for the data which will arise from the use of Talking Mats[®].

4.4.1 Background literature and discussion of data analysis methods

It is anticipated that focus group methods will yield qualitative data in the form of transcripts of discussions in the focus groups alongside ratings data generated by the use of the Talking Mats[®] framework with individuals within the groups. The topics for these discussions will be guided by the research aims, and participants will be encouraged to contribute their personal opinions. It is hoped that this approach will result in a rich understanding of the topics from the perspective of people with aphasia, consistent with the aims of qualitative research (Damico, Simmons-Mackie, Oelschlaeger, Elman, & Armstrong, 1999; Denzin & Lincoln, 1998). The choice of specific methods is made on consideration of issues including the nature of the data and the purpose of the analysis (Richie, Spencer, & O'Connor, 2003).

This section will address methods for analysing the data arising from group discussions and the individual opinions expressed via Talking Mats[®].

4.4.1.1 Analysis of discussion data

Options for analysis of qualitative data range from inductive approaches, relying on interpretation of the data on the part of the researcher to deductive approaches such as Content Analysis, which quantifies participant responses to pre-determined criteria (Bryman, 2008).

Phase 2 of the current study explores participants' views about accessibility of health information and specifically about the presentation of health information on a novel template. The analysis seeks to capture the responses of participants to the template materials presented during the focus groups and to understand the reasons for their responses. It does not aim to provide an account of people's phenomenological experience in relating to information. Previous research into accessible information for people with aphasia has, as discussed, pursued a positivist research tradition, and no studies have yet explored the issues in depth from the perspective of people with aphasia. Since this study aims to explore individual responses within a group context, it needs to allow for different perspectives on the issues discussed. A risk of using a deductive approach, such as Content Analysis, which quantifies the responses given by participants according to the researchers' predetermined knowledge or interest (Bryman, 2008), is that new insights from the data may be missed.

An inductive approach to data analysis relies on the interpretation of the data by the researcher, in a reflexive, more intuitive process (Patton, 2002), the degree of interpretation depending on the nature of the data collected. Methods such as Interpretive Phenomenological Analysis (Smith, Jarman, & Osborn, 1999) are used to analyse data which may be concerned with highly individual experiences or perspectives; for example, families' perceptions of follow-up care of children who had survived cancer (Earle, Davies, Greenfield, Ross, & Eiser, 2005). Whilst attention to the individual perspective is a clear strength of such an approach, it is arguable that the data in this study are unlikely to be sufficiently complex to require a high degree of interpretation.

Framework analysis (Richie & Spencer, 1994; Richie, Spencer, & O'Connor, 2003) is a form of thematic analysis which follows a systematic process, from the data, via the analysis, to the presentation of the findings. It was developed in the area of applied policy research, and was designed to meet the requirement of being accountable to commissioners of research. A principle of Framework analysis is that the data analysis should be transparent, and Richie, Spencer, and O'Connor (2003) emphasise the importance of being able to return directly to the data to show how the analysis has been conducted. The use of such a method answers the potential criticism levelled by Braun and Clarke (2006) at studies using thematic analysis "…of themes 'emerging' from the data…" (p.80) without a clear rationale.

Framework analysis can generate a novel thematic framework, through systematic inductive analysis. Alternatively, data may be applied to an existing "typology" (Ritchie, Spencer and O'Connor, 2003: 248). As Pope, Ziebland, and Mays (2006) note, "... although framework is heavily based in the original accounts and observations of the people studied (i.e. it is 'grounded' and inductive), it starts deductively from the aims and objectives already set up for the study." (p.72). Thus, Framework analysis is typically a more deductive process, seeking answers to the research question rather than, as might occur in a more inductive study, developing new questions on the basis of initial inductive analysis. According to Richie, Spencer, and O'Connor (2003), two broad stages are followed when conducting Framework analysis. The first, data management, involves a methodical process of familiarisation with the data, identification of an initial thematic framework, and the indexing and sorting of the data using this framework. The second stage of analysis involves abstracting the data from the transcripts, collating extracts within identified themes in a process termed "charting" (Richie & Spencer, 1994:182), and starting to develop explanations of the concepts represented in the data. Whilst the specific processes undertaken will depend on the aims of the research, the priority in Framework analysis is the ability to return directly to the data to show how the associations and interpretations are made. In keeping with other qualitative methods of data analysis, Richie, Spencer, and O'Connor (2003) note that the process of analysing the data continues throughout each stage.

4.4.1.2 Analysis of the Talking Mats® scale

As discussed above, Talking Mats[®] provides a framework for participants to express their individual views. With the mat in front of each person, participants also retain a tangible means of referring to their opinions within the group discussion, with the support of a communication facilitator.

In the context of research, Talking Mats[®] can function both as a framework to support participants' communication and as a tool to record participants' views. In considering the issues for the analysis of such data, the reliability and trustworthiness of the scale presented to participants is important.

A Likert scale is widely used in research to gather views (Bryman, 2008). A statement is presented, and the points on the scale represent a continuum relating to the intensity with which a study participant agrees or disagrees with the statement. Scales are constructed to reflect the research aims and then frequently converted to a numerical scale for the purposes of data analysis. The Talking Mats[®] scale, in contrast, represents categorical options including negative, positive, and 'unsure' and participants respond by placing an image on their chosen option.

A number of studies which used Talking Mats[®] to collect data were discussed in Section 4.3.2. Of these, two studies report the data generated from the use of the framework. Methodologically, these two studies report varying treatment of the scale used. Murphy et al. (2005) interviewed ten older people on four topics related to their everyday lives using a three-point Talking Mats[®] scale, representing whether the person felt "unhappy", "not sure" or "happy" (p. 97). The results were reported descriptively, using these terms. Gillespie et al. (2010) explored the views of people with aphasia and their carers regarding their communication using a three-point pictographic scale, indicating positive, negative or unsure (midpoint) responses. For the purposes of data analysis and reporting, the researchers then inferred a five-point numerical scale from the original visual three-point scale.

Murphy et al.'s (2005) data analysis reports directly the choices which participants made during the activity. In contrast, the approach taken by Gillespie et al. (2010) might be criticised for introducing a level of interpretation which could subtly change the meaning of the scale. Firstly, participants did not respond to a five-point scale, so the data reported may not reliably reflect the participants' opinions. Secondly, the conversion to a numerical scale might imply a continuous scale rather than the categorical scale to which participants responded. Finally, the mid-point of the scale as presented to participants can be used to indicate either uncertainty or the mid-point of a continuum. By inferring a numerical scale, the analysis removes the option for participants to express uncertainty.

To ensure the trustworthiness of the Talking Mats[®] data, the meaning of the scale needs to be shared by participants and researchers. Subsequent analysis and reporting of the data will thus reflect the intended meaning of the participants, rather than a researcher-led interpretation.

4.5 Summary of chapter

This chapter has considered the methodology required to ensure a rigorous and trustworthy study. A review of the relevant literature has identified suitable methods for generating and analysing qualitative data to address the study aims.

4.5.1 Aims of Phase 2

In Phase 1 of the study, three literature reviews addressing modified formatting, factors affecting language processing in aphasia, and health information topics required by people with aphasia, were integrated into the methods used in the design phase of the project. The aim of this first phase was to extract principles from the evidence base in each of the three areas to inform the design of the prototype Template 1.

The main aim of Phase 2 is to collaborate with people with aphasia and graphic designers in an iterative, collaborative design process. The aims for the focus group were:

 For people with aphasia to provide qualitative feedback on the accessibility of the template materials in the form of individual supported discussions of preferences for specific aspects of Templates 1 and 2, and related group discussion (see Appendices B, C, and D for examples of template materials). Further aims for Phase 2 were identified as a result of the Phase 1 literature review, reflecting the lack of clear evidence in specific areas, detailed below.

Typography:

- For people with aphasia to denote their preferences for accessibility for specific variations of font size (see Appendix E).
- For people with aphasia to denote their preferences for accessibility for specific variations of font style (see Appendix E).
- For people with aphasia to denote their preferences for accessibility for specific variations of emphasis in font such as bold and italics (see Appendix E).

Image type:

• For people with aphasia to denote their preferences for accessibility for specific variations of image type including photographs and line drawings (see Appendix B).

Number of images per concept:

• For people with aphasia to denote their preferences for accessibility for specific variations of number of images per concept, including one image per concept, versus separate images for each key word up to a maximum of three (see Appendix C).

Set of related concepts:

• For people with aphasia to provide qualitative feedback on the accessibility of a set of cards conveying related concepts in the form of individual supported discussions and related group discussion (see Appendix D).

Within the proposed study design, issues arising from participants will be incorporated into the iterative design process, with identification of further specific aims based on the discussions generated in the first focus groups. In the following chapter, the methods undertaken within the iterative design study will be described.

Chapter 5 Phase 2: Methods

This chapter provides a detailed account of the methods followed in the second phase of the study, which took place between January and April, 2011.

5.1 Participant details

5.1.1 Sampling

Purposive sampling was undertaken for the Phase 2 study, aiming to achieve representation of the population with aphasia, and maximum diversity within that population (Bryman, 2008; Patton, 2002). Such representation of the population is important to ensure the transferability of study findings and concepts (Richie, Lewis, & Elam, 2003). The researcher identified three factors as critical to purposive sampling: age, gender, and severity of aphasia.

The number of participants sampled reflected the aim of achieving data saturation, the point when participants have generated all possible ideas for the subsequent analysis (Corbin & Strauss, 2008). Three focus groups were planned in order to achieve data saturation (Teddlie & Tashakkori, 2009). Each focus group was limited to a maximum of five participants, due to the need for individual communication support within the groups (Finch & Lewis, 2003). Given the need for small groups, a target number of between twelve and fifteen people with aphasia was identified for recruitment, allowing for a maximum of five in each group.

Initial recruitment led to seven men and three women agreeing to take part. To ensure a balance of men and women, four more women were recruited, who were willing to take part but wanted the security of premises known to them.

A possible source of bias within the sample was that some of the participants were known to the researcher, due to previous clinical contact.

5.1.2 Inclusion and exclusion criteria

People were considered eligible to participate in the study if they had acquired aphasia following one or more strokes, or other aetiology, in adulthood. Participants were over 18 years of age, and had acquired their aphasia at least one year before participating in the study. They had no known history of other neurological or psychiatric illness, and had adequate hearing and vision (corrected with aids in some cases) to be able to participate in verbal and visual focus group activities. They presented with normal premorbid language function and level of literacy, and spoke English as their main language to a high level of functionality. People with severe cognitive deficits, such that they were not able to grasp the purpose of the activities, or participate in a group, were excluded. Finally, potential participants needed to have an interest in the topic under discussion, and to live close enough to be able to travel to the focus group meetings.

Ability to participate in a supported focus group was judged by the researcher, an experienced speech and language therapist. This was based on observations of interactions in the Communication Support Groups.

5.1.3 Recruitment

Participants for the study were identified by one of two methods. They were approached either through local voluntary sector Communication Support Groups for people with aphasia, or they were contacted via a database of previous research participants with aphasia held in the university department supporting this study.

At an initial meeting between the researcher and the local Communication Support Group organisers, the aims and the methods of investigation were outlined. The researcher and another member of the research team then visited the two groups on consecutive weeks. Basic information about the background to the project was presented, with pictures and illustrations used to support the information. People who expressed an initial interest in taking part were given an accessible project information sheet to take home (Appendix K). They could discuss this with a family member if they chose, and take time to think about the project.

One week after visiting the groups, the researcher telephoned those people who had expressed an interest in the project and asked if they still wished to be involved. Those who were eligible to participate and said they would like to be involved at this stage were invited to attend a first focus group meeting.

The second form of recruitment involved the researcher contacting previous research participants by email or telephone according to previously expressed preferences. These are people who live independently with aphasia. The aims and methods of investigation for the project were outlined, and those who were interested in taking part expressed their interest and availability via a return telephone call or email.

Following the researcher's visits to the Communication Support Groups, a total of eight people with aphasia expressed an interest in participating. One man was excluded due to severe cognitive impairment, as judged by the researcher following informal discussion. Two further people with aphasia were identified via the database of participants. At this stage, seven men and three women had been identified as eligible and had each expressed an interest in participating.

To ensure a balance between male and female participants, four more women were recruited, who were willing to take part but wanted to meet in premises known to them. Exercising care to avoid coercion in recruitment, the researcher negotiated with this group of four women to participate in the project. As a result, seven men and seven women were identified as eligible and interested participants for the study.

5.1.4 Ethical approval for the study

Ethical approval for the study was granted by the Departmental Research Ethics Committee at the Department of Human Communication Sciences at the University of Sheffield on 17th December, 2010. A copy of the approval form can be found in Appendix J.

5.1.5 Constitution of the focus groups

The participants formed three separate groups. On meeting participants at the recruitment stage, the researcher conversed with each person to form a judgement of the degree of communication support they would need in the focus groups.

In allocating people to groups, various factors were considered, including participants' expressed preferences regarding the day of the meeting and their wish to attend with a friend. Groups were formed to ensure that people with a range of communication support needs were included. The three groups were named Group A, Group B, and Group C to ensure accuracy and consistency in recording the data.

Group A consisted of four men who attended one Communication Support Group and one woman who had previously taken part in research projects at the University. Group B consisted of two men and two women from two different Communication Support Groups and one man who had previously taken part in research projects at the University. Group C consisted of the group of four women from one Communication Support Group, who asked to work together at their regular venue. Each group met on two occasions, four weeks apart.

5.1.6 Informed consent to participate

Informed consent to take part in the study was sought at the start of the first meeting for each of Groups A, B, and C (Kruger & Casey, 2009). Researchers interviewed participants individually, using the project information sheet (Appendix K) as a guide to explain the elements of the project again, and completed an accessible consent form (Appendix N). All were reminded of all elements of the consent process, including the ability to stop their participation at any stage without the need to give a reason. Fourteen people with aphasia gave their independent, informed, written consent to participate.

5.1.7 Aphasia profiling of participants

Standardised assessments of language processing were used to assess the type and severity of aphasia in each participant. In addition, further detail about participants' reading comprehension was sought, due to the focus on reading written information in this study.

Selected subtests from the Comprehensive Aphasia Test (CAT) (Swinburn, Porter, & Howard, 2005) were used to provide an aphasia profile of each participant. Additional reading comprehension data were collected using subtests from the Psycholinguistic Assessment of Language Processing in Aphasia (PALPA) (Kay, Coltheart, & Lesser, 1992). The results for each participant are displayed in Table 6.1.

The assessment of the participants' aphasia was undertaken in their own home at a time agreed with them. An experienced speech and language therapist was employed to contact the participants and conduct the assessments. Each participant underwent one or two assessment sessions. The data were gathered during a single visit between the first and second focus group meetings, and up to six weeks after the second focus group meetings. All responses were noted on the published score sheets of the relevant tests, transcribed phonemically, and scored according to the protocols of the assessments used. Data for three participants were collected after the second focus group meeting, due to difficulties in arranging the visits. In these cases, the data were collected by the researcher and a research colleague, both experienced speech and language therapists. The detailed results of the aphasia testing, together with further details of the participants involved, are presented in Chapter 6.

5.1.8 Overview of the participants

All participants live in a large city in Yorkshire. They are all retired from employment, either prior to or subsequent to their stroke. Participants varied in their independence in terms of both their home lives and their participation in wider activities.

All except one are British nationals, with English as their first language. One woman is a German national who has lived in the UK since 1964. She is a bilingual speaker of German and English.

All had suffered one stroke apart from TM (who had suffered three strokes in the course of ten years), BT (who had suffered three strokes within nine months), and MB, who developed

aphasia as a result of viral encephalitis. All were eligible for inclusion according to the criteria described in Section 5.1.2.

Table 5.1 below summarises the participants by group. Information is given about gender, age, time since stroke, aphasia type, and home circumstances. Diagnosis of aphasia type was achieved by applying scores gained on formal language testing to the Western Aphasia Battery Aphasia Classification (Kertesz, 1979), following the algorithm provided by Davis (1993). The information provided is contemporaneous with the focus group meetings.

Group	Person	Gender	Age	Time since stroke	Aphasia type	Home situation
Group A						
	EC	Male	68	6 years	Broca's	Lives at home with wife
	BT	Male	77	9 years	Broca's	Lives at home with wife
	RW	Male	66	11 years	Broca's	Lives at home with wife
	ТМ	Female	75	6 years	Broca's	Lives at home with husband
	GG	Male	65	13 years	Global	Lives at home with wife
Group B	·					
	RP	Male	58	5 years	Anomia	Lives at home with wife
	SE	Male	67	1.5 years	Transcortical Motor	Lives at home with wife
	NH	Male	68	4 years	Anomia	Lives at home with wife
	OS	Female	67	4 years	Wernicke's	Lives at home with husband
	JB	Female	80	3 years	Transcortical Sensory	Lives alone, supported by family locally
Group C						
	SG	Female	71	5 years	Broca's	Lives alone, independently
	MM	Female	82	15 years	Broca's	Lives alone, supported by family
	MH	Female	76	11 years	Global	Lives alone, supported by family
	MB	Female	80	10 years (viral encephalitis)	Wernicke's	Lives alone, supported by family

Table 5.1 Overview of the participants

5.2 Study design

Phase 2 involved collaborating with people with aphasia and the graphic designers in an iterative design process. The same people participated throughout, each participant attending two focus group meetings. During each focus group meeting, qualitative discussion data and Talking Mats[®] ratings were collected.

In the first focus group meeting, people with aphasia provided structured feedback on the prototype Template 1 materials. Following this meeting, feedback provided to the graphic design team informed the development of the second version of the prototype, Template 2.

In the second focus group meeting, people with aphasia provided feedback on Template 2, after which feedback to the graphic design team informed the third and final template, Template 3. Table 5.2 summarises the iterative design process followed in Phase 2.



Table 5.2 Phase 2 Iterative design process

5.3 Materials and resources

5.3.1 Communication Support for participants with aphasia

The principles for facilitating communication in the focus groups identified in Chapter 4 were followed, and the techniques identified were employed to provide a maximally accessible communication environment for all participants.

5.3.1.1 Focus group leadership

For each topic, one member of the research team was allocated the role of Group Leader. The Group Leader took responsibility for introducing the topic and the associated activity, allocating time for individuals to complete the Talking Mats[®], and leading the subsequent group discussion.

All other staff took the role of communication facilitators. Further duties, for example operating the audio-recorder, were allocated before each focus group meeting.

5.3.1.2 Communication Facilitators

A Communication Facilitator was allocated to each participant, regardless of the type or severity of aphasia. This ensured communication support for all participants, and promoted the principle of equality in the group, since no individual was identified as having a greater need for support than others.

The eight facilitators included three members of the research team (the researcher, a research speech and language therapist, and a family member of a person with aphasia), two speech and language therapists, a Communication Support Group volunteer, a psychology graduate with experience of people with aphasia, and a linguistics MSc graduate with an interest in aphasia research.

All facilitators received training from the researcher in Supported Conversation techniques (Kagan & Gailey, 1993; Kagan, 1998) and Total Communication (Lawson & Fawcus, 1999) prior to participating in the project.

As it was expected that participants might have difficulty reporting back to the larger group in open discussions, it was agreed beforehand that facilitators could report back following the paired discussions between participants and their Communication Facilitators. In these cases, facilitators checked with participants to ensure that they were accurately reflecting the views expressed, and that the interpretation offered to the group discussion was correct. Time was allowed in the group discussion for this collaboration between participants and their communication facilitators, and, when needed, further explanation or re-capping of ideas.

Where the meaning of participants' output was unclear, the facilitator provided a range of possible meanings for them to reject or accept. When reporting back to the large group, the facilitator suggested the meaning, and the participant was invited to confirm or deny the meaning (Luck & Rose, 2007). Communication facilitators were asked to keep brief field notes of issues raised by participants.

5.3.1.3 Communication resources

The researcher prepared accessible written materials to support participants' understanding of the study aims and focus group plans. Several copies of the focus group schedule were printed out on A3 and pasted on the wall; copies of the schedule were printed out on A4 and left on the table top. This information was also provided in poster form on the wall at several points around the room.

Further communication resources were provided on the group table: pens and paper, alphabet charts, and a paper copy of the Talking Mat's[®] visual rating scale. A flip chart and pens were available to record key words from discussions. Participants were asked to bring any specific communication resources to the focus groups.

5.3.2 Talking Mats® materials

5.3.2.1 The Talking Mat® and visual rating scales

The Talking Mats[®] used in the focus groups were designed specifically for the project, with one mat and associated materials provided for each participant to use at the table. Mats were made of thick card in standard A2 (59cm x 42cm) covered with thick dark green felt fabric. The dark background allowed for the materials to be seen clearly. The dimensions allowed for each participant to have their own work space of a comfortable size. For clarity in recording the data, each participant's mat was identified using a printed and laminated card with the participant's name attached at the top left corner.

A visual five-point rating scale was designed for use with participants with aphasia (see Appendix H). A five-point scale was selected as optimal for participants to express a range of views. The scale was made maximally accessible through the use of familiar symbols and colour on each point of the scale. The scale was colour-printed on white paper in Calibri font with the ticks, crosses and question mark in 90pt, and the symbols in 36pt. The scale was laminated, cut into sections, and placed using Velcro[®] across the width of the top of the Talking Mat[®].

5.3.2.2 Talking Mats® practice items

Materials were devised to introduce participants to the use of Talking Mats[®], in preparation for their use in reviewing Template 1, with an emphasis on judging clarity of meaning rather than personal preferences. Picture materials were devised around the concept of an imagined three-course menu. Items were selected on the basis of likely familiarity for all participants, to maximise comprehension of the activity. Three different photographs for each course were selected to represent high, medium, and low degrees of clarity of picture meaning, based on the researcher's subjective judgement. Pictures were colour-printed and presented on individual laminated cards of approximately 8cm x 6cm, and a set of pictures provided for each participant.

5.3.3 Template 1 for focus group meeting 1

The information materials for review in the first focus group meetings were made up of a set of concepts providing background information about stroke and the function and working of the brain, the design of which was described in Chapter 3. The term Template 1 is used here to refer to the design of the materials at this stage of the study. (See Appendices B, C, and D for examples). The materials for Template 1 were professionally colour printed on mid-weight cards measuring 170mm by 110mm. Each card had a white background, and a matt finish.

There were three elements to the information content of each card. Firstly, a coloured banner was printed across the top of the card. A single word or phrase header to identify the topic was printed in Vectora Black at 24 pt, reversed into white font on this banner. Black weight refers to a heavier font than bold rather than describing the font colour. It has the appearance of bold font to the reader, but retains clarity when printed in white on a coloured background. Secondly, below the header, a sentence to convey the information was printed in Vectora Roman at 14pt in black. Thirdly, below the sentence, one, two, or three images which supported the written information were printed. Variations in image style and number of images were provided for review by the participants, as detailed below.

5.3.3.1 Variations in Template 1 presented to participants

As described in Section 3.6, pairs of cards were produced to explore participants' preferences for variations in image style and number of images per sentence. To investigate preferences for image style, cards were presented which conveyed identical concepts, but which were illustrated either by colour photographs or by coloured line drawings.

To investigate participants' preferences for variations in number of images per sentence, pairs of cards were presented which conveyed identical concepts, but which were illustrated either by a single image for the sentence, or by a separate image for each key word in the sentence, up to a maximum of three images.

5.3.3.2 Materials for exploration of text preferences

Text-based materials were devised to explore participants' preferences for a range of font sizes, a range of font styles, and different methods of emphasising keywords in the written text.

These variations were not included in the template but were presented in separate written materials. All text was printed in black ink on white card. Each exemplar was presented on a separate laminated card of 6cm x 4.5cm. Materials were provided for each participant in the focus groups. Examples of these materials are provided in Appendix E.

To investigate font size, a short phrase was printed in Arial font in four different sizes: 10pt, 12pt, 18pt, and 26pt, each presented on a separate card.

To investigate font style, the same phrase was printed in two font styles to contrast serif versus sans serif fonts: serif Times New Roman 12pt and sans serif Arial 12pt. Again this involved separate cards, one with each style.

To investigate emphasis of keywords, the same phrase was printed in Arial 12pt using four different styles of emphasis applied to the final single key word of the phrase: bold, italic, underlined, and large, each style presented on a separate card.

5.3.4 Template 2 for the second focus group meetings

Template 2 was developed from Template 1, with modifications based on the feedback from the first focus group meetings, and were professionally produced in the same way as Template 1.

The information materials for review in the second focus group meetings involved a set of concepts providing information about aphasia and emotions after stroke, depicted using Template 2. The choice of these topics was determined though the literature review (see Section 2.4).

5.4 Procedure

Each participant was invited to attend two separate focus groups, of approximately two hours in duration and one month apart. They underwent language assessment at home in between attending the first and second focus groups.

5.4.1 Preparation for the focus groups

A letter of confirmation was sent to all prospective participants, which was written in an accessible format, giving information about the venue, the transport arrangements, and the start and finish times of the focus group meeting (see Appendix L). Information about the date and time of the second focus group meeting was given to participants at the close of the first session. This was followed up by the transport arrangements and a letter of confirmation one week before the focus group meeting.
The fourteen participants were allocated to one of three groups. As each person was allocated to a specific group, they attended with the same people for both meetings. The focus groups met in the same place on each occasion. The meetings took place in purpose-built facilities, with easy physical access and a quiet environment which ensured uninterrupted discussions.

A structured protocol was developed for each focus group meeting to ensure clarity of purpose and consistency in addressing the research aims across each of the three participant groups (see Appendix O). This specified the focus group topics and activities and their timings, and named the Group Leader for each. Specific equipment and audio recording requirements for each activity were also detailed.

Facilitators were briefed by the researcher before each group to ensure that all were familiar with the research questions and aims, and received a copy of the protocol for their reference during the focus group. For each activity, a designated facilitator was assigned the role of Group Leader, with all others acting as Communication Facilitators, each paired with one person with aphasia.

5.4.2 Communication support within the focus groups

To ensure appropriate communication support for individuals, each Communication Facilitator was allocated to a participant and briefed on their aphasia and likely individual communication needs. At the start of each session, each participant was partnered with their Communication Facilitator, who supported them throughout the focus group meeting using Supported Conversation techniques and Total Communication, where needed. For each activity, these pairs worked together initially. Talking Mats[®] were used as directed by the Group Leader, depending on the aims of the activity. Feedback to the larger group discussion was then given by the participants, where possible, supported by their Communication Facilitators. Frequently, the Communication Facilitator offered detailed feedback from their individual discussions, on behalf of the participant; in these instances, the group leader sought clarification from individual participants that their views were being clearly expressed. Where non-verbal communication was used, the Group Leader reported this verbally for the audio recording, to provide clarity for data analysis. Communication Facilitators also noted any significant non-verbal communication.

5.4.3 Focus group meetings

5.4.3.1 Introductions and settling in

At the start of each focus group meeting, participants were welcomed to the venue and shown the facilities. The process of obtaining final informed consent to participate in the study was completed at the venue, immediately before the first focus group meetings.

Introductions of all participants and facilitators were undertaken by the group leader. Participants were thanked for coming, and all were encouraged to share their opinions as experts in stroke and aphasia. The group aims were explained and an overview of the schedule given in a structured form, to ensure that all were familiar with the plan. Participants were invited to ask questions at any time, and to leave the room for any reason if they needed to. Following their consent to audio recording of the discussions, the recording equipment was pointed out to all members of the groups, with information about when recording would take place.

5.4.3.2 Introduction to the Talking Mats® framework

An introductory activity was carried out, to familiarise participants with Talking Mats[®] and to practise using this framework to rate accessibility of picture materials. The Group Leader directed the Communication Facilitators to present each participant with the three pictures of the same item. Participants could seek clarification of the task and discuss the picture with their Communication Facilitators if they wished. Participants then placed each picture on the Talking Mat[®], underneath the visual scale, to indicate how well each picture conveyed the meaning of the item. The Communication Facilitators reminded participants, if necessary, to focus on the strength of the pictures for conveying menu information, rather than personal preferences for the food. Three pictures were presented at a time for participants to consider, with items from each menu course remaining on the Talking Mat[®]. Once all pictures had been rated, each participant's Talking Mat[®] was photographed by the Group Leader. The Group Leader then clarified that all participants were comfortable with the use of the Talking Mat[®]

5.4.3.3 Presentation of samples of Template 1 for discussion

Participants' responses to specific aspects of Template 1 were sought using the materials devised for the purpose, within a Talking Mats[®] framework. To address each of the aims of the focus group, the same procedure was used in two separate activities, as follows.

In each activity, participants were shown six examples of Template 1. These were presented in the form of three pairs of cards, in which each pair conveyed the same concept, but varied in one dimension, namely image type or number of images per concept.

Participants were shown one pair of cards at a time. They were invited to place the cards on their Talking Mat[®], closest to the point on the visual rating scale which represented their opinion of each card. Time was allowed for participants to complete the rating of each pair of cards, and to discuss this with their Communication Facilitator if they wished. The Group Leader waited for each participant to place each pair of cards before asking Communication Facilitators to present the next pair.

Participants could seek clarification if they were unsure of the task, and Communication Facilitators offered assistance if they thought that their partner might be having difficulty with the task. Time and space was allowed for participants to consider their own views and to talk about the issues with their Communication Facilitator. The importance of participants expressing their own opinion was emphasised.

The Talking Mats[®] were photographed by the Group Leader once all the cards in each activity had been rated. The Group Leader then initiated a whole group discussion, asking participants and their Communication Facilitators to feed back their comments from the Talking Mats[®] activity. Participants were invited to use their Talking Mats[®] as a point of reference, and to comment on the reasons for their preferences. At all times, the Group Leader and Communication Facilitators emphasised the researchers' interest in gaining the individual opinions of the participants, and emphasised that there were 'no right or wrong answers'. Time was allowed for the discussion to develop and it concluded when all who wanted to contribute had done so.

This procedure guided the following activities:

- Seeking participants' preferences for the specific variations of image type of photographs and line drawings. For this, three pairs of cards, each pair conveying the same concept, but using either photographic or line drawing images, were presented (see Appendix B).
- 2) Seeking participants' preferences for specific variations of number of images per concept. For this, three pairs of cards were presented. In one set, the information was presented with a single sentence above a single image. In the other set, the sentence was separated out with space between each of two or three key concepts, and

separate images to represent each key concept were presented below the sentence (see Appendix C).

5.4.3.4 Presentation of materials comparing font size, style and emphasis

Participants' responses to specific variations in font size, style and emphasis were sought using the materials devised for the purpose (see Section 5.3.4) within a Talking Mats[®] framework. A similar procedure to the presentation of Template 1 was used, but here, the full set of materials addressing each issue of font size, font style and emphasis in font were presented to the participants by the Communication Facilitators together, in three separate activities.

For each activity, the same short phrase was used. To support participants' understanding of this phrase, the Communication Facilitators placed a picture to illustrate it at the top of each participant's Talking Mat, above the visual scale.

The participants were invited to compare and rate the accessibility of the written information on the cards using the Talking Mats[®]. The Talking Mats[®] were then photographed by the Group Leader once all the cards in each activity had been rated. As previously, a whole group discussion was then led by the Group Leader.

5.4.4 Compilation of feedback to the graphic designer 1 (first iteration)

After the first focus group meetings, the researcher compiled key information from the data collected (see Section 1.5) to feed back to the designer. This iteration informed the development of Template 2. The qualitative data, Talking Mats[®] ratings, and field notes were summarised.

The researcher met with the designer to provide the feedback. Specific feedback about individual examples of Template 1 was presented in a table showing all examples of the template, with participants' comments and requests for alterations recorded. Any comments applying to the whole set of materials were recorded outside the table, on the same document. Examples of Template 1 were used to illustrate the participants' feedback.

The designer then worked through the written feedback, applying changes to Template 1, to produce Template 2.

5.4.5 Second focus group meetings

5.4.5.1 *Preparation for the second focus group meetings*

At the end of the first focus group meetings, participants were thanked for their contributions. Each participant was given a written sheet with information about the date and venue of their next focus group meeting (see Appendix M). A letter of confirmation was sent to all participants one week before each of the second focus group meetings, and arrangements made for those people requiring transport.

Planning of the second focus group meetings was undertaken by the research team following the first focus group meetings. This allowed flexibility to explore further any significant issues arising from the first focus group meetings, which might be issues highlighted by participants, or noted by the researchers.

Feedback to the graphic designer resulted in the production of a set of Template 2 materials (see Appendix D). The researchers agreed that these materials should be reviewed as a set by the participants, firstly in discussion with Communication Facilitators and subsequently in the larger group, facilitated by the Group Leader.

5.4.5.2 Identification of further aims for the second focus group meetings

Two issues requiring further attention were identified in the preparatory stage for the second focus group meetings. In the first focus group meetings, the Template 1 materials were presented as single concepts on one card. As a result of participant comments and reflection by the researcher, it was agreed that further views on the accessibility of different amounts of information should be sought in the second focus group meetings. The second issue related to the materials used to examine typography preferences. The materials used in the first focus groups to examine preferences of font size, style and emphasis were devised by the researcher and were unrelated to the Template 1 information materials under review. In the light of information gathered in the first focus groups regarding typography preferences, the research team agreed that a re-examination of emphasis in font preferences, presented on the template, was required. The following aims were agreed by the research team:

- To seek participants' preferences for the specific variations in amount of information conveyed at one time, using materials showing one, two or three concepts per page (see Appendix G).
- b) To seek participants' preferences for specific variations of emphasis in font, using materials showing emphasis in bold, large font and large bold font (see Appendix F).

A structured protocol was then devised for the second focus group meetings (Appendix P). As with the first focus groups, all Communication Facilitators were briefed beforehand and received a copy of the protocol for their reference.

5.4.5.3 Materials for the investigation of amount of information and emphasis in font preferences

To investigate participants' preferences for the amount of information conveyed at one time, materials were devised which presented one, two, or three concepts. To present one concept, the researcher selected one information card; to present two and three related concepts on one page, the researcher selected two or three related cards. These were then photocopied in high quality colour onto A4 paper showing two or three concepts on each page (see Appendix G).

To investigate participants' preferences for different methods of emphasis in font within the template, the graphic designer was asked to produce four versions of Template 2 materials, in which the design differed only in the means of emphasis of key words, selected by the researcher, as follows: normal (14pt), bold font (14pt), large font (18pt), and large bold font (18pt) (see Appendix F for examples).

5.4.5.4 Presentation of samples of Template 2 for discussion

Three topics were discussed in the second focus group meetings: participant responses to a set of cards conveying information about aphasia, amount of information per page, and emphasis in font.

5.4.5.4.1 Review of Template 2 presented as a set of cards

As identified in Chapter 2, no previous research has examined concepts presented as a set of related concepts. The evidence concerning language processing in aphasia and effective formatting methods suggests that concepts should be presented singly. Explanations of more complex topics therefore require a set of related concepts.

Participants were shown a series of eight cards, which formed a set of information about aphasia. These were formatted in the Template 2 design (Appendix D).

The Group Leader directed the Communication Facilitators to show the examples to the participants in sequential order 1 - 8. Once all eight cards had been presented, Communication Facilitators asked participants for their opinions on the accessibility of the set of cards. Talking Mats[®] were not employed for this topic, since participants were asked to review the set rather than individual examples.

Once all participants had expressed their views in paired conversations, the Group Leader invited participants and Communication Facilitators to feed back to a whole group discussion. The Group Leader firstly asked for feedback about the concept of a set of cards to convey information. She then invited participants' comments on each card. In each case, feedback was given by both participants and Communication Facilitators and the Group Leader endeavoured to ensure that the discussion encompassed all opinions.

5.4.5.4.2 Amount of information and emphasis in font

A similar, shorter procedure to that followed in the first focus groups was followed to orientate participants to using Talking Mats[®] again. Participants then rated the materials using Talking Mats[®] framework following the same procedure as in the first focus groups for the following two activities.

- 1) Seeking participants' preferences for the specific variations in amount of information per page, materials featuring one, two, and three concepts per page (Appendix G).
- Seeking participants' preferences for specific variations of emphasis in font, including Template 2 materials showing emphasis in bold font, large font, large bold font, and normal font control (Appendix F).

Again, Talking Mats[®] were photographed, and a group discussion was initiated by the group leader. The focus groups were closed with thanks to the participants for their contributions and expertise.

5.4.6 Compilation of feedback to the graphic designer (second iteration)

The same process conducted after the first focus group was completed by the researcher, as outlined in Section 5.4.4.

5.4.7 Production of Template 3

The two iterations described resulted in the production of the final version of the template, Template 3. Full feedback from focus group participants was incorporated in the collaboration between the researcher and the designer, ensuring a completion of the iterative cycle, from the first focus groups to the final template.

5.5 Data recording, preparation and analysis

5.5.1 Overview

The data for the study consist of audio recordings of focus group discussions, associated field notes from facilitators, and digital photographs of Talking Mats[®] completed by individual participants. The audio recordings include the spoken contributions in the groups of all project participants and facilitators. The photographs comprise a record of participants' responses to research questions regarding aspects of information materials, such as image type and number

of images per concept, using a visual rating scale (see Appendix I for examples of the completed Talking Mats[®]).

5.5.2 Audio recordings of discussions

Focus group discussions were recorded using a digital audio recorder. A microphone was placed in the centre of the table, around which group members sat. The recording equipment was placed to one side, operated by an allocated member of the research team, and out of the immediate view of the participants. The recordings took place according to the protocol, capturing the whole group discussions. Periods in the focus groups when participants were completing their individual Talking Mats[®] were not recorded, due to the number of people talking simultaneously. Immediately after the focus group meetings, the recordings were transferred to MP3 files. These files were stored digitally in a password-protected folder. Recordings were stored in separate folders representing each of the focus group meetings, and the different activities within the meetings.

5.5.2.1 Audio-recorded material

As shown in table 5.3, a total of three hours and thirty three minutes of focus group discussion was recorded. The total amount of discussion recorded in the first and second focus groups was similar overall; however, it is notable that in Group C, less data was recorded. This may be a reflection that the participants in Group C required more time to process the information and consider their responses, or had less to say in the discussions.

	Meeting 1	Meeting 2	TOTAL
Group A	35' 57	40' 03	1 hr 16 min
Group B	48' 08	34' 30	1 hr 22 min
Group C	31' 29	23′ 47	55 min
TOTAL	1 hr 47 min	1 hr 37 min	3 hrs 33 min

Table 5.3 Amount of audio data from each group at each meeting in hours, minutes, and seconds

Groups A and B, as shown in Table 5.4 below, generated similar amounts of discussion data. The exception to this is the discussion of font size, style and emphasis, which Group B discussed for twice as long as the other two groups. As can be seen, Group C did not attempt discussions of the topics amount of information or emphasis in font. This group also had less whole group discussion time across all topics, except font size, style, and emphasis.

Meeting 1 discussion topics Meeting 2 discussion topics Template 1 Template 1 Font size, Template Amount of Emphasis Image type Number of style, and 2 Aphasia information in font images emphasis set Group A 11' 32 17' 20 07' 05 31' 23 03' 25 05' 15 Group B 13'00 19' 14 15' 54 27' 56 03' 33 02'01 Group C 09' 24 14' 50 07' 15 23' 47 nil nil TOTAL 33' 56 51'24 30' 16 83'06 06' 58 07' 15

Table 5.4 Amount of audio data from each topic in each group, in minutes and seconds

5.5.2.2 Transcription of audio recordings

The MP3 files of each topic were transcribed by members of the research team, including the Communication Facilitators. This was done to ensure a timely record of the focus group meetings by staff who had been present, with knowledge of the context and different speakers' voices (Poland, 1995). All transcribers received instructions from the researcher regarding the aims, approach, and conventions of the transcription.

All turns of all participants, the group leader, and facilitators were transcribed. In the transcriptions, each turn was labelled with the participant's or facilitator's initials, and a code to indicate whether the speaker was a person with aphasia or a facilitator.

The transcription conventions used were as follows (Atkinson & Heritage, 1984):

• Transcription was orthographic; non-words were transcribed with broad phonemic transcription. Unintelligible content was recorded in round brackets.

- Non-verbal communication was noted in square brackets, e.g. [laughs]
- Brief pauses were recorded with a hyphen, and longer pauses denoted by a sequence of three full stops. No measure of longer delays was undertaken.

• Overlaps between speakers were recorded with both speakers' turns marked in round brackets, on separate lines of transcription. Prosody is not recorded, but obvious questions are denoted by a question mark. Other standard punctuation is used at times in the transcription to add context and meaning; for example, an exclamation mark denotes tone of voice.

To ensure rigour, the researcher reviewed all transcripts whilst listening to the original sound recordings, and corrected all errors (Poland, 1995). A sample of the transcriptions is included in Appendix Q.

5.5.2.3 Data preparation

The researcher prepared the transcribed data for analysis to ensure that the data were complete and accurate. The data were formatted appropriately for ease of initial management and indexing, as well as later retrieval. Transcriptions were stored in Excel workbooks and labelled for each group, each meeting, and each activity within the meeting. The digital photographs of all participants' Talking Mats[®] were inserted at the end of each transcription to add context during the reading and analysis.

5.5.2.4 Data analysis

Framework analysis was selected for this phase of the study due to nature of the data and the research aims, as discussed in Chapter 4. Following Spencer, Richie, & O'Connor (2003), the analysis followed a hierarchical approach, moving from management of the raw data towards a description of the framework and themes, and later, an attempt to interpret the findings. The methods used in each of these stages are described below.

5.5.2.4.1 Familiarisation

The researcher read through the transcriptions of each topic addressed in the focus groups, and listened to all the recorded group discussions, in order to enhance her understanding of individual speakers in context. Notes were made in a 'comments' box within the transcripts.

5.5.2.4.2 Identifying a thematic framework

After several readings of the discussions for each group, notes were made on all emerging issues. This was done to ensure consistency and internal reliability, and to keep any emerging themes close to the data. Overarching themes were avoided, to ensure capture of all data within themes close to their original meaning, and to prevent over-interpretation. Examples were noted of issues in the discussion which fitted within each theme. Constant comparison (Pope, Ziebland & Mays, 2006) was used to ensure that turns allocated to a theme were consistent in meaning within the theme. At this stage, the point of "saturation" (Pope, Ziebland & Mays, 2006:71) was reached as the themes identified in the index covered all issues arising in the data.

The index developed from this list was condensed to a manageable number of headings and subordinate categories (Spencer, Richie, & O'Connor, 2003). A numerical code was then

assigned to the index. All the key issues were represented in this draft of the index, and the process of labelling the data using this index then started.

5.5.2.4.3 Indexing

The researcher applied the numerical index to a section of the data using the prepared transcriptions. This was completed directly into the Excel workbooks saved for this purpose. The index was then applied to all the discussion data covering all focus group meetings.

The transcriptions of different discussions in each focus group were stored as separate worksheets as described above in Section 5.5.2.3. The assigned theme was marked in a separate column on the transcript, adjacent to the relevant sequence of turns. Where a turn was indexed with more than one theme, this cell was copied and inserted.

After indexing each discussion, the researcher sorted the data within each worksheet by theme. Indexed segments were then copied and inserted separately at the end of the worksheet for storage. The text of each indexed turn was copied and re-inserted in red font into the body of the transcription to highlight it. Following this, indexed data from each worksheet were copied into a new worksheet, where it was then sorted again as part of the whole focus group discussion. The merged and sorted data were also retained in the separate focus groups discussions.

During the indexing process, the researcher kept notes both in the comments column of the transcription and in a research journal.

5.5.2.4.3.1 Methods for handling collaborative discussion data

Due to the presence of aphasia, in many instances the meaning an individual wished to convey emerged over a series of turns, rather than in a single turn. This reflects both the collaborative nature of participants with aphasia working with Communication Facilitators to express opinions, and the dynamics of group interactions in which several people contributed to a discussion topic at once. In order to capture authentic and contextualised data, segments of the transcript were indexed together.

5.5.2.4.4 Charting

Initial sorting of data within themes was completed by the researcher as described above. Following this, the researcher created a chart for each theme and each group. Sub-themes were represented in separate columns within the table. At this stage, the researcher recorded themes and primary sub-themes only, so that all segments were visible on the charts, with participants recorded in a consistent order on each chart. Where segments were indexed in more than one theme, this was noted within the table, via a numerical reference underneath the text. As the process of analysis was continuous, this allowed the researcher to reconsider the meanings of the segments. In some cases, segments were then allocated to a single theme rather than multiple themes.

The individual group charts were then collated into whole data set charts. During this process, the data were examined further, particularly any data which were cross-referenced or in the category 'Other'. This completed the initial management of the data, and provided the researcher with a summary of the data for the purpose of a descriptive analysis.

5.5.2.4.4.1 Methods for handling collaborative discussion data in the analytic process

As described above, collaborative turns were indexed together. Retaining the context and meaning of turns into the charting was more problematic, however, since lifting turns out of indexed material into charts resulted initially in a loss of context, requiring the researcher to return frequently to the original transcript for interpretation. To resolve this, the researcher included a description of the collaborative context in the charts.

5.5.2.4.5 Mapping and interpretation

The next phase of the analysis aims to understand the meaning of the themes, abstracted from the raw data, and involves a descriptive analysis of the summarised data (Spencer, Richie, & O'Connor, 2003).

The whole data set charts were examined by the researcher. Working through each chart in turn, an interpretation of the meaning of each segment was noted directly onto the chart.

The descriptions of each segment were then collated and analysed by the researcher to determine overlaps and differences in meaning, then categorised. These categories informed the final version of the framework.

5.5.2.5 Rigour in the data analysis

All stages of the Framework analysis were completed by the researcher, who kept a research journal with notes and comments throughout for later discussion and to ensure a reflexive and transparent process.

To ensure rigour in the analysis of the data, the researcher and supervisor met to talk through this process, with the notes made open to discussion. To test out the framework, a segment of the data was selected for the supervisor to index independently using the thematic framework. Informal training in the framework was given as follows:

1. A description of the process undertaken to date.

- 2. A description of the way the data were handled, capturing collaborative turns between participants with aphasia and facilitators.
- 3. A description of the themes and the emerging framework.

The segment chosen was rich in a variety of themes to allow for the supervisor to use the full framework. This was done in a face-to-face meeting, allowing for discussion and debate where necessary. The researcher had some expectation that this meeting would result in a change to the framework. In fact, there was little need for debate, with this experienced by both researcher and supervisor as a positive and affirmative process, in which the supervisor was readily able to assign indexing codes to the data.

5.5.3 Talking Mats® data

The Talking Mats[®] data for this study consist of participants' ratings of materials using the scale incorporated into the Talking Mats[®] framework. A full record of these ratings was produced in the form of digital photographs of each Talking Mat[®] completed by the participants in the focus group meetings. Participants' initial ratings only were recorded for this purpose.

5.5.3.1 Photographic material generated in the focus groups

This data set consists of 62 digital photographs. As shown in Table 5.5 below, all participants completed a Talking Mat[®] for each of the topics presented. Note that in Meeting 2, Focus group C participants completed a Talking Mat[®] for the topics of Amount of information and Emphasis in font, however as noted above these topics were not followed up with full group discussions.

		Meeting 1 data	Meeting 2 data		
	Template 1 Image type	Template 1 Number of images	Font size, style and emphasis	Amount of information	Template 2 Emphasis in font
Group A	5	5	5	5	5
Group B	5	5	5	5	5
Group C	4	4	4	4	4
TOTAL	14	14	14	14	14

Table 5.5 Number of Talking Mats® data from each group at each meeting

5.5.3.2 Data preparation

The researcher downloaded the digital photographs and stored them in files labelled for each group, each meeting, and each activity. In addition, all photographs were inserted at the end of each section of transcription, so that the two types of data could be viewed together. Each participant's data was identified during the focus group via a card with the participant's name attached at the top left corner of the Talking Mat[®].

5.5.3.3 Data analysis

The photographic material was analysed descriptively. Frequency of responses was noted for each condition presented, to identify trends in the data.

5.5.3.4 Rigour

All photographs were of sufficient quality to be interpreted with ease. Rigour was maintained by completing the analysis directly from the photographic materials. Keeping the photographs with the transcripts also supported the researcher's understanding of both discussion and photographic data.

5.6 Chapter summary

This chapter has provided an account of the methods used in Phase 2 of the study. In the following chapter, the results of this phase of the study are presented.

Chapter 6 Participants

The purpose of this chapter is to describe the people with aphasia who participated in the study, in order to provide context to the qualitative data and the subsequent data analysis.

6.1 Introduction

As described in Chapter 5, three separate groups were formed from the fourteen participants. Some people were already acquainted through the stroke support groups. Some were known to the research team due to previous clinical or research involvement, and this is acknowledged as a possible source of bias in the sample.

In the next section, details of the participants' language and communication will be provided. Reference will be made to language assessment profiles, as well as observations of participants' communication within the focus groups, in order to demonstrate the range of aphasia types and severities represented by the participants in the study.

6.2 Language and communication profiles of participants

Language profiles were constructed from the findings of standardised assessments (described in Section 5.1.7). Subtests from the Comprehensive Aphasia Test (CAT) (Swinburn et al., 2005) and additional subtests from the Psycholinguistic Assessment of Language Processing in Aphasia (PALPA) (Kay et al., 1992) to assess reading comprehension were used. Details of these are provided in Table 6.1 below.

A summary of participants' expressive language ability and communicative effectiveness was made, based on observations during language testing and in the focus group meetings. A judgement regarding the severity of aphasia overall was made by the author using clinical experience, based on a combination of the formal testing and communication performance in the focus groups.

Reading ability was determined with reference to the CAT comprehension of written single words and sentences subtests and the PALPA subtests, as noted above. The language assessment data are summarised in Table 6.1 below.

	Auditory comprehension		Written comprehension			Spoken naming	Repetition		Reading aloud		
Test name	CAT Comprehension of spoken words	CAT Comprehension of spoken sentences	CAT Comprehension of written words	CAT Comprehension of written sentences	PALPA 51 High Imag.	PALPA 51 Low Imag.	CAT Naming objects	CAT Repetition of words	CAT Repetition of nonwords	CAT Reading words	CAT Reading nonwords
n	30	32	30	32	15	15	48	32	10	48	10
Mean SD Range	0.97 0.046 0.83 - 1.00	0.94 0.061 0.81 - 1.00	0.99 0.027 0.90 - 1.00	0.93 0.084 0.75 - 1.00	0.90 0.094	0.82 0.149	0.97 0.035 0.87 - 1.00	0.99 0.021 0.94 - 1.00	0.92 0.160 0.40 - 1.00	0.99 0.022 0.92 - 1.00	0.94 0.120 0.60 - 1.00
Group A											
EC	0.97	1.00	1.00	0.94	0.87	0.80	0.96	0.88	0.20	0.73	0.20
ВТ	1.00	0.91	1.00	0.94	0.87	0.73	1.00	0.94	0.40	0.98	1.00
RW	1.00	0.84	1.00	0.63	0.80	0.27	0.92	1.00	0.70	0.88	0.20
TM	0.93	0.63	0.93	Missing *	0.40	0.15	0.29	0.63	0.40	0.65	0.40
GG	0.43	0.00	0.23	0.06	0.00	0.00	-	-	-	-	-
Group B											
RP	0.90	0.94	1.00	0.81	0.93	0.87	0.88	1.00	0.80	0.94	1.00
SE	0.93	0.94	0.87	0.88	0.67	0.46	0.98	0.94	1.00	1.00	0.80
NH	1.00	0.81	0.97	Missing *	0.87	0.60	0.79	0.88	0.80	0.96	0.60
OS	0.73	0.88	0.90	0.78	0.47	0.47	0.92	0.38	0.40	0.92	0.60
JB	0.90	0.75	0.87	0.75	0.73	0.60	0.67	0.94	0.80	0.92	0.20
Group C											
SG	0.93	0.81	1.00	0.81	0.53	0.40	0.88	0.75	1.00	0.83	0.60
MM	0.93	0.66	1.00	0.66	0.80	0.46	0.46	0.38	0.00	0.46	0.00
MH	0.63	0.06	0.30	0.34	0.33	0.47	0.38	0.84	0.70	0.23	0.70
MB	0.47	0.31	-	-	-	-	-	0.09	-	-	-

Table 6.1 Language assessment data for all participants

Note: Scores marked in bold show where performance is below the published normal range, or fall two or more standard deviations below the mean.

In this section, the language and communication of the participants in each of the three groups will be described. Each group is described in brief. Following this, a vignette for each participant provides an overview of their age, any additional impairments and aphasia type, language and communication performance, and reading ability.

6.2.1 Group A participants

Four participants in Group A were recruited from the weekly Communication Support group, and therefore were familiar with each other prior to the focus groups. One further woman was recruited from the research database, having been previously involved with research projects in the department. All were retired, with ages ranging from 65 to 77, and between six and thirteen years post stroke.

EC is a 68-year-old man with minimal residual physical disability. He has mild Broca's aphasia, with no residual auditory comprehension impairments. His spoken language showed reduced fluency, and slow rate of speech, with reduced syntax and some loss of function words. In connected speech, EC made articulation errors, suggesting a degree of motor speech impairment.

EC's reading comprehension was within normal range on single word and sentence tasks, and for written semantic association tasks.

BT is a 77-year-old man. He uses a wheelchair due to a right-sided hemiplegia. BT has mild Conduction aphasia, with auditory comprehension within the normal range at both single word and sentence levels. He scored at ceiling for object naming, but in connected speech his output was hesitant, with occasional phonological errors in production, and a degree of motor speech impairment.

BT performed within normal range on all measures of single word and sentence level reading, albeit slightly reduced for low imageability words.

RW is a 66-year-old man. He has Transcortical Motor aphasia. Auditory comprehension is within normal range at both single word and sentence levels. His spoken output was effortful, and was restricted to single words and short phrases, displaying reduced syntactic complexity. Although on formal testing, naming performance was within the normal range, he experienced significant word retrieval difficulties in connected speech, representing a moderate level of impairment. RW frequently relied on listeners to interpret his intended meaning. He then repeated their words in confirmation. RW's reading ability was within normal range for single word to picture matching and semantic association of high imageability words. However, he had marked impairment for reading sentences and low imageability words.

TM is a 75-year-old woman. Although she is a native German speaker, she has been a UK resident for over fifty years, and communicates in English, the main language of her household. She has left-sided visual field difficulties. TM has Wernicke's aphasia, with impaired auditory comprehension at sentence level. Her naming ability is significantly impaired, with her spoken output characterised by the use of short, stereotypical phrases, appropriately used in context. Overall, she has moderate aphasia.

Her reading ability is within normal range for a single word to picture matching task. However, she was severely impaired on word semantic association tasks, performing below chance for high and low imageability words. There is no data on her written sentence comprehension but her performance on the CAT spoken sentence comprehension subtest is suggestive of syntactic processing impairment which is likely to be mirrored in written sentence comprehension.

GG is a 65-year-old man. He has right sided hemiplegia, mainly affecting his upper limb, and walks with a stick. GG has severe global aphasia affecting all modalities. On formal language testing, he attempted spoken and written comprehension tasks, but no expressive language tasks.

Auditory comprehension is significantly impaired, scoring at chance at single word level, and nil at sentence level. GG's spoken output was limited to open vowels or single bilabial (m/b/w) consonant and open vowel combinations. GG is able to communicate non-verbally. In focus group discussions he used facial expression, tone of voice, pointing and gesture to convey his responses.

GG's single word to picture matching was severely impaired, and he was unable to attempt either sentence to picture matching or single word semantic association tasks. Despite the severity of his aphasia, he expressed a determined interest in participating in the project, and contributed to all discussions with the support of a Communication Facilitator.

6.2.2 Group B participants

In this group, four participants were recruited from two different support groups and one from the research database. Whilst some were acquainted, they had not worked together as a group before the focus group meetings. Their ages ranged from 58 to 80 years, with all now

111

retired from employment, either prior to stroke or as a consequence. Time since stroke ranged from eighteen months to five years.

RP is a man of 58, with no physical disability. His aphasia profile is consistent with Anomic aphasia, with no auditory comprehension deficits. He experiences word finding difficulties with semantic errors in production, but overall his expressive language is characterised by fluent, phonologically and syntactically intact spoken output representing a mild language impairment.

Although RP's reading comprehension was in normal range for all tasks, he performed less well on written than spoken sentence comprehension, and reported difficulty with reading text level materials, suggesting a level of residual reading difficulty that the assessment tools may not have been sufficiently sensitive to detect.

SE is a man of 67. He has right-sided hemiplegia and walks with the aid of a stick. His aphasia profile is consistent with residual Transcortical Motor aphasia. Auditory comprehension of single words and sentences is within the normal range. His performance on spoken language tasks in formal testing was within normal range. In conversation, SE is a quiet man who tends to respond to questions posed rather than offering his opinion spontaneously. Whilst he appeared fully engaged in the focus groups, it was difficult to gauge whether his taciturnity stemmed from mild aphasia, his natural personality, or a combination of the two.

Although SE's reading comprehension was within normal range at sentence level, he had difficulty with all single word reading tasks. Word association tasks, particularly involving low imageability words, showed marked impairment, suggestive of acquired deep dyslexia.

NH is a man of 68. He has right-sided hemiplegia, and uses a wheelchair. He has Anomia. His auditory sentence comprehension is on the borderline of normal range, whereas single spoken word comprehension is at ceiling. On formal testing, his naming ability is impaired, and in connected speech, his output is fluent but hesitant, representing a moderate impairment.

He performed within the normal range on reading comprehension tasks involving word to picture matching or semantic association of high imageability words, but low imageability written words presented a difficulty for single word reading. Written sentence comprehension data is missing, but is likely to mirror his spoken comprehension.

OS is a woman of 67 with no physical disability. She has Wernicke's aphasia, with marked auditory comprehension impairment, particularly at single word level. She experienced

occasional word finding difficulties, but in general her spoken output was fluent. Overall, her language impairment is moderate, but her communication in the focus groups masks this.

OS's reading comprehension is just within normal range for both single words and sentence picture matching tasks. However, on semantic association tasks for both high and low imageability written words with no picture support, her reading performance was below the level of chance.

JB is an 80-year-old woman. She has mild physical disability and cognitive changes affecting her short term memory. JB's profile is consistent with Transcortical Sensory aphasia, characterised by impaired auditory lexical comprehension, with relatively spared auditory sentence comprehension. Her spoken language output is fluent, with impaired naming, but relatively preserved repetition. In connected speech, JB's fluent output is frequently difficult to understand due to paraphasic errors, and her communication is more impaired than her language assessment data suggest, resulting in a moderate impairment.

Her reading comprehension is just below normal range for single words, and just within normal range for sentences. She had greater difficulty however with semantic association tasks, particularly involving low imageability words, suggesting the presence of semantic impairment in reading.

6.2.3 Group C participants

In this group, all the participants were known to each other as they regularly attended the same support group. They had expressed a wish to work together for this study, and asked that the focus groups be conducted at their regular meeting venue. Their ages ranged from 71 to 80 years. All had been retired prior to the onset of aphasia, with time since stroke ranging from five to fifteen years.

SG is a woman of 71 with a mild right-sided weakness. She has Broca's aphasia. Her auditory comprehension was within normal range at both single word and sentence level, but with a slight disadvantage for sentences, consistent with a pattern of resolving syntactic impairment. Although she scored just within normal range for picture naming, she experienced significant word finding difficulties in connected speech, hindering her ability to express her opinions verbally in group discussions, and suggesting a moderate aphasia.

SG's reading comprehension mirrored her spoken language comprehension on word and sentence picture matching. However, her ability to perform word semantic associations was impaired, and she scored below the level of chance with low imageability words.

MM is a woman of 82 with mild residual right-sided hemiplegia. On language testing, her profile is consistent with moderate Broca's aphasia, with impaired auditory comprehension at sentence, but not single word level. Formal testing showed marked difficulties with naming and repetition. In conversation, however, MM was more able to use her expressive language than formal assessment suggests.

MM's reading comprehension was unimpaired for single word to picture matching and within normal range for word semantic association of high imageability words. However, she had difficulty understanding more complex written sentences, and her semantic association of low imageability words was at the level of chance.

MH is a 76-year-old woman with mild residual right-sided hemiplegia. She has global aphasia, with marked auditory comprehension impairments at single word and sentence level. She showed impairments for all aspects of spoken language processing. In the focus groups, she did not express her opinions spontaneously, but responded to choices provided through facilitated communication. Overall her impairment was severe.

MH has impaired reading comprehension at both single word and sentence level; her performance on word semantic association was at or below chance for low and high imageability words respectively.

MB is an 80-year-old woman who has aphasia resulting from viral encephalitis ten years prior to the project. Although her aphasia results from cerebral infection rather than stroke, she was keen to participate in the project and indicated that she felt the issues under discussion were relevant to her as a member of the Communication Support Group. MB has severe Wernicke's aphasia, characterised by severely compromised auditory comprehension in the presence of fluent spoken output. She attempted spoken single word and sentence to picture matching assessments, scoring below chance, but was otherwise unable to complete any other subtests of the formal test battery. In conversation, MB demonstrates evidence of pure word deafness. She uses lip-reading and requests repetitions and paraphrasing to assist her comprehension of auditory language. She was unable to complete any of the reading comprehension assessment tasks.

6.3 Summary

This study aimed to involve people with a range of aphasia severities, and this chapter has described the breadth of participants' language processing, including the varying degrees of severity and the different aphasia subtypes represented in the sample.

The study included participants with a range of mild, moderate, and severe aphasia. Although the individual profiles varied, four had mild (EC, BT, RP, SE), seven had moderate (RW, TM, NH, OS, JB, SG, MM), and three had severe aphasia (GG, MH, MB).

Seven different aphasia subtypes were represented by participants in the study. Those with non-fluent aphasia subtypes included two participants with Global, three with Broca's, and two with Transcortical Motor aphasia. Of the participants with fluent aphasia, three had Wernicke's, one had Transcortical Sensory, one had residual Conduction aphasia, and two had Anomia.

All participants had some degree of impairment in reading comprehension, even though in some cases this was residual (EC, RP), or mild, affecting only low imageability words (BT). For most participants, sentence and single word comprehension was affected to varying degrees (RW, TM, SE, NH, OS, SG, MM) and some showed severe impairment on the reading comprehension tasks (GG, MH, MB). Several participants reported difficulties reading beyond the sentence level, including those who showed no reading impairments on the tasks for single word and sentence reading (EC, RP). Since the formal testing battery included only single word and sentence level tasks, impairments at the text level were not identified.

In the following chapter, the results of the analysis of the focus group data will be presented.

Chapter 7 Thematic framework

This chapter presents the results of the Framework analysis undertaken of the data generated in the focus group meetings. The focus group discussions generated qualitative data, yielding insights into the experience of people with aphasia in their interaction with written healthrelated information. The qualitative data consist of the transcribed data, described in Section 5.5. All data analysed here are from the three group discussions in both sets of focus group meetings, and are primarily the comments or assisted comments of the participants. Participants' ratings of the accessibility of specific elements of the information were also generated by asking participants to place cards showing Template 1, Template 2, and variations in typography on a Talking Mats[®] rating scale.

The results of the qualitative and ratings data are presented here in an integrated form, in order to avoid repetition, to "tell the main story" (White, Woodfield & Richie, 2003:300), and to provide "hybrid vigor" (Miles & Huberman, 2002:396). The results of the qualitative data analysis provide the themes and narrative, with the ratings data presented at the end of the theme considered to best accommodate this data. Congruence and tension between the qualitative and ratings results are addressed in the discussion.

7.1 Thematic framework

This section provides a description of the themes and sub-themes which were generated from the qualitative data, and their conceptualisation into an understanding of the focus group discussions. The methods used to analyse the qualitative data were derived from Richie, Spencer, and O'Connor (2003), and follow the process outlined in Framework analysis, as described in Section 5.5.2.4. Results from the Talking Mats[®] data were then integrated into the framework.

7.1.1 Development of the framework 1 - Data management

In this section, the results of the initial stages of the process of Framework analysis are reported, to demonstrate how the framework was developed.

7.1.1.1 Identifying a thematic framework

Following the phase of familiarisation with the data, the initial thematic framework was developed by noting recurring themes in the data which were pertinent to the aims of the research. This process was completed methodically, working through the discussion topics within each focus group meeting. The researcher generated six themes from the first discussion topic, and a further seven themes from the subsequent discussions. In total thirteen themes were identified for the initial framework. These are summarised in Table 7.1 below.

Themes generated from first discussions		Themes generated from subsequent discussions			
1. See cl	early	7.	Acceptability		
2. Size o	f text	8.	Pictures		
3. Word	S	9.	Words and pictures together		
4. Perso	nal meaning	10.	Approach to information		
5. Timin	g after stroke	11.	Amount of information		
6. Aesth	etics	12.	Personal experience		
		13.	Individuals		

framewoi	ic fi	thematio	Initial	7.7.1	Table	
framewoi	ic fi	thematic	Initial	7.7.1	Table	

7.1.1.2 Indexing

The index developed from the above initial list was condensed to a manageable number of headings and subordinate categories. To do this, the researcher assigned the thirteen original themes to one of seven categories, for example three themes: 'See clearly', 'Size of text', and 'Approach to information' were merged to form Theme 2 'Looking at the information', with sub-themes to retain the detail of the discussion. The complete index resulting from this process is shown in Table 7.2 below.

1. Everyone's different

- 1.1 Personal experience of stroke
- 1.2 Personal experience of aphasia
- 1.3 Personal style
- 1.4 Personal preferences
- 1.5 Individual differences

2. Looking at the information

- 2.1 'see clearly'
- 2.2 Size of picture
- 2.3 Size of text
- 2.4 Approach to information
- 2.5 Presentation of information

3. The meaning

- 3.1 Single words
- 3.2 Phrases
- 3.3 Sentences
- 3.4 Single picture
- 3.5 Picture meaning
- 3.6 Picture style
- 3.7 Picture detail
- 3.8 Word and picture meaning matches ("cos you can see your arms and legs)
- 3.9 Chunking
- 3.10 Picture per word

4. The look of it

- 4.1 Adult style
- 4.2 Picture meaning
- 4.3 Picture style preference
- 4.4 Colour
- 4.5 Font
- 4.6 Consistency in the set

5. Amount of information

- 5.1 Single concept
- 5.2 Linked concepts
- 5.3 Sequence of concepts
- 5.4 'too much'
- 6. Timing
 - 6.1 Early days
 - 6.2 Now
- 7. Other
 - 7.1 Family
 - 7.2 Emotional responses

Two further sub-themes were added to the final index during this process: 6.3 'Recovery' and 7.3 'Suggestions for changes'.

7.1.1.3 Charting

All indexed data were inserted into the charts representing the seven main themes derived from the process of indexing:

- 1. Everyone's different
- 2. Looking at the information
- 3. The meaning
- 4. The look of it
- 5. Amount of information
- 6. Timing
- 7. Other

The meanings applied to the data at this stage resulted in the development of the final framework, forming the descriptive account provided in the following section. As noted by Spencer, Richie, and O'Connor (2003), the analysis is an ongoing process, continuing throughout the period of description and reporting of the data. This allows the researcher to continue to understand the data, making links, and developing new insights.

7.1.2 Development of the framework 2 – Descriptive account

The process of analysis included a consideration of the meaning of the issues arising in the working framework. This resulted in the identification of four overarching themes, which were ordered to reflect the process individuals may typically follow when interacting with information; thus, the first theme, 'Visual access', reflects individuals' initial visual processing of the information materials. In the second theme, 'The look of it', individuals' engagement with information follows from whether it looks right for them. The third theme, 'Understanding the information', addresses how people approach the different elements of the information to understand the overall meaning. The fourth and overarching theme, 'Everyone's different', captures the impact of the individual experience of stroke and aphasia, as well as individual responses and visual styles. The thematic framework is presented in Table 7.3 below.

Themes	Sub-themes	Issues within sub-themes
Visual access	Seeing clearly	Size Clarity Emphasis
	Amount of information	Single concepts Linked concepts
The look of it	Acceptable	To adults with aphasia To individuals
	Design	Image style Colour and font Consistent design
Understanding the information	The words	Aphasia and reading Understanding the words
	The images	Processing images/visual impact Interpreting image meaning
	Words and images together	Words and images work together Layout of words and images
Everyone's different	Experience of stroke and aphasia	Knowledge of aphasia Aphasia is individual
	Individual responses	Identifying with the information Reacting to the information
	Individual visual style	Looking at information Making sense with support

Extracts from the qualitative data are presented to exemplify issues from the discussion. For maximum clarity to the reader, these are presented orthographically, with broad phonemic transcription used to represent incomplete or unintelligible output. Where appropriate, sequences or interactions rather than individual turns are included, to provide a clear sense of the discussion, rather than decontextualized comments. These are provided particularly where a person with aphasia has been supported to develop their contributions by facilitators or other participants. Turns are presented in speaker order, but overlaps are not recorded as they do not add to the meaning of the data. Pauses in spoken turns were not measured for precise length, but were recorded and are represented here orthographically, according to the transcription conventions outlined in Section 5.5.2.2. Extracts follow a consistent format, with the speaker's initials at the beginning of the turn: participants are referred to by their initials,

Group Leaders as GL and the communication facilitators as F1, F2 etc. to represent individual facilitators.

Where appropriate, reference is made to individual participant's language impairment, where this may clarify the form or content of their contributions. In addition, context of the discussion is provided.

7.2 Theme 1 - Visual access

This theme relates to ease of visual access rather than seeing per se. Although many of the participants wore glasses, and clearly sensory access was an issue for some people, this theme is more about the ease with which materials can be visually processed.

The data contributing to this theme are from group discussions in all three groups across both sets of focus group meetings. The materials discussed include Template 1, Template 2, and the cards designed to elicit ratings of font size, font style and emphasis. All participants contributed in some way, although the amount differed across groups and within groups.

The two main sub-themes within this theme were 'seeing clearly' and 'amount of information,' both of which address people's perceptions of how easy it is to take information in visually. Within 'seeing clearly' the size of materials was important, along with the emphasis of particular elements, and the ways in which that emphasis is made. The theme 'amount of information' related specifically to how many concepts people were being asked to process at one time, and the links between these concepts.

7.2.1 Sub-theme 1 Seeing clearly

Size, clarity, and emphasis were identified as important issues within this sub-theme. Data relating to each of these issues are presented below.

7.2.1.1 Size

The size of text was an important issue for participants. In all discussions about font size, participants expressed the need to be able to see the materials clearly and easily. Several issues arose in the discussions of size of text and images in the materials.

The first of these is that participants talk about larger font sizes as being easier to process. In the extracts below, two participants in separate discussions described how size helped them to make sense of the content. OS's words suggest that the larger font size is more readily understandable to her.

OS: like bigger can you see it - explains

NH's comment adds an impression that larger fonts are also easier for him to process.

NH: if it's ease of choosing I would choose large

Conversely, the smaller fonts were not liked, as expressed by SG.

F2: so if the slightly bigger ones are easier to see - what did people think about the small ones? SG: oh no!

The interaction between size of font and speed of processing was raised by NH, who articulated that larger fonts were more accessible to him. Here, he was referring to the phrase 'cup of coffee' which was presented to participants in varying font sizes.

NH: if it was the big - if it was coffee that said coffee - it - I would - coffee - I would get it - sooner than these that are - smaller

On a similar theme of ease of visual processing, BT raised the issue of font size and motivation to read information. The extract below follows on from a wider group discussion in which individuals gave their opinions on font size. Here, BT has rated a large font size as optimum for him, and went on to contrast this with the much smaller font sizes in newspaper articles, which he reported as difficult to read. He explained that reading smaller font size would require greater effort and motivation on his part.

BT: I can't read all the newspapers I want to, unless I'm really interested and then I'll concentrate on the column - but /a/ that's a size /a/ like

The issue of font size, visual processing and acute stroke was raised by EC. After the Group Leader opened the discussion and asked for general comments about what makes it easier or harder to read text, EC commented that he could now see the larger font clearly, but reflected on his experience soon after his stroke saying:

EC: but when I had first had stroke - I couldn't see that at all

He indicated that he would have found it impossible to process any text at all, regardless of size, and described that he had, in the immediate aftermath of the stroke, required visual information via pictures only.

Finally, the font size needs to be appropriate and look right in the context. In this final extract in this section, JB had selected the second largest font as her preferred size; this was remarked upon by her facilitator, and JB commented that the size is "sensible", contrasting it with the largest size. JB: I think if you see it - that's what you - that seems sensible [points to size 18 font] ... that
[points to size 26 font] makes it - shout out
F2: it's too big - it's shouting
JB: yeah

The size of images as a factor in ease of processing also emerged in one of the group discussions. This was introduced to the group discussion by a facilitator, as the issue had arisen in the preceding discussion between SG and her facilitator.

F2: with this picture - the one that you went for the double tick - am I right in thinking - it was because it was bigger?

SG: yeah

Overall, participants agreed that larger materials are generally easier for them to access visually. As JB's extract illustrates however, the font size needs to be in keeping with the overall materials so that there is a balance between visual access and acceptability, an issue which will be addressed in greater detail in Theme Two 'The look of it'.

7.2.1.2 *Clarity*

Related to size of text and images, participants also expressed that the visual clarity of information materials was important for them to start processing the content. This is summed up in an exchange reported by BT's facilitator from their individual discussion.

F1: the word that BT keeps saying is simple

BT: yeah

F1: simple and clear

In the meetings, participants commented freely on the images in the materials under discussion. In the extract below, when BT articulated his reason for preferring one of the images of a brain, he revealed his preference for an image which is easy to access visually.

BT: I like its simplicity

This was echoed by another participant, who, when the Group Leader asked him to say why he preferred line drawings to photographs said:

SE: I just think they're - lot - plainer to see

EC reported a preference for the line drawing style. He indicated that he might favour photographs, but that this was difficult for him to judge because, to him, the photographs presented here were not sufficiently clear.

EC: /ə/ - better photographs might be good in place of that - but it can't see because - /ə/

Clarity in text emerged as an important factor in the discussions around serif versus sans serif fonts. For the participants who had a strong preference, there appeared to be a difference in visual perception of the two font styles. For BT, the perceived darker appearance of the sans serif font appeared to make it accessible.

F1: there was another point that you noted wasn't there - that number 6 [serif] - you liked number 5 [sans serif] better than 6 - you said it was clearer - easierBT: it's clearer - it's er - it's darker isn't it?

Similarly, darkness as an aspect of clarity was repeated in a comment reported back by a facilitator on behalf of GG. In this discussion, the Group Leader had asked the participants directly whether they preferred to see the text printed in black on a white background, or in white on a coloured background, as on the Template 1 materials. GG expressed to his communication facilitator that he found the text in black font easier to see than the white coloured font.

F2: GG was talking about the black writing and preferring it to the white

The meaning of clarity to individuals may vary, as demonstrated above. Analysis of the data suggested, however, that participants wanted information which was 'simple' and 'plain'. The discussions of the different font styles indicate the importance of material which is straightforward and uncluttered.

7.2.1.3 Emphasis

Overall, participants expressed their preference for key words to be emphasised, as in the extract below. Here, TM, together with the facilitator, explained how the emphasis helped her to process the text, highlighting the visual impact of the bold emphasis.

F2: so you like some sort of emphasis - something to

TM: to see

F2: draw it out - to see

The bold and large fonts were liked by participants. BT explained that the bold succeeds in making the key word stand out from the rest of the short phrase.

BT: that one is er is - I like it [bold font]

- F2: the one with the bold
- BT: /bə/ /bə/ it comes out

Opinion varied among participants: some people preferred to see bold emphasis in normal font size; others preferred to see the key word in both large and bold font. Overall, there was agreement that emphasis is helpful, and that the larger and bolder fonts were most beneficial to visual access.

7.2.2 Sub-theme 2 Amount of information

This sub-theme comes under the overarching theme of 'Visual access', because the issue relates to how much people can look at when interacting with information. Participants described their different responses to the amount of information presented on a page before any attempt to process it began.

Two issues were extracted within this sub-theme, reflecting participants' preferences for the amount of information they find accessible. Firstly, when given the choice, most participants preferred to see only one concept on a page at a time. Secondly, for some participants, the presentation of linked concepts on one page was perceived as useful. Data relating to each of these are presented below.

7.2.2.1 Single concepts

Participants provided several different reasons why information presented as single concepts helped them. One benefit was that single concepts helped people to focus their attention and read information more slowly, and thus understand it more fully. The extract below is an interaction between BT and his communication facilitator in which BT has described that he generally attempts to read information at the same speed as he would have done prior to his stroke. After the stroke, however, he found that he frequently did not fully understand the information. Looking at materials showing information about the brain, BT identified that single concepts helped him to slow down his reading pace and focus his attention on the concepts as they are presented.

BT: my trouble is - I start reading I see - I go brain oh yeah - I - I - before I've - before my brain has absorbed that information - I'm on this one - that's the reason one card's better - because I didn't read properly F2: that's interesting BT: yeah F2: so you're quite quick BT: yeah F2: in how you look BT: yeah F2: but you're a bit quick for absorbing it BT: yes - yeah - at one time I could do it

JB commented that she preferred to look at information with a family member. The provision of single concepts was seen as helpful in pacing a shared engagement with information, as described by the facilitator in the extract below. JB explains that having the information read aloud, one concept at a time, helped her to understand, when she cannot understand by reading information on her own.

F2: we talked about each one of them didn't we and that was good wasn't it - so one at a time I kind of read it to you and we talked about it and then you linked it with your experience JB: yes - when I couldn't do it and when I didn't - yes I can see it

The benefit of looking at information as single concepts together with someone else was also stated by RP, who contrasted the presentation of information on single cards with traditional leaflets. Here, he articulated how single cards help both the person who has had a stroke and the other person to attend to the concepts "one at a time". He added that he thought this especially useful in the early stage after a stroke.

RP: whereas I mean if you've got someone with you - you know family or whatever you can just you know - one at a time go through it all I think particularly the end of when you have your stroke I think they are good

In the extract below, TM commented that if information was presented as single concepts, she was more able to remember information. This would allow her to process each piece of information without the visual distraction of other concepts.

TM: well if you see one that's a lot better than because you then you go to the other one and then you by the time you have forgotten the other one but so you don't really want two what you want is one is first is good better

In summary, participants were positive in their view of single concepts, revealing that information presented in this way helped with engagement with the information, pacing their reading, and accessing and retaining the information. The issue of information which is easy to look at together with a family member was highlighted in these discussions as an important consideration for some people, and is considered further in Theme 4 'Everyone's different', sub-theme 3 'Individual visual style'.

7.2.2.2 Linked concepts

For some participants, the presentation of two concepts on one page was rated as helpful to their processing of information. SE's comment serves to illustrate that it was difficult for him to articulate why, but that he perceived this amount of information positively.

SE: I thought that were good that

The benefit of two concepts was identified more clearly by RP, who responded to the Group Leader's suggestion that the there was a link between the concepts.

GL: are the two things related on there?

RP: for me, yes, 'your brain does maths' - it doesn't now of course - and 'your brain concentrates for you' - makes you think about - oh yeah - it's to do with that stroke

OS agreed with RP, that presenting related concepts together helped her to understand the information.

OS: similar - similar - things you know - to me

A facilitator reported to the group that two participants had found the linking of concepts useful.

F3: MM and SG said that they go together - they follow - that's what they said

Within the theme of 'Visual access', it is important to note that, even where individuals found two linked concepts on a page helpful, they still preferred to see a limited amount of information at a time. In the extract below, NH explained that with too much to attend to visually, he was forced to disengage from the information.

NH: yes - well - I was looking at the brain - at number three - and looking at the - I think the brain is so much different in the one - is says it quite spontaneously - and that is that - and the - it ... [looking at the three concepts on one page] I'm sorry if I can't get it over - but it - when I saw three [three concepts on one page] - I just have to switch off

This opinion was supported by RP, who agreed with NH's comment.

RP: I can't do that - it's too much - what NH was saying [three concepts on one page]

Later in the discussion, RP summarised his view of the different number of concepts per page, and he identified that two concepts per page is optimum for him in conveying the content.

RP: I can manage with the two - but it - there's two things just bring the things out - whereas the one - not quite enough - but the three is too much

The most frequently expressed comment in relation to amount of information was that information that participants encountered in everyday life was generally inaccessible because there was too much to look at on one page. For some participants, this was expressed with some emotion, for example in the extract below, RP talked about internet-based information.

RP: far too many things ... millions - and it just gets annoying

RP, reflecting on what he thinks about information on single cards, commented in a similar way to NH's earlier turn, expressing the risk of disengaging with information when too much is presented at once.

RP: better than leaflets because you've got twenty things to look at - I think a lot of people are fazed -you think oh I can't look at that - what's that mean? - chuck it

These extracts sum up the importance of this issue for people with aphasia who expressed that their visual access was assisted by the presentation of one single or two linked concepts. In contrast, trying to visually process multiple concepts per page was perceived as a significant barrier to accessing information.

7.2.3 Presentation of Talking Mats® data

The fourteen participants completed Talking Mats[®] ratings for a subset of factors which were consistent with the theme of 'Visual Access'. These factors were font size, font style, emphasis in font, and amount of information per page. Each of the participants rated each condition within a given factor on the rating scale (see Appendix E).

7.2.3.1 Size

Each participant was given four cards. The text on each card was printed in one of four font sizes, 10pt, 12pt, 18pt, and 26pt. Participants placed the cards on the Talking Mat[®] underneath the visual scale, according to their opinion of the font size presented. For example, seven participants placed the 10pt card under the **XX®** symbol, and ten participants placed the 26pt card under the **QVV** symbol. Table 7.4 shows the participants' ratings of each font size.

There was a tendency for participants to rate the larger font sizes more highly, with the smaller font sizes less favoured. The smallest font size (10pt) was overall least favoured. One participant rated the two smaller font sizes more highly than the other participants. In the discussions, however, he expressed a preference for a larger font size.

Table 7.7.4 Frequency of font size ratings

	☺✓✓	\checkmark	? 😐	×	* * 🛞	
Font size						Total participants
10	0	0	1	6	7	14
12	1	0	5	6	2	14
18	9	5	0	0	0	14
26	10	3	1	0	0	14

7.2.3.2 Font style

Each participant was asked to rate two cards. Each card had text printed in either sans serif font or serif font. Table 7.5 shows the participants' ratings for each font style. As shown in the table, participants showed a trend towards preferring sans serif font, with no participants rating the serif font as OVV. The large number of mid-scale ratings, however, suggests that participants did not hold strong opinions about font style.

Table 7.7.5 Frequency of font style ratings

	☺✓✓	\checkmark	? 😐	×	⋇≭⊗	
Font style						Total participants
Sans serif	1	7	4	2	0	14
Serif	0	6	4	3	1	14

7.2.3.3 Emphasis

Ratings of different methods for emphasising font in text were collected in the first and second focus group meetings. In the first meetings, participants were given four cards, each one with a key word in the text printed using one of four styles of emphasis: large font, bold font, italic font, and underlined.

The results show a trend towards large font emphasis, as shown in Table 8.6, but the spread of ratings across emphasis types, and the high frequency of mid-range ratings, suggested that further clarity on this issue was required.
Table 7.7.6 Frequency of emphasis style ratings 1

	☺✓✓	\checkmark	? 😐	×	⋇≭⊗	
Emphasis style						Total participants
Large	6	3	4	1	0	14
Bold	3	2	5	2	2	14
Italic	2	2	4	3	3	14
Underlined	1	2	5	6	0	14

In the second focus group meetings, participants were given four cards, each with a key word in the text printed using one of three styles of emphasis, bold font, large font, large bold font, and a normal font control. As shown in Table 7.7 below, this activity demonstrated more clearly that participants favoured emphasis over no emphasis. The font which was both large and bold was highly rated against all other styles.

	©√√	\checkmark	? 😐		××⊗	
Emphasis style						Total participants
Normal	0	1	0	8	5	14
Bold	3	3	4	4	0	14
Large	4	6	2	1	1	14
Large + bold	10	3	1	0	0	14

Table 7.7.7 Frequency of emphasis style ratings 2

7.2.3.4 Amount

Participants were asked to rate the accessibility of information presented with one, two, or three concepts on one page.

The data presented in table 7.8 below show that participants tended to favour materials which contained one or two concepts, and rated more poorly the materials which contained three concepts.

Table 7.7.8 Frequency of number of concepts per page ratings.

	©√✓	\checkmark	? 😐		* * ®	
Number of concepts						Total participants
One concept	9	4	0	1	0	14
Two concepts	4	3	3	4	0	14
Three concepts	0	0	3	5	6	14

7.2.4 Summary of Theme 1

Analysis of the data within this theme demonstrates the importance of initial visual access for people with aphasia. Participants responded more positively to large, clear text and images, using large and bold emphasis in font to highlight key words in the text, with concepts presented either singly, or showing two linked concepts.

The group discussions revealed how these factors contribute to individuals' ease of visual access, and the subsequent impact on their speed and ease of processing information as well as their engagement and motivation to persevere with the information materials. This links with the next theme, in which aspects of the appearance of the information were seen to affect the likelihood of participants engaging with it.

7.3 Theme 2 - The look of it

This second theme reflects participants' views on the appearance of the information. Overall, this theme is not specifically about processing or understanding information, but is concerned with whether it looks right. This subsequently has an impact on participants' engagement with the information.

The data contributing to this theme are from group discussions in all three groups across both sets of focus group meetings. The materials discussed include Template 1 and Template 2, and include ratings of Template 1 materials designed to elicit opinions regarding image style. All participants contributed in some way, although the amount differed across groups and within groups.

Two sub-themes are described in the following sections. The first sub-theme, 'Acceptable', relates to participants' views on whether the information looks appropriate for adults with aphasia, taking into consideration their personal experience of stroke and aphasia. The second sub-theme, 'Design', relates to the way in which aspects of the design, for example, image

style and colour, are used to create information which is attractive to and engaging for people with aphasia. Extracts from the data relating to these themes are presented below.

7.3.1 Sub-theme 1 Acceptable

In this sub-theme, participants expressed the importance of the information appearing appropriate for the intended adult reader. This is based on their experience of aphasia and their personal views concerning the types of images used. The emotional element to the responses seen in some of the extracts indicates the importance to individuals of this aspect of the information.

7.3.1.1 Acceptable to adults with aphasia

The issue of modified materials still maintaining an appropriate adult style emerged in one discussion. The Group Leader summarised this discussion, using the word "simple" to reflect that some participants were indicating that they preferred information with single images and sentences. Following a further discussion about what "simple" information meant to individuals, BT made the point that whilst stroke and aphasia can result in difficulties with understanding, simplified information still needs to reflect an adult style.

BT: yes - it's - when you have a stroke it makes things very simple - got be careful with not making fools out of them [sounds of agreement] ... yet /a/ I think that's good

In the second part of this extract, BT's words suggest that he considered these materials "good". This view was echoed by others when, in subsequent discussion, the Group Leader asked the group whether they thought the modified materials they were looking at were appropriate for adults. Whilst none considered the materials inappropriate, BT identified the risk of producing information which, although accessible, may look inappropriate for adult readers. BT's comment, in the extract below, highlights the risk of stigmatising people with aphasia, if a modified design is interpreted in a childish way.

BT: where do you draw the line? but you can't, /n/ normal person will think, well, that's childish

The tone of materials also influenced whether people viewed them as acceptable or not. Some participants demonstrated that if the information contained a positive message about recovery, their engagement with it would be greater. In the extract below, MM and her facilitator have been looking at a card depicting the concept of aphasia recovery, showing a sentence 'aphasia can get better over time' above a three-part illustration of improving speech (see Appendix D). F2: what did you like about it MM? MM: because they're getting better

MB, MM and SG also referred to this issue.

MB: I said it's going back a long time - but - right from beginning
MM: yeah
MB: I suppose it must have got better
MM: yeah
MB: going back
F2: yeah - and what he's trying to show in the picture is progress - so that you do have progress
MB: yeah - it must have done
SG: done and speak - /ba/ / ba/ - /ə/, fifth time - speak a lot better

SG's concluding comment confirms that there is hope for recovery, but that for her, this followed repeated practice of her speech.

Within this sub-theme, the data indicate that information needs to be acceptable to adults and designed with them in mind, and that it should be positive in tone, conveying hope to the reader.

7.3.1.2 Acceptable to individuals

Several participants reacted strongly to the images used in the materials. One example of this concerned the depiction of blood in the context of information about brain, blood and stroke illness, with the sentence 'blood makes your brain work'. In these materials, blood was conveyed in two different ways: one illustration showed a drop of blood falling from a cut finger; another showed blood vessels between the heart and the brain. In discussions in which the Group Leader invited general comments about the images, some participants did not like the blood pictured in the context of a cut on a finger. For example, in the extract below, MM's facilitator reminded her how she had reacted adversely to this image.

F3: MM - do you mind if I just say - when you - when you looked at that one compared with that one - you said something straight away ... do you remember? When you saw it? MM: /wɛ:/ - blood! [facial expression conveying horror] [laughter]

MM's response suggested that the image of blood was too graphic for her, and this impeded her desire to engage with the information. Similarly, in the extract below, MB indicated a neutral response to the depiction of blood flowing from the heart to the brain, but was evidently distracted by the image of dripping blood. GL: do you like the red?
MB: yeah - well it goes in and out [laughter] ... but If I saw something like that
GL: that's the dripping finger
MB: I'd be thinking - you know - you know [facial expression conveying disgust]

This image of blood was unacceptable to many of the participants, and all, including those with more severe aphasia, were able to express their distaste in verbal or nonverbal terms. At other times, participants experienced a strong emotional reaction to the materials, appearing to have an unpleasant association with the images, but found it difficult to explain their feelings. In the extract below, SG reacted to a photographic image of 'listening', depicting a man cupping his ear in an exaggerated posture.

SG: made me be stupid - yeah - angry

F2: really? So that picture made you feel angry?

- F1: really? so that's the 'listens to speech' man again this man do you know why?
- SG: erm go on no tried no [laughs]
- F1: just brought out that reaction in you

SG: I don't know - I know what I want to say - getting it out doesn't come easily

In the discussion, there followed a fairly lengthy exchange, initiated by MM with SG about the experience of having aphasia and the difficulties of participating in conversations involving those without aphasia, due to topics moving forward too quickly. SG's comment "don't listen" and her immediate engagement with this topic suggests that this is important to her, and may help to explain her response to the image discussed above.

MM: you know when I said talking - erm - talking place today - and they said it about - people they going to ask an - you should have time to listen to them - and they don't - that haven't people - um

SG: yeah - yeah MM: that's SG: I notice what you mean MM: yeah SG: don't listen MM: yeah - they can't be bothered SG: yeah yeah yeah MM: they ant - oth - other people say leave it and let em tell ya SG: yeah yeah MM: and you can't 'cos they'll not - they'll not listen to you SG: yeah yeah - they gone onto something else MM: yeah SG: done

As can be seen in the extract, SG was unable to articulate the exact reasons for her reaction to the image, but the discussion provided some insight into the importance and sensitivity of this topic to her personally. Her negative perception of this image distracted her attention and affected her motivation to engage with the information.

Within this discussion, participants expressed how they found some images in the materials to be unacceptable to them, either through being too graphic (as in the 'blood' image example), or through triggering an unpleasant association (as in the 'listening' image example). These examples demonstrate the importance of providing neutral images which allow the reader to remain engaged positively with the intended message of the information.

7.3.2 Sub-theme 2 Design

In this sub-theme, participants raised issues related to the style of images, the use of colour, font, and the consistency in the design of the materials.

7.3.2.1 Image style

In the focus groups, participants' opinions were sought on their preferences for image style in information materials. It was anticipated that people would express a consistent preference for one style of image over another, and that this would be elicited through the comparison of the two different image styles. Participants were therefore asked to comment on materials showing either photographs or line drawings. Whilst they did express preferences, their reasons varied. For some, the visual appeal of the images seemed most important, whereas for others, the clarity of the image in conveying its intended meaning was more pertinent.

In general, participants appeared to be attracted towards either the line drawing style or the photographic style. For example, responses to the Group Leader's open questioning indicated a simple preference for one style over the other. In the extract below, NH commented that he preferred the line drawing style.

NH: no - I - I - I prefer the - the other photo - not the realistic one but the - the other one

BT's facilitator reported back to the group discussion that he had suggested that he found the photographs more visually appealing, as captured in the extract below.

F2: BT you made a really good point about the photographs as opposed to the line drawings - about ... it was to do with pleasingness

BT noted that both he and GG favoured a photographic image of the brain, and commented on their shared interest in art.

BT: it was funny that we both have art and like that one

For these two participants, the visual appeal of the images seemed to be a key factor in their engagement with the materials. For others, the visual appeal seemed secondary to the clarity in meaning of the image. For example, EC explained that, even though he found photographs more visually appealing, he preferred the clear meaning conveyed in the line drawings.

Some participants made comments which suggested that their preferences changed over time, with recovering language processing. In a discussion about stroke information booklets BT explained that in the early days after a stroke, he would have preferred to look at a limited number of line drawings. Now, he is able to enjoy looking at photographic images in a book.

BT: if I was going to book - at first I wouldn't - but I might go through one or two pages and er -I think this is what I would look at and understand [indicates line drawings] - it's got better - I like that

His final comment conveys that he feels positive that he is once again able to enjoy looking at information in this way.

7.3.2.2 Colour and font

Colour and font were two aspects of design which prompted discussion. Some participants considered the use of colour in the design of information important, whereas others seemed to barely notice it. In the extract below, the issue of colour was initiated by JB's facilitator, reporting back to the group following their individual discussion of materials on which the header word was printed in white on a coloured background. The same colours were repeated sparingly in the images (see Appendices B and C).

F2: we had a little discussion about the colours didn't we?

JB: yes I didn't notice at all [the colours]

Once prompted to consider this aspect of the materials, others, such as NH, expressed a strong preference for colour to be used in information materials.

NH: I would say that you must always try to have colours

In a development of this discussion, RP suggested that a colour-coded index was useful to him when navigating through a set of information.

RP: if if it came a book, or a leaflet or something - I wouldn't know anyway - I want to know something about - I would know which bit to look at GL: hmm RP: it's it's in the red bit GL: hmm F4: yeah RP: you know - a bit - for most people with aphasia - it's awful put /əm/- an index GL: yes RP: that's - I would look - I would look for that [colour coded index]

As with the use of colour discussed above, the different colours of fonts used in the materials did not elicit any spontaneous comments from the participants. Again, the Group Leader asked participants to comment on their preferences. In the extract below, TM indicated that she liked to see the header text in white against a coloured background, but this appeared to relate more to the visual appeal of the coloured banner than the colour of the font.

TM: I like the red one GL: and you like the white writing? TM: yes that's fine

Many of the participants did not appear to notice the colours and fonts used in the design of the materials. Having had their attention drawn to these aspects, some participants recognised them as an important element of visual appeal and accessibility, but for others, they remained a side issue.

7.3.2.3 Consistent design

Consistency in design was raised by participants, in relation to both individual concepts and across a set of concepts. In the first focus group meetings, the participants were asked for their opinions on materials in which the sentence was separated into two parts, with an image presented below each part of the sentence.

OS referred to the materials in which the sentence 'your brain needs blood' is illustrated by two different images of a brain. OS described that she was distracted by the use of different images, and she wanted the image for the concept to be consistent.

OS: it's all the brain like - it's all in the blue - and then when you move it across like - then you get the other diagr-...

The same issue arose in a discussion in which participants were asked to comment on a set of eight cards conveying information about aphasia. The sentence 'stroke can damage language' was illustrated by an image of a brain in the first part of the sentence, and by a speaker with a mark in the brain in the second. BT voiced a similar opinion to OS.

BT: in this one ... why is the brain different for aphasia?

The extract below shows BT's response to the Group Leader's request for a solution to this: that the brain image should be the same for stroke as for aphasia.

BT: just the same brain as on the other side

NH's preference for line drawings emerged from his viewing them as a set of cards together, and experiencing the greater consistency in visual properties, in comparison to photographs which were visually more varied.

GL: so it's interesting that at first you liked the photograph - but when you saw more of them together

NH: together - it - it - it went completely the other way - and went out of the room that one - something - something acceptable to being - I don't know

As discussed in relation to the issue of image style in Section 7.4.2.1, participants expressed preferences for different image styles. These do not appear to be rigid, but are affected by other aspects of the design. In NH's case, the need for consistency outweighed the initial preferred image style presented.

Two aspects of consistent design emerged in this sub-theme: a repeated image to illustrate a repeated concept, and the use of the same style of images to illustrate information materials within a set of materials.

7.3.3 Presentation of Talking Mats® data

7.3.3.1 Image style

Participants compared materials which conveyed concepts using either photographs or line drawings. Participants in Group A rated four cards, of which two contained photographs and two contained line drawings. Participants in Groups B and C rated six cards, of which three contained photographs and three contained line drawings.

A summary of the ratings is provided in Table 7.9 below.

Table 7.7.9 Frequency of image style ratings

	☺✓✓	\checkmark	? 😀	×	* * 🛞	
Photographs						Total participants
Photo 1	5	1	5	3	0	14
Photo 2	2	1	4	4	3	14
Photo 3	3	1	2	3	0	9
Line drawings (LD)						
LD 1	8	2	2	2	0	14
LD 2	6	1	3	2	2	14
LD 3	5	1	3	0	0	9

The data show that for each of the three concepts, more participants gave highly positive ratings to the line drawings, and more gave negative ratings to the photographs. However, many rated the images at the mid-point of the scale, suggesting uncertainty or indifference. Whilst these results appear to show a tentative trend for participants to prefer the line drawings over the photographs, it is not a clear finding.

7.3.4 Summary of Theme 2

The data show how participants' initial impressions of the materials influenced their motivation to persevere with attending to the information. Analysis highlights the importance of information which is acceptable to adults, and designed with people with aphasia in mind. One aspect of this is that information should present a positive view of stroke recovery which conveys hope to the reader.

Visual aspects of design, such as the use of colour, consistency in the overall design style, and the repetition of specific elements such as images, were found to affect participants' engagement with materials. It is noteworthy, however, that not all participants responded in the same way to these elements of the design, suggesting that they are more important for some participants than for others.

Individuals varied in how sensitively they responded to the images used in the materials, but overall it was found that images which were neutral in tone were more likely to retain participants' engagement than those which prompted emotional reactions. Discussions of image style revealed that this is not a straightforward issue. Whilst individuals may tend to prefer one image style over another, their opinions of specific images were complicated by unpredicted features of the images shown, such as their acceptability, visual appeal, and clarity of meaning. The analysis demonstrates how these factors affected participants differently and influenced their opinions of the materials.

7.4 Theme 3 – Understanding the information

This theme captures the data relating to how participants talk about their understanding of the information presented. Whilst the participants were not specifically asked whether they understood the information, they did comment at times on this, expressing variously full, partial, or no understanding of the content.

The data contributing to this theme are from group discussions in all three groups across both sets of focus group meetings. The materials discussed include Template 1 and Template 2, and include ratings of Template 1 materials designed to elicit opinions regarding the number of images per concept. All participants contributed in some way, although the amount differed across groups and within groups.

The sub-themes for this theme related to the words, the images, and the interaction between words and images.

7.4.1 Sub-theme 1 - The words

The issues discussed within this sub-theme were found to link with aspects of other themes, reflecting that reading comprehension is a common experience in people with aphasia, connecting visual access, aspects of design, and individual visual approaches to information.

7.4.1.1 Aphasia and reading

When looking at the template materials, participants made general comments about aphasia and reading, remarking on whether they could understand the written words on the cards. The wide range of reading ability across participants was reflected in their comments.

In the following extracts, a range of participant responses are recorded. In a discussion introduced by the Group Leader about whether people looked at both the text and the images in the materials, MM commented that she looked first at the text because she could make sense of it.

MM: I can read that

MB expressed that she was able to read part, but not all of the information. She used the word "clicking" to denote comprehension of the single word heading, but suggested that the sentence was difficult for her to understand.

MB: we're clicking a bit different on top - in middle

For GG, the severity of his aphasia meant that the written words were beyond his comprehension, as summarised by his facilitator in the extract below. GG expressed that he found the illustration helpful because, as noted by his facilitator, he was not able make sense of the written text.

F2: you were saying that the word doesn't help you

EC raised this issue of changes to reading over time, commenting that in the early days after his stroke he would have found it impossible to read at all, and would have preferred picture material only.

EC: because when I had a stroke in the first place I can't think about words

The comments in this sub-theme show how aphasia can vary in its effect on individuals' ability to read, and that this can change over time following a stroke.

7.4.1.2 Understanding the words

The second issue within this sub-theme relates to factors which affect participants' comprehension of written text in information materials. The factors included font size and emphasis, and the layout of the text. In addition, one participant suggested modifications to the number of words used and the specific choice of vocabulary.

Participants agreed that their reading was helped by fonts which are large in size and include bold font for emphasis of key words. In Theme 1 'Visual access', this was identified as part of the process of visually processing the information. Here, in contrast, participants commented on the impact of typography on their reading comprehension. As previously described in Theme 1, participants consistently favoured larger font sizes for ease of visual access. Here, NH commented that a larger font helped him to understand the text faster. He related this to the experience of feeling under pressure to read and make a choice from a menu.

NH: if it was the big ... if it was coffee that said coffee - I would - coffee ... I would get it sooner than these that are smaller

In a discussion of materials contrasting different styles of emphasis in font, TM noted that the large and bold font was helpful to her reading.

TM: to help - yeah - to read it

Features of the design of the templates were identified in discussions as helpful to reading the materials. Firstly, the use of a header phrase above a concept sentence was initiated in discussion by the Group Leader, who asked participants to give their opinion on this layout. In the discussions, most participants agreed that they found this layout helpful in guiding their reading. In the extract below, SE expressed that this top-down approach helped him to understand the information in stages.

SE: read that and then you - can move down - settle in - your head

In contrast, BT expressed that he had not really noticed the header phrase, or did not consider that it was within his scope for comment.

BT: I thought that was only for you

In one discussion, participants debated whether it was helpful to include words as well as images in the information. RP suggested that the image was primary, and that the sentence could be placed at the bottom of the card. NH argued that he wanted to see the written word at the top of the design, to catch his attention. Here he described a card conveying the concept 'your brain makes your speech' in which the image shows a line drawing of a person and a speech bubble with the word 'hello' inside.

NH: I start at the top - and if it - down at the bottom ... and if it was down at the bottom - I wouldn't bother about it - it's up there where it the brain says hello and - and that's important to you

Two further issues relating to easing the reading demands were discussed in the groups. In the discussion comparing a short phrase contrasting different styles of emphasis, RP expressed a preference for a single word rather than a phrase.

RP: bold's better - but if it's for me I would personally find it a lot easier if it just said 'coffee'

RP raised the issue of suitable vocabulary used in information for people with aphasia on several occasions. He made suggestions for words that seemed more meaningful to him, for example 'communication' rather than 'language'. An example of this is shown below, in which he expressed difficulty making sense of the specific words used in some of the materials. The sentence read 'your brain makes your speech', but RP suggested that the word 'makes' did not fit well in this context. He provided an alternative, as in the extract below, in which he also acknowledged individual variation and preferences.

RP: I preferred 'produces' - but it's - it's very difficult - because we are all - I'm not sure you can find the best word for everybody

This sub-theme reflects the varied experience of people with aphasia when reading after a stroke. Participants reported that this ability may change over time, with resulting changing needs for how information is presented.

Participants mentioned various factors which helped them to deal with the written words on the information materials. These included an increased font size and emphasis of key words, as noted in Theme 1. The layout of the information, with a header phrase and concept sentence helped to guide people in their reading, building their comprehension in stages. There were variations in the extent to which people attended to the different elements of the design, and their preferences for specific layout of the sentence.

7.4.2 Sub-theme 2 The images

This sub-theme records participants' comments regarding the images used in the information materials, and the impact of these on their comprehension of the concepts presented.

Two issues arose from the data in this theme, reflecting firstly participants' responses to different image styles, and secondly how well the meaning of the concepts was conveyed in the images used.

7.4.2.1 Impact of image style

This issue is concerned with the impact of different image styles on participants' understanding, and what they said about processing different styles of line drawings.

Participants' responses to different image styles suggested that individuals generally found one style easier to interpret. Several participants favoured the line drawing style, expressing that they were easier to understand than the photographs. In the extracts below, SE explained why he rated the line drawings more highly.

SE: yeah - I do see the - the - the drawings are much clearer

RP, in contrast, found the photographic style easier to understand, as in the extract below, in which he referred to an image of a man listening.

RP: right - start from the bottom one - 'your brain listens to speech' - the guy doing that to me is just - /komp/- tells me completely what that means

The extract below shows that the same image can be easy for one individual to process and more difficult for another. Here, MB's facilitator supported her to express that this image was difficult for her to process.

F4: MB you didn't like this one - did you? - where it says 'your brain - listens to speech' MB: I put that because I wouldn't - wouldn't be able to

F4: this guy - you said that this - this man didn't click - didn't you - you said that you couldn't understand why

MB: yeah - I wouldn't - for me - I wouldn't be able to - it wouldn't click

The group discussions revealed that participants' image style preferences were not only based on whether these were line drawings or photographs. Differences in style within the line drawings or photographs affected the ease with which the concepts illustrated were understood. Among the line drawings, participants showed preferences for images which could be described as either representational, that is realistic, or figurative, that is less realistic, in style. The designer illustrated the concept of the brain using two styles of line drawings. The first showed a simplified but realistic representation of the brain; the second included an image of cogs and wheels inside the brain in a more figurative illustration. The Group Leader asked participants to comment on which of these styles made more sense to them. Some participants expressed that they found the representational style more accessible. OS's comment suggests that this style fitted more closely with her understanding of the concept. OS: to me it explains /əm/ - it explains the brain

BT's comment about the representational image of a brain reflects the importance to him of a clear image style, linking with 'Clarity' in Theme 1, and 'Image Style' in Theme 2.

BT: I like its simplicity

MM expressed that she preferred the figurative illustration of the brain. Her comment in the extract below suggests that she found it helpful to see an illustration of activity in the brain via the cogs and wheels pictured, and applied this as a way of understanding the workings of the brain after stroke.

F2: you said something really interesting about that - when you looked at that brain MM - when you were looking at the cogs and the wheels ... it made you think - do you remember ... you said /u:/ - do you remember?

MM: well - I know somebody's died [referring to the brain] - I mean - /əm/ - your brain - it's gone ... you know - and - and and others not got as going - some going on [meaning that activity in the brain after a stroke varies]

Not everyone had strong opinions about the image style. In relation to the illustration of the brain, RW indicated that the style of the pictures did not affect his understanding, interpretation, or experience of the information, as his comment illustrates.

RW: I'm not bothered

The presentation of abstract visual material was problematic for some. MB expressed difficulty understanding an image which was intended to convey the concept 'a lot of people', in which stylised human figures were drawn close to each other in varying colours. Her response to her facilitator's comment suggests that she may experience some difficulty with visual processing in addition to having aphasia.

F4: we struggled with that didn't we? You thought this looked a little bit like a pattern or -MB: I wake up in morning - I usually - when I wake up - I get that bit of light - on this side - not long - only a sec - but a bit of darkey

Others, however, also reported difficulty interpreting this image. JB's facilitator reported on their paired discussion about the same card.

F2: the word that you said JB was "it's a bit of a nothing"

In response to this discussion, RP suggested an alternative way of illustrating the concept in a more concrete way, relating to the numbers of people affected by aphasia in the UK, and using words, rather than an image.

RP: you could put there something like 450 thousand people in the UK have aphasia - something - after a stroke - maybe

Additional detail in the images also affected participants' understanding. The impact of illustrative detail was reported in both positive and negative terms, with individuals reporting aspects of the details in the images which seemed to help or to distract them, or interrupt their understanding.

A general problem with use of detail in images may be linked with the issue of amount of information, and having too much to look at, as described in Theme 1. This was expressed by RP, talking about his preference for photographs over line drawings.

RP: there's there's too many different things to look at - got that and that and that

BT expressed that the detail of arrows to indicate movement of arms and legs in an illustration of a body was unnecessary.

BT: those arrows - th – the - take them off - it's the same - can't he move that arm that way

Other participants did not comment that there was too much to look at, but spent time trying to process the details in the images. BT, describing the detail of an eye, drawn in profile, reported that this detail did not make sense to him.

BT: it may be very personal but on the first one - I were ages - I couldn't work out what that 'p' was

In the extract below, NH described materials illustrating stroke and aphasia, in which the first image of the brain was clear to him, whilst the second image of the brain had been faded out by the designer, and the previously clearly drawn cerebellum overlaid with an ear and 'mouthpiece' image, to represent communication.

NH: left me a bit confused as to where they was with the that one - I put it on one side and came back to it and it ... stroke - it he's got the stroke and it something at the back is quite clear there and it goes onto here and it gets an ear in it! [expressing surprise]

NH's facilitator reported further comments from their individual discussion, which illustrates how his attempt to process the detail in the images hindered his understanding. They were looking at a red shape overlaid on the brain to depict the area of stroke damage.

F3: just in relation to that but slightly different you mentioned NH about - /am/ - you didn't really understand why there was a difference between the two red blobs - /a/ - you said the first blob was more brilliant in colour

NH: yes

F3: and the second one was a bit different and

NH: yes

F3: you spent quite a bit of time wondering about what the difference meant NH: yes

Whilst the above example shows NH trying to interpret the image, misinterpretations of images were reported by some participants. BT's comment about the same image suggested that he also tried, unsuccessfully, to interpret the red detail in the brain illustration.

BT: looks like a poppy

Here, both NH and BT attended to details in the images which distracted them from the intended meaning, and BT was distracted further by misinterpreting the image. The details in question were included by the graphic designer with the intention of assisting readers to

interpret the written information, but for these two individuals, the details were counterproductive.

In some cases, participants found it helpful to see extra detail in the images, as in the extract below. A line drawing of the concept 'your brain moves your arms and legs' included arrows drawn near the limbs to denote movement. In the extract below, RW's facilitator supported him to express an opinion he had previously made in an individual discussion.

F5: do you think they make it easier to understand movement with arrows?RW: yeahF5: is it the arrows that you liked?RW: yeah - the arrows is - good

SE clarified that, for him, this detail helped to convey the sense of movement in the image.

SE: well - yes - it is the movement and everything there

Overall, participants demonstrated a preference for the image style which they felt best conveyed the meaning of the concept in the information materials. However, this was more complex than a choice between photographic and line drawing images, and further issues arose in discussion. These included different responses to representative or figurative styles and more abstract images. The presence of additional illustrative detail served to help some participants understand the images better, but distracted others, who struggled to make sense of the extra visual information, and some misinterpretations of images occurred.

7.4.2.2 Interpreting image meaning

This sub-theme reflects what participants said about how well or poorly the images conveyed their intended meaning. Comments included both positive and negative responses to this issue, the reasons for which were explored further in the discussions.

Participants agreed that the image needs to be right for the meaning, and they acknowledged that this is not always easy to achieve. The depiction of some concepts such as stroke and aphasia were discussed in the groups as presenting a challenge for illustration. In the extract below, the Group Leader raised the issue, responding to participants' comments regarding the materials showing that aphasia is different for different people.

GL: OK - so that's a difficult one isn't it - because it's very personal - look at the five of you and it's so different for each of you isn't it - but how do we get that across in a picture?

This issue was acknowledged, but left unresolved. It was summarised by NH, who said, in relation to the issue of picturing stroke:

NH: it's really a big question mark in there - a big question mark with stroke

Participants' contributions to the discussions provided interesting insights into the different aspects of the images which affected their interpretation. In some cases, the comments suggested that the images were ambiguous and this hindered participants' comprehension. For example, OS's response to photographic material showing an image of a human body with arms and legs extended, implied that she attributed to the image a connotation of fitness over and above the intended meaning of 'body'.

OS: I don't - yeah - well - it looks as though you're doing - just a keep fit there - to me - you know

Literal representations of concepts evoked differing responses, as the extract below shows. In a discussion about a card on which the verb 'control' was illustrated by an image of a remote control device, MM explained that the image was confusing to her, since it was unrelated to the sense of controlling one's body. SG clarified to her that the meaning was about 'control', but MM's response suggested that this still did not make sense to her.

F2: yeah - MM wants to say something about the remote control

MM: I don't know what's it doing there? [laughter]

SG: controls your body

MM: but that [pointing to remote control] don't - that don't do that for you - it's your brain

This discussion suggests that for MM, the literal meaning of the image distracted her from the overall meaning of the sentence. For SG, in contrast, the comprehension of the individual word was supported by an image, regardless of its literal meaning.

One aspect of the images which affected participants' interpretation was the context of the image in relation to the intended meaning. For example, participants discussed how best to illustrate the concept of blood in some of the materials. In one version, drops of blood were shown falling from a cut finger. In another version, blood was illustrated by showing vessels between the heart and the brain. RW expressed that, since the sentence was 'your brain needs blood', the cut finger illustration conveyed the wrong meaning.

RW: it's the blood of finger

BT defended this illustration, however, suggesting that this was a recognised way to illustrate the concept of blood.

BT: You couldn't put it any other place could you? - you couldn't put it on the toe or the arm - the finger is the most popular way

In addition, an exchange between BT and his facilitator revealed that the alternative image had not elicited the meaning of 'blood' to BT.

F2: the um - this one - of the brain and the heart ... it didn't really make you think of 'blood' BT: no

F2: but when you looked at the finger with the blood dripping off it ...

BT: yes

F2: that made you think 'blood'

RW's comment suggests that the bleeding finger was a misleading image, because the context was not within the blood supply to the brain. For BT, the context was less important than a clear visual depiction of the word's meaning.

For MB, who has severe aphasia, the context of the image itself was important in helping her to understand the image, as shown in the following extract. When discussing materials showing line drawings of the brain (discussed in relation to their different styles, above) MB's facilitator reported to the group how MB found it easier to understand the image of a brain when it was presented in the context of a head, rather than in isolation.

F4: it helped you to understand it didn't it? whereas in the other one - where it's one - it's just sort of floating in the air - I think MB you said that you likedMB: you what?F4: you liked the brain being inside the head?

MB: them - yeah

When processing images showing people, some participants seemed more likely to relate images of people to themselves. For example, an image of a woman shrugging her shoulders with her hands outstretched illustrated a card which conveyed the concept 'most people have never heard of aphasia'. OS commented that she liked the image as it showed the value of gesture in communication. She confirmed, following a request for clarification from a facilitator, that she interpreted the image as showing herself, when unable to express herself verbally. She did not recognise that this was not the intended meaning of the image.

OS: I liked the one there 'cos it explains - your hands are tee - your hands are speaking for you F2: OS can I just ask you then - what do you - how do you interpret that picture of the woman? Is that you thinking of something to say?

OS: yes - that's how I go - if I can't get the words out - I go [shrugs] and you - you know

In contrast, later in the same discussion, the Group Leader asked NH for his interpretation of this image. His response suggests a negative identification with the image of shrugging, which seemed to convey to him a lack of care on the part of the person pictured.

GL: what did it mean to you NH?

NH: it - so what - I've never heard of aphasic - so what?

GG indicated to his facilitator that a card showing an image of a woman after a stroke did not convey the right meaning for him, as the image was not sufficiently like him. This suggests that the image was too abstract for GG, who would find it easier to process an image which more closely resembled himself in the context of the information.

Overall, participants agreed that it is important for images to provide a clear representation of the concepts in information materials. They acknowledged that this can be difficult to achieve, particularly when the meaning of a concept may vary for individuals, for example 'stroke'. Aspects of the images which were seen to affect comprehension included the degree to which images were open to interpretation, whether participants were likely to be helped or distracted by literal interpretations, and, linked with this, the degree to which the image provided is suitable within the overall context of the topic. Finally, a tendency to see oneself in the image appeared to influence participants' ability to correctly interpret images. Some participants were more literal in their interpretation of images, leading to misunderstandings or distraction from the meaning. Others were able to override the literal meaning of the images, using them to support their interpretation of the information provided.

7.4.3 Sub-theme 3 Words and images together

The descriptive analysis of items indexed within the category of words and images together yielded extensive data. This sub-theme relates to the impact of selection and placement of words and images together on the accessibility of information. Two issues arose: the interaction between words and images and the impact of this on participants' understanding of the information, and the participants' preferences relating to the layout of the words and images.

7.4.3.1 Words and images work together

Participants commented on how their understanding of the information was influenced by the interaction between the text and images. Their observations show that, in general, text and images work together to convey meaning.

Participants agreed on the importance of an exact match between text and images. In a discussion about a card which showed the sentence 'your brain moves your arms and legs'

above a photographic image of a torso with extended arms, participants commented that the absence of legs caused them to respond negatively to this card. As the extract below shows, RW's facilitator supported him to report back to the group a comment that he had made in their paired discussion.

F5: so - what was it you said ... there were no legs? RW: yeah

Similarly, EC explained why he rated the image poorly.

EC: yeah - no legs or owt

JB also pointed out that she found this same card inaccessible. In the following extract, JB's facilitator supported her initially unintelligible comment by referring back to their individual discussion, reporting what she said about the card when she first looked at it. In her final comment, JB indicated that the line drawing, which showed all limbs, was a better example of an illustration for the text.

JB: I think it's - it's very /ɛk/ - it's /ɛkə/ /kə/ - that throws it all off the page
GL: right - so that's the one with the photograph of the moving the arms
F2: it's only the arms - there's no toes is what you said - where are my toes?
JB: yes - it's - it should be more like that [pointing to line drawing showing arms and legs]

RP was the exception, commenting that the image was adequate for him to understand the information, and it was the overall photographic style which he found more accessible than the contrasting line drawing style.

RP: brilliant - and 'your brain moves your arms and legs' - yeah I think great - /prə/ - perhaps some legs down there but ... [laughter] - it's - whereas - incredibly simple

It is not suggested here that an imprecise match between word and image necessarily results in non-comprehension, but that some participants found it distracting and thus detrimental to their ease of processing. For example, EC responded negatively to an image of 'your brain listens to speech' because he thought the image did not convey the concept of listening to speech. In the discussion, he gave several interpretations of the image, and identified that it was "too vague". The extract below shows his response when the Group Leader summed up the issue that the images needed to match the words more closely. RW agreed initially, with EC developing his suggestion for a better image to match the text. EC: same here ... /əm/ - better photograph ... speaking - you know - like an arrow across - might be better than that

EC understood the meaning of the words, and was able to infer the meaning of the image. His comment suggests that the image was a distraction rather than a barrier to understanding.

In the group discussions, participants made comments which showed that there was a variation in the extent to which they needed words and images to work together. RP commented that, since the meaning could be inferred from the image, the written words were not needed. In the extract below, he suggested that the image of a stethoscope to mark the meaning 'medical' rendered the sentence 'aphasia is a medical word' redundant.

RP: I would suggest that where it says aphasia is a medical word could go because it's obviously the doctor 'cos he's got that thing [pointing to stethoscope] - it says 'you have aphasia' so it means that doctor is saying 'you've got aphasia' that's straight

For a person with more severe processing difficulties, working out the implied meaning of the images without textual support may be more problematic. JB reported that for her, it is useful to have the written word to refer to, even if it is difficult to read. In the extract below, she expressed that she preferred to see both words and images together on the information so that she could relate the two to each other.

JB: and - and if definite things that - that - as separate and - and erm that - that what's helps out more and it doesn't - it gets it even more broken that way - but it was right that way

It is worth noting here that JB's access to information relies on having another person to talk it though with her. Thus, given her degree of aphasia, she acknowledged that she was unlikely to understand new information without support.

In summary, participants expected to see both words and images, and these needed to relate exactly to each other to convey the overall meaning. Whilst there were different preferences and abilities among individuals, the perceived match between text and images strongly influenced participants' understanding of the information.

7.4.3.2 Number of images per concept

Participants gave their views on information materials which showed either a single image per sentence, or separate images to illustrate each key word in the sentence (Appendix C). Most participants expressed a preference for a single image per sentence, the reasons for which are explored below.

One aspect of participants' preference for this layout appeared to be related to a perception that this required less processing. For example, EC expressed that the presentation of a single image was "simpler". This was followed up later in the discussion by the Group Leader, who asked participants to say more about this issue. In the extract below, EC had offered the opinion that providing separate images per key word was not simple.

EC: too much information

GL: too much information when you've got two pictures to worry about? EC: yeah

Further evidence for this was provided by an exchange between SE and his facilitator, who reported their individual discussion to the group. In the extract below, SE expressed the opinion that there was less for him to have to attend to when processing a single sentence and image.

F5: and you said to me earlier - you liked the ones where it's all together because you didn't have to - think so hard SE: /ə:/ ... I didn't have to learn anything

These extracts show that layout with one image appears to relate to the issue of amount of information presented at one time, as discussed in Theme 1.

Some participants, in contrast, expressed a consistent preference for the provision of a separate image for each key word. In the extract below, RP identified that an image placed underneath the first key word helped him to orientate to the topic. In contrast, he reported that he found the single sentence difficult to comprehend.

RP: I got the three that are broken down because to me the words is much much better - your brain - see the picture - know what I'm talking about - produces your speech - fantastic - this one - just explained to – [name of facilitator] that the next bit was sort of five letters - your brain makes your speech - doesn't go in

The data revealed differences between participants. Most perceived that a single short sentence, accompanied by a single image, was easier to understand. For a small number of participants however, there was a consistent preference for information with separate images for each key word in the sentence, and this applied where the image was either a direct match for the meaning of the word, or a more distant representation of the word, as in the discussion described above in Section 7.5.3.1, relating to the remote control image.

7.4.4 Presentation of Talking Mats® data

7.4.4.1 Number of images per concept

Template 1 materials, which varied in the number of images per concept, were presented for review by participants. On three cards, the information was presented with a single sentence above a single image. On a further three cards, the sentence was separated out with space between each of two or three key concepts. Separate images to represent each key concept, were presented below the sentence. All fourteen participants rated each card.

The data show that the single image style received the largest number of highly favourable ratings, and was rated very poorly only once. In contrast, the multiple image style was rated across all points of the scale, with the highest number responses at the mid-point. Although ratings of individual cards showed some variation, participants tended to respond consistently in this task, with most participants rating the single image style highly and a smaller group consistently rating the multiple image style more highly. Table 7.10 shows the ratings by frequency.

	☺✓✓	\checkmark	? 😐	*	* * 🛞	
Single image per sentence						Total participants
Card 1	8	0	4	2	0	14
Card 2	7	1	3	3	0	14
Card 3	10	1	1	1	1	14
Image per key word in sentence						
Card 1	3	2	5	2	2	14
Card 2	4	2	4	2	2	14
Card 3	2	3	4	2	3	14

Table 7.7.10 Frequency of ratings for single or multiple images per concept

7.4.5 Summary of Theme 3

This theme brings together all data relating to participants' understanding of the information presented to them during the focus groups.

Overall, participants indicated that they expected to see written words in the materials, even though individual reading ability varied widely. The analysis links with the findings of Themes 1

and 2, suggesting that reading was facilitated by visual presentation, including increased font size, large and bold emphasis of key words, and consistent layout.

The impact of different image styles on participants' perceived comprehension was found to be more complex than a straight choice between two types of image. Variation was found in their preferences for figurative versus representational line drawings, and in their ability to process additional detail or abstract images.

Although participants agreed that images should correctly represent concepts, they varied in their opinions on how best this could be achieved. For some participants, this was complicated by their tendency to process visual information literally. A key issue for some was whether the image was correct in the context of the meaning of the information presented; others seemed to use the images to support their reading more directly, whether their context was congruent or not.

Different processing abilities and approaches, along with varied preferences for image style, contributed to individual responses to the images presented. Whilst the variety of responses makes it difficult to draw conclusions, the breadth of discussion in this theme highlights the complexity of this issue for information design.

Finally, the interaction between text and images was an important element for all participants, a finding which emerged spontaneously in the focus group discussions. To varying degrees, participants expressed that their processing was either distracted or disabled by images which did not exactly match the text. Only one participant expressed that a precise match was not necessary to support his reading.

Alongside this, both discussions and ratings of the provision of single or multiple images per concept showed that most participants favoured the style in which a single image was placed below a short sentence. A smaller group consistently selected the style in which two or three separate images representing the key words were placed below spaced-out text.

The individual ways in which participants responded to the materials is described in the final, overarching theme, 'Everyone's different'.

7.5 Theme 4 - Everyone's different

This theme captures the different responses and individuality of the participants who brought their own life experience, as well as their own experience of stroke and aphasia, to the discussions. The acknowledgement that 'everyone's different' was frequently expressed in all the groups, by both participants and facilitators. The data contributing to this theme are from group discussions in all three groups across both sets of focus group meetings. The materials discussed include Template 1 and Template 2. All participants contributed in some way, although the amount differed across groups and within groups. Three separate sub-themes were identified in the data, representing individual experience of stroke and aphasia, individual responses, and individual visual style.

7.5.1 Sub-theme 1 Experience of stroke and aphasia

The shared experience of stroke and aphasia was the common point of reference for the participants in the focus groups. When looking at the materials, participants drew on their own experiences, both of early days in hospital, and the continuing challenges of living with aphasia.

During the course of the discussions, there was an expressed acknowledgment that stroke is different for everyone, including how people are affected by stroke, the experiences they have, and the recovery they make. Almost all participants commented on their stroke recovery, both in relation to their ability to deal with information, and in a more general sense.

7.5.1.1 Knowledge of aphasia

Whilst there was recognition amongst group members that aphasia is different for individuals, there was a shared understanding of the concept of aphasia. This discussion was largely generated in response to the set of cards about aphasia, in which participants agreed with facts, identified with the information, or expressed opinions about how the information was depicted.

Participants agreed that information must be accurate, reflecting individuals' own experiences, and fitting with their broader knowledge about stroke and aphasia. This was most evident when discussing information about recovery, which was a common but different experience for everyone. One discussion centred on the timescale of recovery shown on one card. The sentence 'aphasia can get better over time' was illustrated by a line drawing of a person in triplicate. Each of these images was accompanied by an image of a calendar page showing three different dates from the months of February, July, and October, and a speech bubble showing the word 'hello', in which the font was initially very fragmented and, in the final image, perfectly formed (Appendix D). Participants in all focus groups indicated that this was inaccurate, and as EC's comment below suggested, potentially misleading to readers.

EC: it takes years for you to get better - but that looks like it's a month

Participants also agreed that the depiction of recovery from stroke had to represent realistically that people rarely recover completely from aphasia. In the extract below, the facilitator reported back to the group on behalf of GG, a man with no expressive language. The facilitator described his nonverbal response to the card described above, in which he expressed that a representation of complete recovery was incorrect. The facilitator's report of GG putting his hand over the perfectly formed font in the final image communicated clearly that this did not represent GG's experience of aphasia recovery.

F2: can GG and I say about - when you looked at card number 8 - you very clearly put your hand over the final one

RP, in the extract below, commented that the timescale for recovery should be represented in years not months, and that the image should show an incomplete recovery which, to him, conveyed the possibility of further recovery over a longer time.

RP: what about if - February to October you're doing incredibly well to get better in that time ... I would say year one - year two - year three ... and it shows it carrying on

BT expressed a preoccupation with the word 'aphasia' and its meaning, remembering that he had not encountered it before his own stroke, and then did not hear the word until he had been in hospital for several weeks. He talked about the importance of knowing the word aphasia for himself, and for others to know it as well, for there to be greater public awareness of aphasia.

BT: people don't know what aphasia is

Whilst all participants had personal experience of stroke and aphasia, two participants also drew on previous professional experience in healthcare in their interpretation and opinions of the information. For example, in the extract below, OS was responding to the Group Leader asking if the red mark on the image of the brain conveyed the idea of a stroke to her.

OS: no - because I'm a medical person - that doesn't say it to me - I know different things from them

Participants' expectations of information about stroke and aphasia were guided largely by their own experiences and by those of others with aphasia whom they had met since their stroke. They were intolerant of information content which they felt did not reflect this knowledge, and made suggestions for improvements in accuracy.

7.5.1.2 Aphasia is individual

The recognition that the experience of aphasia is different for everyone was a recurring theme in discussions. Language processing difficulties, including reading, understanding, and word finding, were described by individuals during the group discussions. These were all mentioned as factors in their everyday lives, including their interaction with information. For example, OS described her difficulty in finding the words:

OS: it's getting the words round

Participants also recognised other sensory and cognitive problems, such as visual processing difficulties, difficulties dealing with numbers, and poor memory. For example, in the extract below, JB described how she was aware that she could not always remember what she had read, even if she felt she had understood it at the time of reading.

JB: I can read it and I'm reading it well - but when I shut it up it's gone - and I can't get it back maybe I can bring it back - but I can't always be sure

In the discussion, participants talked about the realities of having aphasia, sharing the different feelings such as depression, embarrassment, and the loss of confidence they had experienced. They talked about their recovery, comparing their experiences, both acutely after stroke and in their current lives. In the extract below, RW reflected that, ten years after his stroke, he still found it difficult to speak.

RW: I - ten years ago I had a stroke ... I can't - I can't ... speech is - years - ten years ... yeah

Participants also shared a sense of acceptance of aphasia as part of their lives. For example, BT expressed an acceptance of his aphasia and physical limitations.

BT: I don't think I've altered in my speech since the first month in hospital - walking about - I got a little bit better - I'm going downhill now [laughs] - well I don't try now

The concept of hope emerged in discussions and participants expressed a desire that information should be positive in tone, conveying a sense of hope to people new to the experience of aphasia. In a discussion about aphasia recovery, MM drew on her own experience in response to the facilitator asking her if she felt that the template materials conveyed hope for recovery. Her words suggest that she was reflecting back on her experience, and perhaps reappraising her progress, referring to a recent interaction in which her sister had commented that her speech was still improving. MM: I do - yeah - if you couldn't talk before - like us ... I said you've done well - even my sister too - yesterday - you've been really good - you know - for some time

Overall, this issue reflects that whilst aphasia is different for individuals, participants identified with each other's experiences of aphasia as a group. The shared feelings and experiences in the discussion highlighted the value of informing people about aphasia in a realistic but positive way. Individuals' responses to the information content also demonstrated that the materials functioned as a resource for talking about the experience of aphasia, assisting some participants to reflect on and talk about their own experiences.

7.5.2 Sub-theme 2 Individual responses

In this sub-theme, the data reflects the variety of ways in which participants responded to the materials. This is less about accuracy of information and more about the individuals' responses to the materials.

Two issues informed this sub-theme: firstly, some participants' identification to varying degrees, with the words and images in the materials. The second issue relates to the strength of reaction among some individuals to the materials.

7.5.2.1 Identifying with the information

Sometimes, participants showed that they identified personally with the information on the cards. In the extract below, OS commented that she liked the card which showed the concept of reading and writing because this is something that she still liked to do, despite it being difficult for her now.

OS: I like reading and writing - I thought the reading - the person was quite happy 'cos he he looked to me as happy because he's reading the book and the writing you know /ə:/ you try and write so you put it down on paper and pen - which I do try F1: you do try - so that's still part of your life OS: that's part of my life

In a similar way, OS also expressed a positive response to a card which conveyed the concept of listening by depicting an interaction between two people. She commented that for her, language is about communicating with others, and that this is an important part of her life.

OS: it's nice to speak to someone isn't it

For some, there was a sense that the information had to convey the world as they see and understand it. This was most evident in discussions about how to depict language and communication. TM expressed that, for her, the meaning of listening was only conveyed where a two-way interaction was shown, with both a speaker and a listener.

TM: well - because you listen as well what the other one saying

Similarly, in the following extract, JB described that an image to convey the concept of understanding language corresponded with her view of how understanding works in real communication exchanges, with both a speaker and a listener depicted. For her, the interaction shows the listener, the "aim person", and the speaker "the one", with the speech moving between them "it bubbles in".

JB: noticing that - had this because - because I feel that's - there you've got the aim person - there you've got the one - and you've got that there - this one - and it bubbles it in

Participants varied in the extent to which they identified personally with the materials. In the example below, MB appeared to be integrating the information with her own experience and that of others in the group.

MB: it must be something like that that's done what we are now - what's happened to us

As discussed above, this phenomenon was most marked for GG, a man with global aphasia, who expressed nonverbally that the information needed to be 'about me' in order for him to be able to regard it as relevant. A close personal interpretation of the materials resulted at times, however, in participants misunderstanding the intended meaning of the images, as discussed in relation to Theme 3 in Section 7.5.2.2.

The data presented in this section demonstrate how the materials resonated to varying degrees with the participants. Some identified closely with the images and words, almost appearing to see themselves in the information. These individuals valued a portrayal of concepts which closely matched their perceptions. Others needed a stronger personal representation in the materials in order to identify it to themselves. This issue links closely with the issue of interpreting picture meaning in Theme 3, but highlights the degree of variation between individuals.

7.5.2.2 Emotional reactions to the materials

On some occasions during the focus groups, participants reacted with emotion to the materials or the topic under discussion. The researcher identified this issue within this sub-theme, since the responses were individual to different participants. In addition, it was considered noteworthy in the context of health information, which may be intended to be factual and unemotional.

In a discussion comparing different image styles, described in Theme 2 above, NH expressed a strong dislike of one photograph. In the extract below, his response demonstrates his strength of feeling, which resulted in his overall poor rating of the photographic images.

NH: and when number three came along - I couldn't stand that one

SG articulated her negative reaction to the same image. Although she was unable to find the words to explain exactly why, this appeared to be related to the way in which the concept was depicted, by a photograph of a man cupping his ear in an exaggerated pose.

SG: made me be stupid - yeah - angry

In a discussion about the number of concepts conveyed at one time, RP recounted his involvement with a previous research project which involved using the internet. He expressed his irritation at the design of webpages, which displayed large numbers of different elements simultaneously. He acknowledged that this is different for different people, but for him, that this was a barrier to accessing web-based information.

RP: far too many things - millions - and it just - get annoying - but that's me I just - it's whether some people can read some people can't - some people can look - different basically

The above examples demonstrate that information about stroke and aphasia is not an emotionally neutral topic, and that the format of information impacts on individuals personally. Emotional responses were seen to hinder individuals' access to information. For example, when participants perceived that materials were presented with insufficient sensitivity to the challenges of aphasia, they were more likely to disengage from the information. This issue underlines the significance of aphasia as a life-changing experience, and highlights the need for information materials to be designed with awareness of the impact aphasia has on people.

7.5.3 Sub-theme 3 Individual visual style

In the same way that the data showed how individuals responded differently to the format and content of the information, so it emerged that participants approached information with varying visual styles. The first issue reflects the differences in how people approach looking at information independently. A second issue relates to the fact that some participants benefit from support when making sense of information.

7.5.3.1 Looking at information

Participants made comments about how they approached the information. It emerged that whilst many start reading information at the top left and work downwards, this did not apply to all, and different styles were revealed. In one discussion about the layout of Template 1, the Group Leader attended to RP's comment about the amount of information presented together on the internet, and asked whether the information in the template format was easier to manage.

GL: coming back to your point RP about internet and things - all over the place - with taking this in you've got to go brain-sentence-picture-hello - so it's going down and going across - did that work as a kind of following it through logically taking each piece of information in?

Two participants agreed with this description immediately, and following a discussion, most participants identified with the facilitator's description of a step-by-step approach to understanding the words and images. This was summed up by SE, who expressed that he found it helpful to process each element of the information in turn.

SE: read that and then you - can move down - settle in your head

NH agreed with this approach in principle, but qualified this with a comment that he may restrict his focus to the top of the card. In the extract below, he appeared to acknowledge that whilst he was able to read the first part of the information, he could no longer process a larger amount of writing, and wanted the key written information to be presented first.

NH: I start at the top - and if it - down at the bottom - and if it was down at the bottom - I wouldn't bother about it - it's up there where it the brain says hello

Following on from these discussions, participants were asked whether they looked at the words or the images first when engaging with information. They differed in their responses, with some expressing that they wanted to see the image first, whilst others read the header sentence first, and then worked downwards towards the image. When discussing approaches to looking at the cards, most participants agreed that the header sentence helped them to understand the information. One participant's comment, however, suggested that he disregarded the header and focused his attention on the sentence and image only, as captured in the extract below.

BT: yes but I wasn't looking at the title

Another difference emerged during the discussion, when a facilitator asked whether participants were looking at the whole template or at the separate elements. JB replied that she preferred to look at the text and images together.

JB: no - I put them all together - I can manage that together - but I can't get much more than that

This preference for a single view of the information meant that, if two different images were presented, she found it difficult to see the link between them, as in the example below. This comment relates to the photographic style, in which the information content was conveyed using two separate images.

JB: I don't see what that [pointing to first image] has got to do with that [pointing to second image]

In a later discussion, JB acknowledged that although she has looked at the whole card, she has not fully perceived or understood it.

F2: you didn't look at that bit JB: no I just - I looked at it but I couldn't see it

Speed of approaching information emerged as an issue for BT, who identified that he scanned too quickly from one element of the information to the next. He found that he was helped by the presentation of one piece of information at a time, as a strategy to slow down his quick visual scanning.

BT: my trouble is I start reading I see - I go brain oh yeah - I - I - before I've - before my brain has absorbed that information - I'm on this one - that's the reason one card's better because I didn't read properly

Here, individual ways of looking at the information were described by participants, perhaps reflecting previous styles of engaging with information. Overall, discussions revealed differences in how individuals approached the materials. Whilst most identified with the description of reading from the top left of the page, working downwards, some variation emerged. Rather than looking at each element in turn, some participants attempted to process the words and images as a single unit and others processed only part of the content presented.

7.5.3.2 Making sense with support

In the focus groups, discussions centred on factors which contribute to independent access to information for people with aphasia. Alongside this, when looking at the set of cards about stroke and aphasia, some participants raised the issue of support in accessing information.

In the extract below, RP explained that the set of cards could be used for talking through information with a family member, and this was particularly important in the early stages after a stroke.

RP: whereas I mean if you've got someone with you - you know family or whatever you can just you know - one at a time go through it all I think particularly the end of when you have your stroke

He identified the role of families in filtering information to people with aphasia, recalling that he had been unable to take on new information during this time, as reported in the extract below.

RP: it's a blank - the first six months is a blank

For those with more severe language impairment, the enduring difficulty of making sense of information independently was articulated by JB. In the extract below she described how difficult it was for her to understand information fully on her own. The second part of her turn suggested that she was likely to misunderstand an element of the information, and then lose the full sense of it, with her understanding "a further difference", that is, more distant from the original meaning.

JB: maybe if it was that I didn't know what it was what it what when it was there at the time I was being told something now I look back and I know what this is either because I got the other side of it so it's got into a further difference for me

She then agreed with her facilitator's explanation of their collaborative approach to the materials. F2 identified that she had read the information aloud to JB, and they then related this to JB's own experience to ensure that it made sense to her personally.

F2: we talked about each one of them didn't we and that was good wasn't it - so one at a time I kind of read it to you and we talked about it - and then you linked it with your experience JB: yes when I couldn't do it and when I didn't - yes I can see it

This section reports data showing that some participants needed support to access information, despite being designed to be maximally accessible. This was identified as a likely

need in the acute stages after stroke, and also needs to be considered for some people in the long term.

7.5.4 Summary of Theme 4

The data analysed within this theme relate to individuality, capturing differences in participants' responses to, and interaction with, the form and content of the materials. Despite varied experiences of aphasia, there was agreement on the importance of providing accurate and realistic but encouraging information about aphasia. This was found to be important for both accuracy of information, and to allow people to identify with the information, using it as a resource to tell others about their personal experience.

Participants varied in the degree to which they identified closely with the information. Whilst most participants related the information to their own situation, others needed the images to be a clearer mirror of them personally for them to feel meaningful. The emotional reactions displayed in response to the materials underlines the impact of aphasia on individuals, and the need for sensitivity when designing information for this population.

Finally, participants varied in their visual approach to information. When reading independently, participants varied in where they looked, how much they looked at, and how quickly they attempted to process the information. It emerged also that for some participants, independent access was not possible. For some, this related to the experience of acute stage aphasia; for others, this was highlighted as an enduring difficulty, with ongoing support needs identified.

In sum, the individuals and their differences interact with the other three themes, affecting their visual access, what they judge to be appropriate or acceptable, and their understanding of the information materials.

7.6 Chapter summary

This chapter has presented the development of the thematic framework and the results of the analysis of the data within this framework, showing the four overarching themes which reflect how individuals responded to the materials in focus group discussions. This qualitative analysis adds insights to the Talking Mats[®] data, revealing links and interactions between the themes. In the next chapter, the results of the iterative design process will be outlined.
Chapter 8 Outcomes of Phase 2 and design of Template 3

In the previous chapter, the analysis of qualitative and ratings data was presented in a thematic framework reflecting the concerns and opinions of the participants. In this chapter, the key points requiring feedback to the graphic designer are summarised to show how aspects of the template were changed in response to user feedback.

8.1 Results of focus groups: feedback to graphic designer

As introduced in Section 1.5, this study followed an iterative, collaborative design process. In Phase 1, the initial prototype, Template 1, was devised using the findings of three literature reviews into the impact on people with aphasia of the implementation of features of accessible formatting, language processing in aphasia, and health information topics required by people with aphasia.

In Phase 2, participants with aphasia reviewed the prototype templates over two separate focus group meetings. Following the first focus group meetings, the researcher compiled feedback for the graphic designer, the results of which were incorporated into the second version of the prototype, Template 2. These materials were then reviewed by participants in the second focus group meetings, after which the researcher again provided feedback to the graphic designer. The resulting final Template 3 was then produced, incorporating all the feedback from the focus groups.

8.1.1 First iteration

The main issues discussed in the first focus groups concerned variations in typography, image style, and number of images per concept. Overall, comments about the design included that participants valued the adult style in which the information was presented, and found the materials acceptable and pleasing. Few comments were made about the language in Template 1, but most participants found the use of a single word heading helpful, and the written content acceptable. The points raised in the first groups were summarised as follows:

 Typography: the largest appropriate font size relative to page size was welcomed, although participants cautioned against a too large font. A sans serif font style appeared slightly more popular than serif font. Participants favoured both large and bold fonts for emphasis.

- Images: clarity of meaning was of primary importance to participants. The selection of images demanded sensitivity to the experience of aphasia, and required increased consideration of possible misperceptions and negative connotations conveyed by images. Images needed to be completely coherent with text. Participants advised consideration of how to add meaning to images, via detail such as arrows to convey movement, for example. Some participants found that personally relevant images were helpful.
- Layout: a limited amount of information per page was perceived as helpful. The layout of Template 1 helped to focus people's attention, and participants responded positively to the use of a header sentence, as well as a concept sentence, above an image. Most preferred a style of layout in which the sentence was illustrated by a single image, but some participants consistently preferred sentences which provided separate images for each key word or concept. Consistency of design, layout, image style, and representations of key concepts was considered important. Use of colour was considered helpful, although some people did not notice it initially. Participants agreed that visual appeal was important in promoting engagement with materials.

8.1.2 Second iteration

The designer incorporated this feedback, applying changes to the prototype, to produce Template 2. Specific and new issues arising from the discussions in the second focus groups were then summarised and reported back to the graphic designer to inform changes to Template 2.

The second focus group meetings established that participants welcomed the concept of a set of cards as a means of presenting a sequence of single-information concepts. Discussions also concerned the amount of information per page and further details about emphasis in font. The review of Template 2 materials provided specific feedback to the designer, as follows:

- Language: participants advised specific changes to language; for example, the phrase 'Most people have never heard of aphasia' was changed to 'Most people don't know about aphasia' to make it more concrete in meaning.
- Amount of information per page: a key finding from the second focus groups was the almost unanimous preference for the presentation of one single concept at a time.

- Typography: further exploration of font size and emphasis concluded that participants found key words which were emphasised with a combination of larger and bold font more helpful than those with either bold or large font in isolation.
- Images: participants emphasised that images needed to depict positive but realistic information, for example reflecting realistic timescales for recovery. Details within images were sometimes distracting; the graphic designer was advised to simplify illustrations as much as possible, and to be aware of the distracting nature of extraneous details. Some types of image were considered difficult to understand, such as abstract graphic images. Participants advised using more obviously recognisable images relating to concrete nouns.

8.2 The final template

Following the feedback, the designer produced the final template. A checklist containing the criteria for the final template was developed, incorporating all participant feedback, and is presented in Table 8.1 below.

The two iterations described resulted in the final version of a template for accessible information. Template 3 functions as a design template for the production of information for people with aphasia. Its design incorporates key principles for accessible information, derived from the analysis of focus group data.

Language

Banner headings to be single words or short phrases to orient to the topic.

Vocabulary to be short, highest frequency, familiar words.

Sentences to be short, and in canonical forms.

Proforms to be avoided, and, where possible, replaced with content words.

Content words to be repeated within sets of related concepts

Flesch Kincaid readability software to be used to monitor reading grades, ensuring RG5 or lower.

Typography

Banner headings to be printed in Vectora Black in 24pt.

Sentences to be printed in Vectora Roman 18pt in black.

Keywords to be emphasised using bold and font two pt sizes larger than surrounding text.

Images

Each sentence to be accompanied by one or two images to illustrate the meaning clearly.

Images to be professionally-produced line drawings or photographs, suitable for adults.

Images to be fully coherent with the text.

Images to be consistent in style, and to depict concepts in the same way throughout.

Images to depict realistic expectations of recovery.

Images to be sensitive to the experience of stroke.

Images to depict realistic experiences of stroke.

Images to be unambiguous, with attention paid to possible visual misperceptions.

Layout

A white space measuring 170mm by 110mm to contain the content.

White background to the overall space.

Coloured banner with heading text printed in bold black reversed into white and aligned to the left side.

Single sentence below the banner to convey information content, printed in black and aligned to the left side.

One or two images placed centrally, below the sentence.

Borders to separate areas of text and related images.

Consistent use of colour in headings and images within specific topics.

8.3 Chapter summary

This chapter has outlined the iterative design process, in which the participants' feedback was relayed to the graphic design team after the first and second focus group meetings. This took account of both immediate feedback and the results of the data analysis, and thus represents the full collaboration of people with aphasia in the design process.

The following chapter outlines Phase 3 of this study, involving the development of guidelines for making information accessible derived from the template principles, and the user-testing and further refinement of these guidelines.

Chapter 9 Phase three: Development of accessible information guidelines

9.1 Introduction

As identified in Chapter 1, prior to this research no published guidelines previously existed which were evidence-based, user-tested, and made available to clinicians, researchers, and others who produce information for people with aphasia. Chapter 1 highlighted the tensions between national guidelines for Stroke which suggest that information should be accessible to all, and the lack of published guidelines to support such provision. Phase 3 of this study aimed to use the template generated in Phases 1 and 2 to develop such guidelines to enable others outside of this project to integrate the knowledge gained during this project into the production of their own materials.

In this chapter, the process for developing the guidelines from the template, and the subsequent user-testing of these guidelines is outlined.

9.2 Development of the draft guidelines

Phases 1 and 2 of this study resulted in the final version of a template for accessible information. Its design incorporates key principles, summarised in Table 8.1. In this section, the process by which these principles were extracted and organised into a set of draft guidelines for accessible information for people with aphasia is outlined.

9.2.1 Aims

The aim of this phase of the study was to develop concise, practical, and accessible written guidelines which could be used by any health professional, researcher, family member or others working with people with aphasia producing new information materials. Objectives set by the research team in pursuit of this aim included:

- To convey the principles of the final template through clear explanation and modelling of the template criteria.
- To promote a practical approach to enable guideline users to produce accessible information materials.
- To user test the draft guidelines with stroke healthcare professionals.

 To derive the final version of the guidelines incorporating the feedback from the usertesting.

9.2.2 Methods for translating the template criteria into guidelines

A collaborative approach within the research team was taken in the development of the draft guidelines. All members of the team contributed, and consensus on the final draft guidelines was reached through a process of discussion.

In line with the aims and objectives, the research team firstly discussed possible methods for communicating the template design criteria. It was agreed that the criteria should be translated into a series of steps which individuals could follow. It was acknowledged that the criteria, as presented in Table 8.1 do not follow on from each other in a sequence. The guidelines however, aimed to provide a practical method, and therefore required translation into logical, active steps which would reflect the order of work an individual might undertake when producing accessible information.

For each of the design criteria, the research team agreed that the presentation of examples of good practice would support users. Hence for aspects of language, typography, images, and layout, examples were provided throughout the guideline.

It was also agreed that, as far as possible, the formatting of the guideline document should use the template principles. It was reasoned that this approach would perform a dual function, in that it would ensure that the guidelines are accessible and easy to use by a range of different users. Such a presentation style would also provide users of the guidelines with a model of good practice.

The following section will outline the five steps which the research team proposed for the draft guideline, and will indicate how the template criteria are incorporated into these steps.

9.2.2.1 Development of the five step approach

Step 1 A short message

The principles of the template which are summarised in Table 8.1 stand on the assumption that the information content to be conveyed has already been agreed. Step 1 therefore concerns the development of the information content to be conveyed.

A key aspect of the template is the presentation of single concepts, which is based on the need for people with aphasia to be faced with one proposition only at one time (see e.g. Caplan et al., 2007). This step directs users to be aware of how many separate concepts may be inadvertently included in their information content. To ensure that the guidelines themselves were accessible to users, the research team agreed to use the term "message" rather than "concept", since this was considered to be more likely to be understood by all users.

The identification of single concepts within an overall larger set of concepts, leads users towards developing sets of sentences which might be used to explain the various concepts. The research team agreed that the development of sentences should therefore form the next logical step when producing information materials.

Step 2 Simple sentences

The research team agreed that the guidelines should provide clear specification regarding accessible sentence forms, and that these should be explained in lay terms to ensure access to all users. For example, lay users were not expected to understand the term "canonical" in relation to sentence forms. The principle of avoiding passive, compound, and complex sentence forms was therefore addressed by providing examples of each of these sentence forms. In each case, examples of the best practice version were also provided, such as active rather than passive sentences. For compound and complex sentences, examples were provided of how to simplify these sentences, or to separate them into two short sentences. Similarly, the issue of avoiding proforms was explained by the provision of examples of pronouns and examples of sentences showing how they can be avoided. The table below shows how the template criteria were translated into the content for the guideline. The template design criteria are presented in the left hand column, with the content for the draft guideline shown in the right hand column.

Template design criteria	Draft guideline content
Language	
	Make your sentence simple
Sentences to be short, and in canonical	Think about your sentence
forms.	Simple sentences work best
	Check the sentence
Proforms to be avoided, and, where	
possible, replaced with content words.	

Table 9.1Template criteria to guideline – sentences

In the draft guidelines, information about the Flesch-Kincaid readability software was provided within this step, since the research team reasoned that this would be a convenient means of checking the complexity of sentences for users as they worked their way through the steps.

Step 3 Easy words

The research team agreed that the guidelines should prompt users to consider vocabulary choices at an early stage in developing information materials. As with the sentence criteria, users were not expected to have knowledge of terms such as "high frequency" in relation to vocabulary, therefore these criteria are translated into vocabulary considered more accessible to lay users, and examples of "easier" and "harder" words provided, both as single words and within sentences. The table below shows how the template criteria were translated into the content for the guideline.

Table 9.2 Template criteria to guideline – vocabulary

Template design criteria	Draft guideline content
Language	Choose your words carefully
Vocabulary to be short, highest	Choose words people use often
frequency, familiar words.	Choose words most people use

Step 4 Good layout

The research team considered that following steps 1 to 3 would result in users producing a set of related written sentences. The next step is to start to prepare the frame within which the sentences would be presented: this stage would direct users to start to produce information within a document. The step therefore draws on criteria specified within the areas of layout, typography, and images, as shown in the table below. The table below shows how the template criteria were translated into the content for the guidelines.

Note that the template criteria for images that relate specifically to stroke recovery, such as "Images to depict realistic expectations of recovery" were not carried through into the guidelines, due to the wider scope of application of the guidelines beyond just this one topic. Table 9.3 Template criteria to guideline - layout, typography, images

Template design criteria	Draft guideline content
Layout	Space
A white space measuring 170mm by	Postcard size is good
110mm to contain the content.	Half of A4 is good
	A5 is good
White background to the overall space.	Make a border around this space
	Choose a border you like
Typography	Typing
Sentences to be printed in Vectora	Type the sentence top left
Roman 18pt in black.	Font size 14 – 18 Use sans serif font Arial or Calibri
Keywords to be emphasised using bold	Make important information bold
and font two pt sizes larger than	Make important information two font
surrounding text.	sizes bigger
	51205 016601
Images	Pictures
Each sentence to be accompanied by	Choose one or two pictures
one or two images to illustrate the	Use pictures that match the important
meaning clearly.	information
	Choose good quality pictures
Images to be professionally-produced	Use line drawings or photos
line drawings or photographs, suitable	Pictures made for adults work best
for adults.	Colour can add meaning
Images to be consistent in style, and to	
depict concepts in the same way throughout.	
Layout	Writing and pictures
One or two images placed centrally,	Put the picture under the sentence
below the sentence	Check the pictures match the important
	information exactly
Images	
Images to be fully coherent with the	
text.	
Layout	Colour
White background to the overall space.	Use white paper
Consistent use of colour used in	Use dark writing
headings and images within specific	Use colour in pictures to help
topics.	
Typography Sentences to be printed in black	

Step 5 Make a set

This step directs users beyond the single template when producing a set of information on one topic. It directs users to return to Step 1, and to select a second concept, or message, instantiated in their second sentence, to repeat the design process with. The research team agreed that this step would support a methodical approach to developing information, which would assist users to comply with the recommendations.

The design criteria of consistency and repetition are emphasised in this step. The table below shows how the template criteria were translated into the content for the guideline.

Template design criteria	Draft guideline content
Consistency Emphasised throughout across language, typography, images and layout of design template.	Keep it the same Use the same layout for each message Use the same style of writing Use the same type of pictures
Language Content words to be repeated within sets of related concepts	Repetition helps Repetition helps people with aphasia Your sentences can be similar Important words can be repeated
Layout Borders to separate areas of text and related images Consistent use of colour used in headings and images within specific topics	Clear sets Use one border colour for all messages in one set Or use one writing colour for all messages in one set

Table 9.4 Template criteria to guideline - making a set

9.2.3 Production of the draft guidelines

Using the content identified in tables 9.1 to 9.4 above, the draft guidelines were subsequently produced as a 19 page document, printed on white A4 paper. They included a brief introduction, and the five steps described above, including the examples and checklists. The body of the text was printed in Calibri 14pt font, with font sizes 16pt and 20pt for the subheadings and main headings respectively. Images were selected from copyright free internet resources and printed in colour. As stated, the principles of the template that governed the design, outlined in Table 8.1, were used to develop the guideline document itself to maximise accessibility and to demonstrate the principles in the context of an authentic information booklet.

9.3 User-testing of the draft guidelines

9.3.1 Introduction

As previously discussed in Phase 2 of the study, user-testing is essential to ensuring that novel products are required, accessible and usable, as judged by their target users. In this context, potential users of accessible information guidelines were identified as any person working with people with aphasia. A user-testing study was therefore identified as a suitable means of gathering data on potential users' responses to the guidelines.

The data gathered were analysed qualitatively to develop an understanding of the responses of the participants. Secondly, feedback informed the research team regarding specific changes required to the content and design of the guidelines. In this section, the methods and results of the user-testing study are outlined.

9.3.2 Methods

9.3.2.1 Participants

Sampling for this phase of the study was opportunistic. Participants were sought who worked routinely with adults with aphasia at any stage after a stroke and who produced information for people with aphasia as part of their professional role. Two groups were identified for this preliminary user-testing: speech and language therapists and staff employed in a national UK voluntary sector organisation.

9.3.2.2 Recruitment

Potential participants were approached in two ways. Speech and language therapists were contacted via local professional networks and voluntary sector staff were contacted nationally via their regional coordinators. Initial interest in the project was sought via an email with the project information sheet attached. Those people with an interest in the study were invited to respond to the research team by email. Nine speech and language therapists and fourteen voluntary staff expressed an interest in the project.

9.3.2.3 Ethics and Consent

Ethical approval for the study was granted by the Department of Human Communication Sciences at the University of Sheffield.

Staff who responded by email expressing their interest in the project were sent a consent form by Royal Mail. They were asked to complete this and return it to the researcher.

Using this strategy, seven speech and language therapists and eleven voluntary sector staff gave their written consent to participate.

9.3.2.4 Study design

Staff who responded by email expressing their interest in the project were sent a copy of the draft guidelines together with the consent form by Royal Mail.

A survey was selected as the most appropriate design for the purpose of collecting user feedback on the draft guidelines. The research team reasoned that participants would be able to respond at their convenience, having had the opportunity to read and use the guidelines within their own work settings. It was anticipated that the data would accurately reflect participants' individual responses to the guidelines.

9.3.2.5 Materials

The materials for this phase of the study included the draft guidelines described in Section 9.2 above and a ten-item survey (see Appendix S). The survey was developed by the researcher and piloted with three colleagues in the research team using an online survey platform. Following the pilot, minor changes were made to the survey, the questions finalised, and the survey questions uploaded onto the online platform.

For each of the ten survey questions, participants were invited to respond with comments in the text boxes provided, in order to maximise responses and to allow for qualitative analysis of the data. No rating scales were used. No minimum or maximum word limit was imposed, and participants were free to respond as they wished.

Participants were asked to comment on their initial responses to the draft guidelines, on which aspects of the content they found easy or difficult to follow and to use, and whether the content was sufficient or not. Participants were asked if they had used the draft guidelines to produce new information, and for their reflections on how the guidelines supported them to do so. Further questions about access to current guidelines about accessible information production, evidence, and resources used by participants sought to probe what people were currently using to support their practice. This information was gathered for qualitative context, rather than anticipating any impact on the development of the draft guidelines per se.

9.3.2.6 Procedure

Participants received a copy of the draft guidelines and their consent form for completion in the Royal Mail once they had expressed an interest in taking part in the project. They were asked to return their signed consent form.

Once the consent form had been received by the research team, the researcher emailed each participant to explain the activity and provided a link to the online survey for them to complete. Participants were asked to read the guidelines and to use them to produce some information in the recommended format. Information topics were chosen by participants to ensure relevance to their usual practice. Information materials generated by participants were not reviewed as part of this process, since it was their views on the usability of the draft guidelines which were sought. Following this, participants were asked to give their views on the draft guidelines via the online survey. Responses were received anonymously.

9.3.2.7 Data management and analysis

All responses from all participants were retrieved by the researcher from the online survey platform and were collated within a single Word document. Framework analysis (Richie & Spencer, 1994) was then conducted on the data, following the stages outlined in Section 7.1, and applied to the whole data set.

9.3.3 Results

Completed surveys were received from seven speech and language therapists and eleven voluntary sector staff. Across both groups of participants, responses were collected for all the survey questions. This section reports the qualitative analysis of the survey data and the subsequent modifications made to the guidelines.

9.3.3.1 Thematic Framework

Framework analysis resulted in the identification of four overarching themes: the need for the guidelines, content, accessibility, and usability, and associated subthemes, all of which are presented in Table 9.5 below.

In this section, each of the four themes will be outlined, with extracts from the data provided to exemplify the issues. Participants are identified by participant group (SLT or VS) and their participant number. Comments are recorded as they were written by participants on the online survey platform, with no amendments to spelling or grammar. Table 9.5 Thematic framework of the survey data

Theme	Subthemes
The need for guidelines	Current practice
	New information
	Perceived quality
Content	Amount and relevance of information
	What works/is easy to do
	What doesn't work/is hard to do
Accessibility	Overall clarity and comprehensibility
	Language and terminology
	Instructions and examples
	Presentation/layout
	Accessibility for different users
Usability	Overall ease of use
	Time demands/efficiency
	Usability for and with different users
	Attempts at applying the guidelines
	Perceived usefulness
	Resources

Theme 1 - Need for guidelines

All respondents welcomed the guidelines and indicated that they were a valuable resource for those producing accessible information, both for people with previous experience and those new to the task. As the comment below suggests, respondents were supportive of the guidelines being made available to a wide audience.

VS 2 – Clear, logical, concise. Useful. Good that they are evidence based. Would like to see a published booklet be made available for wider distribution to a huge range of organisations who want to make their information more accessible to comply with the Disability Discrimination Act, including both the private and public sector.

Comments regarding current practice suggest that both SLTs and VS respondents have built up their experience through working with people with aphasia, and gathering ideas from a variety of sources. All identified a lack of, and a need for published guidance in this area.

SLT 6 – Knowledge from SLT training or word-of-mouth from research. I feel like it is not explicitly taught in relation to accessible information. And I feel that awareness of accessible information necessary for people with Stroke aphasia is poor, or if present not actively followed.

VS 7 – I don't have any formal guidelines that I refer to. I have generally produced accessible information on the basis of my own knowledge, the knowledge of others who work very closely with people with aphasia and feedback from people with aphasia.

Many respondents felt that the guidelines provided them with new information which they could now implement.

VS 9 – some aspects of the guidelines were really useful and suggested new ways of presenting information that I hadn't considered using before.

SLT 2 – I didn't know that people with aphasia like the key word to be bigger than the other text - this is something new that I'm doing.

Many respondents commented on the quality of the guidelines. Generally respondents felt the guidelines were sound, high-quality, and evidence-based.

SLT 2 – I feel confident using the guidelines as I know they are devised from what people with aphasia have said they want.

Respondents also reported that the perceived credibility of the guidelines made them feel more confident using them with others.

SLT 3 – I feel that having read this guide I will now need to re-evaluate the information I had previously felt to be accessible and that having a clear guide based on evidence rather than my own opinion and brief training gives me more inclination and more support to do this with colleagues.

VS 2 – Having a published set of evidence based guidelines also gives credibility to the content and is therefore useful in influencing the wider teams.

Overall, all respondents perceived the guidelines to be useful and agreed that they would use them in the future, both for their own work, and to support others.

VS 8 – Will use them in the future. They should form part of our training/induction!

VS 5 – I have also been to stroke groups that produce information for members that is not aphasia accessible. I then advise people on how to improve it but having guidelines that people can see would be fantastic, much better than me trying to explain and just giving examples.

Many respondents highlighted areas in which the guidelines would be particularly useful. Specific suggestions included patient and family information leaflets (e.g., health information and advice, explanations of medical procedures, details of service provision, communication strategies), goal setting, letters, emails, meeting minutes, signs, PowerPoint presentations, notices, websites, training materials, advertising and campaigning leaflets, resources for communication groups. Some respondents felt that more information would be needed to adapt certain information, e.g., surveys, evaluation/feedback forms.

Theme 2 - Content

Respondents commented that all of the information contained in the guidelines was required and relevant. While the majority of respondents thought the amount of information included in the guideline was manageable, two respondents felt that document length was an issue. However, this was mainly related to a 'first glance' impression.

VS 10 – Quite daunting at first - it looks like a lot of pages to get through! But once you start reading it, it becomes clear that it's very spread out, and therefore, quick to read. I don't know if there's any way of condensing it at all, I know the examples have to be a certain size, but the rest of the text maybe? It should also be printed front and back, to condense it into more of a 'handy guide'.

Respondents commented on the potential difficulty of providing word processing instructions for software which is used in different versions, but others reported a need for more detailed instructions, reflecting, perhaps, different levels of word processing experience. The issue of further training raised by some respondents is key here.

Suggestions for additional content included background information about aphasia and why it impacts on reading ability, further details on picture resources, and on formatting.

VS 10- Something more about developing the A5 sets of messages. As I want to create newsletters, I want a bit more information about how many pages is too long, the best layout.

Several respondents commented that information about the study which produced the guidelines should be included, to lend authority to the recommendations.

VS 9 - It might be nice to include a bit more about the study...so people realise that the guide hasn't just been produced using guesswork, but with sound scientific backing behind it.

Respondents commented on specific aspects of the guidelines which worked particularly well. The majority of respondents liked the inclusion of a contents page and the five steps approach. Many respondents liked the use of Flesch Kincaid for establishing readability. This was a new tool for all respondents and they reported it could be implemented in a range of communications, such as email, letters, and newsletters.

VS 8 – Have never used Flesch Kincaid before and I believe it should be defaulted (if that is possible) on all our PCs in the Stroke Association. This will enable us to practice what we are preaching.

Some respondents commented on aspects of creating information that was particularly useful, e.g., use of borders to surround information, use of larger font, using active language, avoiding pronouns.

Theme 3 – Accessibility

The majority of respondents reported that the guidelines were very clear and easy to follow and understand. It was apparent that the overall sense of clarity was aided by layout, with comments on specific design features.

SLT 5 – Clearly set out, written in an accessible format. Good examples clearly separated from the instructions by block of colour or border. I like them!

VS 8 – Very simple to follow. I like the way in which they are laid out - it makes it much easier to understand.

Two respondents identified specific terminology used within the text that was ambiguous or unfamiliar. Another respondent commented on the usefulness of word definitions, e.g., pronouns. Some specific suggestions were made, for example,

SLT 5 – There was some terminology that I had to google e.g., Flesch-Kincaid, sans-serif, could there be a footnote relating to these or a glossary at the end?

The majority of respondents felt that the guidelines were written in an accessible format which demonstrated the principles of accessibility clearly.

SLT 1 – ...the guidelines themselves also followed many of the principles you set out for producing easily accessible info for PWA.

Some respondents perceived the accessible format as a positive feature, acknowledging that the guidelines would be accessible to a range of users. However, for some respondents, attempts to make the document accessible, e.g., by using simple language and short simple phrases, were viewed more negatively.

SLT 6 – I found the short sentences used more difficult to follow than typical sentences. I realise that the document itself is likely to be set up as a model for accessible information, however wonder whether if the guidelines are intended for people who do not have aphasia, they would be easier to read in longer, more typical sentence/paragraph explanations. I found the reduced complexity of the language almost off-putting.

Comments relating to instructions and examples indicated that there were clear associations between accessibility and usability. Respondents generally agreed that instructions were clear

and consequently easy to follow. The "5 steps" aided clarity and helped respondents to follow the instructions. SLT respondents liked the use of examples, particularly those relating to the use of pictures to supported written text. Two SLTs commented that the examples made the guidelines easier to use. VS comments relating to examples were more variable. Some VS respondents agreed with SLT respondents commenting that examples were well presented and useful. Others felt that more examples were needed, particularly those demonstrating good examples of adapted information. Some VS respondents also thought the scope of the examples used was limited and did not relate closely to their own role.

VS 6 – Most of the picture examples show statements (e.g., Physiotherapists help with walking). It would be good to have more picture examples of how to adapt practical information (e.g., a letter, details of an out-patient appointment, or asking for views on something).

VS 9 – The examples given seem to concentrate on producing very formal material such as medical information, information about rehabilitation, information about services etc.

Theme 4 - Usability

Most respondents felt that the guidelines were easy to apply. However, not all respondents agreed, particularly on the first attempt to implement the guidelines.

SLT 4 – Generally I found the activity very useful and much harder and more time consuming than I had anticipated. I really had to think through the points and refer back to the guidelines. When doing this I found I had to keep flicking forward and back through the guideline.

Others agreed that time available was a barrier to implementing the guidelines. One SLT commented that making time for adapting information was challenging in the context of their current workload.

SLT 1 – It is hard to make time for this kind of thing when we are constantly faced with so many patients to see on the wards. I know that this is important though and we should organise our time better. However the reality is that other staff, and patients themselves want to be seen by us face to face and we are spread very thinly.

Another SLT suggested that they would be more likely to allocate such a task to an assistant. SA respondents agreed that time was an influencing factor. One VS respondent commented that time demands as well as potential cost would influence the type of document they would attempt to adapt.

VS 1 – There are cost and time implications involved when we produce information for a large audience about, for example, stroke policy and survivors getting involved in campaigning. We

use guidelines to make info more accessible generally but probably will never be able to make most of it accessible for people with aphasia.

Others agreed that particular types of document were more challenging, and consequently more time consuming, to adapt. This was particularly apparent in the responses of VS respondents who were interested in adapting a greater range of information. Some found specific aspects of the process more time consuming, particularly finding appropriate images.

VS 6 - I was able to apply the guidelines to statements of fact but struggled to adapt questionnaires.

VS 10 - I had to stop when I realised translating a newsletter wasn't going to work, and needed more thinking about!

Respondents commented on specific aspects of the guidelines which they found easy or more difficult to use, for example some identified the issue of increasing length of documents if separated out into individual messages.

VS 9 – The accessible information I produce has a lot of space around the text and images but each message does not take up half a page of A4. A simple volunteer role description being produced in this way would stretch to a dozen pages or more!

Most reported that finding pictures was hard and time consuming, and were daunted by the prospect of seeking suitable images in consistent styles to support all concepts. The issue of copyright was also raised.

SLT 10 - It says 'use the same style pictures', but finding good, relevant, simple, adult images which were either all photos, or all line drawings was really difficult / impossible.

VS 6 - It was also impossible to find images for every message - what can I do when I have no picture? Can I just use the text?

Some reported taking their own photographs and using photoshop programmes or picture symbols, and one had had specialist material made. A few commented on how good the pictures section was e.g.

VS 7 - I thought these resources were so useful and have given me extra tools to use in my current role, which is really helpful, and I found the section on good pictures very useful.

Word processing issues relating to guideline use appeared to vary according to individual skill level. Some found the word processing demands reasonable, but others thought the demands were high:

SLT 7 – I think that people producing these guidelines will need medium to high computer skills and this needs to be taken into consideration when thinking about who will produce the information.

Respondents identified that word processing knowledge should not be assumed.

SLT 2 – ... it was difficult to know how to put ideas into a box, how to add a picture. Maybe attach a simple 'how to' for putting pictures and text into Word.

Whilst all agreed on the usefulness of Flesch-Kincaid, participants varied in how easy they found it to set up. This seemed to be linked to the version of Word being used.

Overall, there was disagreement about the ease with which different users might be able to apply the guidelines. Some felt that any health professional would be able to use the guidelines.

SLT 3 – I feel that until a lead is given in how to make info more accessible people will continue to think of it as a 'speech therapists job' and yet guidelines like these make it easy for any member of the team.

VS 2 – Easy to follow and to apply, even for people who do not have much experience of working with people with aphasia.

Other respondents felt that the guidelines in their current format could not easily be used by all. One respondent felt that the guidelines could not be used by a person with aphasia. Specific suggestions for improvement were made including development of cue cards and training packages for non SLT users.

SLT 7 – May be more difficult for someone other than speech and language therapist to use. The guidelines may need to come alongside some further training on aphasia and supported conversation if being used by other AHP's.

Summary

Overall, participants expressed strong support for the guidelines. They identified clearly the need for published guidelines to be made available as a resource for individual use and for staff training. Content of the guidelines was identified as appropriate, with constructive comments received on content which respondents felt would improve the guidelines. Accessibility of the guidelines was rated highly overall, and most participants appreciated the design of the guidelines in modelling good practice in accessible information.

Comments regarding the usability of the guidelines were generally positive, but were honest regarding the challenges participants had faced, including those of time, resources, and individual skills and knowledge. The issue of further training to support the implementation of the guidelines was raised repeatedly.

9.3.3.2 Results of the qualitative analysis and modifications to the guidelines

As outlined above, the survey responses provided support for the publication of the guidelines and for aspects of the draft presentation, particularly in terms of its content, accessibility and usability. In this section, the specific modifications to the guidelines made by the research team on the basis of the user-testing feedback are described.

Participants' comments suggested that guideline users may require more introduction to the document. Therefore an introductory section was inserted, including basic background information about aphasia, and the rationale and purpose of the guidelines.

Participants' comments (see Theme 1 above) suggested that the credibility of the guidelines rested on both the use of the evidence base in developing the template (Phase 1 of this study) and the involvement of people with aphasia in the development of the template (described in Phase 2 of this study). Participants commented that the guidelines should include information about the research underpinning the guidelines, in terms of both the current study and the wider literature. Brief background to the project was provided therefore in a section entitled 'Development of these Guidelines'. In addition, a 'Further Reading' section was inserted at the end of the document.

Participants varied in how much word processing advice they required, therefore all word processing information additional to the basic formatting of the template was collated in a section at the end of the document. Additionally, a web-link for Microsoft Word Help was provided.

The five steps approach was well-received and this was therefore retained. Small changes to ensure consistency and clarity were made to the final version, for example the inclusion of sub-headings within each Step, for example Step 1 – A short message – Make sure you have a clear message; Step 2 Clear sentences – Make your sentence simple; Step 3 Easy words – Use everyday words; Step 4 Good layout – Create a space for your message; Step 5 Make a set – Make a set of messages.

In the draft Guidelines, each of the five steps concluded with a small checklist. For the final version of the Guidelines, these small checklists were replaced by a single final checklist, tickbox in style to encourage user interaction, after the final Step.

Table 9.6 summarises the differences between the draft and final versions of the guidelines in order of presentation in the final document. The left hand column shows the draft Guidelines content, the right hand column shows the content of the final version of the Guidelines. For example, the draft Guideline contained a brief introductory paragraph; in the final version, the Introduction was expanded significantly to include further background about aphasia, the need for the Guidelines and instructions for their use.

Draft Guidelines	Final version of Guidelines
No authors or acknowledgements included in	Authors listed and statement of
draft version.	acknowledgements
Brief, single paragraph introduction.	Introduction
	Background
	What is aphasia?
	The need for Guidelines
	Development of these Guidelines
	Who are these Guidelines for?
	What are these Guidelines for?
	How to use these Guidelines
The Five Steps – different coloured fonts for	Black font.
each step	Orientation to final checklist, Word advice.
Short checklist for each step	Final checklist with interactive tick boxes.
	Checklist points subsumed into instructional
	bullet points within each step.
Variable use of headers within each step	Each step has a header instruction e.g.
	"Use everyday words"
	"Create a space for your message"
Use of language	Increased specification to instructional
	language e.g. "logical order" rather than
	"sensible order"; "choose a simple border"
	rather than "choose a border you like".
	,
Use of images	Examples taken from template materials
	developed in Phase 2, to demonstrate the
	principles in practice.
Step 1: A short message	Steps remain the same, identified as helpful in
Step 2: Clear sentences	user-testing.
Step 3: Easy words	Slight differences as above to increase clarity,
Step 4: Good layout	to demonstrate principles.
Step 5: Make a set	
Using the materials	Included in the Introduction
	Additions to final version:
	Interactive checklist for all formatting
	features.
	Word processing advice for:
	Flesch-Kincaid in Microsoft Word
	2003/2007/2010
	How to: make a box; put a border around a
	box; add a picture; edit a picture.
	Web link to Microsoft Word help site.
	Further reading section
	rather reading section

Table 9.6 Changes to the Guidelines document following user-testing

The modifications made following the user-testing resulted in the final version of the Guidelines. These were agreed by the research team and submitted for publication. The Guidelines were subsequently published by the Stroke Association (Herbert et al., 2012) as a 30-page booklet, available to purchase in print or to download without charge, via the Stroke Association website (see Appendix T).

9.4 Interpretation of main findings

This phase of the study aimed to develop concise, practical, written guidelines, accessible to any user. To achieve this aim, the research team set out to firstly convey the principles of the final template through clear explanation and modelling of the template criteria, and secondly to develop a practical and methodical approach to enable guideline users to produce accessible information materials.

In this phase of the study, the principles for accessible information, embodied in the template developed in Phases 1 and 2, were translated into draft guidelines. The template design criteria provided the basis for the research team to develop a step-by-step approach to creating accessible information. The draft guidelines were then subjected to testing by potential users, who gave their feedback via an on-line survey.

The results of the user-testing survey resulted in an understanding of the key issues for users of the guidelines, such as the need for guidelines, the content, accessibility and usability of guidelines. The results of the qualitative analysis were then used to further refine the guidelines, by generating amendments and additions to the draft document, to produce the final version.

The user-testing findings were largely positive, supporting the need for published, high-quality, evidence-based guidelines in the area of accessible formatting. Overall, participants found the draft guidelines easy to use and they commented favourably on the step by step approach and the use of examples to explain aspects of language and formatting. Results of the qualitative analysis suggest, however, that further training to users on how to use the guidelines may be indicated.

It is interesting to note that, in the guidelines, three separate steps were used to guide users in their composition of accessible language which adheres to the template criteria. In contrast, the criteria relating to typography, images, and layout were managed within two steps. This finding is important, since it highlights the complexity of language impairments in aphasia and the importance of a clear specification for the language used in information materials. As argued in Chapter 2, previous studies (e.g. Brennan et al., 2005; Rose et al., 2003; Rose et al., 2012) have underspecified the language component of accessible information materials for people with aphasia, with greater research focus emphasis placed on issues such as font size and the use of images. Further, whilst the Flesch-Kincaid reading tool can be a useful adjunct, it is insufficient in itself in guiding composition, since it does not reflect the complexity of factors which facilitate comprehension in aphasia. The facilitative features relating to lexical terms, sentence forms, and structural priming introduced in Section 2.3, which were incorporated in the template, and subsequently specified in Steps 1, 2, and 3 of the guidelines clearly demonstrate these complexities. This represents an important contribution to this field.

The template developed in Phases 1 and 2 of this study was designed to convey health information to people with aphasia. User-testing highlighted challenges when extending the scope of the types of information materials, such as questionnaires and newsletters, suggesting a need for further developmental work in formatting different types of documents using the principles of the template.

Interestingly, several of the areas which participants identified as potentially problematic in adhering to the guidelines involved accessing skills and resources beyond the scope of the guidelines themselves, such as word processing skills and access to suitable image resources. As suggested by several survey respondents, supplementary training is indicated, to ensure that users have resources to support their use of the guidelines.

The changes made to the draft guidelines and carried through to the final version reflected the user-testing process. The main changes to the document included the addition of content to orientate users to the guidelines, to support understanding of the principles, and to provide additional word processing instruction. Some content was moved, and in the case of the checklist, consolidated at the end of the document.

As identified in Chapter 1, no user-tested guidelines to support accessible formatting of information for people with aphasia had been published prior to this study. Previous studies in the area of accessible information have presented findings which authors suggest could be used to inform the design of information materials (e.g. Rose et al., 2011a), but none have translated such findings into practical, accessible guidelines. Previous guidance documents have been produced (e.g. Parr et al., 2008), but no studies have undertaken user-testing of these to evaluate either their usability or their impact on information materials. The development and user-testing of draft guidelines described in this phase of the study is therefore a novel contribution to the field of accessible information for people with aphasia.

The publication of the guidelines represents significant progress towards enabling healthcare providers and others to provide information to people with aphasia which meets both the standards required by national policy (NHS England, 2015) and Stroke guidelines (Intercollegiate Stroke Working Party, 2016; NICE, 2013) and the preferences of people with aphasia (e.g. Rose et al., 2011a). Further training to support the implementation of the guidelines is indicated, as identified through the user-testing process outlined above.

The impact of these guidelines on the outcomes of people with aphasia remains unknown, however, and further research will be required to evaluate their effectiveness. Firstly, the quality of information produced by healthcare staff and others will need to be evaluated to determine whether the guidelines are sufficiently well specified for their purpose. Secondly, research needs to explore the responses of people with aphasia to information materials produced according to the guidelines. Finally, responses of people with aphasia to a broader range of types of information produced using the guidelines, including formats such as newsletters and questionnaires are needed. This is particularly important, given that the principles carried into the development of the guidelines were based on health information only.

9.5 Chapter summary

This chapter has outlined the stages involved in the translation of the template principles into draft guidelines and the subsequent user-testing of the draft guidelines. A focused discussion of the findings of this phase of the study has highlighted issues for practice and further research.

The following chapter presents a full discussion of the study in relation to previous and more recent literature.

Chapter 10 Discussion of findings

The purpose of this chapter is to consider the findings of this study, with the aim of developing an "explanatory account" (Spencer, Ritchie, & O'Connor, 2003: 212) of the issues pertinent to designing health information for people with aphasia. It is important to acknowledge here that the methods reported in Phase 1 and Phase 2 of this study were conducted between October, 2010 and April, 2011. Since the original literature reviews to inform the materials and methods for the study were conducted, further research has been published. In this chapter, the more recent research findings will be introduced into the discussion.

A further aim of this chapter is to address issues surrounding the quality of the study, including validity and transferability of the findings, and to reflect on the strengths and limitations of the study. Finally, implications for practice and future research will be considered.

10.1 Aims of the study

The overarching aim of this study was to collaborate with people with aphasia to develop a novel template for conveying health information in an accessible format for people with aphasia. Within the concept of health literacy, such a template may be viewed as an intervention, since it supports access to health information (Coulter & Ellins, 2006).

The aim of Phase 1 was to generate a prototype template based on relevant research evidence. The design of the initial prototype template was informed by reviews of the literature in three areas: evidence about the impact of modified formatting on people with aphasia; evidence regarding the facilitation of language processing in people with aphasia; and evidence about the health information topics required by people with aphasia. The principles derived from the research evidence were instantiated into a physical template through a collaborative design process with a graphic design team and a network of relevant health professionals.

Phase 2 followed a collaborative, iterative design process with the graphic designers and people with aphasia, with two main aims: to explore the views of people with aphasia regarding aspects of the template, and hence to generate the final Template 3. In facilitated focus groups, participants gave their opinions of Template 1, and subsequently of Template 2, via individual supported discussions and group discussion, providing qualitative data and rating scales data. In the first focus group meetings, participants were also asked to give their preferences for aspects of the template, for which clear evidence had not been identified in the literature, namely: image type, including photographs and line drawings; number of images per concept; font size; font style; and emphasis in font.

In the second focus group meetings, participants gave feedback on the revised template. New issues which emerged during the first focus group meetings, including the ideal amount of information presented on one page, and further clarification of emphasis of key words, were also explored.

In Phase 3, the research team aimed to translate the principles instantiated in the final Template into practical guidelines which could be used by anyone working with people with aphasia and producing accessible information for this population. Draft guidelines were developed through a process of collaboration between the members of the research team. Following this, participants working in speech and language therapy and the voluntary sector were recruited to user test the guidelines and to give their opinions via an online survey. Data generated in this phase were used in qualitative analysis and to inform modifications to the draft guidelines.

10.2 Summary of main findings

As discussed in Chapter 2, research into modifications to formatting to increase the accessibility of written information suggests that individual or combined formatting features can facilitate comprehension (Brennan et al., 2005; Rose et al., 2003). Some evidence was, however, unclear, such as the use of images (Brennan et al., 2005; Dietz et al., 2009), and some evidence was missing, such as evidence for emphasis in font. The literature regarding language facilitation in aphasia provided the specification required to identify principles for the template in terms of the choice of lexical terms (e.g. Coltheart, 1981), sentence forms (e.g. Caramazza & Zurif, 1976), and the use of priming to maximise reading comprehension (e.g. Hartsuiker and Kolk, 1998). Finally, examination of the literature into information topics required after stroke found a consistent need for basic information about stroke and aphasia, and information about the emotional consequences of stroke (e.g. Avent et al., 2005; Hanger et al., 1998; van Veenendaal et al., 1996). Thus, although the evidence for aspects of formatting was limited, the findings of the three literature reviews together generated sufficient evidence to identify principles for the design of the prototype Template 1 (summarised in Table 3.1). The Template 1 materials for review in Phase 2 were then produced in collaboration with the graphic designers.

Phase 2 of the study resulted in a successful collaboration with people with aphasia in an iterative design process (Sears & Lund, 1997). People with mild, moderate, and severe aphasia participated in facilitated focus groups to give their views on the prototype design template derived from the current evidence base. The feedback to the designer, collected after each focus group, and the qualitative analysis of the focus group discussion data, resulted in the development of a checklist for the final Template 3 design (Table 8.1).

Phase 3 resulted in the successful collaboration between members of the research team to develop draft guidelines derived from the accessible information principles instantiated in the final Template generated in Phase 2. Following user-testing by speech and language therapists and voluntary sector staff, the guidelines were modified and finalised by the research team. These guidelines were subsequently published by the Stroke Association, and made freely available via the Stroke Association website. They are the first evidence-based, user-tested aphasia-accessible information guidelines to be published, and thus represent a novel contribution to the field of accessible information and aphasia.

In the following three sections, the findings of this study will be analysed in relation to the current research evidence in three main areas. In Section 10.3, the findings of the Phase 2 study which relate to the optimum design of health information for people with aphasia will be considered with reference to contemporary research findings, with the aim of explaining the results within a broader theoretical context. Section 10.4 discusses issues of individual variation, both in relation to the methodology and the findings of this study. In Section 10.5, the methods used for involving people with aphasia in the development of accessible information will be discussed.

10.3 Designing health information for people with aphasia

The evidence reported in Chapter 2 resulted in the development of principles for the design of Template 1. As noted in Section 2.2, the evidence base for specific features of accessible formatting was small, reflecting that little research had at that time been published in this area. Previous studies have explored variations in design in terms of the impact on comprehension and the preferences of people with aphasia, measured by responses to testing, surveys, and interviews.

In the following discussion, issues related to designing health information for people with aphasia will be considered, in order to embed the findings of this study into the current evidence base. This discussion will be structured to reflect aspects of content and formatting incorporated into the template design: topics, language, amount of information per page, typography, use of images, layout, and issues of acceptability.

10.3.1 Topics

Research into the information needs of people after stroke forms a large body of work, addressing issues such as education needs (e.g. Hafsteinsdottir et al., 2011; Kerr et al., 2010), information formats (e.g. Eames et al., 2010; Hoffmann et al., 2007), and the experiences of patients and family members obtaining and using stroke information (e.g. Danzl et al., 2016; Eames et al., 2003). Some of the studies cited above included people with aphasia in their sample, but since the characteristics and needs of this sub-group were not specified, such studies offer limited additional insights into their information requirements. Studies into the experiences and information needs of people with aphasia are limited in number (e.g. Parr et al., 1997; Rose et al., 2010), but provide clarity on specific information needs and powerful evidence of the ways in which people with aphasia are disadvantaged in terms of information provision after stroke.

In the field of accessible information, the materials presented to research participants have varied according to the research aims, including information both related and unrelated to participants' health needs, general knowledge materials, and personally relevant information.

In the current study, topics for the template materials were identified through the literature review reported in Section 2.4. The review found a high level of consistency of findings between the studies, with identified information needs relating to early, rehabilitation and chronic phases, and stroke impairments, activity, participation and psychological well-being. Similar findings have also been reported in studies published subsequently (e.g. Hafsteinsdottir et al., 2011; Danzl et al., 2016).

Only two of the studies reviewed, however, were solely concerned with either the needs of people with aphasia (Parr et al., 1997) or of family members of people with aphasia (Avent et al., 2005). Based on an assumption that people with aphasia share the same information needs as any other person after a stroke, the findings of the other studies reviewed were included to inform the possible topics for the template materials. Topics were selected for the template on the basis of their likely relevance to the focus group participants.

During the focus group discussions, participants were not asked for structured feedback about the topics, but all responded readily to the content, suggesting that they considered the topics relevant and pertinent. This finding was in keeping with the findings of Rose et al. (2011a), who report that participants engaged more with information they perceived as relevant to their own situation. Focus group participants reflected on the difficulties they had experienced obtaining basic information, for example one man reported that he did not find out for several weeks that he had suffered a stroke. Such experiences are in keeping with the literature (e.g. Hanger & Mulley, 1993; Hanger et al., 1998), which reports a fundamental lack of understanding among study participants of basic concepts about stroke, such as the causes of stroke. As well as causing dissatisfaction with services (Hinckley et al., 2013; Tomkins et al., 2013), this also has implications for people's subsequent understanding of risk factors associated with possible future strokes, and their ability to engage in self-management (Coulter & Ellins, 2006). Whilst several authors (e.g. Danzl et al., 2016) report the need for repetition of information over time, an ongoing lack of understanding suggests in addition that the concepts associated with stroke may be inherently complex, requiring carefully modified explanation.

10.3.2 Language

Aphasia is an impairment of language processing, and consideration of the language used in information for this population is therefore vital. Research in the area of accessible formatting for people with aphasia has focused on the use of readability formulae (e.g. Flesch, 1948) and terms such as "simple words and short sentences" (Rose et al., 2003:950) to describe and evaluate linguistic complexity. Research findings indicate the need for simple and straightforward language (e.g. Brennan et al., 2005; Rose et al., 2011a). In the field of applied aphasiology, more specific description of language processing in aphasia has been conducted, with the aim of providing simplified text in newspapers (Devlin & Tait, 1998) and on websites (Singh et al., 1998).

Such work has not carried over into the field of health information, however. Aleligay et al. (2008), in their analysis of the lexical and syntactic aspects of health information materials given to people with aphasia, identified high proportions of low frequency, low imageability words and complex sentences. Information for people with aphasia also continues to be evaluated at reading grade levels in excess of the recommended levels (Surman & Bath, 2013), suggesting that healthcare professionals need more specific guidance on how to write information for readers with aphasia.

The literature review conducted for this study, reported in Section 2.3, resulted in the identification of principles for generating text in ways known to facilitate language processing in people with aphasia (summarised in Table 2.3). These principles were then applied to develop the sentences for the template materials. Concepts were presented singly. Vocabulary choices were guided by evidence of lexical characteristics known to affect single word

processing, including imageability, frequency, and length (e.g. Coltheart, 1981), abstractness (Franklin et al., 1994) and use of non-literal language (Papagno & Caporali, 2007). Sentences were of single clause, canonical structure, avoiding complex structures, non-canonical forms, and reversible sentences, to maximise ease of reading (Caramazza & Zurif, 1976). Proforms were avoided, and replaced by repeated content words (Berndt & Caramazza, 1981). The concepts of structural priming, lexical boost and repetition priming were then applied, by repeating structures and key words (Hartsuiker & Kolk, 1998; Pickering & Ferreira, 2008). The sentences were assessed using the Flesch-Kincaid Reading Grade level tool (Flesch, 1948) to ensure that none exceeded Reading Grade 5.

In this study, participants were not asked to rate the language content on the template, but points for discussion were welcomed. Individuals commented on their own reading ability, with some referring to their comprehension of the template text, and some participants reflected on their early experience of aphasia, when they were unable to read.

Generally, participants commented more on the visual appearance of the text presented than on the language itself. Only one participant commented on the specific vocabulary used, and others did not extend these ideas. Findings from the third sub-theme demonstrate that a clear match between text and image was important, suggesting that participants did read the text, where possible, but did not comment.

As Krueger (1998) notes, the analysis of focus group data should include an account of what participants did not say, as well as reporting on what they did say. Several reasons could account for the paucity of comments in the current study. The application of the evidence base (see Table 2.3) and the systematic development of the sentences could mean that the language level was appropriate and therefore non-controversial. Had the materials incorporated language known to be more difficult to process, such as non-canonical or complex sentences, or low frequency, abstract vocabulary, participants may have responded negatively. Participants were encouraged to comment critically on the template, but the methods did not incorporate a structured way to respond to the language element, as was done with typography and images. Communication facilitators may have read aloud the content during individual discussions, ensuring that participants were able to understand, thus allowing them to focus their critical attention on more visual aspects of the template. Finally, since language is an abstract concept, aphasia itself is likely to limit people's ability to discuss it overtly.

Research findings (e.g. Aleligay et al., 2008) suggest that information about lexical and syntactic processing in aphasia constitutes established knowledge, but that this has been

inadequately applied to the production of health information. In addition, the inherent complexity of information about stroke suggests that explanations require a systematic way of reducing the cognitive load on readers (Wilson & Wolf, 2009). In this study, the application of principles from the literature on structural priming (Hartsuiker & Kolk, 1998; Pickering & Ferreira, 2008) resulted in a novel presentation of text within the template. Such principles have not been previously identified in the field of accessible health information. The potential impact of these principles on readers' comprehension of text, and on their perceptions of acceptability, therefore require further scrutiny.

Further work is now required on two fronts. Firstly, the factors to facilitate language processing in readers with aphasia need to be more clearly specified than in studies to date. The development of principles for generating text in this study contributes to this field.

Secondly, further work is needed to increase the knowledge of healthcare professionals regarding the specific nature of aphasia. Specifically, knowledge about facilitating features of language for readers with aphasia requires more structured explanation in order for professionals who write information for people with aphasia to understand and apply the knowledge. Without this shared understanding, it is likely that there will continue to be a gap between guidance and practice.

10.3.3 Amount of information per page

Although previous studies have aimed to gain consensus on aspects of accessible design, the literature remains underspecified in relation to the amount of information presented per page. Previous research has focused on the length of documents, and no studies have specifically defined the number of concepts per page.

In this study, participants raised the issue of amount of information during the first focus group discussions, identifying that health leaflets or webpages contained too many different ideas on one page. Subsequently, participants rated the accessibility of materials in which one, two, or three concepts were presented on one page at a time. They unanimously rated one or two concepts highly, perceiving any more than this as "too much". Apart from their initial visual reaction to the amount presented, participants preferred single concepts because this guided them to read more slowly than they might otherwise attempt; consequently, this more measured approach supported their understanding and memory of the concepts. In a similar way, where participants rated two concepts per page highly, the linked meaning helped readers to relate the concepts to each other. Presentation of single concepts was also perceived as a support for people early after stroke, or for some with chronic aphasia, as a tool to support looking at information together with a family member.

Previous research findings are mixed, perhaps due to a lack of definition of the concept of amount. Rose et al. (2003) report that some participants did not like the materials adapted along 'aphasia-friendly' principles, because they perceived them as "too long" (p.959) although, as the authors acknowledge, the meaning of this is not clear. Rose et al. (2011a) coded participants' comments on the issue as facilitators (small amounts of information) or barriers (large amounts), but again, these were not specified. Interestingly, in contrast to the earlier (Rose et al., 2003) study, comments were recorded from some participants who expressed a dislike of the "reduced amount of information" (p.339) in the adapted materials. This lack of definition in the research is also inherent in the study by Ghidella et al. (2005), who identified "a reasonable amount of information" (p.1138) as one element of an objective measure of quality of websites designed for users with aphasia. No further description of the concept is provided, however, and no specific findings of it are reported.

Rose et al. (2012) attempted to quantify the issue of document length by asking participants to state their preferences for five A4 documents, varying from 3, 8, 17, 26, and 38 pages. They found that 43.6% preferred the longest booklet, compared with 2.6% preferring the shortest. The remaining participants' ratings were spread evenly across the three other booklet lengths. Since, as the authors claim, the five booklets contained exactly the same information, it may be that the longest document spread the concepts out over the pages in a way that allowed readers to focus on a small number of concepts at a time. These findings help to clarify that a clearly delineated, sequential presentation of concepts is more important for accessibility than length per se for readers with aphasia. This is supported by the findings of more recent studies, for example Knollman-Porter, Wallace, Hux, Brown, and Long (2015) who found that people with aphasia preferred to read short paragraphs of one or two sentences rather than a full page, and Jayes and Palmer (2014), who report that participants preferred to see a limited amount of information per page.

When two people look at information together, the presentation of single concepts provides a means of slowing the rate of information and provides an opportunity for individuals to check their shared understanding as they proceed. This is in contrast to both traditional written leaflets and 'aphasia friendly' materials, where it would be difficult to hold a shared focus of attention on a page containing many concepts. Such presentation is consistent with the notion of supported reading proposed by Knollman-Porter et al. (2015).

It is notable that even the presentation of three concepts per page is less than is generally seen in health information, whether this is adapted for people with aphasia or not. This is therefore an important finding, which helps to define amount (as distinct from length), and has implications for design.

200

10.3.4 Typography

Following Rose et al.'s (2003) definition of the principles of "aphasia friendly formatting" (p.950), research into typography has sought to gain consensus on the optimum font size and styles to use in information for people with aphasia. Rose et al. (2012) explored the preferences of people for different font sizes and styles, whilst Brennan et al. (2005) investigated the impact of increased font size on comprehension of written material. Wilson and Read (2016) compared the impact on comprehension of serif versus sans serif fonts, and sentence case versus capitalised text. The emphasis of key words using bold font is routinely seen in patient information literature, and is accepted as standard practice (e.g. Stroke Association information leaflets, Cottrell & Davies, 2006). However, it is not included in Rose et al.'s (2003) definition, and has not been subject to empirical investigation with people with aphasia.

In this study, participants rated the accessibility of short phrases presented in four different font sizes: 10, 12, 18, and 26pt, of which the two largest sizes were equally and most positively rated. Although they generally found larger fonts to be more accessible, participants cautioned that font size should be in proportion to the materials. They commented on their changing needs over time, for example that a larger font size early post stroke was needed.

There were no marked differences in the ratings of serif versus sans serif font styles in this study. However, a limitation of the methods used was that a short phrase was presented in 12pt, a font size which participants rated less favourably than larger fonts. Had more text been presented, and in a more accessible font size, it is possible that a greater range in participants' ratings might have resulted.

Participants unanimously rated some form of emphasis of key words more highly than none, and the marked preference for emphasis that is both large and bold was a new finding for this study. Overall, participants repeatedly expressed the importance of clear information. On analysis, the breadth of meaning of this term emerged, including that materials should be simple (not complicated), clear (stated obviously) and uncluttered (without too much detail). Literal visual clarity, for example dark printing, also fell within this concept of clarity.

The findings presented here concur with previous studies in which larger font sizes were found to be preferred by people with aphasia (Rose et al., 2011a, 2012) and to impact positively on comprehension of materials (Brennan et al., 2005), but provide more detail on what it is about larger fonts that influenced participants' ratings. Firstly, people reported finding reading easier to attempt when they can see the text clearly. This is not only a visual preference; participants reported that they could read larger fonts more quickly and accurately, and that this allowed
them to read more text. Participants highlighted that this could be subject to change over time, remembering that in the early stage after stroke, larger fonts were needed more than at later stages.

The positive rating of small fonts by a man with severe aphasia is in keeping with the findings of Rose et al. (2003), who report that some of the most severely affected participants stated a preference for the information materials which were not adapted. It is possible that such participants are seeking familiarity rather than accessibility. As Rose et al. (2012) report, some people with aphasia acknowledge that whilst the largest font sizes may be easiest to read, they would not choose to look at information presented in this way. There may therefore be tension between individual preferences and formatting styles which support reading comprehension.

Two factors may influence why larger font sizes are preferred. Firstly, larger font sizes may result in increased engagement because people are not struggling to see, thus increasing motivation. Secondly, if page size remains stable, increased font size will reduce the number of words per line. Rayner and Pollatsek (1989) report that for readers without neurological impairment, "…reading appears to proceed at about the same rate if the type font, size and length of line employed are at all reasonable." (p.119). For readers with aphasia, a reduction in the number of words per line, together with a combination of large, clear font, is likely to benefit attention and comprehension.

In contrast to the equivocal responses to serif and sans serif font styles in the current study, Rose et al. (2012) found a preference for sans serif fonts. This has been recently supported by Wilson and Read (2016), who found a significant advantage to comprehension of the sans serif font Verdana over both Times New Roman and Harrington. It is interesting that the serif styles which were both least preferred and detrimental to comprehension in these two studies were of a light print quality, in comparison to the darker sans serif fonts. It may be, therefore, that serifs alone do not account for ease of processing of specific font styles, but that print darkness is also a factor.

The finding that emphasising key words increases accessibility concurs with those of previous studies such as Rose el al. (2011a), but adds detail about the nature of the emphasis used. The preference for larger and bolder emphasis may be a result of what Heuer and Hallowell (2007) describe as the "pop out effect" (p.884), in which the visual impact of darker font or larger font size causes readers to attend longer to selected words or segments of written materials. This could account for previous explanations of bold font interrupting the flow of reading, to the detriment of comprehension (Hoffman & Worrall, 2004). In fact, some interruption to the flow

of reading may be beneficial to readers with aphasia, to focus attention on words carrying the greatest meaning in a sentence. However, as recently found by Wilson and Read (2016), the use of upper case for emphasis is not suitable, since it is less readable than sentence case. This suggests that that judicious use of bold font is required, to ensure that readers' attention is directed usefully.

10.3.5 Images

Adding images to written text constitutes part of the "aphasia friendly" format, defined by Rose et al. (2003:950). Whilst early research into the effectiveness of such principles focused on whether images do indeed support people with aphasia's comprehension, subsequent studies explored whether different image styles may have a greater impact on reading comprehension. Alongside this, the subjective preferences of people with aphasia have been explored, with a view to obtaining a consensus on the optimum image style for supporting reading in people with aphasia.

In this study, participants commented on the images both in isolation, and in combination with written text on the template. Overall, the line drawings were rated more highly than the photographs, but there was no clear consensus for one style over the other. Although some participants expressed strong aesthetic preferences, participants' responses generally related more to the effectiveness of specific images to convey a particular meaning, rather than whether these were photographs or line drawings. Generally, participants expressed that simple, clear, and uncluttered representational images were easiest to process. Comments made about abstract designs, or images with extra detail, suggested that visual processing sometimes took more time and cognitive effort than was predicted by the design team, and this occurred regardless of aphasia severity.

The portrayal of concepts through images prompted varied responses, demonstrating variability in how participants used the images to support their comprehension of the concepts. The example of how to depict 'blood' showed that while for one person, the literal meaning of the concept was triggered by the image of a bleeding finger, for another, the same image was misleading, since the blood did not refer directly to the blood flow in the brain. Similarly, participants varied in their responses to images of people; one female participant appeared to relate all images of women to herself, resulting in idiosyncratic interpretations of the information. In contrast, a male participant indicated that he could not relate images of women to himself, since they literally did not look like him.

All template materials included both images and text, and whilst many comments focused on the images themselves, discussion also arose from the interaction between the two. All the participants, with the exception of one man with mild aphasia, agreed on the importance of an exact match between words and images. The cards which prompted such discussion were those in which concrete wording was accompanied by images which did not match exactly. Arguably, the images therefore required some degree of interpretation, and it was this which some participants found distracting, and others a barrier to comprehension. Severity of aphasia might be a consideration here; however, participants with all levels of aphasia commented on this issue. Interestingly, where a sentence carried more abstract information, participants did not seem to expect the illustration to match the text precisely. No participants in the current study questioned the presence of images in the template; rather, all seemed to expect images to be used.

The study of aphasia suggests that images are a valuable resource, allowing people with aphasia to utilise preserved visual ability to support a range of communication and language activities. For example, in facilitative approaches to communicating with people with aphasia, pictures have been found to be useful to support interaction. This includes a range of image styles including photographs (Ho et al., 2005), line drawings (Kagan et al., 1998), picture symbols (Murphy, 2004), and drawing (Sacchet, 1999). Knollman-Porter et al. (2015) found that people with chronic aphasia were more likely to select readily-available reading materials which included pictures. They report that looking at the pictures assisted people to decide whether or not the text was of interest or relevance, and supported text comprehension.

Despite this endorsement for the inclusion of images alongside text, studies attempting to clarify the role of images to support reading comprehension have had mixed findings. Brennan et al. (2005), Rose et al. (2011b), and Wilson & Read (2016) found that images did not in themselves support written word comprehension, regardless of image style. Two further studies of reading with pictures in people without aphasia by Thompson et al. (2010) and Liu et al. (2009) were similarly inconclusive. One study by Dietz et al. (2009) demonstrated that the presence of high-context photographs did influence comprehension. As discussed in Chapter 2, the variation in findings may have resulted from differing methods; whereas Brennan et al., Rose et al., and Wilson and Read sourced images to support performance on a cloze task, Dietz et al. developed written paragraphs after identifying suitable images.

A further caution regarding the use of images is indicated by studies exploring visual perception. Multiple levels of visual processing are involved in the correct recognition of images (Warren, 1993) and impairments at any level may disrupt picture recognition, leading to difficulties in understanding the meanings of images. Beaufils et al. (2014) and Liu et al. (2009) both report that older participants experienced greater difficulty understanding pictograms and illustrations than younger participants. This suggests that visual comprehension is a cognitive process which is susceptible to individual variation, including amongst those with aphasia. In this study, some participants struggled to make sense of some images, but whether this was due to a stroke-related visual processing impairment, or a consequence of age, is unclear. A measure of the visuospatial processing abilities of the participants would have provided a means of analysing this in more depth in this study.

Identifying the optimum image style for people with aphasia has been the aim of several studies. Dietz et al. (2009) argued that Brennan et al.'s (2005) use of line drawings accounted for their finding that images do not support written word comprehension. Dietz et al. found that high-context photographs were more successfully matched with written sentences than low-context photographs, but since they did not include a line drawing condition in the experiment, the suggested inferiority of line drawings was not demonstrated.

In an eye tracking study examining a sequence of single word to multiple choice picture matching tasks, Heuer (2016) found that college students without neurological damage processed colour photographs more easily than line drawings. She suggests that increased complexity of photographs versus line drawings may account for this finding, but cautions that people with aphasia may perform differently. Although the current study did not explore the impact of images on comprehension, the data demonstrate that the participants' perceptions of the accessibility of images rests on a variety of factors, of which image style is only one. For example, several participants commented that they would prefer to look at photographs, but stated that the image must be correct for the concept. As Heuer (2016) argues, the type of image selected is determined by the communicative purpose it serves. As Ho et al. (2005) and McKelvey et al. (2010) found, personal photographs were of greater use in supporting communication than pictograms, line drawings, or low-context (non-personalised) photographs. However, such findings are not necessarily transferrable to the communication of printed information, in which concepts are relevant to a population or group rather than individuals, and therefore require less personal specification.

The choice of images in health information materials is therefore of paramount importance. There is agreement in the literature that the target population should be involved in the production of new health information materials (e.g. Coulter et al., 1998; Duman, 2003), and the response of participants in this study and others (e.g. Elman et al., 2003) supports this. Examples of engagement by people with aphasia in developing materials (e.g. Parr et al., 1999, 2004; Cottrell & Davies, 2006) do not, however, provide a clear account of how consensus was achieved in the choice of images. This is concerning, given that in the current study, selection of the correct image was identified as a challenge by participants, and was frequently a source of disagreement. The finding that text and images must match exactly appears to be novel to this study, although Rose et al. (2011a) found a similar, but more general, response, reporting that "Participants also commented that it was helpful when graphics closely linked with the text..." (p.342), and that graphics should have captions. Griffin and Wright (2009) found that older subjects without aphasia read information more quickly when text and graphics were closely allied, but were distracted by decorative graphics which served no communicative purpose.

This discussion has suggested that people with aphasia make use of both text and images in written information. However, evidence for the simplification of language is stronger than the current evidence for use of images, which remains controversial. Whilst images are perceived by many people with aphasia as helpful, research evidence remains inconclusive. This suggests that further knowledge about how best to use images in health information is needed. In the following section, further consideration is given to the layout of written information incorporating both text and images.

10.3.6 Layout

Previous empirical studies have focussed attention on features of formatting, rather than overall layout of the content. Studies by Rose et al. (2011a; 2012) found that participants preferred clear layout and design, but the methods did not allow for an in-depth analysis. Rose et al. (2012) found that, when asked in the abstract, most participants reported white space to be helpful. When looking at information materials, wide line spacing was poorly rated, suggesting that extra white space is unhelpful within sections of text.

Two issues relating to how content is presented were discussed in the current study: overall layout of content, and the number of images per concept.

Overall, participants agreed that the template layout guided their reading, although individuals varied in how they approached the template, and how much of it they looked at consciously. In terms of number of images per concept, two distinct groups were identified. Whilst most participants preferred the presentation of a single image for the concept, a sub-group showed a consistent preference for an image of each key word in the sentence for the concept. Participants with a range of reading ability favoured each style of presentation, and no obvious link between aphasia severity and preference for number of images per concept was found.

Morrow et al. (2012) employed eye tracking technology to examine how older readers without aphasia look at health information containing both text and images. They observed differences in how participants with varying levels of knowledge about their health attended to the materials. Those with higher levels of knowledge attended first to the text, and then to the relevant image, whereas less knowledgeable participants moved between text and both types of images in a less ordered way. The authors postulate that the comprehension strategy of the more knowledgeable was to use the relevant images to reinforce their initial reading of the text. They conclude that designing health information materials which facilitate this strategy could be beneficial, showing readers where to look next and providing materials which prompt different processing styles. Whilst caution is needed transferring findings to a population with aphasia, the observed reading styles of Morrow et al.'s (2012) participants have similarities to those in the current study.

As well as visual and linguistic processing, reading text and images together uses additional cognitive resources, such as allocation of attention and working memory, skills necessary for executive function (Alvarez & Emory, 2006), which is required when drawing inferences between, say, words and images. Since research has demonstrated that attention, working memory, and executive functions have all been found to be vulnerable in people with aphasia (Frankel et al., 2007; Kalbe et al., 2005; Murray, 2012), it is possible that presenting text and images in combination may actually add complexity to the task of reading for some people with aphasia.

In an eye-tracking study comparing participants with and without aphasia, Heuer and Hallowell (2015) found that participants with aphasia had more difficulty allocating attention, and this was exacerbated by increasing complexity in the stimuli. Further, there was no correlation between aphasia severity and ability to allocate attention, although as the authors note, their sample was skewed, with most participants having mild, and only two having severe, aphasia. Thus, although images are intended to support lexical access, they may add complexity to reading by increasing the cognitive load. This issue has been investigated by Hurtado, Jones, and Burniston (2014), who questioned the established 'Easy Read' format for people with intellectual disability. They propose that a "picture only" (p.824) leaflet, when presented with communication support, may enhance comprehension by minimising the cognitive load required to process both text and images.

Wilson and Wolf (2009) argue that design of printed information materials should take account of users' finite working memory capacity. Citing cognitive load theory (Sweller, 1994), they argue that reducing the burden on "extrinsic load" (p.319) (the form of information) releases cognitive resources to process the "intrinsic load" (p.319) (the content). Findings from the current study support this approach, since some participants had difficulty processing the images, which distracted their attention from the content. For example, participants sometimes persisted, despite not being able to draw a clear meaning from the images; or disengaged, either because they could not make any sense of the images, or because they took

207

too long to process. The concept of intrinsic load may also be useful in exploring issues such as how to convey abstract or unfamiliar concepts, and illustrating sentences versus single words in health information materials. How to achieve "germane cognitive load" (Wilson & Wolf, 2009:319), that is, the optimum presentation of text and images to promote learning, is also pertinent to this discussion.

The benefits to cognitive load of developing text content following the principles of structural and repetition priming, as in this study, are worth considering, since researchers into structural priming (e.g. Hartsuiker & Kolk, 1998; Pickering & Ferreira, 2008) argue that priming enhances language processing by reducing cognitive load.

In relation to people with aphasia, further research is needed to understand how people with acquired dyslexia read when images are present, and how the different modalities are integrated for comprehension of the information. It may be that people with aphasia, as demonstrated by Morrow et al. (2012), vary in how they process information when images and text are presented together. While most of the participants in this study seemed to refer to the image as a support for the overall concept, a smaller group preferred to see an image to support their reading of individual words, a finding which may suggest a desire to understand each word in the sentence.

Differences in reading behaviour as well as ability also need to be accounted for. For example, the model of reading proposed by Rayner and Pollatsek (1989) includes behaviours (saccades and eye fixation), processes (word processing), memory (working memory and long term memory), and world knowledge. The impairments to word processing in people with acquired dyslexia and aphasia are known through psycholinguistic assessments. Readers may, however, also display varied visual reading behaviours, cognitive function, and world (or prior) knowledge. Greater understanding of all these would provide a fuller account of individuals' approach to information, and could potentially inform design of materials to maximise comprehension.

10.3.7 Acceptability

In the research literature, the issue of acceptability has been explored in two different ways. Firstly, researchers have sought opinions on whether information modified using 'aphasiafriendly' principles is acceptable to people with aphasia (e.g. Rose et al., 2003; Rose et al., 2011a). Secondly, preferences regarding the acceptability of specific aspects of the content and design of the information have been gathered (e.g. Rose et al, 2011b; Rose et al., 2012). The reported responses of study participants between 2003 and 2012 indicate an increasing acceptance of modified information over time, suggesting that generally, people with aphasia welcome attempts to formulate accessible health information. The change in responses to adapted formats over time may indicate that people with aphasia have been increasingly exposed to aphasia-friendly information. Given that one research unit has a specialist focus on accessible information, this may be particularly relevant. Individual variation proliferates in the area of participant preferences, but appears to be frequently linked to severity of aphasia, with more severe aphasia associated with a lower acceptance.

In the present study, participants identified a need for modifications to design which they hoped would help them access relevant information. Focussing on the template materials, they discussed the importance of information which is simple in design and language but adult in style. Some sensitivity to images which were either too graphic, or were insensitive to aphasic difficulties, was highlighted during discussions. In the same way, participants responded positively to materials which they felt conveyed a sense of hope, both for themselves and for those new to stroke and aphasia.

Participants did not elaborate on what might constitute a childish style of information, since the issue was not raised as a criticism of the materials per se, but as a challenge to design. In previous studies, researchers have sought views based on their own assumptions of this concept. For example, Rose et al. (2102) surveyed participants' responses to information materials which included multiple images. Forty participants were asked to comment on information materials on the topics of stroke and aphasia via a structured questionnaire. These were designed for people with aphasia and included pictures, although no further description of the materials is provided. Participant responses to the open ended question "How do you feel receiving information that has several pictures?" (p.15) were largely positive (as categorised by the authors), suggesting that information which includes pictures does not in itself prompt negative feelings. Interestingly, however, when provided with a closed question format, 40% of the participants rated the materials as "childish" and 35% as "embarrassing" (p.18). It would be interesting to know whether the authors selected such words to force polarised responses. Here, the authors appear to assume that it is the provision of several images which may confer a childish look on information materials. Other aspects of the design, such as the presence of images, or questions of single or multiple image style, are not addressed.

Hope is recognised as an important factor in recovery for people with aphasia after a stroke (e.g. Bright et al., 2013), but no research has been found which specifically investigated the provision of information which provides hope. As Bright et al. (2013) found, "the experience of hope was fluid and changed in response to how people perceived their current situation and future" (p.48). As a highly individual phenomenon, it would be difficult to be confident in

providing hope per se. However, aspects of the literature about stroke and hope may serve to inform the content and tone of written information. Coulter et al. (1998) asked five people who had had a stroke (none with aphasia) to review patient information materials in a focus group. They found that whereas these people preferred information which conveyed a hopeful tone, stroke specialist clinicians and academics were more concerned with whether the information was accurate and evidence-based.

Anderson and Marlett (2004) also discuss the tensions between professionals wishing to provide accurate information about medical conditions, including possible negative prognoses, and the patient and family's need for hope to sustain them through a challenging experience. They advocate a problem-solving approach in contrast to "problem-centred communication" (p.60), acknowledging that "It is a challenge to communicate positively about uncertain recovery, disabilities and the real problems stroke creates." (p.60)

Bright et al. (2013) suggest that a clinical implication of their study is the provision of information "to reduce uncertainty" (p.54), presumably to inform people of services and pathways of care, or next steps in service provision. And whilst the authors also suggest that service providers have a role in developing hope with clients and families through interactive approaches, they do not explore the possible role of information materials in fostering hope.

Adverse reactions to the images of blood support the findings of Coulter et al. (1998), who reported that "gory" (p. 57) images were not welcomed. As reported by Houts et al.(2006), people did not appreciate cartoons which dealt with a serious topic in an inappropriately comic tone. Rose et al. (2012) report strong adverse reactions to some pictographic symbols shown to participants in the course of their study, but did not analyse this further.

The tone of information is carried in language and images, and as Elman et al. (2003) found, sensitivity to this may be heightened for people with aphasia. In a participatory action research project investigating access to the internet, twelve people with aphasia reviewed websites in focus groups over the period of one year. The authors report that some participants reacted strongly to some images, reporting that "The group has been very sensitive to the images of disability presented by websites, and even fleeting, seemingly innocuous images have provoked strong reactions" (p.113).

As Worrall et al. (2007) note, accessibility is an "emotive issue for people with aphasia" (p.135) since people have experienced the traumatic event of stroke and its life-changing consequences (e.g. Edelman & Greenwood, 1992; Parr et al., 1997). This finding suggests that the design of information aimed at people with aphasia requires knowledge of aphasia and sensitivity to the experience. It also lends support for the recommendation that potential

consumers of the information should be involved in developing materials to ensure their suitability and acceptability (Coulter et al., 1998, Duman, 2003). As discussed by Cruice (2007), such an approach is needed to ensure that people with aphasia are properly included in the design of products intended for their use.

10.3.8 Summary

The discussion has demonstrated the interacting issues surrounding the design of accessible information for people with aphasia, including content, amount per page, facilitative language, and formatting features such as typography, images, and layout. The guidelines developed in Phase 3 of the study incorporated all these issues, and propose practical steps towards fulfilling the design criteria of the Template developed in Phases 1 and 2.

Alongside this, it is important to distinguish between design and provision of health information.

Designing acceptable information materials for people with aphasia requires attention to the topics of information required, to visual, cognitive, conceptual, and language processing demands, and understanding of the experience of aphasia.

The provision of health information to individuals demands careful attention to each person's processing abilities, style, and preferences, tailored to their individual current needs.

10.4 Individual differences

Research involving human experience or perceptions must recognise individuality. Previous research in the area of accessible formatting has sought and recognised individual variation in responses to both the provision of information in modified formats (Rose et al., 2003; Rose et al., 2011a) and to specific aspects of formatting, such as image style, typography, or the length of information leaflets (Rose et al., 2011b; Rose et al., 2012). However, the use of surveys, rankings, and structured interviews in these studies has resulted in a tendency to report findings which reflect the majority view. Although this is consistent with the aim of identifying consensus for factors which facilitate access to information, there is a lack of accounting for the full range of individual responses. For example, studies have reported negative or unexpected reponses to modified formatting, but did not include methods to explore the reasons for this (Ghidella et al., 2005; Rose et al., 2003). In their later study, Rose et al. (2012) comment that questionnaire methods limit the interpretation of responses.

In the current study, the data revealed individual variability in opinions about formatting and approaches to information, reflected in Theme 4 'Everyone's different'. Three key issues were

identified within this theme: firstly, the differences between participants in terms of their stroke and experience of aphasia; secondly, different ways in which individuals approached the template materials; and finally, the need for support in accessing information for some individuals.

10.4.1 Individuals with aphasia

Although it seems obvious to state that all participants brought their unique experience and character to the discussions, this is a key element of qualitative research, which aims to understand an issue from the perspective of those affected by it (Denzin & Lincoln, 1998; Simmons-Mackie & Lynch, 2013).

The participants in the Phase 2 study had a variety of types of aphasia, and their presentation ranged from mild to severe. They also differed in their pre-stroke lives and their approaches to living with aphasia. All the participants had had their stroke at least one year prior to the study, but this ranged from eighteen months to thirteen years. People varied in their physical recovery and the degree to which issues such as cognitive impairment affected their lives. Some lived with partners; others lived alone. Participants varied in how talkative they were. Whilst many of these differences resulted from the stroke and aphasia, some were integral to the individuals and their personal situations. It was striking that whilst they drew on their own experiences, participants also demonstrated a wider knowledge about what aphasia might mean for others. They agreed on key issues of accuracy about aphasia recovery, balancing the need for realistic and 'hopeful' information. During the focus group discussions, participants referred to the template content to reflect on their experiences of aphasia. Many issues arose from such discussions, including variations in aphasia, cognitive changes, distress, depression, and acceptance.

Previous research in this area (e.g. Rose et al., 2011a; 2012) has reported on factors of interest to the research question, such as time since onset, aphasia severity and specific language testing scores. The range of variation is highlighted, and some degree of individuality is recognised. However, these aspects of the findings are reported conceptually as relating only to fixed individual preferences. For example, Rose et al. (2011a) used Content Analysis to code interview data using researcher-defined categories, and Rose et al. (2012) used a questionnaire to gather views on aspects of formatting such as typology. Such methods reflect the stated aims of the research in attempting to identify the preferences of people with aphasia, but a possible limitation is that quantitative methods do not provide a means of interpreting the issue from the perspective of the person with aphasia. In addition, when results are analysed only in relation to the severity of participants' aphasia, there is a risk of over stating the role of aphasia in how people respond. Consequently, and unintentionally, other aspects of individuals are under-represented in the analysis. One example of this is that participants with severe aphasia have been reported in the literature to give adverse responses to modified information. The authors explain this only in terms of comprehension (Rose et al., 2003, 2011a), noting that those who disliked adapted information tended to be people with severe aphasia, but that they could not say why. In the current study a man with severe global aphasia also disliked many of the images presented. He also expressed the importance to him of the visual appeal of images. It may be that such visual appeal becomes more significant to people for whom written language is inaccessible. An alternative explanation is that the visual aesthetic is more important for some people than others, regardless of aphasia type or severity. In this study, a man with mild aphasia also spontaneously expressed the importance of aesthetically pleasing images. This insight arose from the discussion, since participants were not asked directly about the importance of the aesthetic per se. This example demonstrates how appropriately facilitated individual and group discussion provided rich data which supported a thorough understanding of participants' responses. And whilst aphasia is common to the participants, it is not their sole defining characteristic, nor does it inform all their preferences.

10.4.2 Individual approaches to information content

In previous studies seeking the response of people with aphasia to health information, participants were asked to look at information materials with a view to reading (Rose et al., 2003), reviewing accessibility (Ghidella et al., 2005) or commenting on the content and format (Rose et al., 2011a). No observations are reported of how participants approached these activities.

A novel finding of this study concerns individual variation in approaches to information. In the course of the discussions, participants revealed differences in how they looked at the template. Most people were guided initially by the header sentence, and then looked at the sentence and images in turn. Some participants limited the number of elements they approached, focusing attention at the top of the page only. In contrast, others seemed to attempt to look at the template as a whole unit. The ensuing discussion suggested that such differences reflect whether participants persist with their pre-stroke visual style, or modify their approach to accommodate their aphasia. For example, whether a person tends read quickly, despite not being able to process the full meaning, or whether people have learnt through having aphasia to approach information more slowly and methodically. Such differences did not appear to have any relation to severity or type of aphasia, but seemed more idiosyncratic.

This suggests that factors, such as cognition and personal reading style, may play a larger role in how people interact with information, an idea supported by the wider literature in aphasia. For example, research into successful text level reading highlights the role of cognitive functions such as working memory and drawing inferences (Chesneau & Ska, 2015; Meteyard et al, 2015), and executive function for self-pacing to maximise comprehension (Caplan et al., 2007). As Caplan et al. note, using such cognitive resources may reduce the capacity available to an individual for other tasks, such as processing language or visual content.

The finding that participants varied in their approach to the template is consistent with the findings of Morrow et al. (2012), who identified variations in how people with and without aphasia look at stimuli containing words and images. Whether this is linked with specific cognitive profiles, or reflects a more general individual style, requires consideration. Further research is also indicated to determine whether readers' idiosyncratic approaches to modified formats have an impact on their comprehension of the content.

10.4.3 Support with accessing information content

Research into modified formatting has focused on individuals' responses to information via testing (e.g. Brennan et al., 2005; Wilson & Read, 2016), surveys (e.g. Rose et al, 2012), ranking materials (e.g. Ghidella et al., 2005), or commenting during individual interviews (Rose et al, 2011a).

Whilst the current study also aimed to explore individuals' views on their independent access to information, some participants identified the need for support in looking at information. This was highlighted as important for people early after a stroke and for some people with chronic aphasia. Participants commented that the template design could help family members or others talk through and explain information at a pace to suit the individual.

The comments made by participants about supported access to information echo what has been reported previously (e.g. Parr et al., 1997), that people with aphasia have different information needs at different times after a stroke.

For people with severe aphasia, or with concomitant cognitive impairments, independent access to information may be an unrealistic target. For such individuals, a more helpful approach may be to provide information in a format which enables their communication partners (family members, service providers) to support access. Knollman-Porter et al. (2015) discuss the concept of "partner assistance" (p.1461). This is described as reading aloud, but the authors propose that training for reading support could be provided in the same way as conversation partner training (Kagan, 1998; McVicker, Parr, Pound, & Duchan, 2009). The template design described here could be used to support such a paired approach to reading, since single concepts are introduced and developed in a step-by-step progression.

10.5 Methods for involving people with aphasia in developing accessible information

As previously discussed in Section 4.1, people with aphasia have been involved in research and development of accessible information in different ways. Developmental work at aphasia centres such as Connect, UK has involved people with aphasia as partners in an approach based on a social model of disability, with the aim of developing information materials (e.g. Parr et al, 1999). Whilst such projects have resulted in widely-used resources, little detail has been provided by authors to display the evidence for the principles used or the methods followed.

Responding to the need to develop a more robust evidence base for accessible information and aphasia, subsequent empirical studies employed a positivist approach, involving people with aphasia as research participants. Such studies aimed to clarify issues, such as the impact of typography on comprehension of text (e.g. Brennan et al., 2005; Wilson & Read, 2016). Participants in these studies were individually tested, surveyed, or interviewed, but limited detail has been reported on the methods used to support their participation, and no studies have reported the impact of the findings on information resources developed in light of the research.

In this section, issues relating to the methodology adopted in this study for developing the evidence into accessible information with people with aphasia are discussed, with reference to the participants, the materials, the data collection methods, and the data analysis methods used in Phase 2 of this study.

10.5.1 Participants

Previous studies of modified formatting have tended to involve participants with mild or moderate aphasia (e.g. Rose et al, 2003; Wilson & Read, 2016), with a paucity of contribution to this field from those with more severe language impairments. Only McKelvey et al. (2010) sought participants with severe aphasia, and this sampling strategy can be attributed to the study aim of identifying optimum images for use in AAC.

The three studies reported by Rose et al. (2011a, 2011b, and 2012) were conducted at the same research centre, and involved the same participants. This can be said to reduce the numbers of people with aphasia who have contributed to the evidence. In addition, it can be

argued that these participants may not be representative of the population with aphasia, since they have been exposed to modified formatting through the studies.

In the current study, purposive sampling (Patton, 2002) was undertaken to ensure that people with mild, moderate, and severe aphasia were included, in order to capture the views of a wider segment of the population with aphasia than has previously been achieved. In common with other studies (Brennan et al., 2005; Dietz et al., 2009; Ghidella et al., 2005; Jayes & Palmer, 2014; Rose et al., 2003; Rose et al., 2011b), all Phase 2 participants had chronic aphasia and commented on their current and earlier information needs with the benefit of experience.

Participants' language was tested in some detail, as described in Chapter 6. On reflection, two aspects of this process would benefit from modification. Firstly, the CAT (Swinburn et al., 2005) reading subtests did not always reflect the participants' reported level of difficulty, suggesting that these tests lacked sensitivity. For example, participants reported greater difficulty with everyday reading activities than was predicted by the test scores. In addition, as noted by Dietz et al. (2009), broader testing across cognition would have provided useful background information, against which to understand individual responses and analyse the data. For example, participants reported difficulties with remembering information, despite no overt impairments to their cognition. Subtle changes to language and cognitive abilities appeared to have an impact on individuals, and therefore require a fuller account.

The current study has contributed the views of people with a broader range of types and severities of aphasia to the evidence for modified formatting. Consistent with the aims of qualitative methods, it is the inclusion of "information-rich cases" (Patton, 2002:230) which is important to the collection of rich and varied discussion data, rather than the numbers of participants included.

10.5.2 Materials

The materials presented to participants in previous research has varied widely, according to study aims and objectives. Published information resources (e.g. Parr et al., 1999, 2004; Cottrell & Davies, 2006) represent the output of collaborative work with people with aphasia. The purpose and practice of such collaborations, which may be described as action research (Patton, 2002), means that traditional research accounts of the development process have not been published.

Previous empirical research (Rose et al, 2003; Rose et al., 2011b; 2012) and one qualitative study (Rose et al., 2011a) sought the opinions of people with aphasia regarding modified

formatting, by showing materials produced according to 'aphasia-friendly' principles. In one study, however, Rose et al. (2011b) asked participants to state their preferences for image styles in health information without reference to any visual materials, suggesting that participants were able to understand the issue and visualise independently to consider their opinions. Topics included in study materials have varied widely, including relevant and nonrelevant health information (e.g. Rose et al., 2003; Ghidella et al., 2005), general interest topics (Brennan et al., 2005; Dietz et al., 2009; Wilson & Read, 2016), and topics of personal relevance (McKelvey et al., 2010).

A novel aspect of this study is the systematic iterative design process described herein. In Phase 1, the prototype materials were developed from the findings of reviews in the literature concerning modified formatting, factors facilitating language processing in aphasia, and the health information topics required by people with aphasia. The materials were then refined through the collaboration with people with aphasia in the iterative design process, described in Phase 2. This approach has resulted in the design of a novel template for health information for people with aphasia, incorporating the principles of accessible language and formatting discussed in Section 10.3.

The production of this template is consistent with health literacy literature, which states that people affected by a health condition should be involved with producing information (e.g. Coulter et al., 1998), and with concepts of inclusion and user-centred design (Cruice, 2007; Wilson et al., 2015).

In realising the final template through the prototype design and iterative design process (Sears & Lund, 1997), the study achieved its overall aim of developing an evidence-based template for conveying health information for people with aphasia. Previous collaborations (e.g. Parr et al., 1999, 2004) have developed new information materials for people with aphasia and previous empirical studies have examined the impact of modified formatting on people with aphasia. This thesis describes a novel study which has combined a collaborative, user-centred approach, embedded within a rigorously designed and reported study.

10.5.3 Focus group methods

Previous studies into accessible formatting have conducted individual testing, surveys and interviews with people with aphasia (e.g. Brennan et al., 2005; Ghidella et al., 2005; Rose et al., 2011a). Focus groups have not previously been used to explore the impact of modified formatting on participants with aphasia and in this regard, this study attempted a novel method. Central to the decision to conduct focus groups for the data collection was the consideration of methods to support the communication of the participants with aphasia, to ensure the comfort of all participants (World Medical Association, 2001) and the quality of the data collection (Finch & Lewis, 2003).

The literature reporting the involvement of people with aphasia in focus groups suggests that researchers utilise a range of strategies, including their clinical skills in interacting with participants (e.g. Kerr et al., 2010), the provision of visual materials to support generic language processing (e.g. Dalemans et al., 2009), and, in the case of Murphy (2006), a group Talking Mat[®].

In the current study, principles for supporting participants' communication were derived from the literature (e.g. Kagan & Gailey, 1993; Lawson & Fawcus, 1999; Lyon et al., 1997; Luck & Rose, 2007) and were followed in the focus groups. Providing individual communication support to all participants, as well as a group leader (Finch & Lewis, 2003; Garcia et al., 2000) aimed to foster a sense of equality in the focus groups, but it was evident that the people with the most severe aphasia required more individual facilitation than those with milder language impairments. Despite the level of communication support, and participants' willingness to respond to the visual materials, there were times when researchers questioned whether some participants were fully able to understand the issues. There were also occasions when participants were unable, despite facilitation, to articulate or develop their opinions.

The use of individual Talking Mats[®] in a group of people with aphasia was a novel approach in this study. Previous research using Talking Mats[®] has been conducted on an individual basis (e.g. Murphy, 2000; Murphy et al., 2005; Murphy, 2006; Murphy et al., 2010). More recently, the use of individual Talking Mats[®] within a group context have been reported to support communicative effectiveness of people with Huntingdon's Disease (Hallberg, Mellgren, Hartelius, & Ferm, 2013).

The data generated during the focus groups indicate that the Talking Mats[®] framework was an effective means of providing both communication support and a tool for individual data collection. The subsequent group discussions were also richer since participants had time and space to develop their own views, with the mat still in place during the discussion as a point of reference. The Communication facilitators were key to this process, functioning both as a supporter for the Talking Mats[®] process and as a conversation partner. Throughout Phase 2, all participants contributed, remained engaged, and attended both focus group meetings, suggesting that they found the groups a supportive environment in which to exchange views. The transcriptions record a number of occasions when group members overtly support each

other to express their views, despite evident word finding difficulties, and the amount of laughter recorded testifies to the good humour shared in the groups. It is possible that hearing others' views encouraged participants to be more critical of the materials than they might have been in an individual interview, as suggested by Finch and Lewis (2003) and Kitzinger (2006). The combination of individual supported Talking Mats[®] conversations and facilitated group discussions in the focus groups allowed for different ways of interacting and expressing views.

In the field of co-design, whereby end users of products or services are involved in the design process, recent research has used focus group methods to involve people with aphasia in designing novel approaches to language therapy (Galliers et al., 2012; Wilson et al., 2015). This suggests that there is a growing confidence in this methodology for including people with aphasia in design projects. Suitable methods for providing linguistic and communication support need to be established, however, as discussed by Galliers et al. (2012), to ensure the credibility of the findings.

10.5.4 Data analysis methods

Decisions regarding data analysis methods follow from the aims and design of research (Bryman, 2008). Previous empirical research addressing the response of people with aphasia to modified formatting (e.g. Ghidella et al., 2005; Rose et al., 2003) or aspects of formatting (e.g. Brennan et al., 2005; Rose et al., 2012) have generated quantitative data. In keeping with the hypothesis-testing approach, the data have been analysed deductively, in response to the research questions. The sole qualitative study by Rose et al. (2011a) generated interview data. The data were analysed using Content Analysis, which records the incidence of participant responses according to the researchers' predetermined coding criteria. On a deductive inductive continuum, Content Analysis is considered to represent a quantitative approach to the analysis of data derived through qualitative research methods (Silverman, 2006).

The analysis of the Phase 2 focus group discussions was conducted using Framework analysis (Richie, Spencer, & O'Connor, 2003), an approach which allows for the use of a deductive analysis, through the use of an existing framework, similar to Content analysis. In the current study, however, the researcher took a more inductive approach, seeking to capture new perspectives on the topic of accessible formatting, and the full range of individual views. The collection of both qualitative discussion data and rating scales data provided a means of triangulating different forms of data, both in the course of data collection and in the subsequent data analysis (Denzin & Lincoln, 1998).

The language processing impairments inherent to aphasia present a challenge to collecting qualitative data with this group, particularly with people who have more severe impairments. It may be partly for this reason that previous studies have used more quantitative methods, which allow participants to select from choices, rather than generating their own responses (e.g. Luck & Rose, 2007). However, as identified by previous researchers (e.g. Rose et al., 2003; Rose et al, 2012), study findings were limited by the lack of insights gained into the reasons for participants' responses. And as discussed in Section 10.5.1, a deductive approach to data analysis may potentially limit the development of new ideas of interpretations of the data.

Inductive methods are often selected to investigate areas of knowledge in which there is limited evidence, to build theories (Bryman, 2008). The use of deductive methods to test theories implies therefore that there is sufficient theoretical evidence in place to adopt empirical methods. The limited evidence base into accessible formatting identified in Chapter 2, however, suggests that this is an area in which further theoretical evidence is still required.

The analysis of the data in the current study has resulted in new findings, including the optimum amount of information per page, the optimum emphasis of key words, and the need for an exact match between text and images. New connections have been made between people with aphasia and aspects of formatting, such as individual approaches to looking at information and the importance of the personal aesthetic. Issues which require further clarification, such as the presentation of images and text together have been identified. Ultimately, the methods, including the approach to data analysis, resulted in the design of the novel template described in this thesis.

The implications for further research are outlined in Section 10.8.2. With a view to developing the theoretical framework for accessible formatting, further focus groups could be conducted and their data analysed using the framework devised in this study in a more deductive method (Pope, Ziebland, & Mays, 2006). Confirmation and building on this framework would then suggest further routes for empirical research, to test the theory (Bryman, 2008).

10.6 Quality, rigour and transferability of findings

The quality of research is judged by the credibility and transferability of the findings (Mays & Pope, 2006). This study has employed techniques from both qualitative and quantitative paradigms, in the pursuit of a well-crafted investigation (Sandelowski, 1995; Symonds & Gorard, 2010). Rigour has been addressed throughout the thesis, with attention given to providing a clear account of the methods of data collection and analysis, to enable readers to make their own judgement of the quality of the research (Seale, 1999; Spencer, Ritchie, Lewis,

& Dillon, 2003). In this section, two key issues, the validation and the transferability of the findings, will be considered.

10.6.1 Validation of findings

Several methods were used to ensure the credibility of the findings. Triangulation of discussion data and Talking Mats[®] ratings provided a means of checking different perspectives, providing a richer account than that afforded by a single perspective (Denzin & Lincoln, 1998). The subsequent presentation of the findings within a single thematic framework demonstrates this clearly, describing and accounting for similarities and differences in the two types of data.

Analysis by one researcher can be criticised as subjective. To validate the interpretation throughout the analysis, data were presented and discussed with supervisory colleagues, who were invited to challenge the researcher's interpretations of the data (Miles & Huberman, 2002).

Validation by participants, to ensure that their views were accurately reported, was incorporated into the study design, due to their involvement throughout the iterative process, in which issues could be discussed at the time they were raised (Seale, 1999). Following this, participants were not subsequently invited to review the results of the data analysis as a separate enterprise, since it was reasoned that such an approach could be problematic for people with aphasia, potentially challenging both people's memory of their contribution, and the linguistic abilities required to confirm or challenge the researcher's interpretations of the data. This position concurs with the caution against respondent validation proposed by Hammersley and Atkinson (1995), who highlight the potential difficulties for participants when asked to remember the details of their contributions to the research.

10.6.2 Transferability of findings

The topic addressed in this thesis is of significance to both people with aphasia and those who work with this population, therefore implicit in this discussion is the aim of what Damico et al. (1999) term "clinical utility" (p.660); that is, the transferability of the findings to wider contexts.

Addressing the issue of transferability of findings, Lewis and Richie (2003) identify two key issues for consideration. Firstly, the quality of the data collection and analysis, and secondly, whether the sample can be said to represent the target population.

This thesis has aimed to provide a transparent account of the research undertaken, including its theoretical context. Chapters 2 and 4 provide the theoretical context, with explanations of how the various literatures have informed both the objectives and design of the study. A

detailed description of the initial design and subsequent iterative development of the template materials has been documented, with the aim of enabling readers to both understand the process undertaken, and to gain a clear picture of the template materials reviewed by the participants. The procedures for collecting and analysing the data, and additional measures to ensure rigour in the conduct of the study, have been recorded transparently in the body of the thesis. Throughout, any difficulties or errors have been documented.

Whether the participants in this study can, or should be said to represent the aphasic population as a whole is a challenging question. No group of fourteen can be said to represent a whole population. In this context, purposive sampling (Patton, 2002) was conducted to ensure opinions could be collected from both women and men, from people of differing ages, and from people affected mildly, moderately, or severely by aphasia. This last was considered of primary importance, since previous research findings have largely represented those with mild and moderate aphasia. This raises a question of whether those most severely affected can genuinely contribute to the findings, due to potential difficulties understanding the aims and tasks, or expressing their thoughts. In this study, methods to maximise the participation of all were developed from the existing literature (e.g. Kagan, 1998; Murphy, 2000) and operationalised in the data collection methods. Participants with severe aphasia did contribute, but their responses showed a high level of individual variation, suggesting that aphasia severity is only one factor in their response to information materials. The analysis demonstrated that factors other than aphasia itself, such as cognition, confidence, and individual style all contributed to participants' views.

The methods used highlighted individuality in an attempt to understand the issues from the perspective of people with aphasia, consistent with the aims of qualitative research (Damico et al., 1999; Patton, 2002; Spencer et al., 2003). There was also an attempt to capture the commonality of responses, to derive implications for practice.

Future research could stratify the sample in greater detail, for example through seeking further detail on aphasia subtypes, cognition and vision, ethnicity and culture, and health literacy behaviours. Such segments would then provide greater opportunity for analysis and would potentially reveal further issues for the transfer of findings.

10.6.3 Reflection on strengths and limitations

One aspect of quality is the demonstrated reflexivity of the researcher (Seale, 1999). Throughout, the reporting of research methods and findings has aimed to provide a transparent account of the study, and reflections on the methods and findings have been included throughout this thesis. Final reflections, based on the strengths and limitations of the whole study, are outlined in this section.

The strengths of this study lie in the methods used in both phases to generate valid and relevant findings. In Phase 1, the prototype template was developed following scrutiny of the evidence base in three areas. The application of research evidence concerning the facilitation of language processing to the principles for the template design provided greater specification than has previously been achieved. The inclusion of principles derived from evidence regarding priming and language processing was, in addition, a novel aspect of this study.

In Phase 2, the collaboration with people with severe, as well as mild and moderate, aphasia means that the perspectives of people with a wider range of aphasia types and severities have been added to the overall evidence base on accessible information. This was achieved in this study via careful consideration of facilitative and inclusive methods. Secondly, and alongside this, the materials developed in Phase 1, produced to a professional standard in collaboration with professional graphic designers, promoted participants' genuine engagement with them as relevant information. The clarity of description of the template aimed to enable readers to envisage the materials and evaluate the data analysis in relation to them more transparently than has been achieved in previous studies.

The methods selected for collecting and analysing the data, combining rating scales and facilitated group discussion, resulted in a clearer understanding of the perceptions and issues for participants. The use of the Talking Mats[®] framework by individuals in a group context, supported by communication facilitators, was a novel approach taken in this study, and the findings generated revealed insights not previously gathered in this topic. The study has therefore extended the evidence base, and generated implications for practice and further research.

There are, however, limitations to the study which require acknowledgement. The study was conducted as part of a larger commissioned project within a limited timescale. With more time, a user group could have been established to collaborate with the initial development of the project, for example, to identify the topics for the materials, and to develop the focus groups formats.

In line with previous studies, participants were profiled based on their linguistic processing abilities. Whilst this was conducted thoroughly, the data analysis would have benefitted from a greater understanding of participants' visual and cognitive processing abilities.

223

Finally, part-time study has lengthened the process of analysis and reporting, resulting in a time-lag between the data collection and the presentation of the thesis. Whilst not an inherent weakness of the study itself, this has resulted in a delay in disseminating the findings more widely, perpetuating a lack of published evidence in the area of accessible formatting for people with aphasia. In the final section, the implications for practice and for further research will be outlined.

10.7 Implications for practice and future research

People with aphasia both need and want to receive information after a stroke, but research demonstrates that they continue to feel under-informed (Hinckley et al., 2013; Rose et al., 2010). Whilst significant progress has been made in the area of interpersonal communication access (e.g. Simmons-Mackie, Raymer, Armstrong, Holland, & Cherney, 2010), the evidence base for how to format accessible information remains limited. Alongside this, knowledge about the use of facilitative language for people with aphasia has been inconsistently applied in the area of health information. This may in part be due to the inherent difficulty of concepts to be communicated, or due to a lack of knowledge about how to use language to support readers with aphasia. These gaps in evidence, knowledge, and skill have resulted in a tension between what healthcare professionals should provide and what people with aphasia receive in practice.

This thesis has framed the issue of accessible information for people with aphasia within two broad concepts: health literacy and accessibility. A wealth of research relating to the improved outcomes for informed patients underpins the concept of health literacy (e.g. Kickbusch et al., 2013). In this context, the development of information which can be understood by service users with aphasia can be regarded as a health literacy intervention (U.S. Department of Health and Human Services, Office of Disease Prevention and Health Promotion, 2010).

Concurrently, the global ambition to increase access to all, articulated in the aims of the United Nations (2008) and the World Health Organisation (2011), includes the concepts of communication and information access. This ambition is now embedded within systematic processes to improve communication and information access to any person with a communication disability, both as a citizen (Equality Act, 2010), and as a user of health and social care services (NHS England, 2015). This thesis contributes to the evidence in both health literacy and information access, with implications for clinical and research practice.

10.7.1 Implications for practice

In health literacy terms, the template is an intervention, which serves a dual purpose. Its first function is to guide the design of information for people with aphasia. As such, it incorporates the key principles of language and formatting accessible information for people with aphasia. The second function of the template is to provide a physical means for conveying health information.

The guiding function of the template has reaching implications for improving the provision of information to people with aphasia, since it involves transferring the findings beyond the current context (Lewis & Richie, 2003). Since the design characteristics of the template serve to illustrate the language and formatting requirements of those affected by aphasia, it has the potential to increase service providers' awareness of aphasia and the needs of this population. The template could serve as a point of reference for adhering to guidelines and policies (e.g. NHS England, 2015), functioning as a means of ensuring good practice. The translation of the template principles into practical guidelines further supports good practice, by guiding the design of new information materials.

As a physical resource, the template has a range of flexible applications, supporting the three levels of health literacy (Nutbeam, 2000). Firstly, by conveying information in a maximallyaccessible format, it contributes to people's understanding of basic health information. Secondly, it functions as a tool to support communication about health, with the template providing a shared point of reference in, say, an interaction between a person with aphasia and a health professional. Thirdly, to support critical health literacy, template materials could be used by people with aphasia as a resource to explain their health to others, thereby increasing awareness of the condition.

Many people with aphasia have language or cognitive impairments, which preclude their independent reading, and they may depend on family members or others to filter information for them. Howe et al. (2012) found that family members wanted to be better informed, to support their partners to understand information after a stroke. This template could be used to support shared reading with others, as well as providing a reference point for later looking. This may be a useful addition to the various ways of providing information after a stroke, which could be tailored to individual need.

This thesis has focused on the factors affecting accessibility of health information. There is, however, potential to use the template to convey other types of information in a range of contexts, and in different media, including electronic and web-based information.

Training healthcare professionals and others to produce accessible information is pivotal to the implementation of good practice (Graham et al., 2006). Such training needs to include knowledge of what aphasia is and its potential impact on individuals, as well as specific skills training in the appropriate use of language and accessible formatting. This could augment training already judged to improve interpersonal communication between people with aphasia and care providers (e.g. Simmons-Mackie, Raymer, Armstrong, Holland, & Cherney, 2010). The guidelines developed in this study support the development of good practice by individual healthcare professionals and others, and could be used as a basis for such training.

10.7.2 Implications for further research

As previously identified, research into accessible information for people with aphasia is at an early stage of its development and the evidence is limited. This study has contributed to the evolving evidence base, but inevitably raises further questions. Key areas for research are outlined below.

Whilst the study gained further insights into the perceptions of people with aphasia, participants' comprehension of the materials was not tested. Having developed a template which is acceptable to people with aphasia, further research should establish the impact of this format on the comprehension of health information when contrasted with traditional aphasiafriendly formatting. Within this aim, the development of methods to check comprehension of information remains a further challenge for future studies.

The guidelines developed in this study require further evaluation. Firstly, research is indicated to examine the quality of information materials produced by users of the guidelines, potentially informing the training needs of staff. Secondly, the responses of people with aphasia to information materials produced using the guidelines is needed, to ensure that the preferences of larger numbers of people with aphasia are included in future research and development in the area of accessible formatting.

Insights into the varied approaches to visual materials were revealed in this study, consistent with previous reports (e.g. Morrow et al., 2012). Further research is indicated to examine in more depth how people with aphasia look at information materials, using eye-tracking technology, for example, to inform optimum layout of concepts.

Given the equivocal findings of this study and others concerning the use of images (e.g. Brennan et al., 2005; Rose et al., 2012), there is also a need to investigate the effectiveness of materials which do not include images. Different groups may benefit from such investigation, for example, people with mild reading impairment, or those for whom images present a significant distraction.

People with chronic aphasia participated in this study, but all acknowledged that their specific information needs had changed over time. Further research is needed to explore accessible formatting with people in the early stages after stroke, a sample which none of the current or previous literature has included.

This study involved only people with aphasia. The literature reports on the separate information needs of families (Avent et al., 2005; Hilton, Leenhouts, Webster, & Morris, 2014), but no studies have investigated how best to provide information to patients and families together, and none have explored the role of family members in mediating information for people with aphasia. Howe et al. (2012) report that some family members expressed the need to be better informed, in order to support their partners with aphasia to understand new information. Further research is therefore required into the ways in which accessible information can be provided to both patients and their family members.

The physical form of the template was developed as a set of cards, allowing participants to handle the materials. Further study into the application of the template into other media, including electronic and web-based information is now indicated. Caute et al. (2016) found that participants' reading confidence and enjoyment increased when using e-readers, on which they were able to adjust the text size and line spacing. The potential for technology to increase individual control in accessing information is vast, but practical solutions require robust investigation.

The template has provided principles for making information accessible, and guidelines are now freely available (Herbert et al., 2012). Issues of knowledge translation now need to be explored, and the training needs of providers identified, to ensure that the social and political will towards inclusive healthcare systems and the wider society is underpinned by adequate knowledge and skills.

The distress and sense of exclusion felt by people with aphasia when faced with inadequate and inaccessible information has been documented (e.g. Parr et al., 1997). Ultimately, the impact of accessible information on the outcomes of people with aphasia after stroke requires investigation, with the aim of improving both the experience of healthcare, and the health literacy of people with aphasia.

References

- Al Mahmud, A., Limpens, Y., & Martens, J.-B. (2013). Expressing through digital photographs: an assistive tool for persons with aphasia. Universal Access in the Information Society, 12(3), 309-326. doi:10.1007/s10209-012-0286-8
- Al Mahmud, A., & Martens, J.-B. (2013). Amail: Design and Evaluation of an Accessible Email Tool for Persons with Aphasia. *Interacting with Computers, 25*(5), 351-374. doi:10.1093/iwc/iws025
- Aleligay, A., Worrall, L. E., & Rose, T. A. (2008). Readability of written health information provided to people with aphasia. *Aphasiology*, 22(4), 383-407. doi:10.1080/02687030701415872
- Alvarez, J. A., & Emory, E. (2006). Executive function and the frontal lobes: A meta-analytic review. *Neuropsychology Review*, *16*(1), 17-42. doi:10.1007/s11065-006-9002-x
- Anderson, S., & Marlett, N. J. (2004). Communication in stroke: the overlooked rehabilitation tool. Age and Ageing, 33(5), 440-443. doi:10.1093/ageing/afh163
- Atkinson, J. M., & Heritage, J. (1984). *Structures of social action*. Cambridge: Cambridge University Press.
- Avent, J., Glista, S., Wallace, S., Jackson, J., Nishioka, J., & Yip, W. (2005). Family information needs about aphasia. *Aphasiology*, *19*(3-5), 365-375. doi:10.1080/02687030444000813
- Berkman, N. D., Sheridan, S. L., Donahue, K. E., Halpern, D. J., & Crotty, K. (2011). Low Health Literacy and Health Outcomes: An Updated Systematic Review. Annals of Internal Medicine, 155(2), 97-+. doi:10.7326/0003-4819-155-2-201107190-00005
- Berndt, R. S., & Caramazza, A. (1981). Syntactic Aspects of Aphasia. In M. Taylor Sarno (Ed.), Acquired Aphasia (pp. 157-181). New York: Academic Press.
- Best, W. (1995). A reverse length effect in dysphasic naming: When elephant is easier than ant. *Cortex, 31*(4), 637-652. doi:10.1016/s0010-9452(13)80017-2
- Best, W., Herbert, R., Hickin, J., Osborne, F., & Howard, D. (2002). Phonological and orthographic facilitation of word-retrieval in aphasia: Immediate and delayed effects. *Aphasiology*, 16(1-2), 151-168. doi:10.1080/02687040143000483
- Bird, H., Franklin, S., & Howard, D. (2002). 'Little words' not really: function and content words in normal and aphasic speech. *Journal of Neurolinguistics*, *15*(3-5), 209-237. doi:10.1016/s0911-6044(01)00031-8
- Bird, H., Howard, D., & Franklin, S. (2003). Verbs and nouns: the importance of being imageable. *Journal of Neurolinguistics*, *16*(2-3), 113-149. doi:10.1016/s0911-6044(02)00016-7

- Brady, M. C., Fredrick, A., & Williams, B. (2013). People with aphasia: capacity to consent, research participation and intervention inequalities. *International Journal of Stroke*, 8(3), 193-196. doi:10.1111/j.1747-4949.2012.00900.x
- Braun, V. & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology, 3*(2), 77-101. doi:10.1191/1478088706qp063oa
- Brennan, A. D., Worrall, L. E., & McKenna, K. T. (2005). The relationship between specific features of aphasia-friendly written material and comprehension of written material for people with aphasia: An exploratory study. *Aphasiology*, *19*(8), 693-711. doi:10.1080/02687030444000958
- Bright, F. A. S., Kayes, N. M., McCann, C. M., & McPherson, K. M. (2013). Hope in people with aphasia. *Aphasiology*, *27*(1), 41-58. doi:10.1080/02687038.2012.718069
- Brown, K., Worrall, L., Davidson, B., & Howe, T. (2010). Snapshots of success: An insider perspective on living successfully with aphasia. *Aphasiology*, 24(10), 1267-1295. doi:10.1080/02687031003755429
- Bryman, A. (2008). Social Research Methods. (3rd ed). Oxford: Oxford University Press.
- Bunge, M., Muehlhauser, I., & Steckelberg, A. (2010). What constitutes evidence-based patient information? Overview of discussed criteria. *Patient Education and Counseling*, 78(3), 316-328. doi:10.1016/j.pec.2009.10.029
- Bunning, K., & Horton, S. (2007). 'Border crossing'' as a route to inclusion: A shared cause with people with a learning disability? *Aphasiology*, 21(1), 9-22. doi:10.1080/02687030600798162
- Byng, S., & Duchan, J. F. (2005). Social model philosophies and principles: Their applications to therapies for aphasia. *Aphasiology*, 19(10-11), 906-922. doi:10.1080/02687030544000128
- Caplan, D., Waters, G., DeDe, G., Michaud, J., & Reddy, A. (2007). A study of syntactic processing in aphasia I: Behavioral (psycholinguistic) aspects. *Brain and Language*, *101*(2), 103-150. doi:10.1016/j.bandl.2006.06.225
- Carling-Rowland, A., Black, S., McDonald, L., & Kagan, A. (2014). Increasing access to fair capacity evaluation for discharge decision-making for people with aphasia: A randomised controlled trial. *Aphasiology*, *28*(6), 750-765. doi:10.1080/02687038.2014.895975
- Caute, A., Cruice, M., Friede, A., Galliers, J., Dickinson, T., Green, R., & Woolf, C. (2016). Rekindling the love of books - a pilot project exploring whether e-readers help people to read again after a stroke. *Aphasiology*, *30*(2-3), 290-319. doi:10.1080/02687038.2015.1052729
- Cave, C. B. (1997). Very long-lasting priming in picture naming. *Psychological Science*, 8(4), 322-325. doi:10.1111/j.1467-9280.1997.tb00446.x

- Chesneau, S., & Ska, B. (2015). Text comprehension in residual aphasia after basic-level linguistic recovery: a multiple case study. *Aphasiology, 29*(2), 237-256. doi:10.1080/02687038.2014.971098
- Cole-Virtue, J., & Nickels, L. (2004). Why cabbage and not carrot? An investigation of factors affecting performance on spoken word to picture matching. *Aphasiology*, *18*(2), 153-179. doi:10.1080/02687030344000517
- Coltheart, M. (1981). THE MRC PSYCHOLINGUISTIC DATABASE. Quarterly Journal of Experimental Psychology Section a-Human Experimental Psychology, 33(NOV), 497-505.
- Coltheart, M., Rastle, K., Perry, C., Langdon, R., & Ziegler, J. (2001). DRC: A dual route cascaded model of visual word recognition and reading aloud. *Psychological Review*, *108*(1), 204-256. doi:10.1037//0033-295x.108.1.204
- Corbin, J., & Strauss, A. (2008). *Basics of qualitative research* (3rd ed.). Thousand Oaks: Sage.
- Cottrell, S., & Davies, A. (2004). Supporting conversation, providing information. *Royal College* of Speech and Language Therapists Bulletin, October, 2004, 12 13.
- Cottrell, S. & Davies, A. (2006) *Stroke Talk- a communication resource for hospital care*. London: Connect.
- Coulter, A., Entwistle, V., & Gilbert, D. (1998). *Informing patients : an assessment of the quality of patient information materials*. London: King's Fund.
- Coulter, A., Entwistle, V., & Gilbert, D. (1999). Sharing decisions with patients: Is the information good enough? *British Medical Journal, 318*(7179), 318-322. doi:10.1136/bmj.318.7179.318
- Coulter, A. & Ellins, J. (2006). Patient-focused interventions A review of the evidence. *Quest for Quality and Improved Performance Programme*. London: The Health Foundation.
- Cruice, M. (2007). Issues of access and inclusion with aphasia. *Aphasiology*, 21(1), 3-8. doi:10.1080/02687030600798071
- Cruice, M., Hill, R., Worrall, L., & Hickson, L. (2010). Conceptualising quality of life for older people with aphasia. *Aphasiology*, 24(3), 327-347. doi:10.1080/02687030802565849
- Cruice, M., Worrall, L., & Hickson, L. (2011). Reporting on psychological well-being of older adults with chronic aphasia in the context of unaffected peers. *Disability and Rehabilitation, 33*(3), 219-228. doi:10.3109/09638288.2010.503835
- Cumming, T. B., Marshall, R. S., & Lazar, R. M. (2013). Stroke, cognitive deficits, and rehabilitation: still an incomplete picture. *International Journal of Stroke*, 8(1), 38-45. doi:10.1111/j.1747-4949.2012.00972.x
- Dalemans, R., Wade, D. T., van den Heuvel, W. J. A., & de Witte, L. P. (2009). Facilitating the participation of people with aphasia in research: a description of strategies. *Clinical Rehabilitation*, 23(10). doi:10.1177/0269215509337197

- Damasio, A. (1981). The Nature of Aphasia: Signs and Syndromes. In Taylor Sarno, M. (Ed.), Acquired Aphasia (pp. 51-65). New York: Academic Press.
- Damico, J. S., Simmons-Mackie, N., Oelschlaeger, M., Elman, R., & Armstrong, E. (1999).
 Qualitative methods in aphasia research: basic issues. *Aphasiology*, *13*(9-11), 651-665.
 doi:10.1080/026870399401768
- Danzl, M. M., Harrison, A., Hunter, E. G., Kuperstein, J., Sylvia, V., Maddy, K., & Campbell, S. (2016). "A Lot of Things Passed Me by": Rural Stroke Survivors' and Caregivers' Experience of Receiving Education From Health Care Providers. *Journal of Rural Health*, 32(1), 13-24. doi:10.1111/jrh.12124
- Davis, G. A. (1993). *A survey of adult aphasia and related language disorders.* New Jersey: Prentice Hall.
- Denzin, N. K., & Lincoln, Y. S. (1998). *Collecting and interpreting qualitative materials*. Thousand Oaks: Sage.
- Department of Health (2003). Toolkit for Producing Patient Information. Retrieved 16 February 2017 from https://www.uea.ac.uk/documents/246046/0/Toolkit+for+producing+patient+informa tion.pdf

Department of Health (2010). Equity and excellence: Liberating the NHS.

- Devlin, S. & Tait, J. (1998). The Use of a Psycholinguistic Database in the Simplification of Text for Aphasic Readers. In J. Nerbonne (Ed) *Linguistic Databases* (pp.161 -173). CSLI Publications.
- Dietz, A., Hux, K., McKelvey, M. L., Beukelman, D. R., & Weissling, K. (2009). Reading comprehension by people with chronic aphasia: A comparison of three levels of visuographic contextual support. *Aphasiology*, 23(7-8), 1053-1064. doi:Pii 909588509 10.1080/02687030802635832
- Doak, C. C., Doak, L. G., & Root, J. H. (1996). *Teaching Patients with Low Literacy Skills* (2nd ed.). Philadelphia, USA: J.B. Lippincott Company.
- Duman, M. (2003). *Producing Patient Information. How to research, develop and produce effective information resources*. London: King's Fund.
- Eames, S., Hoffmann, T., McKenna, K., & Worrall, L. (2008). Community-based stroke information for clients with stroke and their carers: Is there congruency between actual and recommended practice? *Topics in Stroke Rehabilitation*, *15*(4), 295-306. doi:10.1310/tsr1504-295
- Eames, S., Hoffmann, T., Worrall, L., & Read, S. (2010). Stroke Patients' and Carers' Perception of Barriers to Accessing Stroke Information. *Topics in Stroke Rehabilitation*, 17(2), 69-78. doi:10.1310/tsr1702-69
- Eames, S., Hoffmann, T., Worrall, L., & Read, S. (2011). Delivery styles and formats for different stroke information topics: patient and carer preferences. *Patient education and counseling*, *84*(2), e18-23. doi:10.1016/j.pec.2010.07.007

- Eames, S., Hoffmann, T., Worrall, L., Read, S., & Wong, A. (2013). Randomised controlled trial of an education and support package for stroke patients and their carers. *BMJ Open*, 3(5), e002538. doi:10.1136/bmjopen-2012-002538
- Eames, S., McKenna, K., Worrall, L., & Read, S. (2003). The suitability of written education materials for stroke survivors and their carers. *Topics in Stroke Rehabilitation*, 10(3), 70-83. doi:10.1310/kq70-p8ud-qkyt-dmg4
- Earle, E. A., Davies, H., Greenfield, D., Ross, R., & Eiser, C. (2005). Follow-up care for childhood cancer survivors: a focus group analysis. *European Journal of Cancer*, 41(18), 2882-2886. doi:10.1016/j.ejca.2005.08.028
- Edelman, G., & Greenwood, R. (Eds.). (1992). Jumbly Words and Rights Where Wrongs Should Be. The Experience of Aphasia from the Inside. Leicester, UK: Far Communications.
- Egan, J., Worrall, L., & Oxenham, D. (2004). Accessible Internet training package helps people with aphasia cross the digital divide. *Aphasiology*, *18*(3), 265-280. doi:10.1080/02687030344000562
- Ellis, A. W. (1984, reprinted 1989). *Reading, Writing and Dyslexia: A Cognitive Analysis*. London: Lawrence Erlbaum Associates.
- Ellis, A. W., & Young, A. W. (1988). *Human Cognitive Neuropsychology*. Hove: Lawrence Erlbaum.
- Elman, R., Parr, S., & Moss, B. (2003). The Internet and aphasia: crossing the digital divide. In S. Parr, J. Duchan, & C. Pound (Eds.) *Aphasia Inside Out. Reflections on communication disability* (pp. 103-116). Maidenhead, Berkshire: Open University Press.
- Finch, H. & Lewis, J. (2003). Focus Groups. In J. Ritchie & J. Lewis (Eds.) (2003, reprinted 2009, 2010) Qualitative Research Practice. A Guide for Social Science Students and Researchers (pp. 170-198). Sage: London.
- Flesch, R. (1948). A new readability yardstick. *Journal of Applied Psychology*, *32*, 221–233. doi:10.1037/h0057532
- Forster, A., Brown, L., Smith, J., House, A., Knapp, P., Wright, J. J., & Young, J. (2012a). Information provision for stroke patients and their caregivers. *Cochrane database of systematic reviews (Online), 11*, CD001919-CD001919. doi:10.1002/14651858.CD001919.pub3
- Forster, A., Young, J., Nixon, J., Kalra, L., Smithard, D., Patel, A., . . . Farrin, A. (2012b). A cluster randomized controlled trial of a structured training programme for caregivers of inpatients after stroke (TRACS). *International Journal of Stroke*, 7(1), 94-99. doi:10.1111/j.1747-4949.2011.00722.x
- Frankel, T., Penn, C., & Ormond-Brown, D. (2007). Executive dysfunction as an explanatory basis for conversation symptoms of aphasia: A pilot study. *Aphasiology, 21*(6-8), 814-828. doi:10.1080/02687030701192448
- Franklin, S. (1989). Dissociations in auditory word comprehension; evidence from nine fluent aphasic patients. *Aphasiology*, *3*(3), 189-207. doi:10.1080/02687038908248991

- Franklin, S., Howard, D., & Patterson, K. (1994). Abstract word meaning deafness. *Cognitive Neuropsychology*, 11(1), 1-34. doi:10.1080/02643299408251964
- Galliers, J., Wilson, S., Roper, A., Cocks, N., Marshall, J., Muscroft, S., & Pring, T. (2012). *Words are not Enough: Empowering People with Aphasia in the Design Process.* Paper presented at the Participatory Design Conference, Roskilde, Denmark.
- Garcia, L. J., Barrette, J., & Laroche, C. (2000). Perceptions of the obstacles to work reintegration for persons with aphasia. *Aphasiology*, *14*(3), 269-290. doi:10.1080/026870300401478
- Garrett, M. (1980). Levels of processing in sentence production. In B. Butterworth (Ed.), Language Production Vol. 1: Speech and Talk (pp.). London: Academic Press.
- Gernsbacher, M. A. (1984). Resolving 20 years of inconsistent interactions between lexical familiarity and orthography, concreteness, and polysemy. *Journal of Experimental Psychology-General*, 113(2), 256-281. doi:10.1037//0096-3445.113.2.256
- Ghidella, C. L., Murray, S. J., Smart, M. J., McKenna, K. T., & Worrall, L. E. (2005). Aphasia websites: An examination of their quality and communicative accessibility. *Aphasiology*, 19(12), 1134-1146. doi:10.1080/02687030500337871
- Gillespie, A., Murphy, J., & Place, M. (2010). Divergences of perspective between people with aphasia and their family caregivers. *Aphasiology*, *24*(12), 1559-1575. doi:10.1080/02687038.2010.500810
- Glogowska, M. (2011) Paradigms, pragmatism and possibilities: mixed methods research in speech and language therapy. *International Journal of Language and Communication Disorders*, *46 (3)*, 251-260. doi:10.3109/13682822.2010.507614
- Graham, I. D., Logan, J., Harrison, M. B., Straus, S. E., Tetroe, J., Caswell, W., & Robinson, N.
 (2006). Lost in knowledge translation: Time for a map? *Journal of Continuing Education in the Health Professions*, 26(1), 13-24. doi:10.1002/chp.47
- Great Britain (1995) Disability Discrimination Act. Retrieved from http://www.legislation.gov.uk/id/ukpga/**1995**/50
- Great Britain. (2010) Equality Act The National Archives. Retrieved from http://www.legislation.gov.uk/ukpga/2010/15
- Griffin, J., & Wright, P. (2009). Older readers can be distracted by embellishing graphics in text. *European Journal of Cognitive Psychology*, *21*(5), 740-757. doi:10.1080/09541440802155627
- Hafsteinsdottir, T. B., Vergunst, M., Lindeman, E., & Schuurmans, M. (2011). Educational needs of patients with a stroke and their caregivers: A systematic review of the literature. *Patient Education and Counseling*, *85*(1), 14-25. doi:10.1016/j.pec.2010.07.046
- Hallberg, L., Mellgren, E., Hartelius, L., & Ferm, U. (2013). Talking Mats in a discussion group for people with Huntington's disease. *Disability and rehabilitation. Assistive technology*, 8(1), 67-76. doi:10.3109/17483107.2011.644622

- Hammersley, M. & Atkinson, P.(1995) *Ethnography: Principles in Practice.*(3rd ed.) London: Routledge.
- Hanger, H. C., & Mulley, G. P. (1993). Questions people ask about stroke. *Stroke, 24*(4), 536-538. doi:10.1161/01.str.24.4.536
- Hanger, H. C., Walker, G., Paterson, L. A., McBride, S., & Sainsbury, R. (1998). What do patients and their carers want to know about stroke? A two-year follow-up study. *Clinical Rehabilitation*, *12*(1), 45-52. doi:10.1191/026921598668677675
- Hanger, H. C., & Wilkinson, T. J. (2001). Stroke education: can we rise to the challenge? Age and Ageing, 30 (2), 113-114. doi:10.1093/ageing/30.2.113
- Hare, R., Rogers, H., Lester, H., McManus, R. J., & Mant, J. (2006). What do stroke patients and their carers want from community services? *Family Practice*, 23(1), 131-136. doi:10.1093/fampra/cmi098
- Hartsuiker, R. J., & Kolk, H. H. J. (1998). Syntactic facilitation in agrammatic sentence production. *Brain and Language*, *62*(2), 221-254. doi:10.1006/brln.1997.1905
- Helm-Estabrooks, N. (2002). Cognition and aphasia: a discussion and a study. *Journal of Communication Disorders, 35*(2), 171-186. doi:10.1016/s0021-9924(02)00063-1
- Herbert, R., Haw, C., Brown, C., Gregory, E., & Brumfitt, S. (2012). *Accessible Information Guidelines: Making Information Accessible for People with Aphasia*. London: Stroke Association.
- Heuer, S. (2016). The influence of image characteristics on image recognition: a comparison of photographs and line drawings. *Aphasiology*, 30(8), 943-961. doi:10.1080/02687038.2015.1081138
- Heuer, S. & Hallowell, B. (2007). An evaluation of multiple-choice test images for comprehension assessment in aphasia. *Aphasiology*, 21(9), 883-900. doi:10.1080/02687030600695194
- Heuer, S. & Hallowell, B. (2015). A novel eye-tracking method to assess attention allocation in individuals with and without aphasia using a dual-task paradigm. *Journal of Communication Disorders*, 55, 15-30. doi:10.1016/j.jcomdis.2015.01.005
- Hickin, J., Best, W., Herbert, R., Howard, D., & Osborne, F. (2002). Phonological therapy for word-finding difficulties: A re-evaluation. *Aphasiology*, 16(10-11), 981-999. doi:10.1080/02687030244000509
- Hilari, K. (2011). The impact of stroke: are people with aphasia different to those without? *Disability and Rehabilitation, 33*(3), 211-218. doi:10.3109/09638288.2010.508829
- Hilari, K., Needle, J. J., & Harrison, K. L. (2012). What Are the Important Factors in Health-Related Quality of Life for People With Aphasia? A Systematic Review. Archives of Physical Medicine and Rehabilitation, 93(1), S86-S95. doi:10.1016/j.apmr.2011.05.028
- Hilari, K., & Northcott, S. (2006). Social support in people with chronic aphasia. *Aphasiology,* 20(1), 17-36. doi:10.1080/02687030500279982

- Hilton, R., Leenhouts, S., Webster, J., & Morris, J. (2014). Information, support and training needs of relatives of people with aphasia: Evidence from the literature. *Aphasiology*, 28(7), 797-822. doi:10.1080/02687038.2014.906562
- Hinckley, J. J., Hasselkus, A., & Ganzfried, E. (2013). What People Living With Aphasia Think About the Availability of Aphasia Resources. *American Journal of Speech-Language Pathology, 22*(2), S310-S317. doi:10.1044/1058-0360(2013/12-0090)
- Ho, K., Weiss, S., Garrett, K. & Lloyd, L. (2005) The Effect of Remnant and Pictographic Books on the Communicative Interaction of Individuals with Global Aphasia. *Augmentative and Alternative Communication*, *21*(3), 218-232. doi:10.1080/07434610400016694
- Hoffmann, T., McKenna, K., Herd, C., & Wearing, S. (2007). Written education materials for stroke patients and their carers: Perspectives and practices of health professionals. *Topics in Stroke Rehabilitation, 14*(1), 88-97. doi:10.1310/tsr1401-88
- Hoffmann, T., & Worrall, L. (2004). Designing effective written health education materials: Considerations for health professionals. *Disability and Rehabilitation, 26*(19), 1166-1173. doi:10.1080/09638280410001724816
- Houts, P. S., Doak, C. C., Doak, L. G., & Loscalzo, M. J. (2006). The role of pictures in improving health communication: A review of research on attention, comprehension, recall, and adherence (vol 61, pg 173, 2006). *Patient Education and Counseling*, 64(1-3), 393-394. doi:10.1016/j.pec.2006.06.023
- Howe, T., Davidson, B., Worrall, L., Hersh, D., Ferguson, A., Sherratt, S., & Gilbert, J. (2012).
 'You needed to rehab ... families as well': family members' own goals for aphasia rehabilitation. *International Journal of Language & Communication Disorders*, 47(5), 511-521. doi:10.1111/j.1460-6984.2012.00159.x
- Howe, T., Worrall, L. E., & Hickson, L. M. H. (2004). What is an aphasia-friendly environment? *Aphasiology*, *18*(11), 1015-1037. doi:10.1080/02687030444000499
- Humphris, G.M., Ireland, R.S., Field, E.A. (2001). Randomised trial of the psychological effect of information about oral cancer in primary care settings. Oral Oncology, 37 (7): 548-552. doi:10.1016/s1368-8375(01)00017-3
- Hurtado, B., Jones, L., & Burniston, F. (2014). Is easy read information really easier to read? Journal of Intellectual Disability Research, 58(9), 822-829. doi:10.1111/jir.12097
- Ianni, G. R., Cardillo, E. R., McQuire, M., & Chatterjee, A. (2014). Flying under the radar: figurative language impairments in focal lesion patients. *Frontiers in Human Neuroscience*, 8. doi:10.3389/fnhum.2014.00871
- Intercollegiate Stroke Working Party (2012). *National clinical guideline for stroke*. (4th Edition). London: Royal College of Physicians.
- Jayes, M., & Palmer, R. (2014). Initial evaluation of the Consent Support Tool: A structured procedure to facilitate the inclusion and engagement of people with aphasia in the informed consent process. *International Journal of Speech-Language Pathology*, *16*(2), 159-168. doi:10.3109/17549507.2013.795999

- Kagan, A. (1998). Supported conversation for adults with aphasia: methods and resources for training conversation partners. *Aphasiology*, 12(9), 816-830. doi:10.1080/02687039808249575
- Kagan, A., & Gailey, G. F. (1993). Functional Is Not Enough: Training Conversation Partners for Aphasic Adults. In A. L. Holland & M. M. Forbes (Eds.), *Aphasia Treatment. World Perspectives* (pp. 199-225). London: Chapman & Hall.
- Kagan, A., & Kimelman, M. D. (1995). Informed consent in aphasia research: Myth or reality. *Clinical aphasiology*, 23, 65-75.
- Kagan, A., & LeBlanc, K. (2002). Motivating for infrastructure change: toward a communicatively accessible, participation-based stroke care system for all those affected by aphasia. *Journal of Communication Disorders*, 35(2), 153-169. doi:10.1016/s0021-9924(02)00062-x
- Kalbe, E., Reinhold, N., Brand, M., Markowitsch, H. J., & Kessler, J. (2005). A new test battery to assess aphasic disturbances and associated cognitive dysfunctions - German normative data on the Aphasia Check List. *Journal of Clinical and Experimental Neuropsychology*, 27(7), 779-794. doi:10.1080/13803390490918273
- Kalra, L., Evans, A., Perez, I., Melbourn, A., Patel, A., Knapp, M., Donaldson, N. (2004).
 Training carers of stroke patients: randomised controlled trial. *British Medical Journal*, 328 (7448), 1099. doi:10.1136/bmj.328.7448.1099
- Kay, J., Coltheart, M. & Lesser, R. (1992). *Psycholinguistic Assessment of Language Processing in Aphasia*. Hove: Psychology Press.
- Kerr, J., Hilari, K., & Litosseliti, L. (2010). Information needs after stroke: What to include and how to structure it on a website. A qualitative study using focus groups and card sorting. *Aphasiology*, 24(10), 1170-1196. doi:10.1080/02687030903383738
- Kertesz, A. (1982). The Western Aphasia Battery. New York: Grune and Stratton. Inc, 1982.
- Kickbusch, I., Pelikan, J. M., Apfel, F., & Tsouros, A. D. (Eds.). (2013). *Health literacy The solid facts.* Copenhagen: World Health Organization.
- Kitzinger, J. (2006). Focus Groups. In C. Pope & N. Mays (Eds.) *Qualitative research in health care*. (3rd ed.) (pp. 21-31) Oxford: Blackwell.
- Knight, K., Worrall, L., & Rose, T. (2006). The provision of health information to stroke patients within an acute hospital setting: What actually happens and how do patients feel about it? *Topics in Stroke Rehabilitation*, 13(1), 78-97. doi:10.1310/fc6m-p7l0-w3xd-4wae
- Knollman-Porter, K., Wallace, S. E., Hux, K., Brown, J., & Long, C. (2015). Reading experiences and use of supports by people with chronic aphasia. *Aphasiology, 29*(12), 1448-1472. doi:10.1080/02687038.2015.1041093
- Kraat, A.W. (1990) Augmentative and alternative communication does it have a future in aphasia rehabilitation? *Aphasiology*, *4* (4), 321-338. doi:10.1080/02687039008249086

- Krueger, R.A. & Casey, M. A. (2009) (4th Edition) *Focus Groups. A Practical Guide for Applied Research* California: Sage.
- Laiacona, M., Luzzatti, C., Zonca, G., Guarnaschelli, C., & Capitani, E. (2001). Lexical and semantic factors influencing picture naming in aphasia. *Brain and Cognition, 46*(1-2), 184-187. doi:10.1016/s0278-2626(01)80061-0
- LaPointe, L. L., & Horner, J. (1998). *Reading comprehension battery for aphasia*. TX: PRO-ED.
- Law, J., Bunning, K., Byng, S, Farrelly, S & Heyman, B. (2005) Making sense in primary care: levelling the playing field for people with communication difficulties. *Disability & Society, 20*(2), 169-184. doi:10.1080/09687590500059267
- Lawson, R. & Fawcus, M. (1999) Increasing effective communication using a total communication approach. In S. Byng, K. Swinburn, & C. Pound (Eds.) *The Aphasia Therapy File* (pp.61-71). Hove: Psychology Press.
- Leech, G.N. (2001). Word frequencies in written and spoken English: based on the British National Corpus Harlow : Longman.
- Lesser, R. & Milroy, L. (1993). *Linguistics and Aphasia:Psycholinguistic and Pragmatic Aspects* of Intervention. New York: Longman.
- Lewis, J. & Richie, J. (2003). Generalising from Qualitative Research. In J. Ritchie & J. Lewis (Eds.) (2003, reprinted 2009, 2010) Qualitative Research Practice. A Guide for Social Science Students and Researchers (pp.263-286). London: Sage.
- Ley, P. (1988). *Communicating with patients. Improving communication, satisfaction and compliance*. London: Croom Helm.
- Liu, C. J., Kemper, S., & McDowd, J. (2009). The use of illustration to improve older adults' comprehension of health-related information: Is it helpful? *Patient Education and Counseling*, *76*(2), 283-288. doi:10.1016/j.pec.2009.01.013
- Luck, A. M., & Rose, M. L. (2007). Interviewing people with aphasia: Insights into method adjustments from a pilot study. *Aphasiology*, *21*(2), 208-224. doi:10.1080/02687030601065470
- Lynch, K. E., Damico, J. S., Abendroth, K. J., & Nelson, R. L. (2013). Reading performance subsequent to aphasia: Strategies applied during authentic reading. *Aphasiology*, 27(6), 723-739. doi:10.1080/02687038.2012.748182
- Lyon, J. G., Cariski, D., Keisler, L., Rosenbek, J., Levine, R., Kumpula, J., . . . Blanc, M. (1997). Communication Partners: Enhancing participation in life and communication for adults with aphasia in natural settings. *Aphasiology*, *11*(7), 693-708. doi:10.1080/02687039708249416
- Magasi, S., Durkin, E., Wolf, M. S., & Deutsch, A. (2009). Rehabilitation Consumers' Use and Understanding of Quality Information: A Health Literacy Perspective. *Archives of Physical Medicine and Rehabilitation, 90*(2), 206-212. doi:10.1016/j.apmr.2008.07.023
- Mansoor, L. E., & Dowse, R. (2003). Effect of pictograms on readability of patient information materials. *Annals of Pharmacotherapy*, *37*(7-8), 1003-1009. doi:10.1345/aph.1C449
- Mant, J., Carter, J., Wade, D. T., & Winner, S. (1998). The impact of an information pack on patients with stroke and their carers: a randomized controlled trial. *Clinical Rehabilitation*, *12*(6), 465-476. doi:10.1191/026921598668972226
- Mant, J., Wade, D., & Winner, S. (2004). Stroke. In A. Stevens, J. Raftery, J. Mant, & S. Simpson (Eds.), *Health Care Needs Assessment. The epidemiologically based needs assessment reviews* (pp.141-244). (2nd ed.). Oxford, UK.: Radcliffe.
- Marshall, J. (1995). The mapping hypothesis and aphasia therapy. *Aphasiology, 9*(6), 517-539. doi:10.1080/02687039508248712
- Marshall, J. (2002). Assessment and Treatment of Sentence Processing Disorders: A Review of the Literature. In A. E. Hillis (Ed.), *The Handbook of Adult Language Disorders Integrating Cognitive Neuropsychology, Neurology, and Rehabilitation*. Hove: Psychology Press.
- Marshall, J., Pring, T., Chiat, S., & Robson, J. (2001). When ottoman is easier than chair: An inverse frequency effect in jargon aphasia. *Cortex*, *37*(1), 33-53. doi:10.1016/s0010-9452(08)70556-2
- Marshall, J. C., & Newcombe, F. (1973). Patterns of paralexia psycholinguistic approach. Journal of Psycholinguistic Research, 2(3), 175-199. doi:10.1007/bf01067101
- Marshall, J. (2013). Disorders of sentence processing in aphasia. In I. Papathanasiou, P. Coppens, & C. Potagas (Eds) *Aphasia and Related Neurogenic Communication Disorders* (pp. 197-216). Massachusetts:Jones & Bartlett Learning.
- McCann, C., Tunnicliffe, K., & Anderson, R. (2013). Public awareness of aphasia in New Zealand. *Aphasiology*, *27*(5), 568-580. doi:10.1080/02687038.2012.740553
- McKelvey, M. L., Hux, K., Dietz, A., & Beukelman, D. R. (2010). Impact of Personal Relevance and Contextualization on Word-Picture Matching by People With Aphasia. *American Journal of Speech-Language Pathology*, 19(1), 22-33. doi:10.1044/1058-0360(2009/08-0021)
- McKevitt, C., Fudge, N., Redfern, J., Sheldenkar, A., Crichton, S., Rudd, A. R., . . . Wolfe, C. D. A. (2011). Self-Reported Long-Term Needs After Stroke. *Stroke*, *42*(5), 1398-1403. doi:10.1161/strokeaha.110.598839
- McLafferty, I. (2004). Focus group interviews as a data collecting strategy. *Journal of advanced nursing*, *48*(2), 187-194. doi:10.1111/j.1365-2648.2004.03186.x
- McNamara, T.P. (2005). Semantic Priming. Hove: Psychology Press.
- McPherson, C. J., Higginson, I. J., & Hearn, J. (2001). Effective methods of giving information in cancer: a systematic literature review of randomized controlled trials. *Journal of Public Health*, *23*(3), 227-234.

- McVicker, S., Parr, S., Pound, C., & Duchan, J. (2009). The Communication Partner Scheme: A project to develop long-term, low-cost access to conversation for people living with aphasia. *Aphasiology*, 23(1), 52-71. doi:10.1080/02687030701688783
- Merton, R.K., Fiske, M. & Kendall, P.L.(1990). The Focused Interview. London: The Free Press.
- Meteyard, L., Bruce, C., Edmundson, A., & Oakhill, J. (2015). Profiling text comprehension impairments in aphasia. *Aphasiology, 29*(1), 1-28. doi:10.1080/02687038.2014.955388
- Miles, M.B. & Huberman, A.M. (2002). Reflections and Advice. In A.M. Huberman & M.B. Miles (Eds) *The Qualitative Researcher's Companion* (pp.393-397). London: Sage.
- Miller, H., Buhr, H., Johnson, C., & Hoepner, J. (2013). *AphasiaWeb: A Social Network for Individuals with Aphasia*. Paper presented at the ASSETS '13 October 21 – 23, 2013, Bellevue, WA, USA.
- Moffatt, K., McGrenere, J., Purves, B., & Klawe, M. (2004, April). The participatory design of a sound and image enhanced daily planner for people with aphasia. In *Proceedings of the SIGCHI conference on Human factors in computing systems* (pp. 407-414). ACM.

Morgan, D. (1997). *Focus Groups as Qualitative Research* (2nd ed.) London: Sage.

- Morgan, D. L., Krueger, R. A., & King, J. A. (1998). *The Focus Group Kit. Vol. 1-6.* Thousand Oaks: Sage.
- Morrow, D., D'Andrea, L., Stine-Morrow, E. A. L., Shake, M., Bertel, S., Chin, J., . . . Murray, M. (2012). Comprehension of multimedia health information among older adults with chronic illness. *Visual Communication*, *11*(3), 347-362. doi:10.1177/1470357212446413
- Murphy, J. (1998). Helping people with severe communication difficulties to express their views: a low tech tool. *Communication Matters Journal*, *12*(2), 9-11.
- Murphy, J. (1999). Enabling People with Motor Neurone Disease to Discuss their Quality of Life. *Communication Matters Journal, 13* (2), 2-6.
- Murphy, J. (2000). Enabling people with aphasia to discuss quality of life. *British Journal of Therapy and Rehabilitation*, 7(11), 454 – 457. doi:10.12968/bjtr.2000.7.11.13835
- Murphy, J. (2006). Perceptions of communication between people with communication disability and general practice staff. *Health Expectations, 9*(1). doi:10.1111/j.1369-7625.2006.00366.x
- Murphy, J., Tester, S., Hubbard, G., Downs, M., & MacDonald, C. (2005). Enabling frail older people with a communication difficulty to express their views: the use of Talking Mats (TM) as an interview tool. *Health & Social Care in the Community, 13*(2), 95-107. doi:10.1111/j.1365-2524.2005.00528.x
- Murphy, J. & Cameron, L. (2008). The effectiveness of Talking Mats [®] with people with intellectual disability. *British Journal of Learning Disabilities, 36*(4), 232-241. doi:10.1111/j.1468-3156.2008.00490.x

- Murphy, J., Gray, C.M., van Achterberg, T., Wyke, S. & Cox, S. (2010) The effectiveness of the Talking Mats framework in helping people with dementia to express their views on well-being. *Dementia*, *9*, 454 472. doi:10.1177/1471301210381776
- Murphy, P. W., Davis, T. C., Long, S. W., Jackson, R. H., & Decker, B. C. (1993). Rapid estimate of adult literacy in medicine (REALM): A quick reading test for patients. In M. C.
 Radencich (Ed.) Adult literacy: A compendium of articles from the journal of reading (pp. 79 86). Newark, DE: International Reading Association.
- Murray, E., Burns, J., See, T. S., Lai, R., & Nazareth, I. (2004). Interactive Health Communication Applications for people with chronic disease. Cochrane Database of Systematic Reviews. doi:10.1002/14651858.cd004274.pub3
- Murray, L. L. (2012). Attention and Other Cognitive Deficits in Aphasia: Presence and Relation to Language and Communication Measures. *American Journal of Speech-Language Pathology*, *21*(2), S51-S64. doi:10.1044/1058-0360(2012/11-0067)
- Myers, E.D. & Calvert, E.J. (1984). Information, compliance and side effects: a study of patients on anti-depressant medication. *British Journal of Clinical Pharmacology*, *17*(1), 221-225. doi:10.1111/j.1365-2125.1984.tb04993.x
- NHS England (2015). Accessible Information Standard. Retrieved from https://www.england.nhs.uk/ourwork/accessibleinfo/
- National Institute for Health and Care Excellence (NICE) (2013). *Stroke rehabilitation: long term rehabilitation after stroke. Clinical guideline 162.* National Clinical Guideline Centre, 2013. Retrieved from <u>https://www.nice.org.uk/guidance/cg162</u>
- Nickels, L. (2002). Improving word finding: Practice makes (closer to) perfect? *Aphasiology, 16*(10-11), 1047-1060. doi:10.1080/02687040143000618
- Nickels, L., & Howard, D. (1995). Aphasic naming what matters. *Neuropsychologia*, 33(10), 1281-1303. doi:10.1016/0028-3932(95)00102-9
- Nickels, L., & Howard, D. (2004). Dissociating effects of number of phonemes, number of syllables, and syllabic complexity on word production in aphasia: It's the number of phonemes that counts. *Cognitive Neuropsychology*, *21*(1), 57-78. doi:10.1080/02643290342000122
- Northcott, S., & Hilari, K. (2011). Why do people lose their friends after a stroke? *International Journal of Language & Communication Disorders, 46*(5), 524-534. doi:10.1111/j.1460-6984.2011.00079.x
- Nutbeam, D. (2000). Health literacy as a public health goal: a challenge for contemporary health education and communication strategies into the 21st century. *Health Promotion International, 15*(3), 259-267. doi:10.1093/heapro/15.3.259
- Office of Public Sector Information (2005). Mental Capacity Act 2005. London: OPSI.
- Oliver, M. (1996). Understanding disability: From theory to practice. Basingstoke: Palgrave Macmillan.

- O'Halloran, R., Hickson, L., & Worrall, L. (2008). Environmental factors that influence communication between people with communication disability and their healthcare providers in hospital: a review of the literature within the International Classification of Functioning, Disability and Health (ICF) framework. *International Journal of Language & Communication Disorders*, *43*(6), 601-632. doi:10.1080/13682820701861832
- Paasche-Orlow, M. K., McCaffery, K., & Wolf, M. S. (2009). Bridging the International Divide for Health Literacy Research. *Patient Education and Counseling*, 75(3), 293-294. doi:10.1016/j.pec.2009.05.001
- Papagno, C., & Caporali, A. (2007). Testing idiom comprehension in aphasic patients: The effects of task and idiom type. *Brain and Language, 100*(2), 208-220. doi:10.1016/j.bandl.2006.01.002
- Parr, S. (2007). Living with severe aphasia: Tracking social exclusion. *Aphasiology, 21*(1), 98-123. doi:10.1080/02687030600798337
- Parr, S., Byng, S., Gilpin, S., & Ireland, C. (1997). *Talking about Aphasia. Living with loss of language after stroke.* Buckingham: Open University Press.
- Parr, S., Pound, C., Byng, S., Long, B., Connect, & Press, L. (1999). *The Aphasia Handbook*. London: Connect Press.
- Parr, S., Pound, C., Byng, S., Long, B., Connect, & Press, L. (2004). *The Stroke and Aphasia Handbook*. London: Connect Press.
- Parr, S., Pound, C., & Hewitt, A. (2006). Communication access to health and social services. *Topics in Language Disorders, 26*(3), 189-198. doi:10.1097/00011363-200607000-00003
- Parr, S., Wimborne, N., Hewitt, A. & Pound, C. (2008, reprinted 2011). *The Communication Access Toolkit.* London: Connect Press.
- Patton, M.Q. (2002). *Qualitative Research and Evaluation Methods* (3rd ed.). Thousand Oaks: Sage.
- Pearl, G. (2014). Engaging with people who have aphasia a set of resources for stroke researchers. In National Institute for Health Research CRN: Stroke.
- Pickering, M. J., & Branigan, H. P. (1998). The representation of verbs: Evidence from syntactic priming in language production. *Journal of Memory and Language*, 39(4), 633-651. doi:10.1006/jmla.1998.2592
- Pickering, M. J., & Ferreira, V. S. (2008). Structural priming: A critical review. *Psychological Bulletin*, *134*(3), 427-459. doi:10.1037/0033-2909.134.3.427
- Pleasant, A., McKinney, J., & Rikard, R. V. (2011). Health Literacy Measurement: A Proposed Research Agenda. *Journal of Health Communication, 16*, 11-21. doi:10.1080/10810730.2011.604392
- Pope, C. & Mays, N. (Eds) (2006). *Qualitative research in health care* (3rd ed.). Oxford: Blackwell BMJ Books.

- Pope, C., Ziebland, S. & Mays, N. (2006). Analysing qualitative data. In C. Pope & N. Mays (Eds.) *Qualitative research in health care* (3rd ed.) (pp.63-81). Oxford: Blackwell BMJ Books.
- Poland, B. D. (1995). Transcription quality as an aspect of rigor in qualitative research. *Qualitative inquiry*, 1(3), 290-310. doi:10.1177/107780049500100302
- Pound, C., Duchan, J., Penman, T., Hewitt, A., & Parr, S. (2007). Communication access to organisations: Inclusionary practices for people with aphasia. *Aphasiology*, 21(1), 23-38. doi:10.1080/02687030600798212
- Pound, C., Parr, S., Lindsay, J., & Woolf, C. (2000). Beyond Aphasia. Bicester: Speechmark.
- Prior, S., Waller, A., & Kroll, T. (2011). Focus groups as a requirements gathering method with adults with severe speech and physical impairments. *Behaviour & Information Technology*, 1 - 10. doi:DOI:10.1080/0144929X.2011.566939
- Rafi, I., Sullivan, R., & Mathers, N. (2016). The Accessible Information Standard: action and reaction. *British Journal of General Practice, February 2016*, 64-65. doi:10.3399/bjgp16x683545
- Rayner, K. & Pollatsek, A. (1989). The Psychology of Reading. London: Prentice- Hall.
- Richie, J. & Spencer, L. (1994). Qualitative data analysis for applied policy research. In A. Bryman & R.G. Burgess (Eds.) *Analysing Qualitative Data* (pp. 173-194). London: Routledge.
- Richie, J., Lewis, J. & Elam, G. (2003). Designing and Selecting Samples. In J. Ritchie & J. Lewis (Eds.) (2003, reprinted 2009, 2010) *Qualitative Research Practice. A Guide for Social Science Students and Researchers* (pp.77-108). London: Sage.
- Richie, J., Spencer, L., O'Connor, W. (2003). Carrying out Qualitative Analysis. In J. Ritchie & J. Lewis (Eds.) (2003, reprinted 2009, 2010) *Qualitative Research Practice. A Guide for Social Science Students and Researchers* (pp.219-262). London: Sage.
- Riley, E.A., & Kendall, D.L. (2013). The acquired disorders of reading. In I. Papathanasiou, P. Coppens, & C. Potagas (Eds.) *Aphasia and Related Communication Disorders* (pp.157-172). Massachusetts: Jones & Bartlett Learning.
- Rodgers, H., Bond, S., & Curless, R. (2001). Inadequacies in the provision of information to stroke patients and their families. *Age and Ageing*, *30*(2). doi:10.1093/ageing/30.2.129
- Rodgers, J., & Namaganda, S. (2005). Making information easier for people with learning difficulties. *British Journal of Learning Disabilities*, 33(2), 52 - 58. doi:10.1111/j.1468-3156.2005.00341.x
- Rose, T., Worrall, L., Hickson, L., & Hoffmann, T. (2010). Do People With Aphasia Want Written Stroke and Aphasia Information? A Verbal Survey Exploring Preferences for When and How to Provide Stroke and Aphasia Information. *Topics in Stroke Rehabilitation*, 17(2), 79-98. doi:10.1310/tsr1702-79

- Rose, T. A., Worrall, L. E., Hickson, L. M., & Hoffmann, T. C. (2011a). Aphasia friendly written health information: Content and design characteristics. *International Journal of Speech-Language Pathology*, 13(4), 335-347. doi:10.3109/17549507.2011.560396
- Rose, T. A., Worrall, L. E., Hickson, L. M., & Hoffmann, T. C. (2011b). Exploring the use of graphics in written health information for people with aphasia. *Aphasiology*, 25(12), 1579-1599. doi:10.1080/02687038.2011.626845
- Rose, T. A., Worrall, L. E., Hickson, L. M., & Hoffmann, T. C. (2012). Guiding principles for printed education materials: Design preferences of people with aphasia. *International Journal of Speech-Language Pathology*, 14(1), 11-23. doi:10.3109/17549507.2011.631583
- Rose, T. A., Worrall, L. E., & McKenna, K. T. (2003). The effectiveness of aphasia-friendly principles for printed health education materials for people with aphasia following stroke. *Aphasiology*, *17*(10), 947-963. doi:10.1080/02687030344000319
- Rose, T. A., Worrall, L. E., McKenna, K. T., Hickson, L. M., & Hoffmann, T. C. (2009). Do people with aphasia receive written stroke and aphasia information? *Aphasiology, 23*(3), 364-392. doi:10.1080/02687030802568108
- Rowlands, G. P., Protheroe, J., Price, H., Gann, B., & Rafi, I. (2014). *Health Literacy: Report* from an RCGP-led health literacy workshop. London: Royal College of General Practitioners.
- Sacchett, C., Byng, S., Marshall, J. & Pound, C. (1999) Drawing together: Evaluation of a therapy programme for severe aphasia. *International Journal of Language and Communication Disorders*, 34(3), 265-289. doi:10.1080/136828299247414
- Saffran, E. M., & Martin, N. (1997). Effects of structural priming on sentence production in aphasics. *Language and Cognitive Processes, 12*(5-6), 877-882. doi:10.1080/016909697386772
- Sandelowski, M. (1995). Triangles and crystals: on the geometry of qualitative research. *Research in Nursing & Health, 18*(6), 569-574. doi:10.1002/nur.4770180612
- Schwartz, M. F., Saffran, E. M., & Marin, O. S. (1980). The word order problem in agrammatism I. Comprehension. *Brain and language*, 10(2), 249-262. doi:10.1016/0093-934x(80)90055-3
- Seale, C. (1999). The Quality of Qualitative Research. London: Sage.
- Sears, A., & Lund, A. M. (1997). Creating effective user interfaces. *IEEE Software*, *14*(4), 21-24. doi:10.1109/ms.1997.595887
- Serper, M., Patzer, R. E., Curtis, L. M., Smith, S. G., O'Conor, R., Baker, D. W., & Wolf, M. S. (2014). Health Literacy, Cognitive Ability, and Functional Health Status among Older Adults. *Health Services Research*, 49(4), 1249-1267. doi:10.1111/1475-6773.12154
- Silverman, D. (2006). Interpreting qualitative data: Methods for analyzing talk, text and interaction. London: Sage.

- Simmons-Mackie, N., Code, C., Armstrong, E., Stiegler, L., & Elman, R. J. (2002). What is aphasia? Results of an international survey. *Aphasiology*, 16(8), 837-848. doi:10.1080/02687030244000185
- Simmons-Mackie, N., Raymer, A., Armstrong, E., Holland, A., & Cherney, L. R. (2010). Communication partner training in aphasia: a systematic review. Archives of physical medicine and rehabilitation, 91(12), 1814-1837.
- Simmons-Mackie, N., & Lynch, K. E. (2013). Qualitative research in aphasia: A review of the literature. *Aphasiology*, 27(11), 1281-1301. doi:10.1080/02687038.2013.818098
- Simmons-Mackie, N., Worrall, L., Murray, L. L., Enderby, P., Rose, M. L., Paek, E. J., . . . Advisory, C. (2017). The top ten: best practice recommendations for aphasia. *Aphasiology*, *31*(2), 131-151. doi:10.1080/02687038.2016.1180662
- Singh, S., Gedeon, T. D., & Rho, Y. (1998). Enhancing comprehension of Web information for users with special linguistic needs. *Journal of Communication*, 48(2), 86-108. doi:10.1111/j.1460-2466.1998.tb02749.x
- Smith, J.A., Jarman, M. & Osborn, M. (1999) Doing Interpretative Phenomenological Analysis. In Murray, M. & Chamberlain, K. (Eds) *Qualitative Health Psychology. Theories and Methods.* Sage, London.
- Spencer, J., Ritchie, J., Lewis, J., & Dillon, L. (2003). Quality in Qualitative Evaluation: A framework for assessing research evidence. Government Chief Social Researcher's Office, London: Cabinet Office.
- Spencer, J., Richie, J., & O'Connor, W. (2003) Carrying out Qualitative Analysis. In In J. Ritchie
 & J. Lewis (Eds.) (2003, reprinted 2009, 2010) *Qualitative Research Practice. A Guide* for Social Science Students and Researchers (pp.219-262). London: Sage.
- Stacey, D., Légaré, F., Col, N.F., Bennett, C.L., Barry, M.J., Eden, K.B., Holmes-Rovner, M., Llewellyn-Thomas, H., Lyddiatt, A., Thomson, R., Trevena, L., Wu, J.H.C. (2014).
 Decision aids for people facing health treatment or screening decisions, *Cochrane Database of Systematic Reviews*, Issue 1. Art. No.: CD001431. DOI: 10.1002/14651858.CD001431.pub4.
- Stewart, D. W., Shamdasani, P. N., & Rook, D. W. (2007). *Focus groups: theory and practice.* California: Sage Publications.
- Street, R. L., Makoul, G., Arora, N. K., & Epstein, R. M. (2009). How does communication heal? Pathways linking clinician–patient communication to health outcomes. *Patient education and counseling*, 74(3), 295-301. doi:10.1016/j.pec.2008.11.015
- Stroke Association. (2013). *Feeling overwhelmed: the emotional impact of stroke*. Stroke Association.
- Stroke Association. (2017). State of the Nation Statistics, January 2017. Retrieved from https://www.stroke.org.uk/resources/state-nation-stroke-statistics
- Stroke Association (2017). Aphasia and communicating. Retrieved from https://www.stroke.org.uk/finding-support/what-aphasia

- Surman, R., & Bath, P. A. (2013). An assessment of the quality of information on stroke and speech and language difficulty web sites. *Journal of Information Science*, 39(1), 113-125. doi:10.1177/0165551512469775
- Swart, J., & Horton, S. (2015). From patients to teachers: the perspectives of trainers with aphasia in a UK Conversation Partner Scheme. *Aphasiology*, *29*(2), 195-213. doi:10.1080/02687038.2014.961893
- Sweller, J. (2010). Element Interactivity and Intrinsic, Extraneous, and Germane Cognitive Load. *Educational Psychology Review*, 22(2), 123-138. doi:10.1007/s10648-010-9128-5
- Swinburn, K., Porter, G. & Howard, D. (2005). *Comprehensive Aphasia Test*. Hove:Psychology Press.
- Swinburn, K., Parr, S. & Pound, C. (2007). *Including People with Communication Disability in Stroke Research and Consultation. A Guide for Researchers and Service Providers*. London: Connect.
- Symonds, J. E., & Gorard, S. (2010). Death of mixed methods? Or the rebirth of research as a craft. *Evaluation & Research in Education*, *23*(2), 121-136. doi:10.1080/09500790.2010.483514
- Teddlie, C., & Tashakkori, A. (2009). *Foundations of mixed methods research: Integrating quantitative and qualitative approaches in the social and behavioral sciences*. London: Sage.
- Thompson, A. E., Goldszmidt, M. A., Schwartz, A. J., & Bashook, P. G. (2010). A randomized trial of pictorial versus prose-based medication information pamphlets. *Patient Education and Counseling*, *78*(3), 389-393. doi:10.1016/j.pec.2010.01.010
- Thorburn, L., Newhoff, M. & Rubin, S.S. (1995). Ability of subjects with aphasia to visually analyse written language, pantomime and iconographic symbols. *American Journal of Speech-Language Pathology*, *4*(4) 174-179.
- Thurstone, T. (1978) Reading for understanding 2. Chicago, IL: Science Research Associates.
- Tomkins, B., Siyambalapitiya, S., & Worrall, L. (2013). What do people with aphasia think about their health care? Factors influencing satisfaction and dissatisfaction. *Aphasiology,* 27(8), 972-991. doi:10.1080/02687038.2013.811211
- United Nations. (2008). Convention on the Rights of Persons with Disabilities. Retrieved February 12, 2017, from https://www.un.org/development/desa/disabilities/convention-on-the-rights-ofpersons-with-disabilities.html
- U.S. Department of Health and Human Services, Office of Disease Prevention and Health Promotion. (2010). National Action Plan to Improve Health Literacy. Washington D.C: Author. Retrieved February 12, 2017, from https://www.cdc.gov/healthliteracy/planact/national.html
- van Veenendaal, H., Grinspun, D. R., & Adriaanse, H. P. (1996). Educational needs of stroke survivors and their family members, as perceived by themselves and by health

professionals. *Patient Education and Counseling, 28*(3), 265-276. doi:10.1016/0738-3991(95)00853-5

- Wachters-Kaufmann, C., Schuling, J., The, H., & Meyboom-de Jong, B. (2005). Actual and desired information provision after a stroke. *Patient Education and Counseling*, *56*(2), 211-217. doi:10.1016/j.pec.2004.02.012
- Warren, M. (1993). A Hierarchical Model for Evaluation and Treatment of Visual Perceptual Dysfunction in Adult Acquired Brain Injury, Part 1. *The American Journal of Occupational Therapy*, 47(1), 42-54. doi:10.5014/ajot.47.1.42
- White, C., Woodfield, K., & Richie, J. (2003). Reporting and Presenting Qualitative Data. In J.
 Ritchie & J. Lewis (Eds.) (2003, reprinted 2009, 2010) *Qualitative Research Practice. A Guide for Social Science Students and Researchers* (pp.287-320). London: Sage.
- Wiles, R., Pain, H., Buckland, S., & McLellan, L. (1998). Providing appropriate information to patients and carers following a stroke. *Journal of Advanced Nursing*, 28(4), 794-801. doi:10.1046/j.1365-2648.1998.00709.x
- Wilson, E. A. H., & Wolf, M. S. (2009). Working memory and the design of health materials: A cognitive factors perspective. *Patient Education and Counseling*, 74(3), 318-322. doi:10.1016/j.pec.2008.11.005
- Wilson, L., & Read, J. (2016). Do particular design features assist people with aphasia to comprehend text? An exploratory study. *International Journal of Language & Communication Disorders, 51*(3), 346-354. doi:10.1111/1460-6984.12206
- Wilson, S., Roper, A., Marshall, J., Galliers, J., Devane, N., Booth, T., & Woolf, C. (2015). Codesign for people with aphasia through tangible design languages. *Codesign-International Journal of Cocreation in Design and the Arts*, 11(1), 21-34. doi:10.1080/15710882.2014.997744
- Wolf, M. S., Gazmararian, J. A., & Baker, D. W. (2005). Health literacy and functional health status among older adults. *Archives of Internal Medicine*, 165(17), 1946-1952. doi:10.1001/archinte.165.17.1946
- World Health Organisation (2001) International classification of functioning, disability and health: ICF. Geneva: World Health Organisation .
- World Health Organisation (2011). World Report on Disability. Retrieved February 12, 2017, from http://www.who.int/disabilities/world_report/2011/report/en/
- World Medical Association (2001). Declaration of Helsinki. *Bulletin of the World Health Organisation, 2001, 79* (4), 373 – 374.
- Worrall, L., Brown, K., Cruice, M., Davidson, B., Hersh, D., Howe, T., & Sherratt, S. (2010). The evidence for a life-coaching approach to aphasia. *Aphasiology*, *24*(4), 497-514. doi:10.1080/02687030802698152
- Worrall, L., Rose, T., Howe, T., McKenna, K., & Hickson, L. (2007). Developing an evidence-base for accessibility for people with aphasia. *Aphasiology*, 21(1), 124-136. doi:10.1080/02687030600798352

Worrall, L., Sherratt, S., Rogers, P., Howe, T., Hersh, D., Ferguson, A., & Davidson, B. (2011).
 What people with aphasia want: Their goals according to the ICF. *Aphasiology*, 25(3), 309-322. doi:10.1080/02687038.2010.508530

Appendices

Appendix A	249
Appendix B	250
Appendix C	251
Appendix D	252
Appendix E	253
Appendix F	254
Appendix G	255
Appendix H	256
Appendix I	257
Appendix J	
Appendix K	262
Appendix L	
Appendix M	
Appendix N	
Appendix O	
Appendix P	
Appendix Q	270
Appendix R	274
Appendix S	277
Appendix T	278

Appendix A

Торіс	Subtopic	Sentences
Stroke illness	Your brain	your brain controls your body
		your brain controls your arms and legs
		your brain makes your speech
		your brain understands speech
	your blood	your brain needs blood
		blood feeds your brain
		blood keeps the brain healthy
		arteries carry blood to the brain
		arteries need to be clean
	stroke illness - ischaemia	your arteries get blocked
		blocked arteries stop blood
		the brain can't work
Aphasia	background information	language is talking and understanding
		language is reading and writing
		stroke can damage language: this is
		aphasia
	facts	it is difficult to say words
		it is difficult to form sentences
		it is difficult to ask for things
		it is difficult to say yes or no
	what helps	practising the words again and again
		helps
		pointing helps
		using your face helps
Emotions after	facts	stroke can cause depression
stroke		stroke can cause anxiety
		stroke can cause frustration
	what helps	seeing a Clinical Psychologist will help
		telling people how you feel will help

Examples of information sentences generated for Template 1

Appendix B

Examples of Template 1 – variations in image style

All images © The Stroke Association

(examples shown smaller than actual size)

Example 1: Your brain listens to speech





Example 2: Your brain controls your body





Appendix C

Examples of Template 1 – variations in number of images per concept

All images © The Stroke Association

(examples shown smaller than actual size)

Example 1: Blood makes your brain work



Example 2: Your brain produces your speech



Appendix D

Set of Template 2 materials on the topic aphasia

All images © The Stroke Association

(examples shown smaller than actual size)



Appendix E

Focus group 1 - Typography materials, shown actual size.

1. Variations in font size

Arial 10	Cup of coffee
Arial 12	Cup of coffee
Arial 18	Cup of coffee
Arial 26	Cup of coffee

2. Variations in font style

Arial 12	Cup of coffee
Times New Roman 12	Cup of coffee

3. Variations in emphasis. Focus group 1 materials

Arial 12 + 16	Cup of coffee
Arial 12 + bold	Cup of coffee
Arial 12 + italics	Cup of <i>coffee</i>
Arial 12 + underline	Cup of <u>coffee</u>

Appendix F

Focus group 2 – Template 2 materials to explore emphasis

All images © The Stroke Association

(examples shown smaller than actual size)

stroke and your emotions 6 Stroke can cause frustration Image: stroke can cause frustration

Normal font (control)

Bold font



Large font



Large and bold font



Appendix G

Focus group 2 – Template 2 materials to explore preferences for number of concepts per page

All images © The Stroke Association

(examples shown smaller than actual size)

One concept per page



Two concepts per page



Three concepts per page



Appendix H

Visual rating scale provided on Talking Mats® framework

(shown smaller than actual size)



Appendix I

Examples of Talking Mats® data

All images ©The Stroke Association

Talking Mats® examples showing ratings for image style preferences

Example showing preference for line drawings



Example showing preference for photographs



Talking Mats® examples showing ratings for number of images per sentence



Example showing preference for single image

Example showing preference for multiple rather than single images



Talking Mats® examples showing ratings for number of concepts per page



Example showing preference for one concept per page

Example showing preference for two related concepts per page



Talking Mats[®] examples showing ratings for emphasis of key words in Template 2



Appendix J

Ethical approval for the study form



Department of Human Communication Sciences 31 Claremont Crescent Sheffield S10 2TA UK

Head of Department Professor Joy Stackhouse, PhD., F.R.C.S.L.T., C.Psychol., A.F.B.Ps.S

> Telephone: +44 (0) 114 222 2418/ 2402/ 2405 International: +44 (0) 114 222 2418 Fax: +44 (0) 114 273 0547 Email: hcs-support@sheffield.ac.uk http://www.shef.ac.uk/hcs

17th December 2010

Dear Ruth

Title: Helping people with Aphasia to understand information

Thank you for your submission to the HCS Research Ethics Committee. The committee has reviewed your submission and supporting documents and grants you approval to commence the research.

We hope your project proceeds smoothly

Yours sincerely

Bay

Prof R Varley Chair of HCS Ethics Committee

Appendix K

Accessible project information sheet



They are funding this project. **<u>Recordings</u>**

Everything that people **say** in the group is **important**.

We will **record** everything on a sound recorder, and on **video**.

We will **listen** to this afterwards to make sure that we **don't miss** anything.

Everything people say will help with the project.

The recordings will be stored in **computers** in the University.

The recordings will be **confidential.**

Only the researchers will be able to listen to them.

The recordings will be for this project only.

The recordings will be **destroyed five years** after the end of the project.

Getting there

The group will be held at the **University**.

We will arrange **a taxi** to collect you and take you home.

We will pay for the taxis.

We will need to know about your **language**.

We will assess your language

We can **visit your home** to do this,

or you can come to the **clinic**.

You can **choose**.

We will **pay for your taxi** to the clinic and back home.

It will take about one hour.

<u>Stopping</u>

You can **stop** whenever you want. You don't need to give a reason. Just let us know.

Interested?

If you are interested let us know, and tick the box here to find out more

I want to find out more

Appendix L

Pre-group participant confirmation letter

(shown smaller than actual size)

Г

The University Of Sheffield. Helping people with aphasia to	Department Of Human Communication Sciences Med of Department Professor Shelagh Brumfitt, rsc 31 Claremont Crescent Sheffield S10 2TA United Kingdom Methone: +44 (0) 114 222 2418/ 2402/ 2405 Method Sing Sciences Method Sing Sciences Method Sing Sciences Method Sing Sciences Method Sing Sciences Method Sing Sciences Method Sciences
Dear	
Thank you for agreeing to take part in our	study.
You are going to come to the university on	
The taxi is booked and paid for. It will call to:	for you at pm. It will bring you
31, Claremont Crescent S10 2TA	
The taxi will take you home again at 4:00.	
We are looking forward to seeing you again	n.
Best wishes,	
Ruth Herbert, Caroline Haw and Catherine (Researchers)	Brown
Tel 0114 222 2416	

Appendix M

Participant post-group follow up letter

(shown smaller than actual size)

The University Of Sheffield.	Head of Departm Professor Shelagi 31 Claremont Cres Sheffield \$10 27A United Kingdom	ent Brumfitt, PBD scent) 114 222 2418/ 2402/ 2405 (0) 114 222 2418 173 0547 (@sheffield.ac.uk
Helping people with aphasia to	understand information	
Thank you for taking part in ou	r study today.	
Your next focus group will be o	n Friday 18th March at 1:	30pm.
Your transport will be booked a	and paid for.	
We will send you a letter about	t your pick up time.	
We will also phone you.		
We look forward to seeing you	again then,	
Best wishes,		
Caroline Haw C	Catherine Brown	Ruth Herbert
Tel: 0114 222 2416		

Appendix N

Accessible consent form (shown smaller than actual size)

The University Of Sheffield.		Department Human Communica Sciences Head of Department Professor Shelagh Br 31 Claremont Crescen Sheffield S10 2TA United Kingdom Telephone: +44 (0) 11- 2405 International: +44 (0) Fax: +44 (0) 114273 (Email: hcs-support@sl http://www.shef.ac.uk/	umfitt , Pa0 4 222 2418/ 2402/ 114 222 2418 547 heffield.ac.uk
Helping people with aphasia to understand info	ormation		
I understand what the project will involve.			
has explaine	ed what will ha Di	ppen sagree	Agree
	(予	10.000 1
I agree to involvement in the pro	-	X X	$\mathbf{\nabla}$
I understand that my speaking will be recorded			V
SignatureDate Name Researcher name			

Appendix O

Protocol for first focus group meetings (shown smaller than actual size)

	ticipants must sign the consent form before joining the ning the accessible information.	focus group or participa	ition in an intervi	ew
	P PROTOCOL			
All peo	up discussion will be audio-recorded ple working with someone with aphasia on one to one discussion	basis to ensure that all c	omments are fed	into th
NO	ACTIVITY	EQUIPMENT	AUDIO RECORDING?	WHO
1	Introduction: who we all are Convene group. Greetings Caroline to do introductions of everyone.	Name labels Pens	NO	СН
2	State aims of the group. Provide concrete examples of information given after stroke to demonstrate. Encourage all to express their opinions as experts in stroke Provide overview of the afternoon's schedule. Housekeeping.	Standard stroke leaflets A3 schedule x 2	YES	СН
3	 Introduce Talking Mats and the Visual Scale as the means of communicating preferences which will be used in the group. All to be expressed openly for microphone to capture activity for transcription. Menu activity: Caroline to use 1 picture to introduce each course: Like or dislike this food? Place 3 pictures per course, rating how well each conveys what it is, i.e. its meaning. Focus on strength of picture in context of menu (ie information) rather than personal preference for the food item itself. Photograph each TM once picture menu completed. Group discussion Text activity: Cup of coffee picture placed at top of each TM. Participants to rate pairs of written phrases of "cup of coffee" for font preferences. Photograph each TM once all fonts displayed. Group discussion 	TM for each participant One set of food pictures for Caroline to demonstrate Set of food pictures ON VELCRO for each participant Cup of coffee picture x 8 Phrase 'cup of coffee' in range of fonts/sizes etc Visual scale for each participant, velcro'ed to the TM Digital camera Laptop to check photos on to	GROUP DISCUSSION	СН
4	Introduce some of Jon's pictures and word sets and select preferences using TM Compare and rate pairs of line drawings: Your brain makes your speech Your brain controls your body Blood makes your brain work Compare and rate pairs of line vs photo presentations: Your brain moves your arms and legs Your brain needs blood Your brain listens to speech	TM Visual scale Jon's pictures on Velcro Camera Laptop	GROUP DISCUSSION	RH
	Photograph of each TM Open discussion regarding why selections were made for both images and text			

	5	Topic generation ASK PEOPLE to brainstorm what was most	Flipchart to write the topics on	GROUP BRAINSTORM	CB
		important for them after a stroke:	Our list of possible		
		"What does somebody need to know who is	top topics		
		recovering after a stroke?"	Visual scale to rate		
		THEN PRESENT OUR LIST of topics	the topics		
		Participants asked to rate these as very to not			
		important	Camera		
		Participants asked to think again of other topics			
	6	Closing comments	Written information		СВ
	0	Reminder of date of next group	sheet with date of		CD
		Reminder of date of next group	next meeting on and		
			information		
			regarding taxis and		
			that we will ring		
			them beforehand		
			One for each		
			participant		
			P		
	CHODD				
		ING / GET FROM GENERAL OFFICE	warner late of slabeles	ahawta aw Ad	
		abels Pens suitable for people with stroke to use Blank on large piece of paper for people to point to	paper – lots of .alphabe	Charts off A4	
		Mats x 10			
		r on each Mat so we know whose is whose when look a	t photos		
	Tea bag		it photos		
	Coffee	50			
	Milk				
		jugs and tap water fine from kitchen)			
	Good b				
		quipment:			
		art Pens for flip chart Audio recording equipment			
		ecording: F1/ IT support to sort			
-					

Appendix P

Protocol for second focus group meetings (shown smaller than actual size)

	PROTOCOL FOR FOCUS GROUPS FOR ACCESSIE	BLE INFORMATION PRO	DJECT	
GROUI	P PROTOCOL			
-	up discussion will be audio-recorded pple working with someone with aphasia on one to one basis to e sion	nsure that all commen	ts are fed into th	e group
NO	ACTIVITY	EQUIPMENT	AUDIO RECORDING?	WHO
1	Introduction: Convene group. Greetings and thanks	Name labels Pens	NO	
	 State aims of the group: accessible information after stroke a work in progress designer worked based on your comments and opinions 	A3 schedule x2		СВ
	 today asking for your views on what we've done so far plan for the afternoon 			
2	 Open the card boxes and take out one set (aphasia) Ask for individual response to the set and the box (one-on-one discussion) Do you like the cards? Do you like the box? Do you like the name of the box? Does the set make sense? Can you follow all the cards? Are there any that don't fit? Which ones do you like best? 	Box Set of cards about aphasia for each participant Visual scale and yes/no card for each participant pen & paper	NO	RH
3	Open discussion and feedback regarding the set concept Offer options for name of box: Language Box Stroke Box Speech Box Information Box	Set of cards about aphasia for each participant Visual scale and yes/no card for each participant pen & paper	Yes	RH
	TEA BREAK			
4	 Place talking mats on the table. Reminder of visual scale as a way to express preference Use TM to express preferences regarding 'Stroke and your Emotions' Photograph mats and feedback Use TM to express preferences regarding the amount of information per page Photograph mats and feedback 	Emotions cards A4 pages with 3/2/ images from a set on each piece of paper and card with one image on it Talking mats camera		СН
5	What advice would you give to someone who's had a stroke?	Flipchart		СВ
	Group brainstorm	Flipchart stand		
6	 Final comments: Thanks Designer will keep working We will be in touch when the initial set is ready (this is what it will be: 10 topics) hope you can come back for a cup of tea and to see finished product 			СВ

Appendix Q

Sample of transcriptions

Sample 1: Group A discussion relating to the set of cards about aphasia.

RW	Yeah	
	we talked about how different ideas work for some people - different	
F1	pictures work for other people - so that one's ok for you [card no. 3] but not for TM?	
F4	it's the lettering	
ΤM	no - it just doesn't go in properly	
F1	no – ok - so we need to think again about that one	
F2	Can GG and I say about picking up from card number 3 to card number 8. When you looked at card number 8, GG, you very clearly put you hand over the final one.[aphasia can get better over time card]	
GG	ah /mi mi/	
F2	which made me think - this card had really connected for you	
GG	ah /mi/	
F2	because what you're saying is – well - wasn't right [pointing to speech bubble with perfectly formed font 'hello']	
F1	so there's been progress so far.	
GG	Ah	
F1	but not back to	
F2	but not back to normal	
GG	ah mi mi [nodding agreement]	
F2	so what I wonder - is when you looked at this card where there was that progression - that that lettering made sense but that card on its own didn't	
F4	yeah - that was the same for TM – actually - you said 'some days it's OK and other days'	
ΤM	it's better - yeah -some days are better than others	
F2	all right so that's a good card – we like number 8	
F3	it's a good card [no.8] except	
F1	can we just hang on a sec because TM's brought up a different thing there about different days and this one's about progress	14'02
ΤM	Yes	
F1	but it's also applying to some days	
TM	are better than others	
F1	sorry EC	
EC	it takes years for you to get better - but that looks like it's a month	14'25
F1	a month - yeah so that's not doing it for you - is it? that's not the true story	
EC	Yeah	
F1	no - so we can change the dates - that's not a problem - I thought that as well - do people like having the dates there? you just need a different date? or could we get rid of the dates?	
ТМ	you don't need it	
F1	you don't need it - what about you RW?	1
RW	I don't know	1
F1	you don't know	1

RW	Iten years ago I had a stroke I can'tI can't speech is	15'17
F1	still difficult?	
RW	Yeah - ten years - yeah	
F1	yes - so this is a bit too optimistic	
RW	yeah	
F1	Maybe	
RW	I don't know	
EC	years and years	
F2	it's not realistic	
F1	years and years.	
EC	months or years - you don't know.	
F1	no - not months - it's over years -isn't it	
EC	Yeah	
RW	Yeah	
F1	what about having the perfect hello at the end? - which says 'back to normal' - doesn't it?	
BT	but they don't do they - you've always I don't think I've altered in my speech since the first month in hospital - walking about - I got a little bit better - I'm going downhill now [laughter] well - I don't try now	16'23
F1	you don't try now - you've got your wheels [mobility scooter]	
BT	I've had six years of doing this and having blood and I've not made any progress	
F1	no – right	
BT	so but with me I think it's you know - I accept it	17'12
ТМ	Yes	
BT	and I go on - I shave and I get around	
F3	Yes	
GG	ah /mi/	
F2	what we're saying is that we don't think that does make sense –GG - it doesn't - and actually when you first looked at that	
GG	ah /mi/	
F2	you didn't look at the calendar at all - it wasn't necessary - it made sense without the calendar	
GG	ah [indicating agreement]	

Sample 2: Group B discussion relating to the set of cards about aphasia: accuracy and illustration of concepts of recovery and prevalence of aphasia.

F1	it can get better - number 8 - what did you make of that one?	
OS	yes - I think that is – yes - it can - but it were the numbers - I'm not very	
	good with numbers	
F1	Ok	
OS	clocks and numbers - first thing in the morning I cannot tell time [laughter] I	21'30
03	can't no I can't	
F1	Right	
F2	so was that a distraction for you?	
OS	that yeah - and mumbers - numbers are numbers do	
F1	Ok	
OS	numbers are really difficult for me	
F1	Yep - anybody else	

RPwhat about if - february to october you're doing incredibly well to get better in that timeAllLaughterRPI would say year one - year two - year threeF2well that would be very clearRPand it shows it carrying onAlllaughter and general agreementF1does it need a time frame on it?OSit doesn't for meNHno because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that noAll[laughter]OSI have to go with my finger january february what month it is - you know [laughs]	
AllLaughterRPI would say year one - year two - year threeF2well that would be very clearRPand it shows it carrying onAlllaughter and general agreementF1does it need a time frame on it?OSit doesn't for meNHno because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that noAll[laughter]OSI have to go with my finger january february what month it is - you know	
RPI would say year one - year two - year threeF2well that would be very clearRPand it shows it carrying onAlllaughter and general agreementF1does it need a time frame on it?OSit doesn't for meNHno because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that noAll[laughter]OSI have to go with my finger january february what month it is - you know	
F2 well that would be very clear RP and it shows it carrying on All laughter and general agreement F1 does it need a time frame on it? OS it doesn't for me NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
RP and it shows it carrying on All laughter and general agreement F1 does it need a time frame on it? OS it doesn't for me NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
All laughter and general agreement F1 does it need a time frame on it? OS it doesn't for me NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
F1 does it need a time frame on it? OS it doesn't for me NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
OS it doesn't for me NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
NH no because I think it is something that will get there in due course but it is not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
NH not like the 14th of feb or the 31st of october - it's nowhere near that no All [laughter] OS I have to go with my finger january february what month it is - you know	
All [laughter] OS I have to go with my finger january february what month it is - you know	
OS I have to go with my finger january february what month it is - you know	
F2 you didn't look at that bit [addressed to JB]	
JB no I just - I looked at it but I couldn't see it	
F1 bottom that would work - would - does the F2anging speeF2 work for you?	
ves I think so - hallo hallo hallo - that is it getting better as the time goes in	
NH goes on	
ok – okidoke - lovely thank you, any other comments anyone has to make	
F1 about the	
All [talking at once]	
hang on - we'll have IB's thoughts on - [laughter] - on our favourite card –	
F1 ok - number 6 [card 6 aphasia is common]	
JB well she thought it was - she said it looked like [laughing] art form	
F1 looked like?	
well I couldn't make that at all I hadn't worked out what it was - but I	
JB couldn't see it adding anything at all to anybody about anything	
F1 0-k	
F2 the word that you said was 'it's a bit of a nothing' and the word that I said	
was 'wallpaper'	
All [laughter]	
F1 it's very nice wallpaper	
I said a very similar thing to [name of communication facilitator] - same	
thing it's that /fuff/! - what if you did it in black? not too big - 'cos black's a	
	24'02
something like 450 thousand people in the UK have aphasia - something -	
after a stroke - maybe -	
F1 it's a bit vague isn't it	
RP dunno - there are about 450 thousand people - I've never met - we've met	
some of them [laughter]	
F1 you've met quite a lot of them	
RPbut do you know, those le - the numbers 450 thousand people have aphasia after a stroke - it's an idea	
F1 yeah - ok so very specific fact with a number - that might work - yeah	
RP Yeah	
F1 yeah ok -anything else on that one?	
? no –no	
? no -no F1 we've got the thumbs down for the wallpaper	

F1	Brilliant - anything else about the ones you didn't like?	
F2	Yeah - MM wants to say something about the remote control	
F1	Mm	
MM	I don't know what's it doing there [laughter]	07'48
F1	fair enough [laughter]	
F2	it's a fair point [laughter]	
F1	well –yes - so you could just get rid of it	
MM	Yeah	
F1	you'd be happy without it? but SG really liked that - didn't you? yeah?	
SG	controls your body	
F1	yeah, so each step of the (sentence)	
SG	(yeah) – yeah	
F1	controls body - that really	
SG	(yeah)	
F1	(helped) - so again it's different for different people	
MM	what did you mean? how did you mean -anyway? I haven't got that	
SG	Erm –I - talking of erm – erm - oh my god [subvoc] - explain - erm - I don't	
	know	
MM	but that don't - that don't do that for you it's your brain	
unknown	Yeah	
F2	is it that there's a picture for each word?	
SG	Yeah	
F3	for SG it's better for her to have a picture for each word	

Sample 3: Group C discussion of number of images per sentence.

Appendix R

Samples of Template 1 and Template 3, to show changes made as result of the iterative process

All images © The Stroke Association

(examples shown smaller than actual size)

Template 1

Template 3

1. Example showing precise coherence between text and image. Note the inclusion of large and bold font in the Template 3 version.



2. Example showing more acceptable photographic image of the concept 'listens to speech', including interaction between two people, both of whom appear comfortable.





3. Example to show change to single image layout, enlarged font for key message, large, bold font keyword





4. Example showing progression from abstract representation towards a choice of more concrete representations. The first shows this in text form, the second in an image.







5. Example to show realistic timescales for recovery from stroke





6. Example to show a sequence of Template 3 materials.

Each card presents a single concept. In sequence, the cards explain a complex concept.









Appendix S

User-testing survey for Phase 3

- 1. Tell us your first impressions of the guidelines.
- 2. Using the guidelines to produce information: what was **easy** to do? What worked well?
- 3. Using the guidelines: what was hard to do? What didn't work so well?
- 4. Is there **anything missing** from the guidelines? Is there any advice you need that we didn't include?
- 5. Was there anything we included which you felt you didn't need?
- 6. **Resources:** please tell us about your experience of finding and using the resources to make information. For example, using Word or Publisher, using Flesch-Kincaid, getting good pictures.
- 7. Were you able to produce some accessible information?

Did you produce a set of information?

- 8. What guidelines or information do you currently use (if any) to inform your production of aphasia friendly materials? Do you know of any resources for this?
- 9. Would you use these guidelines in future for making information accessible? If yes, please tell us what sort of materials you might use them for.
- 10. Is there anything else you'd like to mention?

Appendix T

Accessible Information Guidelines

Published by The Stroke Association, July, 2012

Please note these Accessible Information Guidelines are © The Stroke Association. They are available to download from:

https://www.stroke.org.uk/sites/default/files/accessible_information_guidelines.pdf1_.pdf

